

A CRITICAL EVALUATION OF THE INTERNATIONAL CRIMINAL COURT'S NEW JURISDICTION OVER THE CRIME OF AGGRESSION

- A milestone for international criminal accountability or a 'renewed licence to wage illegal wars'?*

Suzanne Jeanette Claxton

Doctor of Philosophy

School of Law

October 2021

^{*}Benjamin B. Ferencz, 'Ending Impunity for the Crime of Aggression' (2009)

Declaration:

'I confirm that this is my own work and the use of all materials from other sources has been properly and fully acknowledged.'

Suzanne Claxton

(l. Clasiton

26th October 2021

Acknowledgements

I would like to thank my husband Christopher for his support and patience throughout this long passage. I would like to extend my deepest gratitude to my supervisor, Doctor Graham Melling. His insights throughout the entirety of my research have been thought-provoking and his support and guidance have been instrumental in my completion of this thesis. In addition, I would like to thank Professor Duncan French for his helpful comments in the final stages of my work.

Abstract

The 2010 Kampala definition of the Crime of Aggression represents an historic milestone in the attempt by the international community, beginning at Nuremberg, to define, enable prosecution of and eradicate the 'supreme international crime' of aggression.* The jurisdiction of the International Criminal Court was finally activated as of 17 July 2018, some 20 years after its inclusion as an undefined crime in the Rome Statute 1998.

This thesis critically evaluates the challenges the International Criminal Court faces in exercising its new jurisdiction over this 'supreme' crime, in light of the near debilitating limitations placed on that jurisdiction. These challenges arise, in the first instance, from the highly restrictive definition itself, raising problems of a very high gravity threshold, limitation of prosecution to political and military leaders only, restriction to state-on-state aggression and a finite list of state acts, which are potentially outdated. Modern types of warfare, including cyberwarfare and aggression committed by non-state actors are currently not included.

The second area of difficulty for the Court is posed by the extraordinarily restrictive jurisdictional conditions. Thus, the Court is subject to intense Security Council scrutiny, as well as a significantly reinforced requirement of consent by individual States to

^{*}Robert H. Jackson, 'Opening Statement before the International Military Tribunal', 21 November 1945, *United States Representative to the International Conference on Military Trials*, HMSO, (London 1945), 305.

jurisdiction, even if they are signatories to the Rome Statute. The examination involves a critical assessment of these features and considers whether, in light of the restrictions placed on the Court's determination powers, effective prosecution and jurisdiction over the crime is possible. It will also suggest alternative approaches and potential solutions as to how the Court's reach over the crime of aggression could be widened. These range from judicial extension by close analogy and a re-write of the definition to a re-interpretation of States' right to consent as a duty, as well as the exercise of universal jurisdiction. It is contended that, without extension of jurisdiction through these means, the criminalisation and prosecution of aggression is in danger of irrelevance from the outset, allowing for elective impunity. Extending jurisdiction of the Court through the suggested means, on the other hand, would strengthen the prohibition of the use of force and contribute to the eradication of one of the most serious international crimes, furthering the aims of the Rome Statute and the UN Charter itself.

Table of Contents

INTRODUCTION	1
1. Scope of inquiry	4
2. The role of States at international law	6
3. Positivist underpinnings of state consent	11
4. Can the state consent requirement be circumnavigated?	13
5. Outline of the analysis	32
CHAPTER 1: AGGRESSION IN A WORLD OF SOVEREIGN STATES – THE HISTORICAL A	
1. Introduction	35
2. From the jus fetiale to the jus ad bellum – Early perceptions of aggression	36
3. Attempts to curb aggression in the first half of the 20 th century	39
3.1 The creation of the League of Nations	39
3.2 The Geneva Protocol on the Pacific Settlement of International Disputes	
3.3 The Kellogg-Briand Pact	44
3.4 The Geneva Conference for the Reduction and Limitation of Armaments	46
4. The development of collective security and individual criminal responsibility	48
4.1 The creation of the United Nations and the prohibition on the use of force.	48
4.2 The institution of the Security Council and the anomaly of the veto power	51
4.3 Developing individual criminal liability – The Prosecutions by the Internatio Tribunals	-
5. A Cold War U-turn? From individual accountability to sole reliance on the UN's mechanism of collective security	
5.1 The early record of the United Nations	60
5.2 Cold War efforts to arrive at a definition of aggression	64
5.3 The definition of the State act and its application by the Security Council	67
6. Paving the way for the crime of aggression: The ILC's efforts and a return to the of individual criminal liability	•
6.1 The ILC Draft Codes of Crimes against the Peace and Security of Mankind in 1996	
6.2. The Rome Statute 1998 and a move towards jurisdiction over the crime	75
7. The historical background in context	77
CHAPTER 2: THE GRAVITY THRESHOLD	82
AND STRICT LEADERSHIP CLAUSE	82
1. Introduction	82
2. The gravity threshold	87

		2.1 The ICJ's treatment of the gravity threshold	88
		2.2 The contemporary context - Aggression in an age of 'humanitarian' interventions.	90
		2.3 Negotiating the threshold clause	95
		2.4 Too high a threshold? – Arguments for and against wider determinations	103
	3.	. Limitation to political and military leaders and the 'control or direct' requirement	115
		3.1 The historical roots of the leadership clause	117
		3.2. Private economic actors and the development of a 'shape or influence' test	119
		3.3 The control or direct test	123
		3.4 The practical implications of the choice to apply the control/direct test to the crim aggression	
		3.5 Participation at a lower level and other modes of conduct	135
		3.6 The accessory liability doctrines of Conspiracy and Joint Criminal Enterprise	137
		3.7 An overall evaluation of the strict leadership requirement	140
C	НА	PTER 3: THE CRIME'S STATE-CENTRICITY AND THE LIMITED LIST OF AGGRESSIVE ACTS	144
	1.	. Introduction	144
	2.	. State versus State aggression	147
	3.	. The List of acts deemed to be aggressive	154
		3.1 The relevance of the list of aggressive acts in a modern context	158
		3.2 The (non-)exhaustiveness of the list - a continuing academic debate	162
	4.	. An overall evaluation of the definition of the crime of aggression	168
		APTER 4: CONSENT OF THE POWERFUL – THE ROLE OF THE SECURITY COUNCIL IN GRESSION PROCEEDINGS	172
	1.	. Introducing Articles 15 <i>bis</i> and <i>ter</i>	172
		1.1 The unusual procedural and temporal delays common to Articles 15 bis and ter	176
		1.2 The special jurisdictional conditions in Article 15 <i>bis</i> in comparison with Article 15	
		1.3 The issues to be considered	183
	2.	. The role of the Security Council – exclusivity or mere primacy?	185
		2.1 Security Council responsibility according to the UN Charter	185
		2.2 The stance of the ILC and SWGCA on the contentious question of Security Council monopoly	187
	3.	. Why insist on dependence on the Security Council in the first place?	193
		3.1 Protection of the Security Council's prerogative and the fear of conflicting determinations	. 193
		3.2 Protecting the Court from a political quagmire	210
		3.3 The argument that Security Council involvement expands the Court's reach	215
	1	An evaluation of Security Council involvement	220

STATES STRICT REQUIREMENTS OF MULTIPLE CONSENT AND MEMBERSHIP OF	225
1. Introduction	225
2. Combining Article 15 bis (4) with Article 121 (5) – The multiple consent requirement.	229
2.1 Article 15 <i>bis</i> (4)	229
2.2. The choice to apply Article 121 (5) to the crime of aggression	230
2.3 Why insist on Article 121 (5) instead of (3) or (4)?	233
2.4 Consent by whom exactly? – Two different interpretations of Article 121 (5)	238
2.5 Combining Article 15 bis (4) with the negative interpretation of Article 121 (5)	242
2.6 Yet another 'layer' of consent? – The reasons for the ASP's decision	246
2.7 Assessing the effect of the emphasis on multiple consent by member states	256
3. Article 15 bis (5) and the double membership requirement	259
4. A way forward? - A teleological interpretation, the <i>jus cogens</i> nature of the crime and principle of universal jurisdiction	
CHAPTER 6: TRANSCENDING THE JURISDICTIONAL LIMITATIONS	
1. Introduction	294
2. Revisiting the need to reconceptualise the definition	295
3. Revisiting the stringent jurisdictional restrictions	302
4. Proposals to overcome the difficulties	308
4.1 Rewriting aggression, room for analogy and determinative independence of the C	
4.2 The duty not to withhold consent	313
4.3 The need for Security Council reform	316
4.4 The revival of universal jurisdiction	321
5. Transcending the limitations? – An overall assessment	326
CONCLUSION	332
Suggestions for further research	343
Cases, International Treaties and other documents	346
Bibliography	361

INTRODUCTION

In June 2010, history was made in Kampala at the 1st Review Conference for the International Criminal Court.¹ After more than a decade of deliberations between negotiating states parties and observer states, consensus was finally achieved on a definition of the crime of aggression and its jurisdictional conditions.² The purpose of this definition was to identify individual criminal responsibility for the collective act of aggression.

Vast expectations had, for nearly a century, attached themselves to the birth of this crime, and the hope was expressed that it would capture all types of aggressors responsible for the illegal use of force, constituting a tool for wide jurisdiction of the Court. As the former prosecutor of the Nuremberg Tribunals,³ Benjamin Ferencz, pleaded just prior to the Kampala Conference:

Giving an international criminal tribunal effective jurisdiction over aggression, even if it seems remote today, would be an historical achievement of incalculable significance. Every legal step should be taken that might help deter nations from the incredible horrors of armed conflicts. Aggressors should not be granted a renewed licence to wage illegal wars with impunity... Nuremberg was triumph of

¹ The International Criminal Court ('the ICC' or 'the Court') was established by the Rome Statute of the International Criminal Court (adopted 17 July 1998, entered into force 1 July 2002) 2187 UNTS 38544.

² International Criminal Court, Assembly of States Parties (ASP), Review Conference, 'The Crime of Aggression', ICC Doc. RC/Res. 6 (11 June 2010), (Resolution 6, or the Kampala amendment).

³ Nuremberg Judgment, France and ors v Goering (Hermann) and ors, Judgment and Sentence, The Trial of the Major War Crimes before the International Military Tribunals at Nuremberg [1946] 22 IMT 203, (the IMTs).

Reason over Power. Allowing aggression to remain unpunishable would be triumph of Power over Reason.⁴

Following the acceptance of the Kampala amendments, the inception of the crime of aggression and jurisdiction by the first permanent International Criminal Court was celebrated as an unprecedented innovation and progression of international criminal law.⁵ Claus Kress and Leonie von Holtzendorff, observing the events, described the Kampala consensus on aggression as 'a milestone in the development of international law', and expressed hope that 'world opinion will begin to slowly exert its soft power towards the expansion of the ICC's jurisdictional reach.'⁶

Reaching consensus on the aggression provisions was widely viewed as a major achievement, since the international community had, for decades, been attempting to define and outlaw the illegal use of force. A pre-existing notion of a largely unfettered right to go to war had, during the 20th century, become the subject of scrutiny, leading to condemnation and attempted prosecutions. This change in attitude was the result of two unprecedented world wars and the lingering threat posed by a Cold War, with the potential of culminating in the deployment of nuclear weapons and large scale or even global destruction. However, despite the developing perception of aggression as one of the worst international crimes, an agreement on a definition and conditions for the exercise of

⁴ Benjamin B. Ferencz, 'Ending Impunity for the Crime of Aggression' (2009) 41 Case W. Res. J. Int'l L. 281, 290.

⁵ Claus Kress and Leonie von Holtzendorff, 'The Kampala Compromise on the Crime of Aggression' (2010) 8 (5) JICJ, 1179. See also Jennifer Trahan, 'The Rome Statute's Amendment on the Crime of Aggression: Negotiations at the Kampala Review Conference' (2011) 11 Int'l Crim L. Review 49, 50.

⁶ Kress and von Holtzendorff, *ibid*, 1217.

⁷ For a detailed account, see Chapter 1.

jurisdiction continued to remain out of reach. At the 1998 Rome Conference, aggression was included as a 'blank prose crime' in Article 5 (1),9 without a definition or jurisdictional conditions, which were left to be decided at a later date. 10 The central problem of establishing individual criminal responsibility for the collective state act of aggression had, despite the precedents of prosecutions at Nuremberg and Tokyo, as yet not been resolved. 11 As observed by Antonio Cassese, at the root of the disagreement was not only the political nature of the crime, but also a prevalent reticence of many States to relinquish, even partially, their sovereignty and grant jurisdiction over a new international crime to a new court.¹² As a result, and despite the substantial efforts of the negotiators, the deliberations nearly failed entirely until the last moments of the Kampala Conference, in which a compromise was agreed leading to the incorporation of Articles 8 bis and 15 bis and ter.

Jurisdiction over aggression has only recently been activated, and prosecuting the crime

will be a new challenge for the first permanent international criminal court. For the first

⁸ This iconic term was coined by Michael Glennon, 'The Blank Prose Crime of Aggression' (2010) 35 Yale

⁹ Article 5 (1) of the Rome Statute provides that '[t]he jurisdiction of the Court shall be limited to the most serious crimes of concern to the international community as a whole. The Court has jurisdiction in accordance with this Statute with respect to the following crimes:

⁽a) The crime of genocide;

⁽b) Crimes against humanity;

⁽c) War Crimes;

⁽d) The crime of aggression.'

¹⁰ Article 5 (2) of the Rome Statute provided that '[t]he Court shall exercise jurisdiction over the crime of aggression once a provision is adopted in accordance with Articles 121 and 123 defining the crime and setting out the conditions under which the Court shall exercise jurisdiction with respect to this crime. Such a provision shall be consistent with the relevant provisions of the Charter of the United Nations.' Article 5 (2) was deleted by the Kampala amendment.

¹¹ The IMTs at Nuremberg, (n. 3), and *Tokyo War Crimes Trials*, The International Military Tribunal for the Far East (3 May 1946 to 12 November 1948), established by its Charter, 1946, 14 DSB (Article 5 a)) 361, 362 (1946), Judgment reprinted in reprinted in R. Pritchard (ed); The Tokyo Major War Crimes Trial (1998) 1190.

¹² See Antonio Cassese's concise assessment in *International Criminal Law* (3rd edn, OUP 2013) 138. See also Kress and von Holtzendorff (n. 5).

time in over seventy years, leaders have now been placed on notice that their extant and future acts of aggression may lead to criminal prosecutions. However, as the examination will reveal, the limitations placed on the jurisdiction of the Court are substantial.

1. Scope of inquiry

The main purpose of the inquiry is to assess the potential effectiveness of the Court's new jurisdiction over the crime of aggression, in light of the restrictions placed on it. These limitations, as a direct result of the Kampala compromise, are two-pronged. The first area of concern arises from the overly narrow definition itself, ¹³ which is based on the 1974 'Definition of Aggression' contained in General Assembly Resolution 3314. ¹⁴ Problems range from a high gravity threshold to a restriction to traditional types of warfare only, which may be irrelevant in a modern context. Additionally, participants other than the highest political and military officials have been excluded, and the definition of the crime is exceptionally state-centric, failing to capture the prevalent contemporary phenomenon of aggression committed by non-state groups.

The second area of concern is that the Court's jurisdiction has been severely handicapped through extraordinary jurisdictional conditions.¹⁵ These significant and far-reaching

¹³ The definition is contained in Article 8 *bis* of Resolution 6.

¹⁴ United Nations General Assembly 'Definition of Aggression' (14 December 1974), XXIX, UN Doc A/RES 3314 (hereinafter Resolution 3314).

 $^{^{15}}$ Contained in Articles 15 *bis* and 15 *ter* of Resolution 6, providing for an exception to the regime established in Article 12, see Chapters 4 and 5.

limitations incorporate an amount of Security Council control not commonly present in international criminal prosecutions, as well as an exceptional level of insistence on state consent. The restrictive effect of these specific provisions has been exacerbated by a recent controversial decision to apply the narrowest interpretation, resulting in a requirement for consent to be expressed in multiple ways, by both victim and aggressor state. This exceptional consent requirement, in particular, is capable of undermining jurisdiction of the Court and potentially facilitating elective impunity. The problem of state consent to criminal prosecutions, therefore, poses the greatest problem for wider jurisdiction of the Court, and pervades the discussion throughout.

These remarkable conditions differentiate the crime of aggression from the other crimes under the Rome Statute, and their examination begs the questions: Why was so much effort expended at limiting the ICC's independent jurisdiction? What are the reasons for historical and continuing scepticism of criminal jurisdiction over aggression, resulting in concerted attempts to actively restrain its exercise? Are the grounds for restriction of jurisdiction sufficiently justified? And were the level of Security Council involvement and greatly reinforced state consent requirement necessary? Do these severe restrictions render the newly obtained jurisdiction essentially unfit for purpose, in light of improbable or unsuccessful prosecutions, resulting in a lack of credibility of the Court and ineffective deterrence?

-

¹⁶ See Chapters 4 and 5.

¹⁷ Resolution ICC-ASP/16/Res.5, 16th Assembly of States Parties, 14th December 2017 (Resolution 5). See the discussion in Chapter 5.

¹⁸ Despite the fact that, following a Security Council referral, jurisdiction of the ICC arises automatically irrespective of the consent of victim or aggressor, state consent is also relevant in such situations, because the permanent members of the Security Council (P5) effectively have to consent to the referral. See Chapter 4.

The contribution of this thesis to the debate not only entails an assessment of the likely practical effects of the restrictions on the Court's novel jurisdiction over aggression; it also attempts to suggest alternative approaches and potential solutions to overcome the limitations of the new crime. These range from the practical to the undeniably idealistic: from the proposal to re-amend the outdated definition and pragmatically promote wider voluntary subscription to the Court, to an entire re-think of the state consent requirement, particularly in the criminal branch of international law, and the necessity for reform of the voting structure within the Security Council. ¹⁹ They also consider the revival of the principle of universal jurisdiction over international atrocity crimes at customary law, harking back to Nuremberg and beyond. The more ambitious of these suggestions, in particular, deserve far more attention in a separate research project than I am able to afford them here, focusing mainly on the practical limitations placed on the ICC's new jurisdiction over the crime of aggression. ²⁰ Before embarking on this examination, it may be prudent to set the scene for the arguments presented and set out the theoretical background.

2. The role of States at international law

States are the main actors on the international stage. Acting on behalf of their citizens, they negotiate and enter into international agreements, engage in diplomatic relations with

¹⁹ See the discussion in Chapter 6.

²⁰ See the suggestions for reform in Chapter 6 and for further research in the Conclusion.

other States, and pursue the defence of their territory. At times, States also commit acts of aggression through the participation of individuals. As a result, state centricity is a cardinal feature of aggression,²¹ which has also been transposed into the Kampala definition of the crime,²² despite the fact that the latter is concerned with individual criminal liability. Furthermore, the crime of aggression is subject to the most stringent, triply buttressed state consent requirement within the jurisdictional conditions.²³ It is this reinforced insistence on state consent to criminal jurisdiction over aggression, which constitutes the greatest obstacle to widespread jurisdiction of the Court.²⁴ Andrew Guzman sums up the problem of state consent in international law generally as follows:

[M]odern international law is built on a foundation of state consent. In an era of pressing global challenges, however, this commitment to consent represents a double-edged sword. On the one hand, consent protects the interests of states and supports notions of sovereign equality. On the other hand, it functions as a barrier to effective cooperation in a world of vastly divergent priorities and concerns. A requirement of consent creates a powerful *status quo* bias that frustrates many attempts to solve global problems.²⁵

²¹ See for instance the definition of the state act in Resolution 3314 (n. 14). For detail, see Chapters 2 and 3.

²² Article 8 bis.

²³ Articles 15 *bis* and *ter*.

²⁴ Discussed in Chapter 5.

²⁵ Andrew Guzman, 'The Consent Problem in International Law' (2011), UC Berkeley: Berkeley Program in Law and Economics, available at https://escholarship.org/uc/item/04x8x174, 5, accessed 6 November 2020. See also Stephen Krasner's assessment that the insistence on state sovereignty at international law gravely affects the ability of States to resolve the biggest international problems. Stephen Krasner, 'The Hole in the Whole: Sovereignty, Shared Sovereignty, and International Law' (2004) 25 Mich. J. of Int'l L. 1075, 1077-78.

Notwithstanding the problems the state consent requirement causes in tackling difficult international or global issues, it has long been accepted as the source of international law.²⁶ The Permanent Court of International Justice stated in the 1927 *Lotus* case:

International law governs relations between independent states. The rules of law binding upon states therefore emanate from their own free will as expressed in conventions or by usages generally accepted as expressing principles of law and established in order to regulate the relations between these co-existing independent communities or with a view to the achievement of common aims.²⁷

This interpretation thus insists on voluntarism, with state consent as the source of the binding force and legitimacy of international law.²⁸ States bind themselves legally by granting their consent,²⁹ and – if interpreted strictly – without such consent there is no legal enforceability.³⁰ The consent requirement therefore allows States to 'control both the content and meaning of international law.'³¹ The 1969 Vienna Convention on the Law

²⁶ Anthony Aust, *Handbook of International Law* (CUP 2005), 4. See also James Leslie Brierly, *The Law of Nations: An Introduction to the Modern Law of Peace* (Clarendon 1963) 51-54, and Ian Brownlie, *Principles of Public International Law* (6th edn, OUP 2003), 4.

²⁷ S.S. 'Lotus' (France v Turkey), [1927] PCIJ Rep Series A No. 10, 43-44.

²⁸ See Jan Klabbers, 'Law-Making and Constitutionalism', in Anne Peters, Jan Klabbers, and Geir Ulfstein (eds), *The Constitutionalization of International Law* (OUP 2009), 81, at 100, 114. See also Jean d'Aspremont and Joerg Kammerhofer, 'Introduction: The Future of International Legal Positivism', in Joerg Kammerhofer and Jean d'Aspremont (eds), *International Legal Positivism in a Post-Modern World* (CUP 2014), 4.

²⁹ Anthony Arend, *Legal Rules and International Society* (OUP 1999), 87. Consent itself has been defined as '[t]he role of the procedures of consent to be bound is to constitute a mechanism by virtue of which a treaty becomes binding on states, or, as it was described, acquires characteristics of a "juridical act". (M. Fitzmaurice, 'Consent to be Bound – Anything New Under the Sun' (2005) 74 Nord. J. Int'l L. 484.

³⁰ James Crawford, *Brownlie's Principles of Public International Law* (9th edn, OUP 2012), 20. See also Louis Henkin, *International Law: Politics and Values* (Nijhoff 1995), 28.

³¹ Guzman (2011) (n. 25) 5.

of Treaties confirms the significance of state consent as binding on treaty parties.³² The VCLT also asserts that third parties cannot be bound by a treaty without their consent.³³ Consent to a treaty can be expressed in a number of ways with the effect of binding States.³⁴ The most obvious manifestation of state consent is evidenced by membership of treaties, but the conduct of States may also point to such consent and forms an alternative, although more difficult to determine, source of international law consisting of customary norms. To that effect, Article 38 (1) of the 1946 Statute of the International Court of Justice³⁵ recognises conventions (or treaties), as well as customary international law, evidenced by state practice, *opinio juris*, and 'the general principles of law recognized by civilised nations'.³⁶

The most common form of customary international law is constituted of state practice and *opinio juris*,³⁷ and it is non-consensual in the sense that it is possible for States to be legally bound by customary rules even if they have not individually consented to them.³⁸ The 'general principles' constitute an additional customary source of law,³⁹ and are also recognised in Article 53 of the VCLT:

³² United Nations, Vienna Convention on the Law of Treaties (23 May 1969, entered into force 27 January 1980) 1155 UNTS 331 (VCLT).

³³ Article 34 of VCLT.

³⁴ Articles 11 –15 of VCLT.

³⁵ United Nations, Statute of the International Court of Justice (18 April 1946) 33 UNTS 993 (ICJ Statute). ³⁶ *Ibid.* Article 38 (1).

³⁷ Customary international law has been widely recognised in the jurisprudence of the ICJ, for instance in the *North Sea Continental Shelf cases* (*Germany v Denmark*) [1969] ICJ Rep. 3, and *The Asylum Case* [1950] ICJ Rep. 266. The customary prohibition of aggression, specifically, was acknowledged in *Case Concerning Military and Paramilitary Activities in and against Nicaragua* (*Nicaragua v United States*) (Merits), [1986] ICJ Rep. 14, para 103 (hereinafter *Nicaragua* case). This was confirmed in *Case Concerning Armed Activities on the Territory of the Congo (Congo v Uganda)*, [2005] ICJ Rep. 168, para 223 (hereinafter *Armed Activities* case). National jurisdictions also recognise customary international law. The UK case of *R v Jones* [2006] UKHL 16, for instance, acknowledged the existence of a customary crime of aggression several years before the Kampala Definition (para 19).

³⁸ Michael Akehurst, 'Custom as a Source of International Law' (1975) 47 BYIL, 1, 23.

³⁹ M. Cherif Bassiouni, 'A Functional Approach to 'General Principles of International Law' (1990) 11 Mich. J. Int'l. L. 768, 768-9. See also Hersch Lauterpacht, 'International Law – the General Part' (unpublished) in Elihu Lauterpacht (ed), *International Law: Collected Papers of Hersch Lauterpacht*, (CUP 2009) 1, 74. Whilst there is no definitive list of these norms, it is widely accepted that they include

[A] peremptory norm is a norm accepted and recognized by the international community of States as a whole as a norm from which no derogation is permitted and which can be modified only by a subsequent norm of general international law having the same character.⁴⁰

These customary norms are potentially capable of superseding state consent on account of their compelling and authoritative nature.⁴¹ They are recognised in domestic legal systems as generally applicable on the basis of widespread acknowledgement by other 'civilized nations', but are not necessarily based on the consent of *all* of those States.⁴² As Dworkin states: 'If enough states to constitute "the international community of States" have recognized fundamental rules as peremptory and non-negotiable, then these rules are peremptory and non-negotiable for the whole international community.'⁴³ The 'general principles' are thus not created directly and intentionally through the will of States; rather their pre-existence is merely confirmed by a number of States *ex post facto*. Recently a number of draft conclusions on peremptory norms have been adopted by the International Law Commission, confirming their status.⁴⁴

.

prescriptions relating to piracy and slavery, as well as aggression, the protection of prisoners of war, genocide, torture, war crimes and crimes against humanity (see Bassiouni, *ibid.*) See also International Law Commission, *Report of the International Law Commission on Peremptory Norms of general international law (jus cogens)*, Chapter V, (16th May 2019) UN Doc A/74/10.

⁴⁰ Article 53 VCLT (n. 32).

⁴¹ Peremptory rules have been described as having a 'higher normativity', independent of legal opinion and state practice. See Oscar Schachter, 'Entangled Treaty and Custom', in Yoram Dinstein and Mala Tabory (eds), *International Law at a Time of Perplexity, Essays in Honour of Shabtai Rosenne* (Nijhoff 1989) 717, 734. Note here the differing view of Rosalyn Higgins, reasserting the importance of *opinio juris* and state practice. Rosalyn Higgins, *Problems and Process, International Law and How We Use It*, (2010 reprint OUP), 21.

⁴² Bassiouni, *supra*, (n. 39).

⁴³ Ronald Dworkin, 'A New Philosophy for International Law', (2013) Wiley, Phil. & Pub. Aff., 6.

⁴⁴ See the 2019 *ILC Report* (n. 39).

Notwithstanding the fact that in a given situation the individual consent of a State may not be required for a customary international law rule to apply to that State, these norms do not substantially undermine the common insistence on state consent as the general basis of international law.⁴⁵ All of these sources of legally binding norms, whether treaty based or customary in nature, are thus founded on at least some level of state consent, although not necessarily on the consent of every single State.⁴⁶ Consent is, therefore, widely perceived to be central to international law as a source of legitimacy,⁴⁷ allowing sovereign nations to enter into binding agreements whilst protecting their interests and sovereignty, and consequently it is jealously guarded by States.⁴⁸

3. Positivist underpinnings of state consent

The theoretical underpinnings of the emphasis on state consent, as the basis of international law, stem from the strand of epistemology associated with legal positivism.⁴⁹ According to this traditional view, States are the creators as well as the primary subjects of international law, and their consent is required as of necessity.⁵⁰

⁴⁵ Andrew Guzman, 'Saving Customary International Law', (2005) 27 Mich. J. Int'l L., 115, 141-5.

⁴⁶ Alexander Orakhelashvili, *The Interpretation of Acts and Rules in Public International Law* (OUP 2008), 112. See also Guzman, (2011), (n. 25) 32.

⁴⁷ See Franck's argument that consent amounts to a validation of authority and evidence that this authority is being exercised 'in accordance with right process', Thomas Franck, *Fairness in International Law and Institutions* (OUP 1995), 25-46. For a different view that consent is not necessarily required for legitimacy of a legal rule, citing domestic law and delegation of States' authority to the European Union as support, see Guzman, (2011), (n. 25), at 10-13.

⁴⁸ Guzman, *ibid*.

⁴⁹ See Hans Kelsen, *General Theory of Law and State* (HUP 1945) xiv. Herbert L. A. Hart has since developed international legal positivism to entail greater nuances (see H.L.A. Hart, 'Positivism and the Separation of Law and Morals' (1958) 71 (4) Harv. L. Rev. 593-629. See also H.L.A. Hart, *Essays in Jurisprudence and Philosophy* (OUP 1983), and H.L.A. Hart, *The Concept of Law* (2nd edn. OUP 1994). See also the discussion below.

⁵⁰ See for example Henkin, (1995) (n. 30) 29. See also Arend (1999), (n. 29) 76, and Jochen von Bernstorff, *The Public International Law Theory of Hans Kelsen: Believing in Universal Law* (CUP 2010) 28-30.

International legal positivism does not concern itself with a norm's intrinsic morality or political contribution, validity is simply derived from state consent.⁵¹ This lack of extrinsic considerations attempts to render legal scholarship a true science or 'pure theory' of law, ⁵² devoid of moral considerations and value judgments. Hans Kelsen states:

When this doctrine is called the 'pure theory of law', it is meant that it is being kept free from all the elements foreign to the specific method of a science whose only purpose is the cognition of law, not its formation. A science has to describe its object as it actually is, not to prescribe how it should be or should not be from the point of view of some specific value judgments.⁵³

This view of international law as established simply by state consent, free of any considerations of morality, political value, or social utility, has persisted as the mainstream approach to international law in the 20th century and beyond.⁵⁴ As Paulus and Simma assert, '[l]aw is regarded as a unified system of rules that ... emanate from state will. This system of rules is an "objective" reality and needs to be distinguished from law 'as it should be.' These two simple sentences sum up the basic tenets of the prevailing positivist interpretation of international law, the first stipulating that state consent is its

-

⁵¹ Richard Collins, 'Classical Legal Positivism in International Law revisited', in Kammerhofer and d'Aspremont, (2014), (n. 28) 38.

⁵² This term was coined by Kelsen and is derived from the German word 'Rechtswissenschaft', which is based on 'Grundnormen', or basic norms. See Kelsen, (n. 49) at xiv.

⁵³ See Kelsen (n. 49) at xiv.

⁵⁴ For support see for instance Kelsen (n. 49), Henkin (n. 30), Arend (1999) (n. 29), von Bernstorff (2010) (n. 50), and Crawford (2012) (n. 30).

⁵⁵ Bruno Simma and Andreas L. Paulus, 'The Responsibility of Individuals for Human Rights Abuses in Internal Conflicts: A Positivist View' (1999) 93 AJIL, 302, 304-5.

unquestionable source,⁵⁶ and the second denying the relevance of any moral considerations. Despite representing the mainstream view with apparent certainty,⁵⁷ neither statement is completely unassailable.

4. Can the state consent requirement be circumnavigated?

Rules of recognition and potential customary international law exceptions to state consent

Unlike orthodox legal positivism, its post-modern strand recognises that legitimacy at international law is not necessarily solely derived from state consent. Post-modern legal positivism adopts and develops HLA Hart's theory of law, which identifies the existence of a number of rules. The first category, primary rules of obligation, are rules which impose duties or obligations, however, secondary rules are necessary to provide an authoritative determination and statement of the primary rules. To achieve that purpose, the second category includes rules of recognition, rules of change and rules of adjudication. Following Hart's positivist theory, state consent is thus recognised as *one*

⁵⁶ This perspective also forms the basis for the unwavering emphasis on state consent to the ICC's jurisdiction over the crime of aggression, the greatest obstacle encountered in this discussion. See particularly Chapter 5.

⁵⁷ See the discussion above at pp. 6-9.

⁵⁸ Benedict Kingsbury, 'The Concept of 'Law' in Global Administrative Law', (2009) 20 (1) EJIL, 23, 27. See also Jean d'Aspremont, 'Herbert Hart in today's international legal scholarship, in Kammerhofer and d'Aspremont (2014), (n. 28) 146, stating: 'International legal positivism does not mean that law exclusively emanates from the state... For instance, in the pluralised contemporary international legal system, it would be entirely conceivable for law to emanate from non-state entities.'

⁵⁹ See HLA Hart's theory of 'rules of recognition' in for instance, Hart (1994) and (1983) (n. 49).

of the sources of international law, but not the only one. Despite Hart's wider recognition of a number of potential sources as giving rise to legal norms, including the capability of taking into account a number of social facts, his rejection of moral considerations, ⁶⁰ found prevalent in natural law theories, ⁶¹ follows the classical positivist approach. ⁶²

Consequently, Hart's positivism has been described as offering 'a more sophisticated and nuanced account of legal validity that can incorporate non-consensual and even constitutional norms. They [i.e. post-modern positivists] partially reject the substantive institutional core – rooted in consent – that lies at the heart of the classical variant on the grounds that it is too blunt and empirically insensitive.' A similar critique of orthodox legal positivism has been presented by Jean d'Aspremont, who points out that the traditional insistence on state consent and voluntarism not only has the effect of perpetuating state sovereignty as well as prioritising it over international cooperation and agreements, but also 'fails to offer a satisfying theory to explain the binding character of

⁶⁰ Hart, *ibid*, (1994), 79, and (1983), 89. Hart's positivist theory dismisses morality as a source of legitimacy of law. Morality and law are separate concepts (the 'separation thesis'). (See Hart's stance in the Hart-Fuller debate in Hart (1958) (n. 49), and LL. Fuller, 'Positivism and fidelity to law: A reply to Professor Hart' (1958) 71 (4) Harv. L. Rev., 630).

⁶¹ For natural law claims of 'primal' adherence to universal human values, see, for instance, Mark Murphy, *Natural Law in Jurisprudence and Politics* (CUP 2006), 1. For a more nuanced view of law as 'conventional' principles which are 'rooted in shared beliefs', and the necessity for at least some moral considerations in order to establish a hierarchy of rules, see Patrick Capps, 'International Legal Positivism and Modern Natural Law', in Kammerhofer and d'Aspremont (2014) (n. 28) 229 and 237.

⁶² See Kelsen (n. 49).

⁶³ Capps (n. 61) 217. A similar argument can be derived from a different school of legal theory. To that effect, the New Haven School also devalues the role of States, suggesting that the traditional insistence on state consent as the ultimate source of legitimacy of an international legal norm is too static. Instead, law consists of a social process resulting in policy decisions based on values. See Mc Dougal and Lasswell, 'The Identification and Appraisal of Diverse Systems of Public Order' in Myres S. McDougal and W. Michael Reisman, *International Law Essays* (Minneola 1981). These values include, for example, security, wealth, economic growth and trade, respect, articulation and implementation of human rights, amongst others (*ibid* 35-8). The result of this process of law-making is the establishment of 'authoritative' and 'controlling' rules (*id.* 22). For more on this approach and the concept of a 'minimum public world order', see the discussion below.

international law.'64 D'Aspremont's argument thus revisits the fact that customary international law,65 consisting of state practice and *opinio juris*, the 'general principles recognised by civilized nations', and peremptory rules, or *jus cogens*, 66 are all rules capable of binding States even if they have not individually consented to the acceptance of the legal norm. As long as *some* States have consented to these rules,67 a State can be bound without having granted their consent. Consequently, this phenomenon may be interpreted as contradicting the strict application of the consent requirement.68 Traditional, source-based positivism provides no explanation for this contradiction. It fails to provide an answer to fundamental questions, for instance: How many 'civilised nations' must recognise a rule before it becomes binding on all States? What makes States sufficiently civilised? And which norms are peremptory?69 In an attempt to make sense of the problem, which absolute insistence on state consent in international law entails, Dworkin submits:

These latter difficulties stem from the scheme's perfectly understandable ambition to extend the ambit of international law beyond those communities that have explicitly consented to its principles to include those that have not. International

⁶⁴ See Jean d'Aspremont, in Kammerhofer and d'Aspremont (2014) (n. 28) 145. See also Martti Koskenniemi, *From Apology to Utopia: The Structure of International Legal Argument* (1989 CUP, reissue 2005).

⁶⁵ See Article 38 (1) of the ICJ Statute (n. 35). For case law recognising the existence of customary international law, see (n. 37).

⁶⁶ See the discussion above in section 2.

⁶⁷ Thus potentially becoming 'rules of recognition', following Hart's interpretation (n. 59).

⁶⁸ See Dworkin, (2013) (n. 43) 6. See also Dworkin's critique of legal positivism in 'Natural Law Revisited' (1982), 34 (2) U. Fl. L. Rev., 165-188, *A Matter of Principle* (HUP 1985), and *Law's Empire*, (HUP 1986), amongst others. Dworkin's hybrid theory acknowledges both the rules of source-based positivism and moral principles derived from naturalism, whilst rejecting the separation thesis of Hart's positivism (see n. 60). Dworkin's morality is, however, of a political rather than religious nature, as defended in orthodox naturalist theories. (See for example *Basic Writings of St. Thomas Aquinas*, Vol. 2, A. C. Pegis (ed), (Hackett Publishing Co. 1997), and John Finnis, *Natural law and natural rights* (2nd edn, OUP 2011).

⁶⁹ All of these questions are considered problematic by Dworkin (2013), *ibid*, 6-7.

law could not serve the purposes it must serve in the contemporary world – disciplining the threat some states offer to others, for example – unless it escaped the straitjacket of state-by-state consent. But yielding to that ambition seems to undermine the axiomatic place of consent in the scheme, and thus its assumed jurisprudential foundation.⁷⁰

Non-consensual compliance based on the perception of a 'sovereign duty' or the necessity to achieve a 'minimum world public order'

If one accepts that state consent does not, or at least not in all situations, provide a satisfactory explanation for States' abidance by legal rules, the question arises on what other grounds do States comply? It seems logical to assume that non-voluntarist compliance with international law has to be due to the exertion of some sort of pressure, which may be economic, diplomatic, or perhaps even moral in nature. A State may wish to be seen to be 'doing the right thing', thus improving their international reputation and leading to an advancement of their standing in international relations with other States. This may lead to powerful alliances and advantageous trade deals. A State's motivation may therefore be based on political utility or an adapted political morality with additional benefits.

⁷⁰ *Ibid*, 7

⁷¹ Dworkin refers to this as 'coercive dominion'. *Ibid*, 10.

⁷² Ibid

The question remains, however, how individual States decide that an international rule is 'law', and therefore ought to be legally binding on that State. If the State looks to other States' decisions to determine 'law' for themselves, what considerations have these decisions by other States been based on? And how can the taking into account of any such extrinsic factors not violate the very sovereignty so fiercely defended in the legal positivist interpretation? How can the current government of a State bind future administrations, which are liable to change, without compromising the sovereignty of that future government? All of these questions and problems highlight the difficulty which arises from a strict insistence on state consent and sovereignty in international law.

It may be possible to reduce the dilemma by reinterpreting the sovereign rights of a State as a sovereign duty, exercisable for the benefit of its citizens. ⁷⁴ Following this view, as expressed by Ronald Dworkin, a State gains legitimacy by improving the conditions in which its citizens live, and this includes advancing the stability of international relations, peace and security. ⁷⁵ Thus, States may be deemed to have a duty to protect their own citizens, and promote a peaceful, stable and effective international order. States may also be subject to an onus to intervene, in order to prevent human rights abuses and war crimes, genocide etc. in other countries, and to cooperate internationally 'to prevent economic, commercial, medical or environmental disaster.' ⁷⁶ A similar, if slightly less defined approach had also been taken by early liberal internationalists, who viewed international

⁷³ Many of these points are made by Dworkin, *ibid*, at 8-10, who derives at the conclusion that States enter into political obligations and engage in international law as a result of 'political morality' (*id.* 11-12), rather than because the 'self-limiting consent' of States is the basis of international law (*id.* 10).

⁷⁴ See also Dworkin, *ibid*, 16-18. Admittedly, this interpretation may be more acceptable in democracies than totalitarian regimes or absolute monarchies, based on the notion that a State's government in a democratic system is elected by and on behalf of its citizens.

⁷⁵ Dworkin, *ibid*.

⁷⁶ *Ibid*, 18.

law as a product of a re-evaluation of society as 'directed by a kind of cosmopolitan purpose'. 77 Citing the Swiss jurist Johann Bluntschli, Collins describes the approach as follows:

[I]nternational law concerned the relations between states, but states embodied a collective will of their people, which in turn pointed with 'inner necessity to the higher unity of mankind of which the nations were only members'.⁷⁸

According to this view, international law may manifest itself through States' observation of customs and conclusion of treaties, however it is the result of a 'common juridical conscience'. ⁷⁹ In other words, it is this underlying communal conscience, which affords the State legitimacy as a participant and source of international law in the first place, rather than mere consent of state officials.

Admittedly, the references to a 'higher unity of mankind' and a 'common conscience' do contain moral undertones and therefore are highly likely to be subject to critique, particularly by classical positivists who deny the relevance of any moral considerations in law.⁸⁰ The argument presented here is, nevertheless, that international law, which attempts to regulate behaviour between States and problems affecting the global community at large, is subject to an outward-looking political morality or duty, rather

⁷⁷ Collins (2014) (n. 51), 31.

⁷⁸ *Ibid*, citing Johann C. Bluntschli, *The Theory of the State*, English translation from the 6th German edn (Clarendon 1885), 25.

⁷⁹ Collins, *ibid*, 31. The term 'common juridical conscience' is a direct translation of 'la conscience juridique commune', a term coined by Alphonse Rivier, *Principes du Droit des Gens*, Tome I (Arthur Rousseau 1896), cited by Collins, 27.

⁸⁰ See Kelsen and Hart (both n. 49).

than the inward-looking, subjective morality commonly rejected in traditional positivism. Following this view, as expressed by John Tasioulas, the legitimacy of states, as well as international law itself, is regarded as:

[D]ependant on its furtherance of certain values, e.g. human rights, peaceful coexistence, environmental protection, etc. as tempered by the requirements of the rule of law (predictability, liberty, etc.) and the values of political participation (including the political self-determination of states).⁸¹

Consequently, if States derive their legitimacy and legal capacity from the collective will and benefit of the people, state consent can be a source of international law only if granting it in a particular treaty or situation represents the collective will.⁸² Thus, states can grant valid consent and bind themselves legitimately at international law only if this reflects the people's benefit.⁸³

These arguments echo Dworkin's reinterpretation of States' sovereignty as a sovereign duty, exercisable for the benefit of the people with the aim of improving international stability, peace and security.⁸⁴ Developing this point for the purpose of the discussion in this thesis, it is also tenable that States actually have a duty to eradicate the most heinous

The point of legitimacy of internal state decision-making as a test of valid state consent at international level has been made, for instance, by Capps (n. 61) 218.

⁸¹ See John Tasioulas, 'Customary International Law and the Quest for Global Justice', in Amanda Perreau-Saussine, James B. Murphy (eds), *The Nature of Customary Law* (CUP 2007) 307, 329.

⁸² For support, see Dworkin (2013) (n. 43) 18.

⁸⁴ See Dworkin (2013) (n. 43).

international crimes known to mankind,⁸⁵ including the crime of aggression. Consequently, a State's failure to grant consent to the jurisdiction of the International Criminal Court over the crime of aggression may even be seen as a failure to exercise that sovereign duty.⁸⁶

The specific provisions prohibiting and criminalising aggression can also be viewed as 'necessary rules of coexistence' and 'principles of minimum world public order'. ⁸⁷ This terminology was applied in a number of publications by Myres McDougal and Florentino Feliciano in the late 1950's and early 1960's against the background of the development of nuclear and biological weapons and serious polarisation of the post WWII world into power blocs. Accordingly, McDougal and Feliciano state that '[t]he common interest which sustains the law of war is the interest of all participants in economy in the use of force – in the minimization of the unnecessary destruction of values. ⁸⁸

The 'economy in the use of force', including the prescription of aggression, is thus viewed as a generally applicable rule of coexistence, which is in the interest of the international community at large, and consequently transferred into a duty of States. McDougal and Feliciano view the law of war including the prescription against aggression as essential

-

⁸⁵ The Rome Statute refers to the 'most serious crimes of concern to the international community as a whole', and 'unimaginable atrocities that deeply shock the conscience of humanity'. (See Preamble, Rome Statute (n. 1).

⁸⁶ Admittedly, this discussion raises the problem of who is to be the judge of a State's decisions. Defining the benefit to the people may not be as straightforward as it appears. As Guzman states: 'We live in a world of many States with wildly divergent priorities and concerns.' (2011) (n. 25) 6. For the purpose of this discussion, however, it is contended that the eradication of the worst international crimes is beneficial to the people. See also Dworkin's discussion of states' duty and assessment of benefit by the international community (below at n. 97 and related text).

⁸⁷ Myres S. McDougal and Florentino P. Feliciano, *Law and Minimum World Public Order: The Legal Regulation of International Coercion*, (YUP 1961).

⁸⁸ Myres S. McDougal and Florentino P. Feliciano, 'International Coercion and World Public Order: The General Principles of the Law of War' (1958), 812, https://digitalcommons.law.yale.edu/fss papers/2451/, accessed 25 November 2020.

to ensuring 'the preferred goals of maximum human dignity and minimum destruction of human values.'89 The *status quo* of decentralisation of effective power, and unilateral state supremacy in each State's attempt to retain the maximum power and protection for itself, presents a problem. 90 Thus, States view themselves as competent to take any measures subjectively deemed appropriate in their defence or that of allies, and this problem has to be surmounted in order to guarantee a public world order devoid of aggression. Instead of allowing nations to resort to such unilateral exercise of power, McDougal and Feliciano propose inter-state co-operation and specialised central international institutions to develop and administer rules on 'unlawful coercion', for the benefit of the community as a whole. 91 In order to achieve such a public world order of freedom, an equilibrium of power, and complementary humanitarian policies, international authority has to be organised and centralised, although States are still deemed to be the principal actors at international law. 92 One may ask how such a centralisation of international authority can be compatible with the autonomy of States, however Dworkin's reinterpretation of sovereignty as a duty of cooperation may provide a solution.⁹³

This discussion attempts to make a case for the sovereign duty of States not to deny participation in the international legal process by failing to grant consent, the problem at the heart of the jurisdictional limitations placed on the crime of aggression. Nevertheless, it cannot be denied that it provides an imperfect solution only, leaving the debate still largely unresolved. The concept of an international legal system is very young compared

-

⁸⁹ *Ibid*, 816.

⁹⁰ *Ibid*.

⁹¹ Ibid.

⁹² *Ibid*, 812.

⁹³ See Dworkin (2013) (n. 43).

with the traditional notion of sovereignty as a right of States.⁹⁴ Moreover, compliance with and enforcement of international legal obligations remains a problem in a system originally developed on a voluntarist basis.⁹⁵ Notwithstanding these difficulties, the reinterpretation of the sovereign right to grant consent as a duty to cooperate internationally, where it benefits the citizens of the State in question, may require at least an exceptionally good justification of why consent is being withheld. Thus, political pressure could be used to support and at least partially enforce an international system, requiring less than cooperative States to provide justification for their actions.⁹⁶ The benchmark would be the benefit to the people. Admittedly, this raises the problem as to who should be the arbiter of any existing benefit to the people. Dworkin appears to suggest that the matter should be decided by the international community at large:

If a significant number of states, encompassing a significant population, has developed an agreed code of practice, either by treaty or by other form of coordination, then other states have at least a prima facie duty to subscribe to that practice as well, with the important proviso that this duty holds only if a more general practice to that effect, expanded in that way, would improve the legitimacy of the subscribing state and the international order as a whole. If some humane set of principles limiting the justified occasions of war and means of waging war gains wide acceptance, for instance, then the officials of other pertinent nations have a duty to embrace and follow that set of principles.⁹⁷

⁹⁴ This is explored in detail in Chapter 1.

⁹⁵ 'If no state can be forced to cooperate, they will all have a reason not to participate.' (Dworkin (2013) (n. 43) 18. For a view that even voluntary consent does not ensure compliance of States, see Guzman (2011) (n. 25) 10.

⁹⁶ See Guzman, *ibid*, 8-10.

⁹⁷ See Dworkin (2013) (n. 43) 19.

This view develops an ideal image of a widespread political morality based on utility for citizens and improvement of values which are deemed to be globally prominent. ⁹⁸ These values include peace, security, and economic stability amongst others. ⁹⁹ As attractive as this vision may be, it does gloss over the fact that state interests, in practice, do not always (some might say very rarely) converge, particularly when it comes to resources or territorial claims. ¹⁰⁰ The benefit to a State's nationals may be quite different to a benefit to the global community, and States would be likely to claim that their sovereign duty exists primarily towards their own citizens. ¹⁰¹ Additionally, the judges of benefits to the people appear to be the community of States rather than the citizens themselves, whether national or international.

Consequently, agreement on what constitutes 'benefit' may be problematic, whoever determines its existence. The attempt to dismantle state sovereignty is therefore inexorably imperfect. Admittedly, it is easier to criticise state centrism in international law than to suggest a perfect, unassailable alternative, but that does not mean that the *status quo* should not be questioned. If one accepts that international law *is* politics, law, just like politics, has to be capable of moving with the times and develop in response to contemporary challenges, rather than simply insist on the fixture of consent of the executive of States as a sole source of legitimacy of legal rules.

⁹⁸ *Ibid*, 19. This is what Dworkin calls the 'principle of salience'.

⁹⁹ See McDougal and Lasswell (n. 63) and McDougal and Feliciano (1958) (n. 88).

¹⁰⁰ See for instance Guzman's observations, (2011) (n. 25) 6, and comments in (n. 86).

¹⁰¹ Ibid.

¹⁰² See Kammerhofer's and d'Aspremont's encouragement of this debate (2014) (n. 28).

¹⁰³ Koskenniemi (1989) (n. 64), see also Martti Koskenniemi, *The Politics of International Law* (Hart 2011), Preface I, p. V.

This discussion raises the important question whether international law's obsession with state consent derived from classic positivism is still viable, particularly when prescribing and attempting to eradicate the worst international crimes. International criminal law is concerned with a limited list of international crimes of such gravity that they pose a threat to 'world public order'. In order to create deterrence, these crimes require regulation, enforcement and prosecution, as part of a new and developing cooperative process on a wide international level, rather than the prevailing insistence on state sovereignty as an absolute right, and the requirement of individual state consent. ¹⁰⁴ In a plea to overcome such a static approach in favour of new processes, policies and the interpretation and application of global values, McDougal and Feliciano sum up:

It is by the consistent maintenance of these emphases, in the employment of all his special intellectual skills for inquiry into the basic specific problems of international coercion, that the lawyer who values human dignity may make his richest contribution to moving the general community of mankind from its present incomplete and disorganized structures and processes of authority, harassed by contending exclusive orders, toward a more inclusive world public order in which the values of a safe, free and abundant society are honoured not merely in theory but in practice. ¹⁰⁵

-

¹⁰⁴ Derived from classical legal positivism, which is described by McDougal and Feliciano (1958) as 'the repetitive reiteration of overoptimistic faith in inherited ambiguous technicalities' (n. 88) 844. ¹⁰⁵ *Ibid.* 845.

To that end, the desideratum of a sovereign duty incorporating a certain political morality, and requiring justification of any decisions which affect the international community detrimentally (including the denial of consent to jurisdiction of the ICC generally, and over the crime of aggression specifically), may eventually come to replace a jealously guarded consent system. In the face of increasingly common global crises and shared problems, ¹⁰⁶ States may yet submit either voluntarily or out of necessity to greater cooperation even if the advantage gained is international rather than domestic. In the meantime, even the mere contemplation of States' political duty to cooperate towards the achievement of improvements, and the need for justifications if they refuse, may be able to assert some influence on state behaviour, or at least confirm the existence of a rule. This inadvertent effect was considered by the ICJ in the *Nicaragua* case:

If a state acts in a way *prima facie* incompatible with a recognized rule, but defends its conduct by appealing to exceptions or justifications contained within the rule itself, then whether or not the state's conduct is in fact justifiable on that basis, the significance of that attribute is to confirm rather than to weaken the rule.¹⁰⁷

Thus, the perception of a duty and a justification requirement may act as a persuasive tool, even if there is no absolute compliance pull. In addition to these contemplations,

-

¹⁰⁶ The United Nations Foundation has identified a number of global issues of concern in 2021, including, amongst others, extreme poverty, food and water insecurity, growing violence and a lack of access to healthcare leading to a prediction of 2021 being 'the worst humanitarian crisis year since the beginning of the United Nations'. All of these problems require substantial global cooperation. See the United Nations Foundation, 'Five Global Issues to Watch in 2021', available at http://www.unfoundation.org/blog/post/five-global-issues-to-watch-in-2021/, accessed 5th February 2021.

there may be one other concept, also based on customary international law, which could be capable of circumnavigating the problem of the state consent requirement.

A different approach to circumventing the consent problem? - The concept of universal jurisdiction

The discussion, so far, has considered the waiving of a state consent entitlement based on a reinterpretation of States' sovereign rights as a sovereign duty, the joint responsibility to create an international system which maintains a 'minimum world public order', and the more pragmatic suggestion to exert diplomatic and political pressure and influence on States to consent to jurisdiction of the Court. There remains one other concept which, at least theoretically, may provide a solution to the problem of state consent to criminal jurisdiction in the international arena. To that end, it may be possible to invoke the customary law concept of universal jurisdiction.

The term 'universal jurisdiction' refers to jurisdiction exercised by States over crimes of universal interest, offending universal public policy and attracting universal condemnation. Universal jurisdiction entitles any State to legitimately prosecute a crime in domestic courts in the interest of all States. The first assumption of this concept is that the prosecution of universally deplored crimes is of interest to every single State, as offending all notions of a common perception of humanity. The second element of domestically exercised universal jurisdiction is the exceptional possibility to disregard

.

¹⁰⁸ The perpetrators are deemed to be enemies of all humankind, or 'hostis humanis generis'. See Sharon Williams, 'The Rome Statute on the International Criminal Court – Universal Jurisdiction or State Consent – To Make or Break the Package Deal' (2000) in Michael N. Schmitt (ed), 'International Law across the Spectrum of Conflict', 75 International Law Studies, US Naval War College, 539, 544.

the territory on which the crime occurred, irrespective of where it was committed or the nationality of the perpetrator or the victim. Thus, under universal jurisdiction, States are perceived to have 'the legal competence and jurisdictional competence to define and punish particular offences, regardless of whether that State had any direct connection with the specific offences at issue. The consent to jurisdiction by the State, of which the victim or perpetrator are citizens, is immaterial. Consequently, the concept of universal jurisdiction is capable of avoiding 'a jurisdictional vacuum'.

Traditionally, the most prominent crimes prosecutable in this way were piracy¹¹³ and slavery.¹¹⁴ More recently, the concept is also deemed to include jurisdiction over genocide, crimes against humanity, war crimes, and torture.¹¹⁵ Universal Jurisdiction has also been cited as one of the bases of jurisdiction for the Nuremberg Tribunals,¹¹⁶ granted by states parties to the London Charter,¹¹⁷ who vested their combined domestic

¹⁰⁹ Kenneth. C. Randall, 'Universal Jurisdiction under International Law', (1988) 66 Tex. L. Rev., 758, 788. See also the discussion in Michael. P. Scharf, 'Universal Jurisdiction and the Crime of Aggression', (2012) 53 (2) Harv. Int'l L. J., 358.

¹¹⁰ Christopher Joyner, 'Arresting Impunity: The Case for Universal Jurisdiction in Bringing War Criminals to Accountability' (1996), 59 LCP 153, 167. Jurisdiction over the person appears to be merged with jurisdiction over the offence, see H. Lauterpacht, 'The Law of Nations and the Punishment of War Crimes' (1944), 21 BYBIL 58, at 61.

¹¹¹ See Williams (2000) (n. 108) 545.

¹¹² *Ibid*, 544.

¹¹³ See for instance *United States v Smith*, [1820] 18 US (5 Wheat.) 153. See also *United States v Klintock*, [1820] 18 US (5 Wheat.) 144. See also the recent resort to universal jurisdiction of the Seychelles over Somali pirates arrested by third states in *Republic v Mohamed Ahmed Dahir & Ten* (10) [2010] SCSL. 81 (Sey.).

¹¹⁴ See L.C. Green, *International Law* (1984), 179.

https://www.coalitionfortheicc.org/news/20170519/icc-universal-jurisdiction-two-ways-one-fight, accessed 20th January 2021. For support that genocide, war crimes and crimes against humanity are customary international law prohibitions subject to universal jurisdiction, see *The Prosecutor of the Tribunal v Dusan Tadic*, (hereinafter the *Tadic* case), Case No. IT-94-I-T, Trial Chamber Decision on Jurisdiction, 10 August 1995, paras 46-52. For judicial statements that torture has developed into a *jus cogens* norm subject to universal jurisdiction, see *Prosecutor v Furundzija*, Tribunal for Former Yugoslavia, Case No. 17-95-17/1-T (1998), para 153, and Lord Browne Wilkinson's summary of the majority judgement in *R v Bartle et al.*, *ex parte Pinochet*, [1999] UKHL, 38 ILM, paras 581, 589.

¹¹⁶ The IMTs at Nuremberg and successor trials by the NMTs are considered in detail in Chapter 1.

¹¹⁷ London Charter of August 8, 1945, 82 UNTS, 279, Article 6 (a).

jurisdiction in the international tribunals.¹¹⁸ Consequently, the judiciary at Nuremberg relied on universal standards, established through customary international law, which allowed them to judge aggression committed by the officials of sovereign States irrespective of state consent or national laws.¹¹⁹ They therefore first introduced the equivalent of an 'international rule of law',¹²⁰ independent of nations' consent. The successor tribunals, established under Control Council Law No. 10,¹²¹ also considered their jurisdiction to be universal under customary and treaty law, stating:

[A]n international crime is [...] an act universally recognized as criminal, which is considered a grave matter of international concern and for some valid reason cannot be left within the exclusive jurisdiction of the state that would have control over it under ordinary circumstances.¹²²

¹¹⁸ See Trygve Lie, *UN Secretary-General's Report on the Nuremberg Trials*, UN Doc. A/CN.4/5, UN Sales No. 1949 V.7, at 80 (1949). See also the statement in the Nuremberg judgment (n. 3) that the allied powers had 'done together what any of them might have done singly; for it is not to be doubted that any nation has the right thus to set up special courts to administer law.'

aurrender following defeat. See Madeline Morris, 'High Crimes and Misconceptions: The ICC and Non Party States', (2001) 64 LCP, 13, at 36-7. Note that the International Military Tribunal for the Far East at Tokyo, on the other hand, had the express consent to jurisdiction of the Japanese government (Morris, 37). For the view of Nuremberg establishing a universal 'international rule of law' by international tribunals, see for instance Jack Goldsmith's article 'The Shadow of Nuremberg' in the New York Times of January 20th, 2012, reviewing William Shawcross' book *Justice and the Enemy - Nuremberg*, *9/11, and the Trial of Khalid Sheikh Mohammed*. http://www.nytimes.com/2012/01/22/books/review/justice-and-the-enemy-nuremberg-9-11-and-the-trial-of-khalid-sheikh-mohammed-by-william-shawcross-book-review.html, accessed 15 October 2020.

¹²¹ 'Control Council Law No. 10', Punishment of Persons Guilty of War Crimes, Crimes Against Peace and Against Humanity (1945), (reprinted in Telford Taylor, Final Report to the Secretary of the Army on Nuremberg War Crimes Trials, under Control Council Law No. 10, 250-153 (1949)).

¹²² In re List, 11 Trials of War Criminals 759 (1946-49) (US mil. Trib. Nuremberg 1948), at p. 1241. For similar references to universal jurisdiction, see for instance the *Hadamar Trial*, 1 Law Reports of Trials of War Criminals 46 (1949) (US Mil. Commission – Wiesbaden 1945), p. 53, and the so-called *Einsatzgruppen* case, *United States v Otto Ohlendorf*, (1950) IV Trials of War Criminals Before the Nuremberg Military Tribunals Under Control Council Law No. 10, at p. 462. For a discussion of universal jurisdiction in the Nuremberg tribunals see also Scharf (2012) (n. 109) at pp. 375-9.

The consent of the German aggressor state to the prosecution of its former leaders was neither assumed to exist nor considered to be necessary. ¹²³ Instead, universal jurisdiction was derived from the number and variety of States participating in the prosecution. ¹²⁴ The Nuremberg precedent may, consequently, provide support for the exercise of universal jurisdiction over aggression at customary law. Following this argument, ratifying member states of the Rome Statute ought to be able to vest their domestically exercisable jurisdiction in the Court, as *the* appropriate organ to exercise universal jurisdiction, just as the individual signatories to the London Charter were deemed to have vested their individual jurisdiction in the tribunals. ¹²⁵

In the lead up to the Rome Conference, universal jurisdiction was considered as one of the paths for the ICC's jurisdiction over all of the four crimes in Article 5 in a proposal by Germany. 126 Sharon Williams observed:

The view central to this proposal was that to limit the potential of the ICC by requiring some form of state consent beyond ratification would detract from the effectiveness of the Court and even its rationale and philosophical underpinnings. Thus, the impact of the German proposal would have been to give the ICC universal jurisdiction over the listed crimes with no need for a separate consent of

¹²³ See Roger S. Clark, 'Nuremberg and Tokyo in Contemporary Perspective', in Timothy L. H. Mc Cormack & Gerry J. Simpson (eds), *The Law of War Crimes, National and International Approaches* (Kluwer 1997), 172. For a different view of automatic consent due to the unconditional surrender of Germany, see Morris (n. 119).

¹²⁴ 19 States took part in the Nuremberg prosecutions and vested their individually exercisable jurisdiction in the tribunals under universal jurisdiction. See Clark, *ibid*, and Chapter 1.

¹²⁵ See Williams (2000) (n. 108), 545-546, and 559 at n. 46.

¹²⁶ 'German Proposal of 1998', placed before delegations in the Committee of the Whole and contained in the Draft Statute for an International Criminal Court, UN Doc. A/Conf. 183/2/Add.1, Articles 6 (b), 7, 9 (further option, and further option for Article 7. Available at https://www.documents-dds-ny.un.org, last accessed 07 December 2020.

interested States. As Germany indicated in Rome, the universal principle's application would have eliminated loopholes.¹²⁷

The proposal for universal jurisdiction of the ICC failed to attract sufficient support in Rome, resulting instead in the inclusion of Article 12, which established the preconditions to jurisdiction of the Court. 128 The crime of aggression, however, still required its own definition and jurisdictional conditions, and the subject was therefore not entirely closed. The possibility of domestic prosecutions and potential delegation of universal jurisdiction to the ICC sans state consent continued to be a matter of great concern particularly to the observer state of the United States of America during the negotiations. ¹²⁹ Consequently, the US insisted on the inclusion of an understanding that the amendments 'shall not be interpreted as creating the right or obligation to exercise domestic jurisdiction with respect to an act of aggression committed by another State.' This understanding attached to the Kampala aggression amendment, attempts to expressly exclude the crime of aggression from the exercise of such universal jurisdiction by States. Nevertheless, according to Amnesty International's 2012 report, 163 out of 193 UN Member States claim the domestic exercise of universal jurisdiction over one or more crimes under customary international law, 131 and the matter may therefore not be as closed as it appears. 132 Admittedly, like the other arguments introduced above, this view is

¹²⁷ Williams, (2000) (n. 108) 545.

¹²⁸ Article 12 of the Rome Statute (n. 1). This article is discussed in detail in Chapter 5.

¹²⁹ See for instance Kress & von Holtzendorff (2010) (n. 5) 1216. See also Scharf (2012) (n. 109) 365.

¹³⁰ Understanding No. 4 *bis*, Annex III, Understandings regarding the amendments to the Rome Statute of the International Criminal Court on the crime of aggression, attached to Resolution 6 (n. 2).

¹³¹ Amnesty International, *Universal Jurisdiction: A Preliminary Survey of Legislation Around the World*, 2012 update, available at www.amnesty.org/en/library/info/IOR53/019/2012, accessed 15th November 2020. See also International Justice Resource Center, *Universal Jurisdiction*, available at https://ijrcenter.org/cases-before-national-courts/domestic-exercise-of-universal-jurisdiction/, accessed 20th November 2020.

¹³² See the discussion in Chapter 5.

controversial, since it consciously de-emphasises the state consent requirement, the pervading problem for wider jurisdiction of the Court.

The mainstream view prevails that state consent constitutes the source of international law, and any attempt to dismantle this consent requirement is undoubtedly contentious and likely to encounter objections. In the face of this strong and traditional resistance, any solutions attempted in this thesis cannot claim to be perfect and unassailable themselves, despite the extent of the research and the pursuit of such a solution with genuine passion. Consequently, a bittersweet caveat is expressed that any suggestions for extension of the Court's jurisdiction are, inevitably, as imperfect as the situation they are attempting to address. This does, however, not detract from the fact that the retention of a greatly reinforced state consent requirement in the crime of aggression provisions practically encourages non-membership, non-ratifications, and opting out even post-ratification. State consent is also a potential obstacle to Security Council referrals on account of its internal voting structure, based on the fact that members are unlikely to vote for a referral, where their interests or those of an ally may be affected adversely. P5 members, in particular, have the capacity to frustrate a referral to the Court in a given situation, potentially for good, through use of the 'veto'. 133 As a result of the state consent problem, deterrence from the commission of the infant crime, and effectiveness of the Court itself are weakened immeasurably.

¹³³ The veto power and internal Security Council structure is discussed in detail in Chapter 1.

5. Outline of the analysis

This work is divided into six core chapters and a conclusion. Chapter 1 provides an indepth consideration of the roots of the crime of aggression. This focuses on the historical development from the traditional perception of a sovereign right to go to war to the prohibition of the use of force in Article 2 (4) of the Charter of the United Nations, ¹³⁴ the ground-breaking post-World War II prosecutions for the 'supreme international crime' of aggression, ¹³⁵ and the attempts of (and resistance to) achieving a definition of the crime since.

Chapters 2 and 3 examine the provisions contained in Article 8 *bis* of the 2010 Definition of the Crime of Aggression. The particular features to be investigated in Chapter 2 are the threshold an act of aggression must surpass in order to give rise to criminal liability, the limitation of possible perpetrators to political and military leaders only and the culpable conduct itself. Chapter 3 examines the state centricity of the definition and the list of aggressive acts based on traditional warfare. To what extent do these definitional limitations restrict the Court's jurisdiction? How would a wider definition have extended the jurisdiction of the Court, and why was there such a great level of resistance to such an extension?

Chapters 4 and 5 will consider the limitations placed expressly on the Court's jurisdiction in Articles 15 *bis* and *ter*, which are essentially consent-based. Chapter 4 first dissects the

¹³⁴ United Nations, Charter of the United Nations, 24 October 1945, 1 UNTS XVI, Article 2 (4), discussed in Chapter 1.

¹³⁵ Robert H. Jackson, 'Opening Statement before the International Military Tribunal', 21 November 1945, Jackson, Robert H., *United States Representative to the International Conference on Military Trials*, HMSO, (London 1945), 305.

jurisdictional conditions for both articles, providing the foundation for the discussion. It then examines the particular relationship between the Security Council and the Court, and States' resistance to independent jurisdiction, which resulted in wide control of the Council over criminal prosecutions. Was this necessary? How does it compare to the other crimes of the Statute? And is the high level of control of a political organ, subject to the consent of its members, particularly the P5, over prosecutions of a crime desirable? How is this feature likely to affect the efficiency of such prosecutions? The second of the jurisdictional chapters (Chapter 5) examines the specific consent-related restrictions placed on the Court's jurisdiction in Article 15 bis (4) and (5). These include nonjurisdiction over non-consenting states parties and non-jurisdiction of the Court over nonsignatories, unless there is a Security Council referral. States can withhold their consent to the Court's jurisdiction in three ways, first by remaining non-members, second by becoming members, but failing to ratify the aggression amendments, and third by opting out of jurisdiction, despite being a member and having ratified. Additionally, the jurisdiction of the Court has been restricted even more through the application, and most recently the narrowest possible interpretation, of Article 121 (5) of the Rome Statute. 136 This avoidable decision by the 2017 ASP to apply the negative understanding of Article 121 (5) requires the threefold consent to be granted by both, aggressor and victim state. These jurisdictional conditions are extraordinary and again raise the question: Why the insistence on such extraordinary restrictions? And how effective can the Court's jurisdiction possibly be in the face of these unusual and far-reaching limitations?

¹³⁶ Through the decision by the 2017 Assembly of States Parties (ASP) to interpret application of Article 121 (5) in the narrowest possible way. See Chapter 5.

Finally, Chapter 6 will attempt to make suggestions on how to widen the Court's jurisdiction, based on the analysis of the restrictions in the previous chapters. Much of this may remain a wish list on account of the practical difficulties likely to be encountered. Nevertheless, the contemplation and promotion of at least some of these improvement suggestions is highly necessary to develop and retain relevance of the new crime in modern aggression scenarios, and aid deterrence. It is argued that only if the Court allows itself wider interpretation powers to counteract at least some of the restrictions, will there be any chance of successful prosecutions. A relaxation of the state consent requirement at international law is fundamental to these improvement suggestions. Less controversially, the promotion of wider membership of the Court, particularly of all of the permanent members of the Security Council, and ratifications without reservations, ought to be pursued in order to advance deterrence from the commission of aggression and eradicate the perception of impunity for one of the worst international crimes.

CHAPTER 1: AGGRESSION IN A WORLD OF SOVEREIGN STATES – THE HISTORICAL AND GEOPOLITICAL CONTEXT

1. Introduction

The Kampala consensus on a definition of the Crime of Aggression and jurisdictional conditions represents a significant breakthrough for international criminal law, ¹³⁷ establishing jurisdiction by the first permanent international criminal court over the illegal use of force. It is also remarkable because of the intrinsic difficulty of arriving at this consensus in light of the historical and geopolitical context, particularly of a 20th century marred by two harrowing World Wars and a lingering Cold War, in addition to continuing disaccord amongst States. ¹³⁸ This historical context sets the scene for the discussion, central to this thesis, of the limitations placed on the jurisdiction of the Court. It also provides an insight into the difficulties international institutions have in the past encountered in their censorship of state acts of aggression, mostly because of a requirement of state consent to such condemnations.

¹³⁷ Kress and von Holtzendorff (2010) (n. 5) 1179. See also Trahan (2011) (n. 5) 50.

¹³⁸ Benjamin Ferencz, *Defining International Aggression, The Search for World Peace* (Oceana 1975) II, 5. See also Antonio Cassese (2013) (n. 12) 138.

The following discussion traces the change in States' perceptions from an inherent entitlement to go to war, subject to some form of justification or authorisation, to the widely practiced exercise of a virtually unfettered right to commit aggression and subsequently to emerging notions of aggression as a punishable international crime in the first half of the 20th century.

2. From the *jus fetiale* to the *jus ad bellum* – Early perceptions of aggression

The antecedents to the establishment of formal rules regulating the use of force in the relations between States, known as the *jus ad bellum*, can be traced to at least as far back as the conduct of the Roman Empire. Throughout the Roman Empire, the rules of hostilities had been governed by the *jus fetiale*; legal rules established by a body of priests. Authorisation was necessary for any use of force, and it was provided by these priests if there were 'sufficient grounds justifying the outbreak of hostilities (e.g., violation of a treaty or the sanctity of ambassadors, infringement of territorial rights or offences committed against allies)'. He were sufficient grounds and authorisation had been granted, the perception of the hostilities was one of a 'just war' (*bellum justum*), as opposed to an 'unjust war' (*bellum injustum*), or the first concept of an act of aggression.

¹³⁹ See Arthur Nussbaum, A Concise History of the Law of Nations (Macmillan, 2nd edn 1954), 10-11.

¹⁴⁰ Yoram Dinstein, *War, Aggression, and Self-Defence* (CUP 5th edn 2011), 65, quoting from Coleman. Phillipson, II *The International Law and Custom of Ancient Greece and Rome* II, at 182, 328.

¹⁴¹ See Nussbaum (1954) (n. 139) 11.

In the 4th century AD, St. Augustine began to develop the so-called 'just war doctrine', according to which the motivation and justification for the use of force was considered morally acceptable, as long as it was authorised in the name of God.¹⁴² In St. Augustine's world 'the resort to violence that is inherent in war is undertaken, not as a means of self-defence, but as a punitive effort initiated by lawful authority'.¹⁴³ Thus, war efforts were exercisable as a Christian duty on the authority and on behalf of a government.¹⁴⁴ A use of force subjectively authorised as morally acceptable, was therefore deemed to be lawful rather than constituting an unacceptable act of unlawful aggression.¹⁴⁵

Christian theology followed St. Augustine's ideas and the 'just war' theory was subsequently developed further in order to justify the Crusades and later on the discovery and often violent subjugation and colonisation of the New World in the name of Christianity, particularly the Americas. In the 13th Century, St. Thomas Aquinas, whilst admitting that war was contrary to peace, propounded that peace was not always a just order worth preserving. After Moreover, he argued that war could be a means to achieve a real ('just') peace and a means to break a false ('unjust') peace. Thomas Aquinas elaborated on the existing 'just' war theory by establishing three principles. The first principle asserted that war could only be declared by the sovereign authority. The second principle necessitated a 'just' cause for the war, i.e. a fault element by the attacked State. The aim of the use of force was therefore to avenge a wrong, punish a nation, and/or

¹⁴² See St. Augustine, *Contra Faustum Manichaeum*, around 400 AD, in John Langan, 'The Elements of St. Augustine's Just War Theory' (1984), 12 The Journal of Religious Ethics (1), 20, available at www.jstor.org/discover/10.2307/40014967, accessed 3rd March 2020.

¹⁴³ *Ibid*.

¹⁴⁴ *Ibid*.

¹⁴⁵ *Ibid*, 22.

¹⁴⁶ St. Thomas Aquinas, *Summa Theologiae* (XXXV Blackfriars 1972) 80-3.

¹⁴⁷ *Ibid*.

restore what had been seized 'unjustly'. The third element required a 'just' intent, i.e. to avoid harm or advance 'the good'. 'The good' here refers to a good result in the sense of a morally justifiable result. Whilst these early principles insisted on some form of moral justification in order to establish legitimacy of the use of force, to be distinguished from aggression, any assessment of these criteria was inevitably subjective and concluded by the sovereign of the State who intended to go to war. 149

Over the centuries, theologians, philosophers and jurists expanded on the permissible causes of war in an attempt to justify their rulers' use of force. Instead of generally restricting prolific use of force, the 'just war doctrine' therefore simply insisted on subjective justification. Thus, Grotius and Gentili were accepting of the fact that 'just' war could be waged on both sides, depending on the opponents' perceptions. This developing view acknowledged subjectivity as inevitable in individual assessments of a situation and signified a change in attitude towards the use of force, placing increasingly less emphasis on the need for objective justification. This led Thomas Hobbes to come to the conclusion that the state of nature was generally a state of war, and peace merely arose from the need for self-preservation. The prolific use of force, based on a broad range of subjective grounds or none at all, blurred the distinction between 'just' (or justified) wars and wars of aggression.

¹⁴⁸ *Ibid.* See also Anthony C. Arend & Robert J. Beck, *International Law and the Use of Force: Beyond the UN Charter Paradigm* (Routledge 1993), 14.

¹⁴⁹ St. Augustine in Langan (1984), 1887:301.

¹⁵⁰ Ibid.

¹⁵¹ Grotius, De Jure Belli ac Pacis II (Clarendon 1925), para XXIII, XIII, 565-6.

¹⁵² Gentili, De Jure Belli I, para VI, 48-52 (Clarendon 1933), para VI, 31-3, 48-52.

¹⁵³ Thomas Hobbes, *Leviathan*, (Touchstone 2008), Ch. 13.

By the end of the 19th and the very early 20th century, States were embarking on war freely. In the words of Dinstein: '[O]nce war qualifies as objectively 'just' on the part of both adversaries, there is scarcely a reason why any State should feel inhibited from going to war at will'. ¹⁵⁴ The diminishing emphasis on the need for justification to go to war therefore led to the use of aggressive force becoming more and more widespread, at a time when the science of warfare also developed exponentially, enabling wars to be fought on a much larger scale and ultimately leading to the two greatest wars of aggression of the 20th century.

3. Attempts to curb aggression in the first half of the 20th century

3.1 The creation of the League of Nations

Colonial expansion, the unification of Germany in the Second Empire under Bismarck with extensive militarisation, and shifting alliances between the European Powers led to mounting international tension towards the end of the 19th century. In an attempt to obtain some form of security from acts of aggression, several States began to consider entering bilateral agreements on dispute settlement, however such treaties were subject to time limits and had no effect on third parties. The signatories to the multi-national Hague Conventions of 1899¹⁵⁶ and 1907¹⁵⁷ agreed to resort to mediation before

¹⁵⁴ Dinstein, War, Aggression, and Self-Defence (6th edn CUP 2017), 71.

¹⁵⁵ David Stevenson, 'Europe before 1914', British Library archives, https://www.bl.uk, accessed 4th March 2020.

¹⁵⁶ Hague Convention (I) of 1899 for the Pacific Settlement of International Disputes (29 July 1899).

¹⁵⁷ Hague Convention (II) of 1907 for the Pacific Settlement of International Disputes (18 October 1907).

embarking on aggressive war.¹⁵⁸ Nevertheless, these attempts to limit the outbreak of hostilities were unable to prevent the First World War. The atrocities of the First World War were so shocking to the world that the victors established the *Commission on the Responsibility of the Authors of the War and on Enforcement of Penalties* in order to determine responsibility for the acts of aggression committed.¹⁵⁹ In its report to the Preliminary Peace Conference, the Commission stated:

The premeditation of a war of aggression, dissimulated under a peaceful pretence, then suddenly declared under false pretexts, is conduct which the public conscience reproves, and which history will condemn, but by reason of the purely optional character of the institutions at The Hague for the maintenance of peace [...] a war of aggression may not be considered as an act directly contrary to positive law, or one which can be successfully brought before a tribunal such as the commission is authorized to consider under its terms of reference.¹⁶⁰

This quote demonstrates that the Commission regarded the First Wold War as a morally condemnable act of aggression, despite not offending positive international law of the time. As a result the Commission recommended future principles of international law outlawing similar acts of aggression and suggested the future criminalization of such acts, concluding that '[i]t is desirable that for the future penal sanctions should be provided for such grave outrages against the elementary principles of international law'. ¹⁶¹

¹⁵⁸ See Dinstein (2017) (n. 154), 83. See also Sergey Sayapin, *The Crime of Aggression in International Criminal Law – Historical Development, Comparative Analysis and Present State* (Asser 2014).

¹⁵⁹ Commission on the Responsibility of the Authors of the War and on Enforcement of Penalties, 1919-1920

¹⁶⁰ Report of the Commission (ibid), of 29 March 1919. See Claus Kress, 'On the Activation of ICC Jurisdiction over the Crime of Aggression' (2018) 16 (1) JICJ, 2.

¹⁶¹ Ibid.

Significantly, for the first time the trial and punishment of a sovereign, the German Emperor Wilhelm II, was sought for 'the supreme offence against international morality and the sanctity of treaties', ¹⁶² namely the crime of aggression. Despite the fact that this trial never took place, the attempt represents a watershed in history, since the international community (or at least the allied victors) perceived aggressive war-making as a punishable crime rather than a sovereign right. ¹⁶³

The First World War involved more States than ever before and resulted in a then unparalleled number of casualties. The figures vary, but are generally accepted as being in the region of 9.7 million dead soldiers, 21 million wounded, and more than 10 million civilian deaths. As a result, the international community attempted for the first time in history to limit the right of States to wage aggressive war and to create a permanent international organization dedicated to the maintenance of world peace through disarmament and arbitration. To this end, the League of Nations was created in 1919. In its covenant, member states contracted to 'respect and preserve as against external aggression the territorial integrity and existing political independence of all Members of the League'. The Covenant provided for arbitration by the League's Council in case of disputes, and members agreed to refrain from entering hostilities for at least three months following an arbitration attempt. Accordingly, Article 12 stipulates:

The Members of the League agree that if there should arise between them any dispute likely to lead to a rupture, they will submit the matter either to arbitration

¹⁶² Treaty of Peace between the Allied and Associated Powers and Germany (28 June 1919), UKTS 14, Cmd. 1746, Article 227. See the account in Cassese (2013) (n. 12), 253-5.

¹⁶⁴ Centre Europeen Robert Schuman, http://www.centre-robert-schuman.org, accessed 5th March 2020.

¹⁶⁵ Covenant of the League of Nations (1919), UKTS 4, Cmd. 153.

¹⁶⁶ *Ibid*. Article 10.

or to enquiry by the Council, and they agree in no case to resort to war until three months after the award by the arbitrators, or the report by the Council. In any case under this Article the award of the arbitrators or the judicial decision shall be made within a reasonable time, and the report of the Council shall be made within six months after the submission of the dispute.¹⁶⁷

Article 12 therefore provided for specific waiting times, amounting effectively to a 'cooling off' period. After the expiry of this, States were still free to embark on war.

Moreover, Article 15 of the Covenant reserved *inter alia* the right to resort to war, if the Council failed to reach unanimous agreement. In particular paragraph 7 stated:

If the Council fails to reach a report which is unanimously agreed to by the members thereof, other than the Representatives of one or more of the parties to the dispute, the Members of the League reserve to themselves the right to take such action as they shall consider necessary for the maintenance of right and justice. 168

Furthermore, the Council was not entitled to make recommendations or pass judgment on domestic matters, and if a dispute arose from such issues, the State in question was free to resort to any action it deemed appropriate, including war. ¹⁶⁹ Importantly, as an instrument for the prohibition of war, the Covenant of the League of Nations was therefore ineffective, since it did not outlaw aggressive wars *per se*. Instead, States' rights to resort to war were preserved subject to the said conditions. Consequently, the Covenant

¹⁶⁷ *Ibid*. Article 12.

¹⁶⁸ *Ibid*, Article 15 (7).

¹⁶⁹ *Ibid*, Article 15 (8). For commentary, see Dinstein (2017) (n. 154) 84-6.

of the League of Nations was, at best, a weak attempt at limiting aggressive wars but foremost an agreement of intention of its member states to attempt initial arbitration before resorting to the use of force. The Covenant did not contemplate individual criminal responsibility for aggression.

3.2 The Geneva Protocol on the Pacific Settlement of International Disputes

In an attempt to rectify the shortcomings and loopholes of the Covenant, the British and French Prime Ministers proposed to the League of Nations acceptance of the Geneva Protocol on the Pacific Settlement of International Disputes in 1924.¹⁷⁰ Its preamble declared aggression to be 'an international crime', ¹⁷¹ and it demanded the settlement of disputes through submission to the Permanent Court of International Justice. ¹⁷² Member states contracted 'in no case to resort to war' except where there was consent by the Council of the League of Nations or in resistance to aggression. ¹⁷³

Notwithstanding the naming of aggression as 'an international crime', no elements for individual criminal responsibility by participants were considered. In reality, the Protocol, like the Covenant for the League of Nations, addressed state responsibility, and sanctions for offending member states were limited to financial and commercial sanctions only. The Protocol also failed to regulate the resort to aggressive war by non-signatories entirely. Despite gaining preliminary approval by all 47 member states of the League of

¹⁷⁰ Geneva Protocol for the Pacific Settlement of International Disputes, 1924, 2 *Int.Leg.* 1378, 1379.

¹⁷² *Ibid*. Article 3.

¹⁷¹ *Ibid*, Preamble.

¹⁷³ *Ibid.* Article 2.

¹⁷⁴ *Ibid*, Articles 11 and 15.

Nations, the Protocol never entered into force and, as a practical result, the sovereign right of States to resort to aggressive war remained, at least technically, lawful.¹⁷⁵ The prohibition of acts of aggression thus continued to be limited to a voluntary agreement by States to arbitrate, and therefore in practice remained in a vacuum between weak and non-existent.

The Geneva Protocol, and the reasons behind the failure to ratify it, later came under scrutiny by the judges of the Nuremberg Tribunal, arriving at the conclusion that '[t]he principal objection appeared to be in the difficulty of defining the acts which would constitute "aggression", rather than any doubt as to the criminality of aggressive war'. This problem foreshadowed the same difficulties incurred by the authors of the Rome Statute in 1998 and the negotiating parties leading up to the Kampala Conference in linking individual criminal responsibility to the state act of aggression.

3.3 The Kellogg-Briand Pact

Despite the difficulty of achieving consensus, international efforts to outlaw aggressive war continued, leading to the conclusion of the so-called Kellogg-Briand Pact in 1928, ¹⁷⁸ which contained an express 'renunciation of war as an instrument of national policy' by

1

¹⁷⁵ Oona A. Hathaway and Scott J. Shapiro, *The Internationalists: How a radical plan to outlaw war remade the world* (Simon & Schuster 2017), 117-19.

Nuremberg Judgment (n. 3), page 221. This point is considered in Noah Weisbord, 'Prosecuting Aggression' (2008) 49 Harv. Int'l. L. J. 161, 164.

¹⁷⁷ See the discussion in Chapters 2 and 3 in particular.

¹⁷⁸ League of Nations, General Treaty for Renunciation of War as an Instrument of National Policy (*Kellogg-Briand Pact*), 27 August 1928, LNTS (1929) XCIV, 58.

the signatories.¹⁷⁹ This pact represented the first comprehensive attempt at a prohibition of aggressive war, and immediately before the outbreak of the Second World War in 1939, it boasted an impressive 63 Members.

Nevertheless, the Kellogg-Briand Pact did not contemplate individual criminal responsibility for the state act. Furthermore, it had several significant shortcomings and its effectiveness is questionable. Firstly, it is doubtful that the members actually abandoned their right to use of force short of actual war. 180 Secondly, the pact failed to provide an actual definition of prohibited acts of aggression. Thirdly, it also lacked provisions for enforcement or sanctions in case of violation. A fourth flaw is that self-defence was generally understood to be permissible without any parameters, thereby creating a potentially large loophole. Finally, and most importantly, it did not prevent expanding militarisation or aggressive use of force in the inter-war period, despite the strong statement of intent of the signatories to renounce war. Examples of such acts of aggression by signatories, without expressly declaring war, are Japan's invasion of Manchuria in 1931, Italy's invasion of Abyssinia in 1935, the Soviet invasion of Finland in 1939, and the German invasion of Poland in the same year. The Kellogg-Briand Pact was never terminated, however it is deemed to be superseded by the United Nations Charter. 183

¹⁷⁹ *Ibid*, Preamble, also Article 1.

¹⁸⁰ Dereck Bowett, *Self-Defence in International Law* (Praeger 1958), as quoted in D. J. Harris, *Cases and Materials on International Law*, (7th edn, Sweet & Maxwell 2010) 722. See also Dinstein (2017) (n. 154), 87-89.

¹⁸¹ See Stephen Neff, 'A Short History of International Law', in Malcolm Evans (ed), *International Law* 3rd edn, (OUP 2010) 22.

¹⁸² *Ibid.* See also Dinstein (2017) (n. 154), 87.

¹⁸³ Charter of the United Nations (n. 134). The UN Charter is discussed in detail below.

3.4 The Geneva Conference for the Reduction and Limitation of Armaments

In a separate, contemporary project, and particularly out of concern arising from continuing militarisation by States, the League of Nations together with the United States, congregated in Geneva for the Conference for the Reduction and Limitation of Armaments.¹⁸⁴ Once again, the purpose was to limit the possibility of aggressive wars, but the talks proved to be unsuccessful. States penalised by the Versailles Treaty, led by Germany, insisted on a right to equal militarisation, ostensibly in order to establish a power equilibrium as the basis of international security. France, as the main victim of previous German aggression, insisted on continuing limitations placed on German military expansion.¹⁸⁵ In light of the emphasis on equal militarisation rather than actual disarmament and joint renunciation of aggression, the failure of the negotiations is unsurprising. Nevertheless, a noteworthy draft definition of aggression emerged during these talks and the contemporary London Conventions for the Definition of Aggression in 1933, ¹⁸⁶ which was submitted by the Soviet Delegation. Art 1 of the Soviet Draft provided that:

The aggressor in an international conflict shall be considered that State, which is the first to take any of the following action:

- a) Declaration of war against another State;
- b) The invasion by its armed forces of the territory of another State without the declaration of war;

1

¹⁸⁴ League of Nations, Conference for the Reduction and Limitation of Armaments, Geneva, 1932-4.

Thomas Davies, 'France and the World Disarmament Conference of 1932-4' (2004) Dipl. Statecraft 15 (4) 765

¹⁸⁶ 1933 Disarmament Conference, 147 LNTS 67, 148.

- Bombarding the territory of another State by its land, naval or air forces or knowingly attacking the naval or air forces of another State;
- d) The landing in or introduction within the frontiers of another State of land, naval or air forces without the permission of the Government of such a State, or the infringement of the conditions of such permission, particularly as regards the duration of sojourn or extension of area;
- e) The establishment of a naval blockade of the coast or ports of another State. 187

Whilst the Soviet draft provisions failed to be adopted due to lack of consensus, the wording made several significant reappearances in the attempt to define aggression throughout the decades. Most notably this draft definition was used as the basis for the General Assembly's 1974 definition of the state act. ¹⁸⁸ The acts of aggression listed in the Soviet draft also found their way into the definition of the crime in Art 8 *bis* (2) of the Rome Statute describing the essential state acts of aggression, for which criminal responsibility arises. ¹⁸⁹

Ultimately, the inter-war era signalled the beginning of a change in attitude that 'there shall be no violence' by States. ¹⁹⁰ For the first time, comprehensive attempts to limit and outlaw aggression were undertaken. A notion of aggression as an international crime had begun to appear, particularly through the consideration of prosecution of the German Emperor ¹⁹¹ and in the wording of the Geneva Protocol, ¹⁹² however this was not pursued.

¹⁸⁷ See M. Litvinoff, Union of Soviet Socialist Republics, 'Definition of "Aggressor": Draft Declaration', Doc. Conf D/C.G.38, *Monthly Summary of the League of Nations*, XIII: 2 (1933), Annex.

¹⁸⁸ Resolution 3314 (n. 14).

¹⁸⁹ See Article 8 bis (2) Rome Statute, discussed in Chapter 3.

¹⁹⁰ Hersch Lauterpacht, *The Function of Law in the International Community*, (OUP 1933) 64, as quoted by Thomas Franck, *Recourse to Force* (CUP 2002) 1.

¹⁹¹ See n. 162.

¹⁹² See n. 170.

All of the inter-war instruments and conferences focused on the prevention of aggression and sanctions for such acts committed by States. Despite these efforts, neither the existence of the League of Nations nor the Kellogg-Briand Pact were able to prevent another war of aggression on a massive international scale, the Second World War.

4. The development of collective security and individual criminal responsibility

4.1 The creation of the United Nations and the prohibition on the use of force

The immediate aftermath of the Second World War heralded a new era. Following the harrowing and devastating outcome of this unprecedented war of aggression, a genuine agreement prevailed amongst States that a worldwide collective security system was required in order to prevent future wars of aggression. Consequently, States came together to form the United Nations with the aim of preventing aggression and promoting peace and stability. For the first time in history there existed a real gravitation towards a comprehensive international security system, and a serious prohibition of the use of force. Art 1 (1) lists as the first purpose of the United Nations:

To maintain international peace and security, and to that end: to take effective collective measures for the prevention and removal of threats to the peace, and for the suppression of acts of aggression or other breaches of the peace [...]. 193

٠

¹⁹³ Article 1 (1) UN Charter (n. 134).

The inception of the United Nations was born out of the alliance of the major victorious powers of the Second World War and aimed to replace an ineffectual League of Nations. At the core of the United Nations was the Security Council, originally consisting of the major victors of the Second World War, namely France, Russia, the UK, and the US, plus China. These nations awarded themselves permanent seats on the Council. It was hoped that they would 'continue their wartime alliance in perpetuity as a collective bulwark against future aggressors'. Once more, war was renounced by the international community, but this time a strict prohibition on the threat of or use of force, i.e. aggression, was enshrined in the Charter of the United Nations. Thus Art 2 (4) provides:

All Members shall refrain in their international relations from the threat or use of force against the territorial integrity or political independence of any state, or in any other manner inconsistent with the purposes of the United Nations.¹⁹⁵

This provision goes further than the Covenant of the League of Nations in condemning aggressive violence, as noted by Malcolm Evans, 'by prohibiting not only war as such, but also 'the use of force' in general – thereby encompassing measures short of war, such as armed reprisals'. Unlike the Covenant, Art 2 (4) also prohibits threats of violence. Another notable difference is that, under the Covenant, States did not actually abandon their perceived prerogative to enter war, and consequently it was unsuccessful in preventing States from engaging in aggressive wars.

¹⁹⁴ Evans (2010) (n. 181), 24

¹⁹⁵ Article 2 (4) UN Charter (n. 134).

¹⁹⁶ Evans (2010), *supra*, 24.

¹⁹⁷ The Geneva Protocol would have outlawed war and declared aggression to be an 'international crime', however it never entered force (see section 3.2).

The UN Charter prohibition of the use of force is now deeply enshrined in customary international law through state practice, as evidenced for instance by wide international subscription of 193 member states, and *opinio juris* particularly by the UN's own International Court of Justice.¹⁹⁸ Despite the fact that its prohibition on aggressive use of force has thus become a prominent part of international law, the UN Charter itself provides no definition of such acts of aggression. An attempt had been made to incorporate such a definition of aggression at the United Nations Conference on International Organisation in San Francisco in 1945, however it failed.¹⁹⁹ Whilst some States²⁰⁰ approved of a definition along the lines of the 1933 Soviet draft,²⁰¹ the committee, which had been assigned to make recommendations on a definition of aggression, faced objection by a majority of States led by the United States and the United Kingdom.²⁰² As a result, no consensus was reached, and the committee's report suggested

_

- a) Invasion of another State's territory by armed forces.
- b) Declaration of war.
- c) Attack by land, sea, or air forces, with or without declaration of war, on another State's territory, shipping, or aircraft.
- d) Support given to armed bands for the purpose of invasion.
- e) Intervention in another State's internal or foreign affairs.
- f) Refusal to submit the matter which has caused a dispute to the peaceful means provided for its settlement.
- g) Refusal to comply with a judicial decision lawfully pronounced by an International Court.

See United Nations Commission on International Organisation, United Nations, New York, 1948, Vol. 3, 577.

¹⁹⁸ For a discussion of the prohibition on the use of force at customary international law, see for instance the *Nicaragua* case (1986) (n. 37), paras 172-86. See also Claus Kress' comprehensive summary of ICJ case law in 'The International Court of Justice and the "Principle of Non-Use of Force", in Marc Weller & Jake Rylatt (eds), *The Oxford Handbook of the Use of Force in International Law* (OUP 2015), 567-70. For a review of customary international law on aggression and domestic application, see *R v Jones* [2006] UKHL 16 (n. 37), para 19.

¹⁹⁹ See *Commission III*, Security Council, Committee 3, *Enforcement Arrangements*, Report of Mr. Paul-Boncour, Rapporteur, on Chapter VIII Section B, at The United Nations Conference on International Organization, Doc. 881 III/3/46, 10th June, 1945.

²⁰⁰ See for instance Bolivia's draft suggestion, which is based on but would extend the 1933 Soviet draft to include

²⁰¹ See n. 187.

²⁰² See the *Report of Committee 3* (n. 199).

to omit any definition of aggression from the Charter and 'to leave to the Council the entire decision as to what constitutes a threat to the peace, a breach of the peace, or an act of aggression'. The UN Charter therefore contains neither a definition of the state act of aggression, nor does it concern itself with individual criminal responsibility for such acts. Nonetheless, under the UN Charter a widely subscribed international system was created specifically to promote collective security and detect and prevent threats of aggression. The execution of such acts of aggression was strictly prohibited with the possibility of imposing wide-ranging actionable consequences under the Security Council's Chapter VII powers, if the prohibition was ignored. On the council of the council of the council of the council of the prohibition was ignored.

4.2 The institution of the Security Council and the anomaly of the veto power

The powers delegated to the Security Council for the use of collective measures in dealing with threats or actual aggression were comprehensive. Thus Article 39 of the UN Charter provides not only for determination powers of a threat or use of unlawful force, but also for appropriate measures to enforce the Charter provisions:

The Security Council shall determine the existence of any threat to the peace, breach of the peace, or act of aggression and shall make recommendations, or decide what measures shall be taken in accordance with Articles 41 and 42, to maintain or restore international peace and security.²⁰⁵

²⁰³ *Ibid*, section I (c).

²⁰⁴ See Chapter VII, UN Charter (n. 134).

²⁰⁵ *Ibid*, Article 39.

Consequently, the powers of the Security Council under Chapter VII include economic and political sanctions, ²⁰⁶ as well as, significantly, the authorisation of members' use of force in enforcement of its decisions. ²⁰⁷ This provision represents yet another noteworthy difference to the UN's forerunner, the Covenant of the League of Nations. ²⁰⁸ The Covenant had merely allowed for economic sanctions, although these sanctions were automatic. Security Council sanctions, by comparison, are not automatic; they are open to debate and a vote. Whilst this allows for substantial flexibility in Security Council responses, measures proposed have in the past often been thwarted by the use of the veto power.

The term veto is widely used,²⁰⁹ if slightly misleading. Substantive decisions of the Security Council require a majority of nine out of 15 votes, including the unanimous votes of the five permanent members (P5).²¹⁰ Consequently, a negative or dissenting vote by a permanent member will prevent the passing of a resolution, effectively amounting to a veto. An abstention, however, will not defeat it.²¹¹ Whilst the use of the veto power had originally been considered as a rare possible occurrence,²¹² and a tool in aid of the exercise of the responsibility for the maintenance of international peace and security, reality soon proved otherwise. Thus use of the veto was prolific and applied for reasons

²⁰⁶ *Ibid*, Article 41.

²⁰⁷ *Ibid*, Article 42.

²⁰⁸ See section 3.1.

²⁰⁹ See for instance *Practices, Procedures and Working Methods of the Security Council*, as described on the United Nations own website, available at www.un.org/securitycouncil/content/procedures-and-working-methods, accessed 10th May 2020.

²¹⁰ Article 27 (3) UN Charter (n. 134).

²¹¹ See *Security Council Report on the UN Security Council Working Methods*, 'The Veto', 30th October 2017. Also see *Research Report on the Veto*, 19 October 2015, both available at http://www.securitycouncilreport.org/un-security-council-working-methods/, accessed 12th May 2020.

²¹² See Statement of the Four Sponsoring Governments on Voting Procedure in the Security Council, UNCIO Document 852, III/1/37(1) [8] UN Doc A/578 (7 June 1945).

of national interest even from the very early days.²¹³ Professor Ferencz places the situation into historical context:

[I]t was a political reality that the victorious allies were not prepared to give up their military power to any untried international body. They assumed the responsibility to maintain the peace but insisted upon a special right to veto any enforcement action.... Amendments to correct inequitable Charter provisions, as was promised in 1945, never materialised.²¹⁴

The voting structure in the Security Council has been widely criticised as outdated and unfair, as well as open to abuse by the P5 to vote in their particular interest or that of an ally.²¹⁵ Reform proposals have suggested a re-organisation of the Security Council to create a better power equilibrium.²¹⁶

Notwithstanding the fact that use of the veto power has, from the outset, had the potential to obstruct the work of the Security Council, their role as the executive organ of the United Nations was of crucial importance. The enforcement by the Security Council of the comprehensive prohibition on the use of force contained in the Charter, with the ability to impose wide-ranging actionable consequences if this prohibition was ignored, sent a warning to actual and potential aggressors. The unequivocal message of deterrence was particularly indicative of the spirit of the international community at the end of the Second

53

²¹³ Andrew Carswell, 'Unblocking the UN Security Council: The *Uniting for Peace* Resolution' (2013) 18 (3) J.C.&S.L. 453-80. For a full account of veto use see UN Documentation Research Guide, *Veto List*, as compiled by the Dag Hammarskjöld Library, available at http://www.research.un.org, accessed 20th January 2021.

²¹⁴ Ferencz (2009) (n. 4), 285-6.

²¹⁵ See the discussion on Security Council reform in Chapter 6.

²¹⁶ *Ibid*.

World War, instrumental in creating this new, widely-subscribed international system specifically designed to promote collective security and detect and prevent threats of aggression.

In addition, for the first time in history, there existed a real sense of a need for personal accountability at the end of the Second World War, giving rise to prosecutions of individuals through the innovative development of international criminal law. 217 Consequently, the creation of the United Nations coincided with the first comprehensive attempt to prosecute individuals for state acts of aggression. This coincidence of the first wide-ranging international prohibition of the use or threat of force in the UN Charter, and the first international criminal prosecutions for such acts of aggression, was not accidental. Instead, in the aftermath of the most atrocious international war of aggression to date, these events represented a notable and deliberate change in perspective.

4.3 Developing individual criminal liability – The Prosecutions by the International

Military Tribunals

The trials of individuals for the crime of aggression by the International Military Tribunals represent a watershed.²¹⁸ The intention was to send a strong message to States and their leaders that acts of aggression were now truly outlawed and the bearers of

²¹⁷ Resulting in accusations of retrospective law-making by the prosecution, see n. 229 below.

²¹⁸ Article 6 of the London Charter (n. 117), provided for the criminalisation of aggressive war. This article became the legal basis for subsequent prosecutions by the Nuremberg Military Tribunals (NMTs) under *Control Council Law No. 10* (n. 121), and the International Military Tribunal for the Far East (IMTFE) (see n. 11).

responsibility would not escape punishment.²¹⁹ To that effect, after the Allied victory in the Second World War, unprecedented trials of German and Japanese leaders in their States' acts of aggression took place at Nuremberg and Tokyo.²²⁰ Preparations for the trial of German leaders had begun as early as 1943, when the Allies established the UN War Crimes Commission to investigate the possibility of prosecutions.²²¹ According to Professor Schabas, '[b]y far the most important issue of substantive law to be studied by the Commission and its Legal Committee was the question of whether aggressive war amounts to a criminal act'.²²² This statement is unsurprising when considering that both the absolute prohibition of aggression and the criminal prosecution of individual participants were in reality new and borne out of the unparalleled devastation caused by the Second World War.

Notably, aggression was included as an undefined crime against peace in the London Charter.²²³ Under Article 6 (a), the prosecutors were charged with the investigation and prosecution of 'leaders, organizers, instigators and accomplices' for '[c]rimes against the peace: namely, planning, preparation, initiation or waging of a war of aggression [...]'.²²⁴ It has been suggested that, by separating the individual crimes against the peace (planning, starting and waging a war of aggression), the 'Allies could maintain the principle of individual criminal responsibility... without actually having to define a "war of aggression" in categorical terms'.²²⁵

²¹⁹ See n. 235 below and related text.

²²⁰ See n. 218.

²²¹ See William Schabas, 'Origins of the Criminalization of Aggression: How Crimes Against Peace Became the "Supreme International Crime", in Mauro Politi, *The International Criminal Court and the Crime of Aggression* (Routledge 2004), 17, 22.

²²² *Ibid*, at 22.

²²³ In Article 6, see n. 218.

²²⁴ Ibid

²²⁵ See Jackson Maogoto, 'Aggression: Supreme International Offence still in search of definition' (2009), 5 (1) South. Cross Univ. L. Rev., 20-21.

The subsequent prosecution of 24 German defendants, of whom 12 were found guilty on at least one count, was subject to heavy criticism of violating the legality principle, since it was argued that such a crime had previously not existed. Nevertheless, the Tribunal accepted an argument by the Chief Prosecutor at Nuremberg, Justice Robert Jackson, that the Kellogg-Briand Pact, the Geneva Protocols, and resolutions of the Assembly of the League of Nations as well as the Conference of American States had established a customary international law prohibition against waging aggressive war. The retrospectivity argument was, therefore, rejected by the IMT. As Dinstein explains:

The International Military Tribunal at Nuremberg held that Art 6 (a) of the London Charter [i.e. 'the planning, preparation, initiation or waging of a war of aggression'] is declaratory of modern international law, which regards war of aggression as a grave crime. Hence, the Tribunal rejected the argument that the provision of the Article amounted to *ex post facto* criminalization of the acts of the defendant, in breach of the *nullum crimen sine lege* principle.²³⁰

²²⁶ The defendants found guilty were Hermann Goering, Rudolf Hess, Joachim von Ribbentrop, Wilhelm Keitel, Alfred Rosenberg, Erich Raeder, Alfred Jodl, Konstantin von Neurath, Wilhelm Frick, Walther Funk, Karl Doenitz, and Arthur Seyss-Inquart, see Nuremberg Judgment (n. 3), pp. 272-331.

This incorporates the maxims *nullem crimen sine lege, and nulla poena sine lege,* respectively meaning 'no crime without law', and 'no punishment without law'. According to these principles, 'a person should not face criminal punishment except for an act that was criminalized by law before he/she performed the act'. See Wex Legal Dictionary, http://www.law.cornell.edu, accessed 3 March, 2020.

²²⁸ Robert H. Jackson (n. 135) 305. See also Keith Petty, 'Sixty Years in the Making: The Definition of Aggression for the International Criminal Court' (2008), 31 (2) Hastings Int'l & Comp. L. Rev., 4.

²²⁹ Nuremberg Judgment (n. 3), 219-223.

²³⁰ Dinstein, (2017) (n. 154), 134.

The principles of *nullum crimen*²³¹ and *nulla poena*²³² were deemed to be 'at any rate general principles of justice which should not apply where the defendants ought to have known that the acts they committed, or ordered to be committed, were wrong'.²³³ Interestingly, the Tribunal applied a very low standard of *mens rea* for the crime, stating that the defendants ought to have known of the wrongfulness of their acts, rather than requiring the demonstration of actual knowledge.²³⁴ The Tribunal famously declared:

War is essentially an evil thing. Its consequences are not confined to the belligerent states alone, but affect the whole world. To initiate a war of aggression, therefore, is not only an international crime; it is *the supreme international crime* differing only from other war crimes in that it contains within itself the accumulated evil of the whole.²³⁵

This statement demonstrates the strong condemnation of aggression and the resulting atrocities by the international community, applying a remarkably low standard of *mens* rea for the determination of culpability. Someone had to be held responsible, and individual criminal liability for aggression was born out of this need for accountability.

²³¹ See n. 227.

²³² *Ibid*.

²³³ Maogoto (2009) (n. 225), 23.

²³⁴ *Ibid.* The standard is now one of actual awareness, see Article 8 *bis*, Elements of the Crime of Aggression, Rome Statute.

²³⁵ Final Judgment of the Nuremberg Tribunal (n. 3). See also Office of the United States Chief of Counsel For Prosecution of Axis Criminality, *Nazi Conspiracy and Aggression*, US GPO, Washington D.C., 1946, Vol. I, p. 16, emphasis added.

Shortly after the establishment of the original International Military Tribunal, Control Council Law No. 10 instituted the authority of national tribunals to prosecute individuals for participating in aggressive state acts, ²³⁶ and specifically included jurisdiction over private economic actors as well as complicit third-state officials.²³⁷ This provision is remarkable, since it results in much wider jurisdiction than the modern Crime of Aggression in Art 8 bis of the Rome Statute, under which the prosecution of private economic actors or third party state officials will be extremely difficult, if not impossible. 238 Once again, this wide jurisdiction of the tribunals demonstrates the great appetite, at the time, for developing individual accountability for aggression. Consequently, the Nuremberg Tribunal (NMT) began the prosecution of officials of industry, military officers of rank lesser than 'leaders', and government and party officials. Jurisdiction was derived from the London Charter itself in Article 6, and Control Council Law No. 10,²³⁹ as well as their interpretation in the *Goering* decision that 'Hitler could not make aggressive war by himself. He had to have cooperation of statesmen, military leaders, diplomats, and business men.'240 The NMT held in the 'Industrialist', specifically the Farben and Krupp cases, that private economic actors had the capacity to commit the crime of aggression, as long as they had knowledge that their actions were furthering the aggressive war, despite acquitting the defendants on the facts.²⁴¹ Furthermore, the level of leadership in the *High Command*²⁴² and *Ministries*²⁴³ cases was

²³⁶ See n. 121.

²³⁷ *Ibid*, para 2 (f).

²³⁸ See Chapter 2.

²³⁹ See n. 218.

²⁴⁰ Goering and others, Nuremberg Judgment, (n. 3), at pp. 171-341.

²⁴¹ United States v Krauch et al., Military Tribunal VI, Judgment 29th July 1948 (Farben case), pp. 6-11, and United States v Krupp, Von Bohlen und Halbach et al., Nuremberg Military Tribunal (NMT) III, Judgment 31st July 1948 (Krupp case), pp. 76-7. Industrialist cases is the summary term for NMT cases against private economic actors.

²⁴² United States v Von Leeb et al., Military Tribunal XII, The High Command Case, Case No.12, Judgment 27th October 1948, p. 12, pp. 68-9.

²⁴³ United States v Von Weizsaecker et al., Military Tribunal XI, The Ministries Case, Case No. 11, Judgment 11th April 1949.

a mere 'responsib[ility] for the formulation and execution of policies'. It was not necessary to be in a position to 'control or direct' a State's political or military action, merely 'to shape or influence the policy of his State'.²⁴⁴

In evaluation, the Nuremberg tribunals were ground-breaking for developing an unprecedented concept of individual criminal liability for the state act of aggression. Furthermore, they established that '(1) non-governmental actors can commit the crime of aggression; (2) aggression is a policy-level crime; and (3) an individual is at the policy level if he is in a position to 'shape or influence' a State's political or military action'. ²⁴⁵ The prosecutions also highlighted, because of their novelty, the need for a proper definition of aggression, not only in order to serve as a deterrent for possible aggressors, but also in order to avoid accusations of judicial *ex post facto* law-making in the future. Times were changing, however, and the post-war spirit of unity and consensus began to fade.

Kevin Heller, 'Retreat from Nuremberg: The Leadership Requirement in the Crime of Aggression' (2007) 18 EJIL, 486, quoting from the *High Command* Judgment (n. 242) at 490-491.

Heller, *ibid*, 488. This standard of mere ability to 'shape or influence' rather than 'control or direct' was also adopted by the IMTFE in Tokyo. See n. 11, at 1190.

5. A Cold War U-turn? From individual accountability to sole reliance on the UN's mechanism of collective security

5.1 The early record of the United Nations

Following the creation of the United Nations, the inclusion of the comprehensive prohibition of aggression in their Charter and the criminal prosecution of responsible individuals in the post war tribunals, the apparent consensus of the international community soon dissipated.²⁴⁶ The newly found appetite for developing individual criminal liability all but evaporated.²⁴⁷ The concept of aggression was still lacking a definition, both as a state act and as an international crime.

Contemporary events had the additional effect of shifting the focus of the international community away from the divisive subject of a definition and the development of individual criminal accountability. The end of the Second World War heralded the beginning of an ideological conflict between the communist Soviet Union and Western powers, the so-called Cold War,²⁴⁸ which would eventually develop into a nuclear arms

²⁴⁶ See, for instance, the discussion in Thomas Bruha, 'The General Assembly's Definition of the Act of Aggression', in Claus Kress and Stefan Barriga (eds), in *Crime of Aggression Library*, *The Crime of Aggression: A Commentary*, I, (CUP 2017) 147-8.

²⁴⁷ Ibid, referring to the ILC's First Report on a Draft Code of Offences against the Peace and Security of Mankind, UN Doc. A/CN.4/25, YB ILC (1950), vol. II, 253, and Second Report on a Draft Code of Offences against the Peace and Security of Mankind, including commentary, YB ILC (1951), II 133. Both reports failed to provide a useable definition of aggression, and the debate was shelved in light of the developing Cold War disagreements. See the discussion below in section 5.2, and Bruha, *ibid*, at 148. See also M. Cherif Bassiouni, 'Draft Code of Offenses Against the Peace and Security of Mankind', Proceedings of the Annual Meeting (ASIL), 80 (9-12 April 1986) (OUP 1986) 120.

²⁴⁸ Referring to an ideological, political, economic, military and strategic contest between the United States and its allies and the Soviet Union and its allies. See http://www.oxfordbibliographies.com, accessed 12th January 2021.

race. The political world became increasingly polarised. Disagreements between States led to the increased reliance on self-defence arguments, as the only charter-based exception to the prohibition of unauthorised use of force. The new collective security system, entrusted to the United Nations, was the only mechanism potentially capable of decelerating mounting threats of aggression, and yet the frequent use of veto powers often paralysed the Security Council. 250

This impasse within the Security Council came to particular prominence against the backdrop of the 1950's crisis on the Korean peninsula. The deadlock appears to have been worsening over time, since initially five resolutions were passed referring to aggression in Korea. The Security Council's emphasis on the subject of aggression in Korea gradually dwindled. A sixth resolution, passed in January 1951, removed the issue from the agenda of the Security Council, despite the continuing hostilities. In the meantime, and in the face of the Security Council's inaction, the General Assembly felt compelled to develop their own way to deal with threats to the peace, where the Security Council had failed to act. Consequently, they adopted and utilised the Uniting for Peace Resolution, which allowed and forced the General Assembly to immediately consider threats to the peace where the Security Council failed to exercise their primary

²⁴⁹ Article 51 UN Charter (n. 134).

²⁵⁰ See Benjamin Ferencz, 'The UN Consensus Definition of Aggression: Sieve or Substance?' (1975) 10 J Int'l & Econ. 701, 707. See also Dinstein (2017) (n. 154) 342-53.

²⁵¹ United Nations Security Council 'Complaint of aggression upon the Republic of Korea'. UN Doc S/RES/82 (25 June 1950) condemned 'the armed attack on the Republic of Korea by forces from North Korea' (Preamble, para 3), constituting 'a breach of the peace' (Preamble, para 4). UN Doc S/RES/83 (27 June 1950) and UN Doc S/RES/84 (7 July 1950) entitled 'Complaint of aggression upon the Republic of Korea' used identical language, whereas UN Doc S/RES/85 (31 July 1950) referred to 'aggression' in the title and an 'unlawful attack' in the Preamble, para 1. UN Doc S/RES/88 (8 November 1950) refers to 'aggression' in the title only.

²⁵² UN Doc S/RES/90 (31 January 1951), although still entitled 'Complaint of aggression upon the Republic of Korea'.

²⁵³ United Nations General Assembly 'Uniting for Peace' (3 November 1950) UN Doc. A/Res/377 (V) (the *Uniting for Peace* Resolution or U4P).

responsibility due to the exercise of veto power. According to the Uniting for Peace Resolution, the General Assembly:

Resolves that if the Security Council, because of lack of unanimity of the permanent members, fails to exercise its primary responsibility for the maintenance of international peace and security in any case where there appears to be a threat to the peace, breach of the peace, or act of aggression, the General Assembly shall consider the matter immediately with a view to making appropriate recommendations to Members for collective measures, including in the case of a breach of the peace or act of aggression the use of armed force when necessary, to maintain or restore international peace and security[...].²⁵⁴

This new mechanism therefore allowed the General Assembly to determine acts of aggression and make recommendations for collective action in the face of a Security Council deadlock. In the first application of this new procedure, the General Assembly condemned China, which 'by giving direct aid and assistance to those who were already committing aggression in Korea... ha[d] itself engaged in aggression in Korea'. Several years later the resolution was applied in an attempt to defuse the Suez Crisis, which entailed the invasion of Egypt by Israeli, British and French troops and threats of a Soviet nuclear missile strike in retaliation. The Uniting for Peace Resolution was

²⁵⁴ *Ibid*, para 1.

²⁵⁵ UN General Assembly 'Intervention of the Central People's Government of the People's Republic of China in Korea' (5 November 1951), UN Doc. A/Res/498 (V).

²⁵⁶ See for instance UN General Assembly, 1st Emergency Spec. Sess. (2 November 1956) UN Doc A/RES/997; UN Doc A/RES/998 (2 November 1956); UN Doc. A/RES/999 (4 November 1956), all of which applied to the 1956 Suez Crisis.

²⁵⁷ See Barnaby Crowcroft, 'Egypt's Other Nationalists and the Suez Crisis of 1956', (2016) 59 (1) Hist. J., http://www.cambridge.org/core/journals/historical-journal/article/abs/egypts-other-nationalists-and-the-suez-crisis-of-1956/0C35FA1FA5A92491568487DCBB29C255, accessed 8th October 2020.

subsequently utilised on several other occasions in ten Emergency Special Sessions.²⁵⁸ U4P, although not applied very frequently, represents a significant tool available to the General Assembly to address, in a subsidiary capacity, international crises including aggression, which would otherwise be allowed to fester because of inaction by the Council.²⁵⁹

Importantly, U4P has contributed to keeping the Charter prohibition of the use of force alive, through condemnation of aggression in times of Security Council silence as a result of political differences.²⁶⁰ Nevertheless, none of the resolutions passed by the General Assembly in the Emergency Special Sessions actually went as far as actively authorising counter measures, for instance the use of force to counteract aggression. The General Assembly may suggest enforcement action,²⁶¹ however authorisation remains the prerogative of the Security Council.²⁶²

The Security Council's own record of exercising their primary responsibility through the censorship of flagrant acts of aggression continued to be inconsistent, reinforcing the impression of the Council's reticence to determine such an act.²⁶³ Consequently, express condemnations of aggression have been extremely rare compared to the considerable

²⁵⁸ In total, the General Assembly has so far convened ten Emergency Special Sessions under the Uniting for Peace Resolution, dating from 1956 to 1997. The tenth ESS was never terminated and last resumed in June 2018 (see General Assembly of the United Nations, Emergency Special Sessions, available at http://www.un.org/en/ga/sessions/emergency.shtml, accessed 10th October 2020.

²⁵⁹ See Graham Melling and Anne Dennett, 'The Security Council veto and Syria: Responding to mass atrocities through the "Uniting for Peace" Resolution' (2017) 57 Indian J. Int'l L., 285, 298.

²⁶⁰ Carswell (2013) (n. 213), 474.

²⁶¹ See Article 11 (2) UN Charter (n. 134), providing for GA recommendations and referral to the Council. See for instance UNGA 'Duties of States in the event of the outbreak of hostilities' (17 November 1950) UN Doc A/RES/378 (V).

²⁶² See Chapter VII UN Charter (n. 134).

²⁶³ See Nicolas Strapatsas, 'Rethinking General Assembly Resolution 3314 (1974) as a Basis for the Definition of Aggression Under the Rome Statute of the ICC', in Olaoluwa Olusanya (ed), *Rethinking International Criminal Law: The Substantive Part* (Europa Law, 2007), 155.

number of actual or arguable acts of aggression.²⁶⁴ Underlying reasons for Security Council inactivity are inequalities as a result of the weighted voting structure, coupled with geopolitical schisms between some of the P5, resulting in frequent use of the veto and paralysis.²⁶⁵

5.2 Cold War efforts to arrive at a definition of aggression

Notwithstanding the occasional use and rare condemnation of the term aggression by the Security Council and, at times, the General Assembly, an actual definition continued to be elusive. It has already been noted that, during the United Nations Conference in 1945, no consensus on a definition of the state act could be reached, leading to the comprehensive determination powers of the Security Council. The post-war tribunals prosecuted individuals for aggression in the contemporary spirit of developing individual criminal responsibility for the international crime, however neither the original London Charter, nor the Charter for the successor tribunals at Nuremberg, established by Control Council Law No. 10, had defined either the crime itself or the underlying state act.

Despite the increasing emergence of political differences between the former allies, and now protagonists of the Cold War, the General Assembly did not abandon their efforts

²⁶⁴ Between the early Security Council resolutions on Korea and 1974, when a definition of the state act of aggression was adopted in Resolution 3314 (n. 14), only three other Security Council resolutions refer to aggression. UN Security Council 'Complaint concerning acts of aggression against the territory and civilian population in Cambodia' (4 June 1964) UN Doc. S/RES/189; 'Measures to Safeguard Non-Nuclear-Weapon States Parties to the Treaty on the Non-Proliferation of Nuclear Weapons', (19 June 1968) UN Doc. S/RES/255; 'Complaint by Senegal' (15 July 1971) UN Doc. S/RES/294.

²⁶⁵ See Strapatsas (2007) (in Olusanya, n. 263), Carswell (2013) (n. 213), 456-9, and the account in Dinstein (2017) (n. 154), 342-353.

entirely to arrive at a consensus definition of aggression. In 1950, they referred the issue to the International Law Commission (ILC),²⁶⁶ who included aggression as an international crime in their 1954 Draft Code of Offences against the Peace and Security of Mankind.²⁶⁷ However, the ILC failed to define either the crime or the state act, merely stating that aggression included the use of armed force by one State against another. Thus Article 2 (1) of the 1954 Draft Code includes the offence of:

Any act of aggression, including the employment by the authorities of a State of armed force against another State for any purpose other than national or collective self-defence or in pursuance of a decision or recommendation of a competent organ of the United Nations.²⁶⁸

The 1954 Draft Code was submitted to the General Assembly, but the project was tabled because of the lack of a definition of aggression and continuing divisiveness of the subject.²⁶⁹ Cold War disagreements between States meant that the political climate was less than conducive to an agreement on a definition of the Crime of Aggression, not to mention investigations and prosecutions thereof. As Ferencz notes, '[p]owerful States, primarily concerned with preserving their own power, hesitated to accept the idea that aggression was a punishable international crime'.²⁷⁰ The majority of States viewed the subject of aggression as a political hornets' nest, which could only be addressed through the collective security system.

²⁶⁶ In resolution A/RES/378 (V) (n. 261).

²⁶⁷ International Law Commission, *Draft Code of Offences against the Peace and Security of Mankind*, (1954), UN Doc. A/CN.4/44 (the 1954 Draft Code).

²⁶⁸ *Ibid*, Article 2 (1).

²⁶⁹ Bassiouni (n. 247), 120.

²⁷⁰ B. Ferencz (2009) (n. 4), 286.

Meanwhile, in the year of 1952, the General Assembly had decided to set up a Special Committee of 15 members, in a separate project, to arrive at a definition of the state act of aggression.²⁷¹ This first of three Special Committees failed to come to a consensus. A second Special Committee of 19 members was established in 1954.²⁷² It submitted its report to the General Assembly in 1956, but the issue was once again postponed, on account of great disagreement within the Assembly as to whether such a definition was at all desirable.²⁷³ As Benjamin Ferencz sums up, '[w]ith fighting going on all over the globe – including in India, Pakistan, Cyprus, the Congo, Cambodia, Vietnam, and the Middle East – it was clear that it was easier to commit aggression than to define it.'²⁷⁴

The discussion in the General Assembly was not resumed for a decade, when a third Special Committee was tasked with the attempt to define aggression. ²⁷⁵ Between 1968 and 1974, this third Special Committee held seven sessions. The work of the third Special Committee coincided with the 25th anniversary of the United Nations in 1970 and the adoption, during a commemorative session, of the so-called 'Friendly Relations Declaration'. ²⁷⁶ This resolution stated in its third principle that a 'war of aggression constitutes a crime against the peace, for which there is responsibility under international law'. The adoption of this resolution signalled a renewed focus on the effort to define and outlaw aggressive use of force, which was mirrored by the growing consensus within the

2

UN Doc. A/RES/2625 (XXV).

²⁷¹ UN General Assembly 'Question of defining aggression' (20 December 1952) UN Doc. A/RES/688 (VII).

²⁷² UN Doc. A/RES/895 (4 December 1954).

²⁷³ UN Doc. A/RES/1181 (XII) (29 November 1957).

²⁷⁴ Ferencz, (1975) (n. 250), 707.

²⁷⁵ UN General Assembly 'Need to expedite the drafting of a definition of aggression in the light of the present international situation' (18 December 1967) UN Doc. A/RES/2330 (XXII). This Special Committee consisted of 35 members, reflecting 'the principal legal systems and geographic areas of the world'. See Benjamin Ferencz, 'Defining Aggression – The Last Mile' (1973) 12 Colum. J. Transnat'l L., at p. 431. ²⁷⁶ UN General Assembly 'Declaration on Principles of International Law concerning Friendly Relations and Cooperation among States in accordance with the Charter of the United Nations' (24 October 1970)

Special Committee. Thus, the Committee eventually adopted a draft definition of aggression, which it submitted to the General Assembly in 1974. This definition gained consensus within the General Assembly and was adopted unanimously without a vote in Resolution 3314.²⁷⁷

5.3 The definition of the State act and its application by the Security Council

Resolution 3314 generically defines the state act of aggression as 'the use of armed force by a State against the sovereignty, territorial integrity or political independence of another State, or in any other manner inconsistent with the Charter of the United Nations...'²⁷⁸ The definition includes a non-exhaustive list of acts, for the purpose of illustration, ²⁷⁹ providing guidance to the Security Council in their determinations of aggression. ²⁸⁰ The Council is, however, not bound by the definition, retaining primary responsibility to determine other acts of aggression by virtue of Article 39 of the Charter. Whilst the 'first strike' may serve as evidence in the determination of aggression, the Security Council may decide otherwise. ²⁸¹ Importantly, this definition of the state act was perceived as a prelude to future criminal prosecutions for aggression. Thus, Article 5 (2) of the resolution asserts the same sentiments expressed in the Friendly Relations Declaration ²⁸² that 'a war of aggression is a crime against international peace', giving rise to 'international

²⁷⁷ Resolution 3314 (n. 14). For an in-depth discussion of this definition, and comparison with the definition of the crime in Article 8 *bis* of the Rome Statute 1998, see Chapters 2 and 3.

²⁷⁸ Resolution 3314, *ibid*, Article 1.

²⁷⁹ *Ibid*, Article 3.

²⁸⁰ *Ibid*, Preamble, para 4.

²⁸¹ *Ibid*, Article 2.

²⁸² A/RES/2625 (n. 276).

responsibility'.²⁸³ The definition of a state act of aggression in Resolution 3314 was a remarkable achievement after decades of deadlock, serving 'as a warning to decision-makers that there are perimeters to the lawful use of force... and an important stepping stone on the road to enduring peace in a rational world'.²⁸⁴

Nevertheless, its status is not binding on the Security Council, ²⁸⁵ and, consequently, the Council's record of determining acts of aggression continued to be disappointing even after the adoption of Resolution 3314. ²⁸⁶ Between 1974 and the end of the Cold War in 1990, a mere 32 Security Council resolutions were passed, which descriptively referred to 'acts of aggression'. ²⁸⁷ In the Falklands War, the Council determined the existence of a breach of the peace without specifying who had committed the armed attack, or naming the act as aggression. ²⁸⁸ Similarly, in the Iran-Iraq War, a breach of the peace was determined, however not until seven years after hostilities broke out, and no actual reference to aggression was made by the Council. ²⁸⁹ Many other acts of aggression can

²⁸³ Resolution 3314 (n. 14), Article 5 (2).

²⁸⁴ Ferencz (1973) (n. 275), 463.

²⁸⁵ See Maogoto (2009) (n. 225), 28. See also Weisbord (2008) (n. 176), 168.

²⁸⁶ See Dinstein (2017) (n. 154), 343. See also Strapatsas (2007) (n. 263).

²⁸⁷ Twenty of these resolutions condemned South Africa directly some also concerning other African countries: UN Doc S/RES/387 (31 March 1976); S/RES/418 (4 November 1977); S/RES/428 (6 May 1978); S/RES/447 (28 March 1979); S/RES/454 (2 November 1979); S/RES/475 (27 June 1980); S/RES/507 (28 May 1982); S/RES/527 (15 December 1982); S/RES/535 (29 June 1983); S/RES/546 (6 January 1984); S/RES/554 (17 August 1984); S/RES/567 (20 June 1985); S/RES/568 (21 June 1985); S/RES/571 (20 September 1985); S/RES/572 (30 September 1985); S/RES/574 (7 October 1985); S/RES/577 (6 December 1985); S/RES/580 (30 December 1985); S/RES/581 (13 February 1986); and the last of these is S/RES/602 (25 November 1987). Another 6 resolutions condemned acts of aggression by the former Southern Rhodesia: S/RES/326 (2 February 1973); S/RES/328 (10 March 1973); S/RES/386 (17 March 1976); S/RES/411 (30 June 1977); S/RES/423 (14 March 1978); S/RES/424 (17 March 1978); S/RES/445 (8 March 1979); and S/RES/455 (23 November 1979). Two other resolutions condemned Israel: S/RES/573 (4 October 1985), and S/RES/611 (25 April 1988), and another the State of Iraq, namely S/RES/667 (16 September 1990). Three other resolutions did not identify the aggressor: S/RES/405 (14 April 1977); S/RES/419 (24 November 1977); and S/RES/496 (15 December 1981). For a full list of developments relating to aggression, until mid-2001, see Historical Review of Developments Relating to Aggression (UN Publications, 2003), 225-36.

²⁸⁸ UN Security Council 'Falkland Islands (Malvinas)', UN Doc. S/RES/502 (3 April 1982). The Falklands war took place from April till June 1982.

²⁸⁹ UN Security Council 'Iraq-Islamic Republic of Iran' UN Doc. S/RES/598 (20 July 1987). The Iran-Iraq War took place from September 1980 till August 1988. For a discussion, see Dinstein (2017) (n. 154), 343.

be identified within this timeframe, which have not been condemned for diplomatic and political reasons, frequently due to an internal deadlock because of use of veto powers by the P5.²⁹⁰ An incomprehensive list of acts of aggression, mostly uncensored by the Council, includes the US invasion of Grenada in 1983, actions by the US in Nicaragua in 1986 and a vast number of transgressions by and against Israel over the decades.²⁹¹

Since 1990, the Security Council has been even more reticent in actually using the word aggression. In 1990 nine resolution were passed concerning the Iraqi invasion and occupation of Kuwait, all bar one avoiding the term aggression. In 1992 Instead, generally, they refer to a 'breach of international peace and security'. Subsequent Security Council resolutions in other situations of aggression entirely avoid the word, referring to the 'use of armed force', the provision of (political, financial and logistical) 'support and training to armed opposition groups', 'attacks of armed opposition groups' amounting to 'threats' to the country in question and the international community, and the condemnation of attacks by armed opposition groups, constituting 'a terrorist threat' to the country and region in question, as well as the international community. A recent textbook example

²⁹⁴ Examples include the Security Council's condemnation of Armenia for the use of armed force against

²⁹⁰ See for instance the account in Dinstein, *ibid*, 342-53. See also Strapatsas (2007) (n. 263).

²⁹¹ Israeli violations against Palestinian territory and vice versa have been manifold and are well documented. For a comprehensive analysis see for example James L. Gelvin, *The Israel-Palestine Conflict* – *One Hundred Years of War*, 3rd edn. 2014, CUP.

²⁹² UN Security Council 'Iraq-Kuwait' (16 September 1990) UN Doc. S/RES/667. The other eight resolutions concerning the same situation follow the general pattern of avoiding the term aggression: S/RES/660 (2 August 1990); S/RES/661 (6 August 1990); S/RES/662 (9 August 1990); S/RES/664 (18 August 1990); S/RES/665 (25 August 1990); S/RES/666 (13 September 1990); S/RES/669 (24 September 1990), and S/RES/670 (25 September 1990).

²⁹³ *Ibid*.

Azerbaijan (UN Doc. S/RES/822 (30 April 1993) (Preamble, paras 4 and 5); S/RES/853 (29 July 1993) (Preamble, paras 5 and 6) and S/RES/884 (12 November 1993), in paras 4 and 5 of the Preamble). The actual words used here were 'escalation in armed hostilities', and the endangering of 'peace and security in the region'. Other examples are the condemnation of Rwanda and Uganda for the use of armed force against the DRC (S/RES/1234 (9 April 1999)). Here the words used are 'continuation of hostilities' (Preamble, para 2) and 'in violation of the national sovereignty and territorial integrity of the country' (para 5), constituting 'a threat to peace, security and stability in the region' (para 10), and reiterating the parties'

^{&#}x27;obligation to refrain from the threat or use of force' (operative para 1). S/RES/1304 (16 June 2000), S/RES/1341 (22 February 2001) and S/RES/1355 (15 June 2001) used similar language in respect of the

of unequivocal aggression, uncensored by the Security Council due to use of the veto, are the Russian military intervention in and annexation of the Crimea and the violation of territory in Eastern Ukraine.²⁹⁵

Ultimately, the fact is hard to deny that the record of the Security Council's application of Resolution 3314, resulting in actual condemnation of acts of aggression, has been minimal. Nevertheless, in the wake of the end of the Cold War, efforts continued to arrive at a definition for the crime, in order for it to be prosecuted by the first permanent international Criminal Court.

ongoing situation in the DRC, without expressly referring to 'aggression'. The issue of the use of armed force in the DRC has more recently been revisited, for instance in S/RES/2136 (30 January 2014). Thus the Security Council strongly condemned 'any and all internal and external support to armed groups active in the region, including through financial, logistical and military support' (emphasis added, see Preamble, para 7). The same language was used in S/RES/2211 (26 March 2015) (see Preamble, para 5) and S/RES/2360 (21 June 2017) (see Preamble, para 8). Further examples of the Security Council's condemnation of the use of armed force, without expressly naming it 'aggression', are the condemnation of 'Eritrea's military action against Djibouti' (S/RES/1862 (14 January 2009) (in Preamble, para 2, and operative para 5), and in S/RES/1907 (23 December 2009) of 'all armed attacks' in that situation (Preamble, para 8), and Eritrea's provision of 'political, financial and logistical support to armed groups' (para 7). The situation in the Ukraine since the Russian annexation of the Crimea and ongoing fighting in Eastern Ukraine has appeared before the Security Council on a number of occasions, but resulted in only two Security Council resolutions without being officially stigmatised expressly as 'aggression.' Thus S/RES/2166 (21 July 2014) condemned the downing of MH17 on 17 July 2014, without apportioning blame or calling it 'aggression', whilst S/RES/2202 (17 February 2015) called for observation of the Minsk Agreement ('Package of Measures', 12 February 2015), particularly paragraph 10, stipulating the 'withdrawal of all foreign armed formations, military equipment, as well as mercenaries from the territory of the Ukraine...' (Operative paragraph 3). The annexation of the Crimea following an illegal referendum would have been condemned in draft resolution S/2014/189 (15 March 2014), but was vetoed by the Russian Federation. See https://www.securitycouncilreport.org/atf/cf/%7B65BFCF9B-6D27-4E9C-8CD3-

<u>CF6E4FF96FF9%7D/s_2014_189.pdf</u>. These resolutions by the Security Council are mere examples, however they demonstrate an important trend of avoiding the actual term 'aggression'.

²⁹⁵ See the vetoed draft resolutions of the Security Council, *ibid*. See also *Veto List* (n. 213).

6. Paving the way for the crime of aggression: The ILC's efforts and a return to the concept of individual criminal liability

The 1974 consensus on a definition of the state act of aggression removed the technical impediment standing in the way of the project of the Draft Code of Offences, which had been postponed in 1954,²⁹⁶ and ultimately the development of an International Criminal Court. However, as the above discussion shows, the Cold War continued to hamper efforts to codify the criminalisation of aggression. As Bassiouni notes:

The technical reasons [i.e. the lack of a definition of aggression prior to 1974] for tabling the project were only a cover for the real reason – the 'cold war'. During that period, neither the Eastern nor the Western bloc wanted much to do with the codification of ICL or its enforcement.²⁹⁷

Therefore it was not until after the end of the Cold War that serious efforts were resumed to criminalise aggression and incorporate the crime in the statute for the first permanent International Criminal Court. The end of the Cold War also coincided with events which renewed the focus on international crimes in general and the development of individual criminal responsibility for these crimes. The Balkan civil war and genocides in Kosovo and Rwanda in the 1990's involved some of the most large-scale reprehensible atrocities since the Second World War, and re-focused the international community's attention on war crimes, crimes against humanity, and genocide as well as the crime of aggression. It

-

²⁹⁶ See n. 267-9 and related text.

²⁹⁷ M. Cherif Bassiouni, 'International Crimes: Rationae Materiae', in Bassiouni, *Introduction to International Criminal Law*, (Nijhoff, 2nd edn. 2013), 140.

is against this background, that a renewed appetite for bringing individual perpetrators to justice before the World's first permanent International Criminal Court emerged, after decades of sole focus on the collective security response system.

6.1 The ILC Draft Codes of Crimes against the Peace and Security of Mankind in

1991 and 1996

In 1991 the ILC produced a Draft Code which included a consideration of individual criminal responsibility for aggression, once the UN Security Council had made a determination of such an act.²⁹⁸ According to the related commentary, it was at this time undecided whether only government officials could be prosecuted, or also others with political or military responsibility, or even private actors of sufficient economic or financial influence who enabled the act of aggression.²⁹⁹ The draft code was unclear as to whether the future International Criminal Court should have independent or concurrent determination powers to those of the Security Council, or none at all. The ILC adopted largely the 1974 definition of the state act contained in Resolution 3314,³⁰⁰ although there was disagreement as to whether that resolution, serving as a guide to the Security Council, should be used as a basis for criminal prosecution.³⁰¹ Consequently, no final decision was made as to whether Resolution 3314 should be included in the final definition of the crime in its entirety, and whether the acts listed in it should be exhaustive, or whether judges

²⁹⁸ International Law Commission, *Ninth Report on a Draft Code of Crimes against the Peace and Security of Mankind* (hereafter 1991 Draft Code), UN Doc. A/CN.4/435, and Add. 1 (Feb 8 and Mar 15, 1991).

²⁹⁹ Yearbook of the ILC (1991), Vol. II (Part Two), 72-3, in paragraph 1. This point is of great importance, and will be discussed in detail in Chapter 2.

³⁰⁰ In Article 15 of the 1991 Draft Code (n. 298).

³⁰¹ See the commentary in the 1991 ILC Yearbook (n. 299), paragraph 2.

should be free to determine other acts.³⁰² As the following chapters show, the ILC's indecision on these points would be mirrored in States' negotiations all the way to Kampala.³⁰³

The second attempt at a Draft Code of Crimes against the Peace and Security of Mankind in 1996, unlike its 1991 predecessor, rejected the definition of aggression in Resolution 3314 for the purposes of criminal prosecution, and instead looked to the Judgments at Nuremberg and the UN Charter as 'the main sources of authority with regard to individual criminal responsibility for acts of aggression'. Article 16 of the 1996 Draft Code provides simply that '[a]n individual who, as leader or organizer, actively participates in or orders the planning, preparation, initiation or waging of aggression committed by a State shall be responsible for a crime of aggression'. The 1996 Draft Code and its commentary offer no other explanation or details of a definition, or the individual's conduct, and no reference to the definition of the state act in Resolution 3314.

In evaluation of the ILC's draft codes, Professors Bassiouni and Ferencz complained that they were so fatally weak on account of vagueness that 'a comparative analysis of the ILC's ... effort to define aggression demonstrates the inconclusiveness of this undertaking'. Similarly, James Crawford described the ILC's endeavours as 'a history of dead ends, evasive or circular formulations, conceptual overreach and repeated

³⁰² *Ibid*, paragraph 3.

³⁰³ See Chapters 2, 3, and 4.

³⁰⁴ ILC, Draft Code of Crimes against the Peace and Security of Mankind (1996 Draft Code), Report of the International Law Commission on the Work of its 48th Session, May 6-July26, 1996, at 83-85, UN Doc. A/51/10 (SUPP) (Jan 1, 1996).

³⁰⁵ *Ibid*, 83, Article 16.

³⁰⁶ M. Cherif Bassiouni & Benjamin Ferencz, 'The Crime against Peace' in M. Cherif Bassiouni (ed), *International Criminal Law* (Nijhoff, 2nd ed. 1999), 342.

deferrals to some future conference or session'. In search of an explanation, he continues:

Some of the reasons for this are political and perhaps intractable: a more precise definition of aggression for example, increases the chances of impugning acts by States that are prone to intervene abroad militarily... Ultimately, this project foundered. From an initial formulation that did little to specify what acts might constitute the 'offence', it swung too far in the opposite direction – towards an overly expansive provision – before coming to rest on a less ambitious one that in effect deferred the question.³⁰⁸

These assessments, therefore, lead to the conclusion that the ILC's efforts at a draft definition were plagued by indecision and lack of clarity on vital elements, thus hindering a certain and efficient definition of criminal aggression and sensible jurisdictional conditions. In a separate, but contemporaneous project, the ILC was preoccupied with the development of the future Statute of this International Criminal Court. The ILC presented its final Draft Statute for the creation of an International Criminal Court to the General Assembly in 1994, including the Crime of Aggression as one of the core crimes under the future Statute.³⁰⁹

³⁰⁷ James Crawford, 'The International Law Commission's Work on Aggression', in Kress and Barriga (2017), (n. 246), 233.

³⁰⁸ *Ibid*, 234.

³⁰⁹ In draft Article 20, 1994 ILC Draft Statute (with Commentary), *Report of the International Law Commission*, 46th Session, Sept. 1, 1994, UN Doc. A/49/355; GAOR, 49th Sess., Supp. No. 10 (1994).

6.2. The Rome Statute 1998 and a move towards jurisdiction over the crime

The above discussion has shown that, particularly since the Second World War and the Nuremberg and Tokyo tribunals, the international community has exhibited the strong conviction that a) a State's act of aggression against another is a prohibited use of force, and b) individuals responsible can and ought to be held criminally liable. The realpolitik of the 20th century, however, also engendered substantial resistance by some States aiming to protect their own sovereignty and perceived impunity, whilst continuing to engage in old and new conflicts, including the Cold War.³¹⁰

This resistance led to the tabling of the 1954 Draft Code, the long delay between this Draft Code and the 1974 Definition of Aggression in Resolution 3314, and the subsequent postponement of the ILC's efforts until the end of the Cold War. This resistance also hampered the efforts to criminalise aggression, and even in the lead-up to the Rome Conference some States continued to oppose the inclusion of the crime in the Rome Statute. For example, the United States and a number of its traditional allies, including the United Kingdom, strongly opposed the new ICC's jurisdiction over aggression. By contrast, a considerable number of non-aligned and Arabic States, as well as some industrialised countries, i.e. Canada, Germany, Greece, Italy, and Japan, pressed for the inclusion of the crime in the Rome Statute, as did the European Union.

³¹⁰ See the discussion above and, for instance, Dinstein 2017 (n. 154), 342.

³¹¹ See the discussion in Bassiouni and Ferencz (1999) (n. 306), 346.

³¹² *Ibid*.

At the Rome Conference in 1998, it proved impossible for States to agree on a definition of the crime or jurisdictional procedure. Finding consensus on a mechanism for individual accountability for aggression, in particular, remained as problematic and intractable as it had been throughout the decades since Nuremberg. Despite the divisiveness of the issue of aggression, the offence was included in Article 5 (1) (d) of the 1998 Rome Statute³¹³ as a 'blank-prose' crime, 314 to be defined at a later point. Article 5 (2) provided for the future jurisdiction of the Court in accordance with the amendment procedures of Articles 121 and 123, once a provision was adopted. 315 A provision was included in the Final Act of the Rome Conference for the Preparatory Commission to create its own draft and consider States' proposals on the definition and conditions for the exercise of jurisdiction. The Preparatory Commission established the Working Group on Aggression in its third session at the end of 1999, culminating in the Coordinator's 2002 Discussion Paper, which summarised the main proposals.³¹⁶ Subsequently, the work of the Preparatory Commission was taken over by the Special Working Group for the Crime of Aggression established by the Assembly of States Parties. Negotiations took place in a multilateral forum and considered the views of member states and non-members alike. They were collated in a number of papers presented at regular intervals for consideration by the Assembly of States Parties.317

-

³¹³ For Article 5 (1) Rome Statute, see n. 9.

³¹⁴ Glennon (2010), n. 8.

³¹⁵ Art 5 (2) Rome Statute, see n. 10.

Discussion Paper Proposed by the Coordinator, 11 July 2002, UN Doc. PCNICC/2002/WGCA/RT.1/Rev.2 in *Report of the Preparatory Commission for the International Criminal Court*, UN Doc. PCNICC/2002/2/Add.2, Part II, (the 'July 2002 Coordinator's Paper').

³¹⁷ The most important summaries, discussion papers, Coordinator's and Chairman's report papers of the Special Working Group have been collated in Stefan Barriga and Claus Kress (eds), *Travaux Preparatoires of the Crime of Aggression*, Crime of Aggression Library (CUP 2012). They will be discussed in more detail in the Chapters 2-5.

The extensive negotiations ultimately resulted in the 2009 Discussion Paper, which was the basis for the proposed definition of the Crime of Aggression as presented at the First Review Conference of the International Criminal Court in Kampala in 2010.³¹⁸ At this Review Conference, the definition of the crime, including its elements and jurisdictional procedures, was finally adopted and incorporated into the Rome Statute. Article 8 *bis* contains the definition, and the jurisdictional conditions were included in Articles 15 *bis* and *ter*.³¹⁹ The provisions are the culmination of twelve years of hard fought negotiations, compromise and no little controversy.³²⁰ As the following chapters will demonstrate, they also have the effect of greatly limiting the Court's jurisdiction, at the same time as enabling it in the first place.

7. The historical background in context

Any enquiry into the difficulty of achieving consensus on the definitional and jurisdictional parameters of the Crime of Aggression, and its resulting limitations, is inextricably linked to the historical and geopolitical background discussed above. The following points are particularly salient for the remaining examination, and will be revisited throughout this thesis.

³¹⁸ 'Discussion Paper on the Crime of Aggression Proposed by the Chairman: Revision January 2009', 19 February 2009, ICC/ASP/7/SWGCA/INF.1 (the '2009 Discussion Paper', not published in *ASP Official Records*), as considered in the *Report of the Special Working Group on the Crime of Aggression*, ICC-ASP/7/SWGCA/2, *ASP Official Records*, ICC-ASP/7/20/Add.1, Annex II, 20.

³¹⁹ These were incorporated into the Rome Statute by virtue of Resolution 6 (n. 2), and will be examined in great detail in the following chapters.

³²⁰ See Chapters 2, 3, 4 and 5.

Individual criminal responsibility for illegal aggression was until recently still a relative novelty, the notion of an unfettered right to go to war having remained prevalent amongst nation states until as late as the beginning of the 20th century.³²¹ Attempts to prosecute aggression after the Second World War were subject to the problems of developing a new definitional content of the crime, as well as establishing the jurisdictional parameters for its prosecution. Consequently, granting jurisdiction over the novel Crimes against the Peace to the ad hoc IMTs of Nuremberg and Tokyo,³²² led to accusations of retrospectivity and violation of the *nullem crimen sine lege* principle, although these were refuted by the tribunals.

The discussion also showed that post Nuremberg, the development of an international jurisdiction over the crime remained controversial and was too politically sensitive to find easy agreement. The Security Council was frequently deadlocked, and often failed to determine and condemn state acts of aggression, as a result of which the General Assembly occasionally resorted to determinations of aggression. Eventually a definition of a state act of aggression was developed for the guidance of the Security Council. Additionally, the ICJ passed a small number of judgments and opinions pertaining to aggression. None of these references to and condemnations of acts of aggression, had, however, the effect of creating consensus on the general parameters for the jurisdiction of a permanent future international criminal court over the crime itself.

³²¹ See section 3 above.

³²² See the discussion in section 4.3 below.

³²³ See the account of Bassiouni and Ferencz, (n. 306), 346.

³²⁴ See section 5 above.

³²⁵ Applying the Uniting for Peace Resolution (n. 253), see section 5.1.

³²⁶ Resolution 3314 (n. 14).

³²⁷ Most notable is the ICJ judgment in the *Nicaragua* case (1986) (n. 37), passing opinion (at para 103) that Article 3 (g) of Resolution 3314 was considered 'to reflect customary international law'. This was confirmed in the *Armed Activities* (2005) (also n. 37), at para 223.

Nevertheless, the Crime of Aggression was included as one of the core crimes in the 1994 Draft Statute for an International Criminal Court,³²⁸ despite the substantial disagreement between States about its inclusion, as well as the role to be played by the Security Council.³²⁹ In search of a reason, this lack of consensus has been explained as the result of prevailing and wide-felt scepticism as to whether the ICC was the right organ to have jurisdiction over the crime due to the political nature of aggression, and whether aggression should remain solely the purview of the Security Council. Antonio Cassese's assessment helps to clarify the situation:

It is striking that in the negotiations leading to the adoption in 1998 of the Statute of the ICC, no agreement was reached on the definition of aggression. [...] It would seem, however, that the main bone of contention was about the role to be reserved to the UN Security Council. It was a matter of discussion whether its determinations were binding upon the Court, whether it could thus stop the Court from prosecuting alleged cases of aggression, or whether the Court should instead be free to make its own findings, whatever the deliberations of the supreme UN body.³³⁰

Cassese's description of the situation just before the Rome Conference demonstrates not only the existence of fundamental disagreement between States on whether criminalising aggression was a worthwhile endeavour at all, but also the more practical concern of the extent of freedom of jurisdiction to be granted to the Court, particularly vis-à-vis the

³²⁸ See n 309

³²⁹ See the account by Bassiouni and Ferencz (1999) (n. 306 and section 6.1), 346.

³³⁰ Antonio Cassese (2013) (n. 12), 138.

functions and powers of the Security Council. Essentially, determining the jurisdictional conditions for the International Criminal Court over the Crime of Aggression involved a reassessment of the role of the new Court within the established framework of international institutions, considering and assigning determination powers over an act of aggression by the Security Council, the General Assembly, the International Court of Justice and the ICC itself. Additionally, States were also being asked to relinquish, at least partially, their sovereignty by granting their consent to criminal jurisdiction over political acts committed by their state officials. Taking into account these difficulties, it is remarkable that, despite the inability of States to come to an agreement on either a definition or jurisdictional procedure, a compromise was negotiated in Rome, providing for the inclusion of aggression as a still undefined core crime in Article 5 (1) (d) of the Statute.³³¹

After Rome, States and the Special Working Group on the Crime of Aggression continued their negotiations for a definition and jurisdictional conditions. The prevailing, most contentious issue remained, however, whether the International Criminal Court should have automatic and independent jurisdiction, or jurisdiction depending on Security Council authorisation. A number of powerful States continued to resist any jurisdiction of the Court, based on an argument for Security Council prerogative, as well as fundamental scepticism whether the attempt to criminalise aggression was a worthwhile endeavour. Despite the considerable efforts of the SWGCA, consensus on the jurisdictional conditions for the Crime of Aggression seemed out of reach. Negotiations

³³¹ See n. 9.

³³² The Special Working Group took over from the Working Group on Aggression in 2002.

³³³ See Kress and von Holtzendorff (2010) (n. 5), 1194.

³³⁴ See the statements, for instance, by the P5 members, in the 2010 Explanations of Positions in the *Travaux Preparatoires* (n. 317), from p. 810.

nearly failed entirely until the very end of the Kampala Conference due to continuing disagreements and a wide chasm between proponents of Security Council monopoly on the one hand and advocates of independent determination powers of the Court itself.³³⁵ This issue had continued to be a particular problem throughout the negotiations.³³⁶ Eventually, however, a compromise was reached allowing for limited jurisdiction of the Court over a very narrowly defined crime,³³⁷ subject to substantial Security Council control,³³⁸ as well as a reinforced double requirement for state consent, even by signatories.³³⁹

The remainder of this thesis evaluates whether these provisions are likely to be successful and workable or whether, in fact, they result in rendering the new jurisdiction of the first permanent International Criminal Court ineffectual. Additionally, some suggestions will be attempted as to how the provisions could be altered to make the crime of aggression more appropriate in modern day aggression scenarios.

-

³³⁵ See Kress and von Holtzendorff (2010) (n. 5), 1194.

³³⁶ *Ibid.* See also Cassese (2013) (n. 12) 138.

³³⁷ Discussed in Chapter 3.

³³⁸ Discussed in Chapter 4.

³³⁹ Discussed in Chapter 5.

CHAPTER 2: THE GRAVITY THRESHOLD AND STRICT LEADERSHIP CLAUSE

1. Introduction

The previous chapter demonstrated that the path leading to the criminalisation of aggression and jurisdiction by the first permanent international criminal court was by no means smooth or unproblematic. Consensus on a definition had been difficult to achieve in the face of a Cold War which lingered for several decades. The post Second World War focus on jointly developing jurisprudence over atrocious international crimes all but disappeared in favour of reliance on the collective security system of the United Nations, until the establishment of the ad hoc international tribunals of the 1990's³⁴⁰. Controversy continued to surround the inclusion of aggression in the jurisdiction of the International Criminal Court. Nevertheless, it was listed in Article 5 (1) (d) of the Rome Statute, as a 'blank prose' crime still in search of a definition. Eventually, on 11 June 2010, a consensus agreement on a definition and jurisdictional conditions was achieved at the First Review Conference in Kampala. The purpose of this definition was to identify individual responsibility for the collective act of aggression, and hope was widespread

³⁴⁰ UN Security Council 'The International Criminal Tribunal for the former Yugoslavia' (ICTY) (22 February 1993) UN Doc. S/RES/808, and 'Establishment of the International Criminal Tribunal for Rwanda' (ICTR) (6 November 1994) UN Doc. S/RES/955.

³⁴¹ See Bassiouni and Ferencz (1999) (n. 306), 343.

³⁴² See n. 9.

³⁴³ See n. 8.

that it would become an effective mechanism to hold individuals to account for their responsibility in a State's pursuit of acts of aggression.³⁴⁴

This chapter and the next together examine whether the definition arrived at is capable of achieving this high aim, assessing how effectively modern acts of aggression and the culpability of individual aggressors for those acts are captured by the definition. Article 8 *bis* raises four particular issues, the first two arising from Article 8 *bis* (1),³⁴⁵ and the latter two from Article 8 *bis* (2).³⁴⁶ The first two together set a very high bar for the crime, limiting its application. Thus, an act of aggression firstly has to satisfy an elevated gravity threshold, amounting to a manifest violation of the UN Charter to the effect that lesser violations are not covered. Secondly, Article 8 *bis* contains a very stringent requirement of leadership of only high-level politicians or members of the military, to the exclusion of actors in a lesser or different capacity. These two elements of the crime are considered in detail in this Chapter.

The next Chapter assesses two other characteristics of the crime contained in Article 8 *bis* (2), which are based on the types of act included in the definition of the crime. The first of these is the application of the crime to state-on-state aggression only, at a time when most modern threats originate from non-state actors. Secondly, it will be considered whether the limited list of acts, describing outmoded forms of state aggression as perceived in 1974,³⁴⁷ is suitable for the prosecution of modern day acts of aggression.

-

³⁴⁴ See pp. 1-3.

³⁴⁵ Article 8 *bis* (1) Rome Statute, see below.

³⁴⁶ Article 8 *bis* (2) is discussed in the next chapter.

³⁴⁷ Resolution 3314 (14). Note that only Articles 1 and 3 of Resolution 3314 have been incorporated into Article 8 *bis* of the Rome Statute, which is therefore silent on other evidence and circumstances (Article 2 of Resolution 3314), the non-exhaustiveness of the list of acts of aggression (Article 4) (discussed in the next chapter), and the non-availability of political, economic or military justifications for an act of aggression (Article 5).

Ultimately, the discussion of all four of these characteristics raises the question whether the narrow definition results in overly limiting effective jurisdiction of the Court, and whether this consequence could have been avoidable. An examination of the negotiating process, ³⁴⁸ as well as academic concerns expressed in the build-up to the Kampala Conference, ³⁴⁹ reveals that this narrow interpretation was by no means unchallenged. Whilst some States and academics preferred a limited, narrow definition, others viewed Article 8 *bis* as far too restrictive, arguing that a wider definition would capture aggressors more effectively, thus expanding the Court's jurisdiction. ³⁵⁰ Realistically, however, consensus on a wider definition may have proved impossible, considering the divisiveness of the subject.

The effectiveness of the definition of the crime of aggression will undoubtedly be tested in the future. In the meantime, these two chapters provide a critical analysis of Article 8 *bis*, evaluating its limitations and aiming to predict practical implications for such future prosecutions. Additionally, the benefits of a wider interpretation of the definition of aggression, thus extending the ICC's jurisdiction, will be weighed against likely accusations of illegality and retrospective law-making, offending the principle of *nullem crimen sine lege*. ³⁵¹

³⁴⁸ See the discussion below at section 2.3.

³⁴⁹ See the discussion below at section 2.4.

³⁵⁰ *Ibid*.

³⁵¹ See n. 227.

Article 8 bis (1) in detail

The first two distinct limitations contained in Article 8 *bis* arise from its first paragraph.

Article 8 *bis* (1) provides:

For the purpose of this Statute, 'crime of aggression' means the planning, preparation, initiation or execution, by a person in a position effectively to exercise control over or to direct the political or military action of a State, of an act of aggression which, by its character, gravity and scale, constitutes a manifest violation of the Charter of the United Nations.³⁵²

This single sentence establishes a number of conditions, all of which have to be satisfied before individual criminal liability arises. Thus, the definition firstly insists on a state act of aggression, which amounts to a manifest violation of the Charter of the United Nations on account of its character, gravity and scale. This description implies that those three assessment criteria together have to be of such severity that they amount to a manifest violation. The questions arise whether this threshold is too high and whether violations less than manifest could and should be included in order to aid jurisdiction of the Court. Additionally, individual criminal liability ensues only once the initial state act of an unlawful and manifest violation of the UN Charter through the use of armed force or aggression has taken place. This criterion of a completed state act of aggression distinguishes the Crime of Aggression from the other crimes under the Rome Statute, which begin with the act of an individual and are therefore not dependent on such a prior

³⁵² Article 8 bis (1) Rome Statute 1998.

³⁵³ The state element is greatly reinforced in Article 8 bis (2) and therefore discussed in Chapter 3.

act committed by a State.³⁵⁴ This element of state centricity, prevailing throughout the entire Crime of Aggression provisions, also establishes a connection to the examination of the jurisdictional conditions in the second half of this thesis, since it is this multiple reinforcement of consent by States which serves as the basis of jurisdiction under Article 15 *bis*.³⁵⁵

The second prominent feature of Article 8 *bis* (1), examined in this chapter, is the inclusion of a stringent requirement of a high level of leadership. Criminal responsibility of the aggressor depends on his or her political or military position as an indicator of their ability to control or direct the state act of aggression.³⁵⁶ If an act of aggression is committed by a person exercising a lesser degree of control, or that person has control over actions other than political or military (i.e. economic or mere policy-making), the definition does not apply to that person, regardless of the outcome of their actions.³⁵⁷ Here, the question arises whether a lesser degree of control, or in fact a standard of 'shaping and influencing', as coined by the Military Tribunals after the Second World War,³⁵⁸ should have been applied, in order to capture modern day aggressors and make the definition more timely and effective, thus aiding jurisdiction of the Court.

A related concern is whether the description of the culpable behaviour, i.e. the 'planning, preparation, initiation and execution of an act of aggression', is too narrow. Any behaviour which falls short of these levels of participation, or taking part in an act which

-

³⁵⁴ See Article 6 for Genocide, Article 7 for Crimes against humanity, and Article 8 for War Crimes, Rome Statute 1998.

³⁵⁵ See the discussion of the limitations arising from the jurisdictional conditions in Articles 15 *bis* and *ter* in Chapters 4 and 5.

³⁵⁶ See sections 3.1-3.4 below.

³⁵⁷ *Ibid*.

³⁵⁸ See section 4.3 in Chapter 1.

amounts to a less than manifest violation of the UN Charter is, by virtue of this definition, not captured.³⁵⁹ This gives rise to the interesting consideration whether the criminal doctrines of conspiracy and Joint Criminal Enterprise could be utilised to widen the Court's jurisdiction and extend liability to participants at a lower level, for instance accessories and instigators of acts of aggression. Before embarking on the detailed consideration of this leadership clause, let us first examine the high threshold applied to the Crime of Aggression.

2. The gravity threshold

The definition of the crime of aggression provides that an act of aggression constitutes by its character, gravity and scale a manifest violation of the Charter of the United Nations.³⁶⁰ Although Article 8 *bis* is largely derived from the definition of the state act contained in Resolution 3314, this original source requires mere inconsistency with the Charter rather than its violation at a manifest level. Thus, Article 1 of Resolution 3314 provides:

Aggression is the use of armed force by a State against the sovereignty, territorial integrity or political independence of another State, or in any other manner inconsistent with the Charter of the United Nations, as set out in this Definition.³⁶¹

³⁶⁰ Article 8 *bis* (1) Rome Statute (see p. 85).

87

³⁵⁹ See sections 3.5-3.6 below.

³⁶¹ Article 1, Resolution 3314 (n. 14).

Article 8 *bis* (1) of the Rome Statute, by comparison, stipulates that the act of aggression has to amount to a manifest violation. The threshold for individual criminal liability is therefore much higher than mere inconsistency. Additionally, the crimes of genocide, war crimes and crimes against humanity have no such qualifier,³⁶² so the need for the 'supreme international crime' of aggression to be subject to a 'manifest' qualifier does indeed appear to be an oddity.³⁶³ The questions arise why this high threshold was chosen, and whether this was entirely necessary, since the Rome Statute already limits prosecution to only the most serious international crimes.³⁶⁴ A further question worthy of consideration is whether lesser violations, which are inconsistent with the Charter, but do not amount to manifest violations, could or should attract individual criminal responsibility under any circumstances. In attempting to find an answer to these questions, it may be helpful to begin by examining the ICJ's treatment of the subject, providing guidance on what amounts to a state act of aggression and the gravity threshold such an act has to surpass.

2.1 The ICJ's treatment of the gravity threshold

The ICJ has, in the past, addressed the sliding scale of acts potentially amounting to aggression in the context of state acts. Although not attempting to determine state acts for the purpose of establishing criminal responsibility, the ICJ's findings are relevant in the context of the definition of the crime, because it is the state act itself which amounts to

³⁶² Articles 6-8 Rome Statute, see n. 354.

³⁶³ The term 'supreme international crime' is derived from Robert Jackson's opening statement (n. 135) and the final judgment of the Nuremberg Tribunal (n. 3), page 186.

³⁶⁴ See the Preamble, Article 1 and Article 5 (1) of the Rome Statute (n. 1).

the manifest violation of the Charter in Article 8 *bis*. It is also this very state act which triggers the criminal responsibility of the individual in charge of it. Therefore, the ICC will be required to establish the link between the state act, which must be of sufficient gravity, and the act of the individual, in order for criminal responsibility to arise.

Significantly, the ICJ's treatment of acts of aggression establishes a clear relationship between 'armed attack' and act of aggression, impliedly equating the meanings of the two terms. The case of *Nicaragua* is of particular importance, since it specifically establishes a hierarchy of attacks, pitching armed attacks as the gravest use of force against less grave forms. The ICJ derived authority for its equation of aggression with 'armed attack' from the early *Corfu Channel* case, and confirmed its view in the *Oil Platforms* case, the *Armed Activities* case, the *Nuclear Weapons Opinion*, and the *Wall Opinion*. Interestingly, in the *Oil Platforms* case, the ICJ considered even the mining of a single vessel to amount to an 'armed attack', highlighting the fact that what amounts to an act of aggression may depend on the circumstances of each case. Nevertheless, in all of the above cases, aggression was only peripheral to the main proceedings, and in none of them did the ICJ actually declare an express state act of

³⁶⁵ See Dapo Akande and Antonios Tzanakopoulos, 'The ICJ and the Concept of Aggression', in Kress and Barriga (eds) (2017) (n. 246), 214-232.

³⁶⁶ The *Nicaragua* case (Merits) (1986), (n. 37), para 191.

³⁶⁷ Corfu Channel case (United Kingdom of Great Britain and Northern Ireland v Albania), [1949] ICJ Rep 4, page 22, cited in Nicaragua (ibid) at paras 215, 219..

³⁶⁸ Oil Platforms (Iran v US) [2003] Judgment of 6 November 2003, ICJ Rep 161, (Oil Platforms case), para 51.

³⁶⁹ Armed Activities (2005) (n. 37), (Merits Decision), 19 December 2005, para 142.

³⁷⁰ Legality of the Threat or Use of Nuclear Weapons (1996), Advisory Opinion of 8 July 1996, ICJ 2, [hereinafter The Nuclear Weapons Opinion], para 38.

³⁷¹ Legal Consequences of the Construction of a Wall in the Occupied Palestinian Territory, Advisory Opinion of 9 July 2004, [2004] ICJ Rep 136, (The Wall Opinion), paras 127, 139.

aggression, instead coming to findings of the legality of the use of force and assessing States' adversarial claims under Article 2 (4) and 51 of the UN Charter.³⁷²

Despite considering the gravity of acts amounting to aggression, it is worth noting that the ICJ provided no clear criteria for such a threshold, still potentially leaving the matter open to discussion. Nevertheless, following the ICJ's treatment of the subject, it stands to reason that, at the high end of the scale, actual 'armed attacks' will satisfy the threshold criteria of gravity, character and scale for a manifest violation of the UN Charter and the purpose of establishing the *actus reus* for the crime of aggression. This still leaves the question whether acts at the lower end of the scale, which are inconsistent with the Charter but fall short of an actual armed attack, could or should attract individual criminal responsibility, i.e. where the exact boundaries of aggression lie. The other issue worthy of contemplation is why the high threshold of 'manifest violation' was chosen in the first place. The consideration of the contemporary background helps to shed some light on the contentious debates surrounding these issues.

2.2 The contemporary context - Aggression in an age of 'humanitarian'

interventions

The negotiations for the International Criminal Court coincided with an era in which the necessity of international interventions on humanitarian grounds was suddenly pushed into the headlines. The genocides in Rwanda and the Balkans in the 1990's, and the

³⁷² For Article 2 (4) of the UN Charter, establishing the prohibition of the use of force, see n. 195 and for Article 51, providing for self-defence, see n. 249.

international community's non-existent or at best untimely response resulted in deep international concern over non-intervention in the face of grave humanitarian crises.³⁷³ In response, and in the face of deadlock within the Security Council and failure to authorise action to prevent further atrocities, NATO took an independent decision to respond to the developing crisis in the Balkans, in particular Kosovo. Despite the initially unauthorised use of force in the name of humanitarian intervention, NATO's actions later gained legitimacy through retrospective authorisation in Resolution 1244.³⁷⁴ Several years later, and as an extension to an operation against terrorist groups responsible for the atrocities of September 11, 2001, the US and UK decided to invade Iraq on alleged humanitarian grounds in the year of 2003.³⁷⁵ The legitimacy of this unauthorised invasion has been called into question extensively,³⁷⁶ but it is against this particular background of the emerging claims of justifiable humanitarian interventions that the negotiations and debates amongst the member states of the Rome Statute must be viewed. As the following discussion shows, States and academics alike wrestled with the question of what amounts to a manifest violation and where the boundaries should lie, the issue having gained particular relevance in light of the controversy surrounding the humanitarian interventions of the late 1990's and the new millennium.

-

³⁷³ See the discussion on pp. 71-2.

³⁷⁴ UN Security Council, 'The Situation in Kosovo' (10 June 1999), UN Doc. S/RES 1244.

The airstrikes began on 20 March 2003, see the Council on Foreign Relations, http://www.cfr.org/timline/iraq-war, accessed 20th January 2020.

³⁷⁶ See for instance Alex Bellamy, 'Responsibility to Protect or Trojan Horse? The Crisis in Darfur and Humanitarian Intervention after Iraq' (2005), 19 (2) Ethics Int. Aff., 31, 54. See also Kenneth Roth, 'War in Iraq: Not a Humanitarian Intervention', in Richard Wilson (ed) Human Rights in the 'War on Terror', (CUP 2005) and Kenneth Roth, 'Was the Iraq War a Humanitarian Intervention?' (19 August 2006), 5 (2) J. Mil. Ethics, https://doi.org/10.1080/15027570600711864, accessed 12 March 2020. For a different view of Iraq as a valid intervention, see Robert Heinsch, 'The Crime of Aggression after Kampala: Success or Burden for the Future' (2010), 2 Go. J. I. L., 713, 726-7. For determination of its unlawfulness in the absence of a second Security Council Resolution, see Elizabeth Wilmshurst' resignation letter of 18 March letter', 2003, 'Wilmshurst resignation BBC News, March 2005. http://new.bbc.co.uk/1/hi/4377605.stm, accessed 11 August 2020, and the 2016 Chilcot Report, John Report Inquiry, Executive Summary. Chilcot. The of the Iraq 128-135. http://www.iraqinquiry.org.uk/media/247921/the-report-of-the-iraq-inquiry executive -summary.pdf, accessed 11 August 2020.

The only two Charter-based exceptions to the prohibition on the use of force in Article 2 (4) are self-defence under Article 51³⁷⁷ and authorisation of action by the Security Council under Article 42.³⁷⁸ Unauthorised use of force, apart from justifiable self-defence, remains illegal under the Charter. Humanitarian intervention has been described as 'the threat or use of force by a state, group of states, or international organisation primarily for the purpose of protecting the nationals of the target state from widespread deprivations of internationally recognised human rights.'³⁷⁹ If unilateral or collective humanitarian intervention is unauthorised, it may indeed appear to be a violation of the UN Charter's Article 2 (4),³⁸⁰ on account of the use of force being contrary to the Charter's prohibition.

The Charter does not provide for an implied exemption of humanitarian reasons for an unauthorised military intervention,³⁸¹ a fact confirmed by the jurisdiction of the ICJ. Thus, the ICJ held in the *Corfu Channel Case* that the parameters of the Charter and the prohibition of the use of force in Article 2 (4) did not include implicit exceptions.³⁸² In the *Nicaragua* case, the ICJ reiterated the validity of the use of force prohibition in the Charter and at customary international law, and that resort to force was not an appropriate means to ensure respect for human rights.³⁸³ The *Nuclear Weapons* opinion confirmed that self-defence under Article 51 and Security Council authorisation of use of force under

³⁷⁷ Article 51 of the UN Charter, recognising an 'inherent right of individual or collective self-defence' following an armed attack (n. 249).

³⁷⁸ Article 42 of the UN Charter providing for authorisation of the use of force 'to maintain or restore international peace and security' (n. 207).

³⁷⁹ Sean D. Murphy, *Humanitarian Intervention: The United Nations in an evolving world order*, University of Pennsylvania Press (1996), 11-12.

³⁸⁰ Article 2 (4) of the UN Charter (n. 195). For a discussion, see Thomas Franck (2002) (n. 190), 136.

³⁸¹ Franck, *ibid*. See also Ian Brownlie's statement that humanitarian intervention was generally insufficient grounds to go to war at customary international law. I. Brownlie, *International Law and the Use of Force by States*, OUP (1963), 340-341.

³⁸² *Corfu Channel* case (n. 367) (Merits), pp. 34-5.

³⁸³ *Nicaragua* case (n. 37) (Merits) (1986), paras 257-269.

its Chapter VII powers were the only two valid exceptions to the prohibition of the use of force, thus implicitly precluding humanitarian intervention as a valid third exception.³⁸⁴ In *The Armed Activities* case the ICJ held that even Security Council recognition of states' responsibility for peace did not equal authorisation and legitimisation of the use of force on another state's territory.³⁸⁵ These authorities demonstrate that any intervention, and particularly those involving military measures, should, under international law, be authorised by the Security Council in order to be distinguishable from an illegal use of force.³⁸⁶ If no agreement on sanctions or collective action can be reached within the Security Council,³⁸⁷ it is possible for the General Assembly to pass resolutions under the Uniting for Peace principle.³⁸⁸ In the past, these have fallen short of authorising forceful interventions by States, but this route is not entirely barred. Authorisation by either organ would avoid undermining the existing collective security system.

Despite the fact that unauthorised humanitarian interventions are *prima facie* illegal, States have, since Kosovo, demonstrated a growing appetite for unauthorised unilateral or collective interventions involving military action.³⁸⁹ It is conceivable that evolving

³⁸⁴ The Nuclear Weapons opinion (1996) (n. 370), paras 37-50.

³⁸⁵ The Armed Activities case (2005) (n. 37), para 143.

³⁸⁶ Article 41 of the UN Charter (n. 206) provides for the authorisation on non-military measures, and Article 42 (n. 207) for the authorisation of military interventions.

³⁸⁷ See for instance the situation in Syria, where resolutions attempting to authorise collective action have been repeatedly thwarted by use of the veto power. See, for instance, Draft Resolutions UN Doc. S/2011/612 (4 October 2011); UN Doc. S/2012/77 (4 February 2012); UN Doc. S/2012/538 (19 July 2012); UN Doc. S/2014/348 (22 May 2014); UN Doc. S/2017/172 (28 February 2017); and UN Doc. S/2017/315 (12 April 2017). For a discussion, see also Aidan Hehir, 'The Impact of the Security Council on the Efficacy of the International Criminal Court and the Responsibility to Protect' (2015), 26 Criminal Law Forum, 153. ³⁸⁸ Uniting for Peace Resolution 1950, see n. 253 and related discussion.

³⁸⁹ See for instance the unauthorised 2003 coalition invasion of Iraq (n. 375-6). The 2011 Libyan intervention, beginning on 19 March 2011, by contrast, was authorised by UN SC/Res 1973, but has raised the question whether NATO's action went beyond the mandate (see, for instance Geir Ulfstein and Hege Christiansen, 'The Legality of the Nato Bombing in Libya' (2013) 62 (1), ICLQ, 159, 169. Unilateral and even multilateral unauthorised air strikes in Syria by France, the UK and the US, despite attempts to rely on a justifiable need for humanitarian intervention, remain unlawful (see for instance Marko Milanovic, 'The Syria Strikes: Still Clearly Illegal' (EJIL:Talk! 15 April 2018) http://ejiltalk.org.the-syria-airstrikes-still-clearly-illegal/ accessed 12 December 2020. In the Yemen, the externally supported civil war (the government being backed by a coalition of Arabic States led by Saudi Arabia with substantial support of

state practice may bring about a change at customary international law, a possibility which was hinted at by the ICJ in *Nicaragua*.³⁹⁰ In this case, the ICJ stated that 'reliance by a State on a novel right or an unprecedented exception to the principle might, if shared in principle by other States, tend towards a modification of customary international law.'³⁹¹ Thomas Franck, in examining a number of humanitarian interventions, came to the conclusion:

In this practical reconciliation [of the pursuit of peace and justice] we can detect a pragmatic range of systemic responses to unauthorised use of force, depending more on the circumstances than on strictly construed text...., mitigat[ing] the consequence of such wrongful acts by imposing no, or only nominal, consequences on states which, by their admittedly wrongful intervention, have demonstrably prevented the occurrence of some greater wrong.³⁹²

Similarly, Leclerc-Gagne and Byers argue that *bona fide* humanitarian intervention ought to be exempt from the possibility of prosecution for the crime of aggression on account of falling short of the *mens rea* requirement in Article 30 of the Rome Statute of intent and knowledge.³⁹³ Nevertheless, whilst humanitarian interventions may be morally excusable, existing international law interprets an action against the sovereignty of a foreign state, which is not justifiable as an act of self-defence under Article 51 and not

the United States, whereas the Houthis are backed by Iran) has raised issues of aggression in the form of a proxy war, with little or no forthcoming relief of the humanitarian crisis, see for instance Richard Reeve's article 'War in Yemen' (17 April 2015), available at https://www.oxfordresearchgroup.org.uk/war-in-yementhe-african-dimension, accessed 12 December 2020.

³⁹⁰ Nicaragua (1986) (n. 37), para 206.

³⁹¹ *Ibid*, at para 207.

³⁹² T. Franck (2002) (n. 190), 139.

³⁹³ Elise Leclerc-Gagne and Michael Byers, 'A Question of Intent: The Crime of Aggression and Unilateral Humanitarian Intervention' (2009), 41 (2) Case W. Res. J. Int'l Law, 378, 381.

authorised by the Security Council under its Chapter VII powers, *prima facie* as an act of aggression. States participating in the negotiations for the crime of aggression were concerned that their own involvement in such interventions could be construed as an act of aggression capable of incurring criminal responsibility.³⁹⁴ Only if the measuring bar for the crime of aggression was declared to be so high that humanitarian intervention would not be captured, would this threshold effectively provide an extra-charter based exemption to aggression for such interventions. As the following subsections examine in detail, this reasoning explains the insistence, particularly by powerful states involved in contemporary interventions, on the high manifest threshold, in order to distinguish these acts.

2.3 Negotiating the threshold clause

Even prior to the Rome Conference, and in face of the controversy surrounding even the very idea of including the crime of aggression in the ICC's jurisprudence, states submitted proposed definitions of the crime which included an 'object' or 'result' clause in addition to the already narrow manifest requirement. An early proposal by Germany, for instance, makes the following suggestion for *actus reus* of the crime of aggression:

For the purpose of the present Statute, the crime of aggression means either of the following acts committed by an individual who is in a position of exercising control or capable of directing political or military action of a State

³⁹⁴ This concern arose specifically as a result of the unauthorised intervention in Iraq, which involved the illegal use of force under the pretext of a humanitarian intervention (n. 376).

- (a) initiating or
- (b) carrying out

an armed attack directed by a State against the territorial integrity or political independence of another State when this armed attack was undertaken *in manifest* contravention of the Charter of the United Nations and resulted in the effective occupation by the armed forces of the attacking State or in the annexation by the use of force of the territory of another state or part thereof.³⁹⁵

The delegation derived the object or result clause from Article 3 (a) of GA Res. 3314,³⁹⁶ and in its almost identical 1999 proposal explained the rationale for combining the manifest requirement and the object or result clause. Thus, they considered it necessary for the definition to 'focus on and try to **cover only the obvious and indisputable cases of this crime** (such as the aggressions committed by Hitler and the one committed against Kuwait in August 1990)'. ³⁹⁷ The rationale continues:

This limitation seems indispensable, in particular for the following reasons: 'It is of utmost importance that the definition does not lend itself to possible frivolous accusations of a political nature against the leadership of a Member State. Furthermore, it must be avoided that the definition somehow negatively affects the legitimate use of armed force in conformity with the Charter of the United

³⁹⁵ 'Proposal by Germany', 1997, A/AC.249/1997/WG.1/DP.20 (emphasis added).

³⁹⁶ Article 3 (a) of Resolution 3314 (n. 14), providing for the following example of an act of aggression: '[t]he invasion or attack by the armed forces of a State of the territory of another State, or any military occupation, however temporary, resulting from such invasion or attack, or any annexation by the use of force of the territory of another State or part thereof'.

³⁹⁷ 1997 'Proposal by Germany' (n. 395), emphasis included.

Nations whose necessity – maybe unfortunately – cannot be ruled out in the future.'398

Germany's 2000 proposal went further by suggesting an even more restrictive definition, limiting it to large scale 'wars' of aggression of the 'utmost gravity' in 'manifest violation' of the UN Charter.³⁹⁹ This proposal required the armed attack to be of 'particular magnitude and dimension', and 'frightening gravity and intensity', leading to 'the most serious consequences.'⁴⁰⁰ The intended effect was to further narrow the definition of the crime of aggression to wars of aggression, military occupation or annexation, and, importantly, to distinguish humanitarian intervention.⁴⁰¹

Whilst these early proposals represented the prevailing view of the majority that the crime of aggression should be restricted to only the gravest violations of the UN Charter, particularly in order to differentiate humanitarian interventions, some states proposed alternative views suggesting a wider definition with a lower threshold. Egypt and Italy, for example, proposed that aggression needed no threshold at all, since it was already considered the 'supreme international crime', ⁴⁰² and the Rome Statute itself only applied to the most serious international crimes in any case. ⁴⁰³ Thus, simple inconsistency with the Charter involving an act of 'sufficient gravity' was argued to be more appropriate. ⁴⁰⁴ The delegations of Greece and Portugal expressed a preference for the simple term

³⁹⁸ The '1999 Proposal by Germany', as incorporated in PCNICC/1999/INF/2, closely mirrored the earlier one (n. 395) and added the above rationale.

³⁹⁹ 'Proposal submitted by Germany: the Crime of Aggression – A Further Informal Discussion Paper', 13 November 2000, as incorporated into PCNICC/2000/WGCA/DP.4.

⁴⁰⁰ *Ibid*.

⁴⁰¹ *Ibid*.

⁴⁰² See n. 135.

⁴⁰³ 'Proposal by Egypt and Italy on the definition of aggression', A/AC.249/1997/WG.1/DP.6, as incorporated into PCNICC/1999/INF/2. The Rome Statute limits the Court's jurisdiction to the most serious crimes only, see the Preamble (n. 85), Article 1 and Article 5 (1) Rome Statute (n. 1).

⁴⁰⁴ *Ibid*.

'violation' without a reference to the gravity or a manifest qualifier. 405 Colombia suggested using 'illegitimate use of force', which could include *any* use of force in contravention of Article 2 (4)), rather than a 'manifest violation'. 406

Another interesting suggestion for a wider definition was submitted by the Cuban delegation. This proposal referred to inconsistency with the Charter rather than insisting on a manifest threshold, and included acts which *directly or indirectly* affected the sovereignty, territorial integrity or the political *or economic* independence of another State. The proposal is remarkable because it would have encompassed less obvious acts of aggression, including potentially the aggressive influencing and interference in another country's sovereignty, whether political or economic. A definition along these lines may also have covered acts of cyber warfare and interference in another country's democratic process, as well as vital systems disruptions, situations which are a growing threat and which are currently not covered by the definition of aggression. The cubar and the country of the definition of aggression.

All of these minority proposals are noteworthy for promoting a wider definition, however the most outstanding alternative submission was made by Samoa. Their proposal dismissed both the manifest requirement and an object or result clause, on account of the fact that the definition was already narrow, and supported total independence of the ICC in deciding whether the threshold had been crossed.⁴⁰⁹ This proposal would have provided

⁴⁰⁵ 'Proposal by Greece and Portugal', 1999, PCNICC/1999/WGCA/DP.1

⁴⁰⁶ 'Proposal submitted by Colombia on the definition of the crime of aggression and on conditions for the exercise of jurisdiction of the Court with regard to this crime', 1st March 2000, PCNICC/2000/WGCA/DP.1.

⁴⁰⁷ 'Proposal submitted by Cuba', 6th February 2003, Assembly of States Parties, First session, ICC-ASP/1/L.4 (emphasis added).

⁴⁰⁸ See for instance the discussion in Chapter 3 and 6, in particular.

⁴⁰⁹ 'Elements of the Crime of Aggression – Proposal submitted by Samoa', 2nd June 2002, PCNICC/WGCA/D.P.2.

for a wide definition and liberal determination power of the ICC in order to counteract *any* acts of aggression. Had this suggestion been adopted, the definition would have benefited from being adaptable to developing methods of warfare and changing types of aggressors. Such developing methods of warfare could include proxy or remotely conducted wars, as well as cyberwars and other aggressive interference in another state's sovereignty, the effects of which may be just as devastating as those of a conventional war.⁴¹⁰ The Samoan proposal also may have included aggression committed by non-state actors, a group which has over the last two decades emerged as the main instigators of prevalent aggression around the globe.⁴¹¹

The Samoan proposal is particularly noteworthy amongst the minority views, because had it gained consensus - it would have led to much wider jurisdiction of the ICC by allowing the Court itself to determine which acts of aggression ought to be prosecuted. This thesis argues that such wider jurisdiction, in appropriate cases, would also help to reinforce the purpose of the general prohibition of the use of or threat of force contained in Article 2 (4) of the UN Charter itself. The discussion has already shown that many of the reservations by the majority of States were based on the determination to exclude extra-charter based interventions on humanitarian grounds from the jurisdiction of the Court. The determination of whether an act of humanitarian intervention was indeed genuine or an act of aggression under pretext would, under the Samoan proposal, be left to the Court, able to consider all the available facts and criteria outside of the political arena. The prevailing scepticism of the Court being able to function as an impartial and independent arbiter of alleged or potential acts of aggression is, in the writer's view,

⁴¹⁰ See the arguments presented in Chapters 3 and 6.

⁴¹¹ See the discussion in Chapter 3.

exaggerated and misplaced. Wider judicial capacity would not necessarily have to be seen as interfering in matters of the Security Council, since the Court could be viewed as acting complementarily to the Council and other political organs rather than as usurping their powers. Following the approach advocated by Samoa, the determination of a state act of aggression would be merely incidental to the determination of criminal responsibility, therefore fears of judicial interference in matters of politics would be unfounded. The Samoan approach is therefore, this thesis argues, highly persuasive.

Nevertheless, this view was not shared by the majority of states who preferred to restrict the Court's jurisdiction over the crime of aggression by insisting on an extremely high threshold. On examination of the *Travaux Preparatoires*, it becomes apparent that this preference for limiting the crime of aggression to only the most serious violations of the utmost gravity became increasingly influential amongst member states, and this slowly emerging majority view began to make an appearance in the summaries of the Working Group and later the Special Working Group for the Crime of Aggression. The Coordinator's Paper of April 2002 presented a number of options, the first of which defined the crime of aggression as an act, which 'by its characteristics and gravity amounts to a war of aggression'. Option 2 defined the crime as having 'the object or result of establishing a military occupation of, or annexing, the territory of another State or part thereof,' and option 3 added the words 'in manifest violation of the Charter of the United Nations'. The Coordinator's Paper of July 2002 suggested the term 'flagrant violation' of the UN Charter, with or without the options of an object or result clause.

-

⁴¹² The Special Working Group took over the deliberations from the Working Group in 2003 (see *Travaux Preparatoires* n. 317).

⁴¹³ '2002 Coordinator's Paper (April)', 'Definition of the Crime of Aggression and Conditions for the Exercise of Jurisdiction: Discussion Paper Proposed by the Coordinator', 1 April 2002, UN Doc. PCNICC/2002/WGCA/RT.1, as reproduced in the *Travaux Preparatoires*, *ibid*, 398-99.

⁴¹⁴ 'July 2002 Coordinator's Paper' (n. 316), and Travaux Preparatoires (n. 317), 412-14.

These distinctions reappeared in later recommendations of the Special Working Group, for instance the Annex to the 2007 Chairman's Paper, 415 and the 2007 Princeton Report. 416 Paragraphs 56 and 57 of the 2007 Princeton Report shed light on the different views expressed in the discussion, although States are not named expressly. 417 Thus some States expressed a preference for an amendment of the suggested threshold clause to add 'when the act of aggression in question has been committed in a particularly grave and largescale manner.'418 Whilst these States admitted that all acts of aggression are in fact a violation of the Charter, they felt the need to differentiate between the definition itself, and which level of aggression should give rise to criminal jurisdiction. Others argued that the object or result clause should limit jurisdiction further by insisting on a war of aggression or military occupation or annexation, all highly traditional forms of aggression. 419 Despite the fact that neither the Chairman's Paper, 420 the 2007 Princeton report, 421 nor the 2007 SWGCA Report of December, 422 attempt to explain the insistence by some States on an object or result clause, the reasoning is likely to be as expressed in the German Proposals of 1997 and 1999, as wishing to differentiate between prosecutable acts of aggression and borderline transgressions and humanitarian intervention.

⁴¹⁵ 'Discussion Paper Proposed by the Chairman', 16 January 2007, ICC-ASP/5/SWGCA/2 (not published in ASP Official Records), see pp. 525-529 in the *Travaux Preparatoires* (n. 317).

⁴¹⁶ 'Informal Intersessional Meeting of the Special Working Group on the Crime of Aggression', Princeton University, 11-14 June 2007, (the 2007 *Princeton Report*), ICC-ASP/6/SWGCA/INF.1 in *ASP Official Records*, ICC-ASP/6/20, Annex III, 96.

⁴¹⁷ *Ibid.* Please note that the reports and papers often refer to States generically as holding the 'majority' or 'minority' view. I have followed these generic references out of necessity, where no other information was available. Wherever it has been ascertainable which particular States tended particular proposals or entertained particular views, I have included the State names for accuracy.

⁴¹⁸ 2007 Princeton Report, ibid, paras 56 and 57.

⁴¹⁹ *Ibid*.

⁴²⁰ n. 415.

⁴²¹ n. 416.

⁴²² Report of the Special Working Group on the Crime of Aggression (December 2007) (ICC-ASP/6/SWGCA/1, ASP 6/20, Annex II, 87.

Nevertheless, the SWGCA reports in general and the 2007 Princeton Report in particular show that by this time the 'characteristics and gravity' reference in the Chairman's Paper had become more favoured and was considered to sufficiently narrow the definition, whereas general support for an 'object' or 'result' clause was diminishing. Therefore it was suggested to delete the object or result clause entirely and retain the threshold.

Observing non-member states, most notably the United States, also insisted on the highest of thresholds. 424 Throughout the negotiations and even as late as the Kampala Conference, the US expressed their dissatisfaction with draft Article 8 *bis*. Consequently, they demanded the explicit exclusion of certain state acts of aggression, specifically forceful humanitarian intervention, from the umbrella of the crime of aggression. 425 The US delegation was eventually pacified by the inclusion of Understanding No. 7 that the ICC's assessment of whether a violation of the UN Charter is manifest must include consideration of the three elements of character, gravity and scale of the use of force, albeit not necessarily in equal measure. 426 In the meantime, the controversy surrounding the threshold clause continued to reappear in discussions all the way up to the Kampala Review Conference, 427 although the arguments remained largely unchanged. Finally, both the manifest requirement and the reference to the character, gravity and scale were incorporated into Article 8 *bis* (1).

⁴²³ See n. 415-6.

⁴²⁴ See Kress and von Holtzendorff (2010) (n. 5), 1192.

⁴²⁵ *Ibid*

⁴²⁶ 'Understanding No. 7', included in Annex III of Resolution 6 (n. 2), see *Handbook on the Ratification* and *Implementation of the Kampala Amendments on the Crime of Aggression*, http://crimeofaggression.info/docments/1/handbook.pdf, accessed 19th March 2020.

⁴²⁷ See the account in Kress and von Holtzendorff (2010) (n. 5), 1192-5.

2.4 Too high a threshold? – Arguments for and against wider determinations

In practice, the very existence of the threshold clause requires both the Prosecutor and the Pre-trial Division to be satisfied that it has been surpassed before a prosecution can be initiated, unless the Security Council has referred the case to the Court. 428 This may lead to undesirable delays and makes little sense, bearing in mind that all of the elements of the crime, including the threshold, have to be proven later at trial in any event. 429 At the same time, the high threshold applied in the definition of the crime has the effect of greatly restricting determination powers and ultimately jurisdiction of the Court. 430 This point raises the question whether imposing such a high threshold was reasonable and necessary, since the Rome Statute itself is limited to the most serious international crimes already. 431 Should the Court, instead, enjoy wider determination powers in order to decide, on a case by case basis whether an act is capable of giving rise to prosecution for the crime of aggression?

It has already been noted that the exclusion of humanitarian intervention continued to be of the greatest concern to many States, constituting the main reason for the limitation to only the gravest violations of the Charter. It may, however, have been possible to distinguish interventions on genuinely humanitarian grounds from criminal acts of aggression without such a high threshold of the crime. Where a situation arises in which

⁴²⁸ This is due to the interplay between Articles 8 *bis* and 15 *bis*, insisting on a 'filter' in the form of the Pre-trial Division.

⁴²⁹ See Keith Petty, 'Criminalizing Force: Resolving the Threshold Question for the Crime of Aggression in the Context of Modern Conflict' (2009), 33 Seattle University Law Rev. 105, 108, 116.

⁴³⁰ *Ibid.* See also Noah Weisbord, 'Judging Aggression' (2011) 50 (1) Col. J. Transnat'l L., 82, 93,103, 164, and Maogoto, (2009) (n. 225), 21.

⁴³¹ See Preamble (n. 85), Article 1, as well as Article 5 (1) of the Rome Statute (n. 1). See also the argument presented in the 'Proposal by Egypt and Italy' (n. 403).

⁴³² See section 2.2.

intervention may become necessary to avoid or limit the extent of a grave humanitarian crisis and/or extensive loss of lives, there is generally evidence of some consensus by the greater international community, even if authorisation of action may not be forthcoming because of disagreement within the Security Council. This consensus may be evident from expressions of concern or condemnations by the Security Council, the General Assembly, individual States, NGO's, and Human Rights Watch and other civil organisations. There may be support for an intervention by regional organisations. International condemnation of an unauthorised intervention on humanitarian grounds may be muted, signifying at least some implied consensus on the action undertaken. The intervention may even be authorised retrospectively, following the Kosovo precedent. Aggressive interventions under humanitarian pretext would also have to be extremely well disguised in order to detract from the fact that the humanitarian element is, at best, incidental to the main military or political goal. The deployment of civilian relief organisations to alleviate a humanitarian crisis is a main feature of interventions for

⁴³³ See for instance African Union support for the unilateral intervention in Mali by France. For a discussion, see Susanna D. Wing, 'French intervention in Mali: strategic alliances, long-term regional presence?' (5 February 2016) 27 (1) Small Wars & Insurgencies, 59, https://doi.org/10.1080/09592318.2016.1123433, accessed 18 August, 2020.

⁴³⁴ See the retrospective authorisation of the intervention by the Multidimensional Integrated Stabilization Mission in Mali (MINUSMA) in Resolution 2100 UN Doc. S/Res/2100 (25 April 2013). Note, however, criticism of this operation as a 'permanent military intervention' under a pretext of counter-terrorism, rather than for humanitarian or peacekeeping reasons, 'The Military Intervention in Mali and Beyond: An Interview with Bruno Charbonneau (Sahel specialist and scholar)', 28 March 2019, http://www.oxfordresearchgroup.org.uk/blog/the-french-intervention-in-mali-an-interview-with-bruno-charbonneau, accessed 20 August 2020.

genuinely humanitarian purposes, ⁴³⁵ and an absence of relief efforts is likely to create an impression of action taken for entirely different purposes. ⁴³⁶

The safest distinction of genuine humanitarian intervention from illegal use of force may be derived from the Kosovo Report itself, which looked to evidence of 'severe violation of human rights or humanitarian law on a substantial basis', and 'state failure, subjecting civilians to great suffering and risks'. 437 Intervention in these circumstances must have the direct purpose of protection of civilians, must be aimed at a quick end to the conflict and the safeguarding of civilians, and the use of force must be necessary. 438 Peaceful attempts should first have been made to resolve the conflict, and UN Security Council authorisation ought to have been at least attempted, particularly on account of the fact that States' unilateral decisions cannot help but contain an element of subjectivity. A useful distinction between genuine humanitarian intervention and aggression has been provided by Noah Weisbord, describing the former as 'time-limited, multi-lateral action,

⁴³⁵ This was considered a main objective in a number of reports on an emerging Responsibility to Protect. See International Commission on Intervention and State Sovereignty (ICISS), *The Responsibility to Protect* (2001), at VIII. See also *A More Secure World: Our Shared Responsibility, Report of the High-Level Panel on Threats, Challenges and Change*, UN Doc. A/59/565 (2004), available at http://www.un.org/hlp_more_secure_world.pdf (the *High-Level Panel Report*), and *In Larger Freedom: Towards Development, Security and Human Rights for All*, Report of the Secretary-General, UN Doc. A/59/2005, paras 16-22 (2005), available at http://www.un.org/largerfreedom/contents.htm, all accessed 18 August 2020. See also the 2005 *World Summit Outcome*, GA Res. 60/1, paras 138-9 (October 24, 2005).

⁴³⁶ See the discussion above at n. 376 and related text.

⁴³⁷ See *The Kosovo Report - Conflict - International Response - Lessons Learned*, (OUP 2000), p. 4. See also *Peace and Governance Programme*, *Kosovo and the Challenge of Humanitarian Intervention*, (Albrecht Schnabel & Ramesh Thakur eds., 2000), available at http://archive.unu.edu/p&g/kosovo full.htm, accessed 12 August 2020.

⁴³⁸ *Ibid.* The developments since the Kosovo intervention have been described as the growing recognition of a 'paradigmatically justified' right or even duty to intervene. See Larry May, *Aggression and Crimes against Peace* (OUP 2008), 276. See also Petty, (2009), (n. 429), 106. The ICJ opinion in the so-called *Genocide Case* referred to an 'obligation of all States parties [] to employ all means reasonably available to them, so as to prevent genocide so far as possible.' See Case concerning the *Application of the Convention on the Prevention and Punishment of the Crime of Genocide (Bosnia & Herzegovina v Serbia),* 2007 ICJ General List No 91 (26 February 2007), para 166. Note that this thesis does not attempt to provide a moral assessment as to whether unauthorised humanitarian interventions are justifiable or not. It is merely attempted to demonstrate that States concerns, leading to the extremely high threshold, were at least partially exaggerated and unnecessary.

divorced from territorial or economic goals and followed by robust humanitarian aid to a supportive victim population'. 439

Consequently, the failure to install supportive humanitarian infrastructure following an intervention may be a good indicator that an intervention is less than genuine. 440 These are all factors which could be taken into consideration by the Court, in order to determine whether a state act had surpassed the threshold for the crime of aggression, or whether it amounted to a bona fide intervention for humanitarian reasons which should be exempt.

Additionally, there is room for argument that a genuine humanitarian intervention by its nature is not inconsistent with the UN Charter, since the Charter itself promotes humanitarian purposes. 441 Consequently, the view can be defended that there was no need in the first place for negotiating states to raise the threshold of aggression to the high level of a manifest violation, with the potential result of excluding other acts of aggression.⁴⁴² If genuine humanitarian interventions do not violate the UN Charter, there was no need to exclude them at all through the high threshold.

The reverse side of the coin is that the threshold of the crime may now be so high to exclude even unauthorised aggressive interventions under a humanitarian pretext. 443 It has already been acknowledged that an element of subjectivity is inevitable in States'

⁴³⁹ Weisbord, (2011) (n. 430), 164.

⁴⁴⁰ See for instance the ongoing humanitarian crisis in the Yemen, with little sign of relief but continuing involvement of external forces ((n. 389), see particularly Reeve (2015)).

⁴⁴¹Article 1 (3), and 55 of the UN Charter.

⁴⁴² Petty (2009) (n. 429) 123

⁴⁴³ A prime example is the 2003 Iraq invasion. See for instance the *Chilcot Report*, n. 376.

unilateral decisions to intervene. 444 Because of this inevitability of subjectivity, combined with the fact that States are unlikely to volunteer for action if this directly counteracts their own interest, it is an unfortunate reality that unauthorised unilateral interventions may, at times, be utilised for purposes other than providing humanitarian relief, for instance regime change or subjective territorial or political advantages. 445 Even the unauthorised but later legitimised Kosovo intervention gave rise to concerns of abuse in other situations, capable of potentially undermining the existing 'imperfect, yet resilient security system... setting dangerous precedents.'446 And yet, because of the high threshold of the crime, such interventions under pretext may escape criminalisation.

All of these considerations raise concerns that the threshold of a 'manifest violation' may in certain cases be too high, and determinations of what amounts to an act of aggression in a particular situation ought to have been left for the Court to decide, taking into account all of the facts. 447 Admittedly, this may give rise to concerns of legality in criminal proceedings, since liberal determination powers without a clear threshold may make it impossible for a defendant to foresee that his actions could give rise to a prosecution for aggression. 448 Nevertheless, the Court ought to be more than capable of distinguishing acts of aggression, taking into account the criteria of an attempt to seek authorisation, the aim of protection of civilians, threat to or likelihood of loss of life without intervention,

⁴⁴⁴ *Ibid*.

⁴⁴⁵ Petty, (2009) (n. 429), 124.

⁴⁴⁶ See Kofi Annan's comments in his Report of the Secretary-General on the work of the Organization, UN GAOR, 54th Sess., 4th plenary meeting at 2, UN Doc. A/54/PV.4 (20th September, 1999), https://documents-dds-ny.un.org/doc/UNDOC/GEN/N99/858/23/PDF/N9985823.pdf? concerns expressed earlier, see Ian Brownlie, 'Thoughts on Kind-Hearted Gunmen', in Richard Lillich (ed), Humanitarian Intervention and the United Nations (UPVA 1973) 139, 147-8. See also concerns raised by Lou Henkin, in How Nations Behave: Law and Foreign Policy, 144-5 (Columbia University Press 2nd ed. 1979) and Oscar Schachter, International Law in Theory and Practice, 123-5 (Nijhoff 1991).

⁴⁴⁷ Weisbord, (2011) (n. 430), 163.

⁴⁴⁸ See for instance, Andreas Paulus, 'Second Thoughts on the Crime of Aggression', (2009) 20 (4) EJIL, 1117, 1121.

and creation of a supportive infrastructure post-intervention.⁴⁴⁹ States' trepidations that genuine humanitarian interventions could be accidentally mistaken for such acts, exposing their leaders to prosecutions for the crime, may have been exaggerated, if such a 'checklist' is used by the Court.

Additionally, the Rome Statute already provides its own safety net against unwarranted and unjustified prosecutions, and States' concerns that genuine humanitarian interventions could be mistaken for acts of aggression appear to have been exaggerated. Thus, as mentioned above, Article 1 limits application of the Statute to the most serious international crimes, 450 excluding crimes of lesser concern. The Court could simply decide for itself that the violation of the prohibition on the use of force did not cross the gravity threshold to be considered one of those most serious international crimes. Furthermore, bona fide humanitarian intervention ought to be exempt from the possibility of prosecution for the crime of aggression on account of falling short of the mens rea requirement in Article 30 of the Rome Statute of intent and knowledge. 451 Thus, a genuine humanitarian intervention could be distinguished from an act of aggression on the basis that there was no intent by the intervening State to violate the sovereignty, territorial integrity or political independence of the other State, as stipulated in Article 8 bis (2). 452 Article 53 (1) (c) of the Rome Statute also provides the Prosecutor with the discretion to decide not to proceed where a prosecution 'would not serve the interests of justice', 453 and importantly there, therefore, appears to be a place for flexibility when considering all the circumstances. All of these arguments rely on an inherent capacity of the ICC to make

⁴⁴⁹ See the Kosovo Report (n. 437), and Weisbord (2011) (n. 430), 123.

⁴⁵⁰ See the Preamble (n. 85) Article 1 and 5 (1) of the Rome Statute (n. 1). See also the argument presented in the 'Proposal by Egypt and Italy' (n. 403).

⁴⁵¹ Leclerc-Gagne and Byers (2009) (n. 393), 381.

⁴⁵² For a detailed discussion of Article 8 bis (2), see Chapter 3.

⁴⁵³ Article 53 (1) (c) Rome Statute 1998.

an independent assessment based on the individual facts, employing their own determination and interpretation skills. Coming to these findings, the court should also be allowed to conclude that applying a wider threshold than 'manifest violation' is, in certain circumstances, justified.

The need for wider judicial powers to determine whether a violation of the UN Charter is of sufficient gravity to cross the required threshold can also be envisaged in situations other than relating to humanitarian interventions. Thus, it may be arguable that an apparently aggressive act should not fall within the scope of Article 8 *bis*, where the political, economic, or military circumstances are such that they may serve as justification for the act, for instance if the so-called aggressor state has been subjected to some form of severe coercion. One such example might be where a country's supply route is blockaded, and this country then resorts to the use of force to break the blockade for the supply of essential goods.

An argument could also be posed that alternative forms of aggression, for instance economic or other forms of coercion (i.e. the cutting off of energy supplies), might be taken into account as 'other relevant circumstances' to find that such a prior act of coercion led to an act of legitimate self-defence as distinguishable from aggression. This argument is, to some extent, based on Stone's consideration of loopholes in the 1974 definition of the state act. ⁴⁵⁴ Additional support for the availability of other considerations to the judges of the ICC can be derived from the fact that Article 8 *bis*, which is largely based on the definition of the state act of aggression in Resolution 3314, unlike its

⁴⁵⁴ Julius Stone, 'Hopes and Loopholes in the 1974 Definition of Aggression' (1977) 71 AJIL 224, 234 (1977).

predecessor does not expressly exclude justifications for the use of force. Thus, Resolution 3314 had contained a provision in its Article 5, which expressly excluded other considerations potentially amounting to justifications for what would otherwise constitute an act of aggression. Article 8 bis, on the other hand, does not contain a provision akin to Article 5 of Resolution 3314. It is possible to view this omission as deliberate, and therefore as an implied inclusion of the admissibility of other considerations in the determination of whether the threshold of an act of aggression which gives rise to criminal responsibility has been surpassed. The hypothetical ability of the Court to determine justification grounds, based on the absence of a provision corresponding to Article 5 of Resolution 3314, is an interesting consideration, and could be defended in principle by reasoning that the Court makes findings for the determination of criminal liability, not political condemnations of state acts per se.

Notwithstanding the plea for wider determination powers of the Court presented in this thesis, a number of counter-arguments presented by critics ought to be considered. The first and most obvious, discussed in detail above, is that the exclusion of humanitarian interventions was paramount to some States.⁴⁵⁷ The insistence on a narrow manifest threshold thus represents their express intentions and, specifically, the attempt to ensure that exclusion. Article 31 (1) of the Vienna Convention of the Law of Treaties insists that interpretations of treaty provisions require to be given their 'ordinary meaning,' 458 and

.

⁴⁵⁵ Article 5 of Resolution 3314 (n. 347), which states that 'No consideration of whatever nature, whether political, economic, military or otherwise, may serve as a justification for aggression.' ⁴⁵⁶ *Ibid*.

⁴⁵⁷ See section 2.3.

⁴⁵⁸ VCLT (n. 32), Article 31 (1).

proponents of strict treaty interpretations would therefore argue for rigorous adherence to that threshold.⁴⁵⁹

Other concerns are based on a fundamentally protectionist attitude towards the Court, wishing to save it from venturing into the political arena. Thus, Claus Kress for instance, argues that the existing narrow threshold clause 'shields the Court from the need to deal with uses of force that raise questions, which, though amenable to an international *legal* analysis, continue to be a matter of considerable international *legal policy controversy* among states.'460 It would, for instance, be feasible for the Court to come to a potential judicial finding of actual justification for an act of aggression, following the consideration of political, military or economic reasons. Sceptics may perceive such a finding as amounting to a sanction of the use of force by the Court, 461 which is the remit of the Security Council. 462

It is, nonetheless, arguable that the requirement of the Court to conduct a legal analysis of the threshold may inevitably lead to the need for wider determinations. Thus, even despite the high threshold and the necessity for the 'qualitative' and 'quantitative' and 'quantitative' requirements to have been satisfied, the intensity threshold does not necessarily exclude

⁴⁵⁹ See for instance Professor Akande, defending strict adherence to the text of the treaty, even over and above the expression of the underlying intentions of treaty parties. Dapo Akande, 'Treaty Interpretation, the VCLT and the ICC Statute: A response to Kevin Jon Heller & Dov Jacobs', August 25, 2013, http://www.ejiltalk.org/treaty-interpretation-the-vclt-and-the-icc-statute-a-response-to-kevin-jon-heller-dov-jacobs/. For a differing view, see Dov Jacobs 'Why the Vienna Convention should not be applied to the ICC Rome Statute: a plea for respecting the principle of legality' (August 24, 2013), *Spreading the Jam*, https://dovjacobs.com/2013/08/24/why-the-vienna-convention-should-not-be-applied-to-the-icc-rome-statute-a-plea-for-respecting-the-principle-of-legality/ both accessed 20 March 2020. See the discussion below, at pp. 112-5.

⁴⁶⁰ Claus Kress, 'The State Conduct Element' in Kress and Barriga (eds) (2017) (n 246), 543 (emphasis in original). For other protectionist arguments against wider jurisdiction, see Chapter 5.

⁴⁶¹ Heinsch (2010) (n. 376), 742, raising concerns of legality, amongst others.

⁴⁶² Article 39 of the UN Charter (n. 134).

⁴⁶³ The 'character' criterion in Article 8 bis (1), see Kress (n. 460), 519.

^{464 &#}x27;Gravity and scale' in Article 8 bis (1), Kress, ibid.

acts of aggression of lesser consequences, for instance 'limited wars'. 465 Consequently, judges may in certain circumstances come to the conclusion that the failure to reach the level of 'full-scale hostilities' and even entirely 'bloodless invasions', violating the sovereignty of the invaded State, are capable of surpassing the threshold. 466 Therefore, Kress' argument, despite highlighting the need to protect the Court from getting embroiled in political controversies, in fact, actually supports wider determination powers.

Other arguments against expansive exercise of judicial discretion also require acknowledgement. Thus, admittedly, the proposition of greater determination powers of the Court is open to the likely criticism that such a wide interpretation of the definition would broaden the 'objective' standard of the 'manifest' requirement in the Elements of the Crime, ⁴⁶⁷ and this, on its own, could give rise to accusations of uncertainty. Resistance is likely to be incurred, because discretional determinations and justifications potentially offend the principle of *nullum crimen sine lege*, according to which a person should not incur criminal liability for an offence which was not already criminalised at the time of commission of the act. ⁴⁶⁸ Liberal determination by ICC judges of what amounts to a manifest violation may, therefore, fall foul of criticism of retrospective law-making by extending the provisions, if the criminalisation of the act was unforeseeable. ⁴⁶⁹

Related to the question of legality of retrospective determinations is the argument that Article 31 (1) of the VCLT establishes that interpretations of treaty provisions require to

⁴⁶⁵ *Ibid*.

⁴⁶⁶ *Ibid*, 523. Kress cites the 2014 invasion and annexation of Crimea, 607.

⁴⁶⁷ 'Elements of Crime', 8 bis, (2) in Resolution 6 (n. 2).

⁴⁶⁸ See n. 227.

⁴⁶⁹ Similar complaints were raised against the IMTs at Nuremberg, see section 4.3 of Chapter 1. See also Dinstein (2017) (n. 154), 134.

be given their 'ordinary meaning.' ⁴⁷⁰ If this provision is read strictly into the Rome Statute, it may prevent the Court from exercising wider determination powers. Thus, Resolution 3314, the resolution from which Article 8 *bis* of the Rome Statute is derived, did allow for the consideration of 'other circumstances', permitting some discretion. Article 2 of this resolution provided:

The first use of armed force by a State in contravention of the Charter shall constitute *prima facie* evidence of an act of aggression although the Security Council may, in conformity with the Charter, conclude that a determination that an act of aggression has been committed would not be justified *in the light of other relevant circumstances*, including the fact that the acts concerned or their consequences are not of sufficient gravity.⁴⁷¹

Article 8 *bis*, whilst following Article 1 and 3 of Resolution 3314 almost verbatim, does not include the other articles contained in the 1974 resolution. If strict adherence to treaty provisions is sacrosanct, the fact that Article 2 of Resolution 3314, in particular, has been omitted from Article 8 *bis* of the Rome Statute is likely to be read as intentional, potentially precluding the Court's consideration of 'other relevant circumstances'.⁴⁷²

The strict application of the VCLT to the Rome Statute has been upheld, for instance, by Dapo Akande.⁴⁷³ In response, Dov Jacobs has raised the interesting question whether the VCLT ought to apply to the Rome Statute at all, as an instrument of criminal law. He states:

⁴⁷⁰ Article 31 (1) VCLT (n. 458).

⁴⁷¹ Article 2 of Resolution 3314 (n. 347).

⁴⁷² Following Article 31 (1) VCLT (n. 458).

⁴⁷³ See for instance Akande's engagement with the subject in EJIL:Talk! (August 2013) (n. 459).

I've always found the question of the applicable rules of interpretation to international criminal law statutes to be an underdeveloped aspect of the literature on the work of the tribunals [on international criminal law]. As international lawyers, the VCLT is our default go-to document to look for rules of interpretation of international documents. But I believe this fundamentally ignores the specific nature of international criminal law and the central role of the principle of legality.⁴⁷⁴

Thus, Jacobs argues that it is the very requirement of legality in international criminal law which should result in non-application of the 'broad and ultimately discretionary rules of interpretation of the VCLT.' The Rome Statute should be interpreted as an internal rule book for the judges of the ICC rather than as a strictly applicable treaty, since the Rome Statute contains *lex specialis* rules, rather than reiterates the *lex generalis* rules of the VCLT. The provision for strict construction in Article 22 (2) of the Rome Statute, according to Jacobs, applies to the Court's interpretations over and above the VCLT. If, however, a judicial analogy could be perceived to have been foreseeable by a defendant, i.e. it is *that* close, it is not in violation of Article 22 (2). Following Jacob's view, this interpretation of the legality argument may, in fact, support wider determination powers of the Court, as long as the analogy applied is sufficiently close.

Ultimately, this discussion has shown that the vagueness of the term 'manifest violation' is likely to require careful and incremental clarification. As long as discretion is used in

⁴⁷⁴ Jacobs (August 2013) (n. 459).

⁴⁷⁵ *Ibid*.

⁴⁷⁶ *Ibid*.

⁴⁷⁷ *Ibid.* Article 22 (2) of the Rome Statute provides for strict construction of a crime's definition and non-extension by analogy.

⁴⁷⁸ Jacobs (n. 459).

order to avoid the creation of random justifications or inclusion of unforeseeably novel acts of aggression beyond simple analogy and comparison, it should be possible to fend off accusations of violation of the legality principle. 479 Such careful and incremental interpretations are therefore perhaps less likely to encounter resistance, and could at least be a step in the right direction in order to render an extremely narrow definition more workable and applicable to modern aggressive scenarios. Arguably, this would have the benefit of leading to greater effectiveness of the provisions and their general application, thus reemphasising the overall prohibition on the use of force contained in Article 2 (4) of the Charter. This benefit, this thesis argues, outweighs the dangers associated with wider determination powers in the view of some of the above critics.

Nevertheless, the narrow threshold is, in reality, only one of the hurdles presently built into the definitional conditions, which simultaneously enable and limit jurisdiction of the Court. As the following section shows, Article 8 bis (1) also entails the highest of leadership requirements.

3. Limitation to political and military leaders and the 'control or direct' requirement

Article 8 bis (1) imposes a stringent requirement of leadership of only high-level politicians or members of the military, to the exclusion of actors in a lesser or different capacity. Thus, Article 8 bis (1) describes the perpetrator of the crime of aggression as 'a

⁴⁷⁹ For support see Marko Milanovic, 'Aggression and Legality: Custom in Kampala' (2012), 10 (1) JICJ, 165, 171.

person in a position effectively to exercise control over or to direct the political or military action of a State...', 480 and this leadership requirement is reiterated in the Elements section and Article 25 (3) *bis*, limiting accessory liability. Whilst the original Article 25 (3) of the Rome Statute dealt with accomplice liability for the other crimes under the Rome Statute, 481 a specific problem arose for the crime of aggression, because of the specific leadership requirement stipulated in Article 8 *bis*. Therefore a new Article 25 (3) *bis* was included, stating:

In respect of the crime of aggression, the provisions of this article shall apply only to persons in a position effectively to exercise control over or to direct the political or military action of a State.⁴⁸²

As pointed out by Graziano and Mei, Article 25 (3) *bis* therefore, 'on its face, limits the individual responsibility for crimes of aggression in an extraordinary way, excluding the possibility of accessory liability except for those individuals who are in a position 'effectively to exercise control over or to direct' state and military action.'⁴⁸³ This represents an exceptional provision and restriction compared with the other crimes under the Statute, none of which carry an equivalent leadership requirement, nor is accessory liability for these crimes limited in the way it is in aggression cases.⁴⁸⁴ In the search for an explanation for this unusual clause, and attempting to assess its effect on the jurisdiction of the Court, its origin provides a good starting point.

⁴⁸⁰ See p. 85.

⁴⁸¹ Article 25 (3) of the Rome Statute (n. 1).

⁴⁸² Article 25 (3) *bis*.

⁴⁸³ Mackennan Graziano and Lan Mei, 'The Crime of Aggression Under the Rome Statute and Implications for Corporate Accountability' (2017), 58 (4) Harv. Int'l L. J., 55, 57.

⁴⁸⁴ See Article 25 (3) Rome Statute (n. 481), which generally provides for accessory liability for genocide, crimes against humanity and war crimes.

3.1 The historical roots of the leadership clause

A review of the history of the crime of aggression and its predecessor, crimes against peace, 485 shows that the international community had throughout the 20th century battled with the question of how far down the command chain responsibility for the crime should be applied. The crucial issue was whether culpability should arise merely at policy-making level, or whether it ought to capture the executors of that policy, possibly even all the way down to the foot soldier following orders. 486

The historical precedents of the Nuremberg and Tokyo tribunals confirmed the limitation of the crime to high level policy makers. Robert Jackson's opening statement at Nuremberg expressed that the Prosecution had 'no purpose to incriminate the whole German people', but was aiming to prosecute only 'the planners and designers, the inciters and the leaders.... of this terrible war. The crimes against peace precedent was based on a common plan or conspiracy, as well as the individual defendant's effective knowledge of the plan, and actual participation. Article 6 of the London Charter defined crimes against the peace as 'planning, preparation, initiation or waging of a war of aggression, or a war in violation of international treaties, agreements or assurances, or participation in a common plan or conspiracy for the accomplishment of any of the foregoing. Nevertheless, Article 6 also expressly included '[I]eaders, organizers, instigators and accomplices participating in the formulation of a Common Plan or

⁴⁸⁵ See section 3, Chapter 1.

⁴⁸⁶ See section 4.3, Chapter 1.

⁴⁸⁷ *Ibid*.

⁴⁸⁸ See n. 135.

⁴⁸⁹ Article 6 (a) of the London Charter (n. 117).

Conspiracy' to commit any of the crimes in the Charter. Successor tribunals were established by Control Council Law No. 10, 491 of which the Nuremberg Charter was an integral part, and which provided authority to the allies to prosecute German war criminals other than the major ones dealt with by the original Nuremberg Tribunal after the Second World War. Article II (2) describes accessory liability for crimes against peace (the crime of aggression forerunner), war crimes and crimes against humanity as applicable to anyone who 'was (a) a principal or (b) was an accessory to the commission of any such crime or ordered or abetted the same or (c) took a consenting part therein or (d) was connected with plans or enterprises involving its commission or (e) was a member of any organization or group connected with the commission of any such crime.... '493 Subparagraph (f) continued with a reference to the high political, civil or military position or high position in the financial, industrial or economic life. 494

These provisions demonstrate the fact that at Nuremberg a wider range of actual power over the acts of the State, and a lesser degree of individual participation in the act of aggression was considered as sufficient for prosecution. Nevertheless, a position at policy level and actual knowledge of the aggressive plot were essential. The following section examines the ground-breaking prosecutions of instigators, accomplices, and economic leaders in addition to political and military actors.

⁴⁹⁰ *Ibid*, emphasis added to demonstrate criminal liability of accessories and instigators of aggression.

⁴⁹¹ See Control Council Law No. 10 (n. 121).

⁴⁹² See section 4.3, Chapter 1.

⁴⁹³ Article II (2) of Control Council Law No. 10 (n. 121).

⁴⁹⁴ *Ibid*.

3.2. Private economic actors and the development of a 'shape or influence' test

Despite restricting culpability for aggression to policy makers, the tribunals also had to consider the liability of private economic actors. As Noah Weisbord describes:

The tribunals wrestled with the status of business leaders, non-state actors, who did not hold formal positions or exercise effective control in the Nazi's bureaucratic apparatus, but whom the Allied populations felt should be held criminally responsible nonetheless.⁴⁹⁵

Consequently, the Nuremberg and Tokyo Charters did not restrict jurisdiction to political and military leaders only and included complicit accessories and private economic actors, such as industrialists, as long as they had a sufficiently 'high political, civil or military [...] position ... or held high position in the financial, industrial or economic life' of one of the belligerent countries. The Nuremberg Tribunal justified these prosecutions by stating that 'Hitler could not make aggressive war by himself. He had to have cooperation of statesmen, military leaders, diplomats, and business men. The had to have cooperation actors were prosecuted, a high number of them were acquitted, unlike their political and military counterparts, due to the difficulty of attributing the requisite level of knowledge and active participation in the atrocities to them.

⁴⁹⁵ Noah Weisbord, 'Conceptualizing Aggression' (2009), 20 Duke J. Comp. & Int'l. L. 1, at 45. See also Heller (2007) (n. 244), 480.

⁴⁹⁷ Nuremberg Judgment, United States v Goring et al., (n. 3), at para 223.

⁴⁹⁶ See Control Council Law No. 10 (n. 121).

The successor tribunal in the so-called *Farben*⁴⁹⁸ case held that any person 'in the political, military, [or] *industrial* fields [...] who [was] responsible for the formulation and execution of policies' could satisfy the leadership requirement.⁴⁹⁹ The tribunal continued:

The defendants furnished Hitler with substantial financial support which aided him in seizing power and contributed to keeping him in power [...] and worked in close cooperation with the Wehrmacht in organizing and preparing mobilization plans for the eventuality of war.⁵⁰⁰

Despite this contribution to Hitler's efforts, they were, however, acquitted because they were followers rather than leaders. Consequently, they did not have the requisite *mens rea* of knowledge for the leadership crime of waging aggressive war. The so-called *Krupp* case confirmed that industrialists could be convicted of aggression as long as they had knowledge that their actions were aiding the aggressive war. ⁵⁰¹ The tribunal went further in this case, by holding that actual participation in planning, preparing, or initiating the war of aggression was not required, and mere knowledge coupled with actual aiding was sufficient. ⁵⁰² On the facts of the case, however, the defendants in *Krupp* were also acquitted. Nevertheless, both of these cases established specifically that private economic actors could be convicted of aggression, as long as they had the requisite knowledge and a sufficiently high level of influence.

⁴⁹⁸ The *Farben* case (n. 241), at 1124.

⁴⁹⁹ *Ibid*.

⁵⁰⁰ *Ibid*, at 1297.

⁵⁰¹ The *Krupp* case (n. 241).

⁵⁰² *Ibid*, at p. 450. See also Heller (2007) (n. 244), 482-88.

The subsequent *High Command*⁵⁰³ and *Ministries*⁵⁰⁴ cases, on the other hand, concerned military and political leaders rather than private economic actors and placed high emphasis on the leadership requirement. The tribunals reaffirmed that the defendants clearly had to be in a position of policy-making in order to satisfy this leadership standard. Importantly, however, it was considered to be sufficient that they had the ability to 'shape or influence' the policy rather than exercise actual control over it. The *High Command* case confirmed:

It is not a person's rank or status, but his power to shape or influence the policy of his State, which is the relevant issue for determining his criminality under the charge of crimes against peace.⁵⁰⁵

The tribunal continued by stating that '[i]nternational law condemns those who, due to their actual power to shape or influence the policy of their nation, prepare for, or lead their country into or in an aggressive war.' The tribunal also rejected claims made by the defence that the crime was being prosecuted retroactively, and considered that aggression had been a crime at customary international law prior to the Second World War, on the basis of the pre-war treaties of the League of Nations and the Kellogg-Briand Pact. The *Ministries* case applied a similar leadership standard to the *High Command* judgment. Here, the tribunal found the defendant guilty because of 'the wide scope of his

⁵⁰³ The *High Command* Judgment (n. 242), p. 489.

⁵⁰⁴ The *Ministries* Judgment (n. 243), p. 425.

⁵⁰⁵ High Command (n. 242), p. 489.

⁵⁰⁶ *Ibid*.

⁵⁰⁷ *Ibid*, pp. 490-1.

authority and discretion in the positions he held, and which enabled him to shape policy and influence plans and preparations for aggression.'508

Consequently, these cases emphasised that aggression was a leadership crime, which could only be committed by policy makers. The *High Command* case also re-asserted that the criminal act was the actual 'planning, preparation, initiation and waging of war or the initiation of invasion' rather than the mere aiding of such an act, as applied in the *Farben* case. Despite this disagreement with *Farben* on the required level of participation, the *High Command* case confirmed that the ability to 'shape or influence' was the minimum standard rather than the capacity to assert actual and direct control over the actions of a State. The International Military Tribunal for the Far East, followed the Nuremberg reasoning and also applied the 'shape or influence' test, enshrining it further in customary international law. Thus, the question for the Tribunal was whether the defendant had 'attained a position which by itself enabled him to *influence the making of policy*... and had become aware that Japan's designs were criminal. Dissenting opinions centred on the argument that aggression was, at the time, not a crime at international law, rather than voicing disagreement with the actual test applied.

These precedents demonstrate that the jurisprudence of the tribunals included private economic actors, and applied a standard of 'shape or influence' rather than a more restrictive 'control or direct' test, although it was also clearly established that aggression

⁵⁰⁸ Ministries case (n. 243), p. 425. Even Judge Powers' dissenting judgment referred to a position of influence rather than 'control', p. 889.

⁵⁰⁹ High Command, (n. 242), pp. 490-491.

⁵¹⁰ *Ibid*, 489, confirmed in *Ministries* (n. 243), 425.

⁵¹¹ The *Tokyo* Judgment (n. 11), 1190-1991.

⁵¹² *Ibid*, emphasis added.

⁵¹³ *Ibid*, see for instance Justice Pal of India's dissenting opinion. This argument had already been rejected in the *High Command* case (n. 242), p. 489.

is indeed a leadership crime. This discussion raises the question where the control or direct test originated, if not in the post-war tribunals.

3.3 The control or direct test

The origins of a narrower 'control or direct' test, and specifically one of 'effective control', can be found in the civil proceedings of the much later *Nicaragua* case. ⁵¹⁴ Here, the International Court of Justice applied the phrase as a test to determine whether the United States had committed a state act of aggression against Nicaragua through their extensive support of the Contra Guerrillas. Nicaragua had brought the case before the ICJ, alleging that the United States were effectively in control of the Contras and their activities against the State. It is in response to this specific allegation that the ICJ applied Nicaragua's own term of effective control, by asserting that 'whether the United States Government at any stage devised the strategy and directed the tactics of the *contras*' depended on the extent to which the United States had 'effective control' of the military or paramilitary forces. ⁵¹⁵ The ICJ thus developed a connection between the level of control of the party aiding the act of aggression and directing the tactics involved, and the dependence of the party executing the act of aggression on its controller. Legal responsibility would only arise, if this level of control could be established.

The International Criminal Tribunal for the former Yugoslavia (ICTY) adopted a different 'control' test in the case of *Tadic*, modifying it into a wider one of 'overall

⁵¹⁴ *Nicaragua* (1986) (n. 37), paras 110, 115.

⁵¹⁵ *Ibid*, paras 110, 115.

control'. 516 Although the main issue considered in Tadic was whether the armed conflict

in Bosnia at the time was internal or international, it was a relevant and important question

for the tribunal whether the acts of an organized and hierarchically structured group could

be attributed to a State, provided that that State had 'overall control' over the group. The

Appeals Chamber held:

The requirement of international law for the attribution to States of acts performed

by private individuals is that the State exercises control over the individuals. The

degree of control may, however, vary according to the factual circumstances of

each case. The Appeals Chamber fails to see why in each and every circumstance

international law should require a high threshold for the test of control.⁵¹⁷

Thus, the Tadic standard of overall control allowed for the fact that subordinates can act

independently, and yet remain under the overall control of the State in question.⁵¹⁸ The

Appeals Chamber continued:

The 'effective control' test propounded by the International Court of Justice as an

exclusive and all-embracing test is at variance with international judicial and State

practice: such practice has envisaged State responsibility in circumstances where

a lower degree of control than that demanded by the Nicaragua test was

exercised.519

⁵¹⁶ *Tadic* (n. 115), para 117.

⁵¹⁸ Dinstein (2017) (n. 154), 238.

⁵¹⁹ *Tadic* (n. 115), para 124.

124

The ICTY therefore clearly distanced itself from the higher standard of effective control applied in *Nicaragua*. Nevertheless, the ICJ subsequently confirmed its own *Nicaragua* standard in the 2005 *Armed Activities case*, ⁵²⁰ and in the 2007 *Genocide case*, ⁵²¹ rejecting the 'overall control' test advocated in *Tadic*.

The ILC's commentary on Article 8 of the 2001 Draft Articles on State Responsibility also considered the 'effective control' standard of *Nicaragua* as appropriate, ⁵²² despite conceding that the ICJ's opinion concerned State responsibility rather than the individual criminal responsibility considered by the ICTY in *Tadic*. ⁵²³ This view is astounding, because it is in direct contrast to the ILC's earlier approach in their 1996 Draft Code. Not only does Article 16 of that Draft Code apply the term 'leader or organizer' as taken directly from the Charter of the Nuremberg Tribunal, ⁵²⁴ but related commentary clearly confirms an interpretation in line with the Nuremberg precedent:

These terms must be understood in the broad sense, that is to say, as referring, in addition to the members of a Government, to persons occupying high-level posts in the military, the diplomatic corps, political parties and industry, as recognized by the Nuernberg Tribunal. 525

⁵²⁰ Armed Activities (2005) (n. 37), para 226. The ICJ here cited the ILC's endorsement in support.

⁵²¹ The Genocide case (n. 438), paras 207-210. The level of control required was interpreted as 'complete dependence' on the State in this case (at para 207), an even higher degree of control than the one suggested in *Nicaragua* (1986) of a 'degree of dependence' (n. 37, at para 62), see Dinstein (2017), (n. 154), 240.

⁵²² Draft Articles on State Responsibility for Internationally Wrongful Acts (with commentaries), Report of the International Law Commission, 53rd Session, 2 (2) Yearbook of the International Law Commission, 2001, Annex to GA Resolution A/RES/56/83 (12 December 2001), at 105.

⁵²³ *Ibid*, at 108. For a discussion see Dinstein, (2017), (n. 154), 239.

⁵²⁴ Article 6 (a) London Charter (n. 117).

⁵²⁵ Para 2 of Commentary on 1996 ILC *Draft Code of Crimes against the Peace and Security of Mankind*, Yearbook ILC (1996), Vol. II, 15, 55, drawing on a reference to the *Nuremberg* Judgment (n. 3).

This demonstrates that the ILC in 1996 considered the 'shape or influence' standard to be correct, and therefore were able to envisage the potential prosecution of economic leaders and possibly even spin doctors, as long as they were in a position of high influence. The ILC does not provide an explanation for its change in stance, however it seems logical and likely that by 2001 they paid heed to the majority of States' preference for severe restrictions on the Court's jurisdiction over aggression.

As for the different stances taken by the ICJ and the ICTY, it was noted above that the ICJ concerned itself only with state responsibility and the connection between the State and the military or paramilitary group in question. The ICTY, on the other hand, had to consider individual criminal responsibility within a group, as well as its nexus to state responsibility. See Because of this application to individual criminal responsibility, and despite the narrower views of the ICJ on state responsibility, the ICTY's advocacy of an overall control standard appears to be more appropriate for the crime of aggression, although it is still far removed from the even wider Nuremberg precedent, which had been recognised as appropriate in the 1996 ILC Draft Code.

Nevertheless, *Nicaragua's* 'effective control' was to become enshrined as the new leadership standard for aggression in the 2010 definition of the crime of aggression. This is extremely narrow and certainly far more restrictive than the Nuremberg standard of being able to 'shape or influence' policies. Consequently, and because the *Nicaragua* case considered state responsibility at civil law rather than individual criminal liability, a plausible argument can be made that its 'effective control' test serves badly as a definitive precedent for the crime of aggression. The Nuremberg test of 'shape or influence' would,

⁵²⁶ *Tadic* (n. 115), para 137.

therefore, have been more appropriate for the determination of individual criminal responsibility and, this thesis argues, should have been applied by the SWGCA. 527

3.4 The practical implications of the choice to apply the control/direct test to the

crime of aggression

The difference between the Nuremberg test of 'shape and influence', *Nicaragua's* 'effective control' and *Tadic's* 'overall control' is more than semantics. Nuremberg's broad concept, applying a 'shape or influence standard', acknowledges the influence of policy makers on the conduct of States, therefore capturing war mongers, spin doctors and potentially economic supporters of and profiteers from war efforts. The 'overall control' standard in *Tadic*, on the other hand, is more restrictive. *Tadic* viewed the link between state and group acts in the context of individual criminal responsibility thus:

Control by a State over subordinate *armed forces or militias or paramilitary units* may be of an overall character (and must comprise more than the mere provision of financial assistance or military equipment or training...The control required by international law may be deemed to exist when a State (or in the context of an armed conflict, the Party to the conflict) *has a role in organising, coordinating or planning the military actions of the military group,* in addition to financing, training and equipping or providing operational support.⁵²⁸

⁵²⁷ For support, see Weisbord (2008) (n. 176), 165-6.

⁵²⁸ *Tadic* (Appeals Chamber, Merits) IT-94-1-A, para 137 (emphasis made in original text).

This concept is therefore narrower than the Nuremberg standard, but still considerably wider than the view taken by both the ICJ and latterly the ILC, ⁵²⁹ which applied the highest standard of a requirement for 'effective control or direct[ion]', as eventually adopted in Kampala for the definition of the crime of aggression. This standard creates individual criminal responsibility only where a person is capable of directly controlling the political or military action of a State, at the highest possible level, rather than exercising general or overall control. Even if accessory or lesser participation hadn't been expressly excluded in Article 25 (3) *bis*, ⁵³⁰ the extremely stringent requirement of a high level of control would also serve to exclude such liability by accomplices of a lesser capacity.

This strict leadership requirement, now formally implemented in the Crime of Aggression, has been criticized as representing a substantial retreat from the principles of the International Military Tribunals at Nuremberg and Tokyo. ⁵³¹ The Special Working Group's elevation of the standard to a 'control or direct' requirement also paradoxically contradicts their own statement that the IMT's and NMT's jurisprudence 'codified customary international law'. ⁵³² Thus the Special Working Group had applied the 'control or direct' standard throughout their discussions, but only in the misguided belief that the IMT and NMT utilized this standard rather than the one of 'shape or influence', which was actually applied by the tribunals. ⁵³³ This erroneous perception by the SWGCA is highly puzzling, since it appears that they failed to take account of its predecessor's own

⁵²⁹ See the 2001 ILC Draft Articles on State Responsibility (n. 522 and related text).

⁵³⁰ Article 25 (3) *bis* (n. 482).

⁵³¹ See Petty (2008) (n. 228). See also Heller (2007) (n. 244), and Ferencz (2009) (n. 4).

⁵³² Report of the Special Working Group on the Crime of Aggression, ICC-ASP/4/SWGCA/1 (1 Dec. 2005) Annex II.A, para. 26.

⁵³³ Heller (2007) (n. 244), 479, 488.

Historical Review.⁵³⁴ This lengthy document examined the jurisprudence of the IMT's at Nuremberg and Tokyo in great detail, and confirmed time after time that the standard applied in those precedents was that of 'shape or influence'.⁵³⁵

No reasonable explanation for the SWGCA's perplexing misinterpretation has been encountered during the research for this thesis, but it is likely that they succumbed to the pressure of States' preference for the narrower standard, wishing to keep the definition and its leadership clause highly limited. Even as early as 2002, the Coordinator's Discussion Papers of April⁵³⁶ and July⁵³⁷ expressed confidence that general agreement appeared to have been reached over the leadership clause defining the defendant as being in a position 'to exercise control over or direct the political or military actions of a State'.⁵³⁸ This confidence in general agreement on the clause was based on a number of States' Proposals, demonstrating wide support for this vision of aggression as a highly limited leadership crime. Examples are for instance a number of Proposals by

Preparatory Commission for the International Criminal Court, Working Group on the Crime of Aggression, 'Historical review of developments relating to aggression, prepared by the Secretariat', PCNICC/2002/WGCA/L.1., particularly the set of tables, demonstrating the Nuremberg 'shape or influence' standard, in PCNICC/2002/WGCA/L.1/Add.1, available at http://www.un.org/law/icc/documents/aggression/aggression/aggressiondocs.htm, accessed 12 October 2020.

⁵³⁵ *Ibid*.

⁵³⁶ n. 413.

⁵³⁷ n. 316.

Proposed by the Chairman' (n. 415) at I(1)(a)-(b), ICC-ASP/5/SWGCA/2, Annex (ICC, Assembly of States Parties, Resumed 5th Sess., New York, Jan 29-Feb 1, 2007), available at http://www.icc-cpi.int/iccdocs/asp_docs/SWGCA/ICC-ASP-5-SWGCA-2_English.pdf, and the 'Non-Paper by the Chairman on Defining the Individual's Conduct', in ICC, Assembly of States Parties, Special Working Group on the Crime of Aggression, 6th Sess., New York, Nov. 30-Dec. 14, 2007, *Report of the Special Working Group on the Crime of Aggression*, Annex, ICC-ASP/6/SWGCA/1, http://www.icc-cpi.int/iccdocs/asp_docs/SWGCA/ICC-ASP-6-SWGCA-1_English.pdf.

Germany,⁵³⁹ the Arab States,⁵⁴⁰ Greece and Portugal,⁵⁴¹ Colombia,⁵⁴² Guatemala,⁵⁴³ and Belgium, Cambodia, Sierra Leone and Thailand.⁵⁴⁴

Notwithstanding this majority position, it is noteworthy that some delegations of the Assembly of States Parties to the Rome Statute began to recognise the misinterpretation and dichotomy of the SWGCA's application of the 'control or direct standard' and attempted to widen it to one of 'shape or influence' in line with the Nuremberg precedents. For instance in 2002, Colombia's delegation distanced itself from its earlier insistence on the control or direction requirement,⁵⁴⁵ and suggested that criminal liability for aggression should be extended to persons 'in a position to contribute to or effectively cooperate in shaping in a fundamental manner political or military action by a State.⁷⁵⁴⁶ The use of the words 'contribution', 'cooperation', and 'shaping' suggests a far wider standard of policy making capacity than 'control or direct', and applies similar terminology to the Nuremberg standard of 'shape or influence.'⁵⁴⁷ It is possible that this more encompassing definition of the Colombian delegation in 2002 could have captured other actors essential to an act of aggression, for instance policy makers, spin doctors, and economic investors and profiteers from war. The Samoan delegation also preferred a wider leadership standard and took the view that the perpetrator need not even necessarily

⁵³⁹ 'German Proposals' of 1997 (n. 395), 1999 (n. 398), and 2000 (n. 399).

⁵⁴⁰ 'Proposal by the Arab States and Iran', 1 July 1998, UN Doc. A/CONF.183/C.1/L.37

⁵⁴¹ 'Proposal by Greece and Portugal' (1999) (n. 405).

⁵⁴² 'Proposal by Colombia' (2000) (n. 406).

⁵⁴³ 'Proposal by Guatemala', PCNICC/2001/WGCA/DP.2, 26 September 2001.

⁵⁴⁴ 'Proposal submitted by Belgium, Cambodia, Sierra Leone, and Thailand', 8 July 2002, PCNICC/2002/WGCA/DP.5

⁵⁴⁵ 'Proposed text on the definition of the crime and act of aggression submitted for the Delegation of Colombia' (1 July 2002), PCNICC/2002/WGCA/DP.3 ⁵⁴⁶ *Ibid.*

⁵⁴⁷ See section 3.1 and 3.2.

be a formal member of the government or military, in order to control or direct of the political or military actions of a State.⁵⁴⁸

This latter position represents an interesting compromise. Whilst applying the narrow control or direct requirement preferred by the majority, the Samoan proposal would also enable capturing a wider group of possible aggressors in the spirit of the IMTs. Thus the Nuremberg Charter had considered not only the culpability of 'leaders' and 'organizers', but also 'instigators' and 'accomplices participating in the formulation or execution of a Common Plan or Conspiracy' to commit crimes against peace, war crimes or crimes against humanity. ⁵⁴⁹ Similarly, the Samoan proposal suggests indirectly that the influence of the defendant over the political and military actions of a State and the nexus between his or her actions and the State act of aggression are far more important than his or her actual position.

Such a more comprehensive concept, had it been adopted, would have widened the scope of the narrow Article 8 *bis* significantly, thereby potentially expanding the Court's jurisdiction. An interesting variation of the control or direct standard can also be found in Cuba's 2003 proposal, according to which the definition should apply to individuals 'in the position of effectively controlling or directing the political, *economic*, or military actions of a State.'550 This appears to slightly widen the influence level of the perpetrators by including economic actors, whilst still insisting on the requirement to exercise actual control. These latter proposals can be distinguished from the mainstream preference for a narrow leadership requirement, by potentially capturing *all* persons in a position to

^{548 &#}x27;Proposal by Samoa' (n. 409).

⁵⁴⁹ Article 6 of the London Charter (n. 117).

⁵⁵⁰ 2003 'Proposal by Cuba' (n. 407).

decisively influence the policies of a State, including political, social, business and even spiritual leaders. This leadership standard would have captured culpable instigators of aggression, and incorporating this level of influence would also have been far more in line with the Nuremberg and Tokyo precedents.⁵⁵¹

Despite eventually succumbing to the majority preference for the application of the stricter 'control or direct' test, the Special Working Group repeatedly acknowledged the existence and validity of the argument that non-military or non-political leaders capable of shaping or influencing the actions ought to be captured. For example, the 2006 Princeton Report referred to 'consensus among participants that aggression should be understood as a leadership crime', but conceded 'that the leadership clause should refer to the ability to influence policy'. The Chairman's Discussion Paper describes the state of negotiations in 2007 as follows:

Some states' objective had been to replace the phrase 'being in a position... to exercise control over or to direct' with 'being in a position... to shape or influence' and to thus assure in particular the inclusion of private economic actors such as industrial leaders... The Chair pointed out that it had always been understood that the 'control or direct' formula covered this group in any event.⁵⁵³

⁵⁵¹ See the discussion of the *Farben*, *Krupp*, *Ministries* and *High Command* cases (in section 3.2), and Heller (2007) (n. 244).

⁵⁵² 'Informal Intersessional Meeting of the Special Working Group on the Crime of Aggression', held at the Liechtenstein Institute on Self-Determination, Woodrow Wilson School, at Princeton University, New Jersey, United States, from 8 to 11 June 2006, (the 2006 *Princeton Report*), ICC-ASP/5/SWGCA/INF.1, at para 88.

⁵⁵³ See Coalition for the International Criminal Court, 'Observations about the Discussion Paper Proposed by the Chairman', ICC-ASP/5/SWGCA/2 (25 Jan. 2007), at 5.

This statement reiterated the same point made at the 2006 Intersessional Meeting 'that it had been always understood that the leadership clause would reach just as far [as covering industrialists and financiers] and that it had never been limited to heads of state or individuals in the military'.⁵⁵⁴

Even as late as 2009, the Special Working Group recognized in its report of its final session the existence of the view amongst some States that the leadership clause ought to be sufficiently wide to include non-governmental actors who held 'effective control over the political or military actions of a state,' for instance high-level industrialists.⁵⁵⁵ Notwithstanding this acknowledgment, a provision representing this view failed to be incorporated into the final definition of Article 8 *bis* due to the vehement resistance of a number of powerful States against a wider clause. Consequently, Article 8 *bis* employs the narrowest of leadership standards, restricting potential liability to political and military leaders at the highest level only. The prosecution of private economic actors now appears to be entirely precluded.

In practice, the limitation of the definition to only the highest political and military leaders is likely to amount to a vast restriction of the Court's jurisdiction. If interpreted literally, the definition may prove to be inapplicable in a great number of modern aggression scenarios, which are often executed through the exertion of soft influence rather than the exercise of old-fashioned hard power through traditional politics and military operations. The formal position and effective control of individuals may in reality be neither apparent

⁵⁵⁴ 'Informal Intersessional Meeting of the Special Working Group on the Crime of Aggression', ICC-ASP/5/SWGCA/INF.1 (5 Sept. 2006).

⁵⁵⁵ Report of the Special Working Group on the Crime of Aggression (7th session, 2nd resumption, New York, 9-13 February 2009).

nor essential when committing acts of aggression.⁵⁵⁶ An example is presented, for instance, by Robert Heinsch, pointing to the modern problem of religious leaders,⁵⁵⁷ who may be in a position to influence a State's actions towards aggressive use of force, a situation which has arisen for some time in Iran and Afghanistan, and recently in the Middle East and North Africa following the Arab Spring uprisings. Other influencers could be spin doctors or war mongers supporting aggressive war efforts for power or financial gain. Armament and weapons manufacture is also a profitable business, and the prospect of profit is likely to provide a strong motive for an incitement of aggressive war efforts. Large contributions to election campaigns and lobbying by the armament industry may influence a political leader to consider an act of aggression they might otherwise not be tempted to undertake.

Allowing the ICC discretion to view the political or military position as a mere indicator of the defendant's presumed influence over state acts would be, this thesis argues, a beneficial alternative. Intellectual or economic influencers inciting aggressive state acts could be deemed to be instrumental to such acts, giving rise to potential prosecutions, just as they did in the Nuremberg precedents. Support for the inclusion of such highly influential individuals in a non-political or non-military capacity can also be derived from the application of a similar logic in the crime of genocide, which has been declared as falling within 'the category of crimes so serious that direct and public incitement to commit such a crime must be punished as such.'558 The crime of aggression has been

.

⁵⁵⁶ See, for instance the arguments presented by Dov Jacobs, 'The Sheep in the Box: The Definition of the Crime of Aggression at the International Criminal Court' (2010) The Review Conference and the Future of the International Criminal Court, Proceedings of the First AIDP Symposium for Young Penalists in Tuebingen, Germany, Kluwer Law International, 131, 144.

⁵⁵⁷ Heinsch (2010) (n. 376), 722, 723.

⁵⁵⁸ See *Prosecutor v JP Akayesu*, Case No. ICTR-96-4-T, Judgment, 2 September 1998, para 562. Liability for incitement to genocide arises from Article 25 (3) (e) of the Rome Statute.

deemed to be 'the supreme international crime', and as such merits sincere and effective prosecutions of persons instigating aggressive use of force. The Rome Statute already restricts possible prosecutions to only the worst perpetrators of the most serious crimes, ⁵⁵⁹ greatly limiting the possibility of frivolous prosecutions. An excessive emphasis on and restriction to political and military leadership is as unnecessary as it is inappropriate in the face of modern types of aggressors in an age of a growing phenomenon of cyberwars and information warfare. A wider clause incorporating the potential liability of intellectual, spiritual, or economic influencers at a high-level would have been more appropriate.

3.5 Participation at a lower level and other modes of conduct

An issue closely related to the leadership clause is the question of exactly what conduct amounts to an act of aggression by a person in control over the political or military action of a State. Article 8 *bis* (1) settles this question by simply naming this conduct as 'the planning, preparation, initiation or execution' of an act of aggression. During the negotiation stage, however, some suggestions were made to include other forms of participation, which took into account the possibility that, just as in respect of the other crimes in the Rome Statute, there existed a variety of participation levels including, as considered in Article 25 (3) (b), 'order[ing], solicit[ing] or induc[ing]' the commission of aggression. ⁵⁶⁰

⁵⁵⁹ See Preamble (n. 85), Articles 1 and 5 of the Rome Statute (n. 1).

⁵⁶⁰ Article 25 (3) (n. 481). For the suggestion of applying these modes of participation to the crime of aggression, see the '2007 Discussion Paper on the crime of aggression proposed by the Chairman', ICC-ASP/6/SWGCA/INF.1, and Annex.

In particular during the 6th session of the Special Working Group in 2007, a new differentiated approach proposed that the offender '(leads) (directs) (organizes) and/or directs) (engages in) the planning, preparation, initiation or execution of an act of aggression/armed attack.'⁵⁶¹ The differentiated conduct clause, and the inclusion of other modes of participation did, however, not gain the support of the majority, who insisted on a strict leadership requirement in order to exonerate from criminal responsibility persons of a lesser capacity. Instead, there was much support for the inclusion of the new draft Article 25 (3) *bis*,⁵⁶² creating an exception to the other modes of participation in Article 25 (3) and excluding expressly the liability of mere soldiers obeying orders, or aiders and abettors.⁵⁶³ Despite continuing resistance by the minority, the majority view was eventually adopted into the 2010 definition.⁵⁶⁴

This exception and exclusion of other modes of liability can be seen as another retreat from the principle laid down in the precedents of Nuremberg and Tokyo. Article II (2) of Control Council Law No 10 had described accessory liability for crimes against peace (the forerunner of aggression), war crimes and crimes against humanity as follows:

Any person without regard to nationality or the capacity in which he acted, is deemed to have committed a crime as defined in paragraph 1 of this Article if he

⁵⁶¹ ICC-ASP/6/SWGCA/INF.1, and Annex, ICC-ASOP/5/SWGCA/2, pp. 3-5, Variant (a). A monistic option was retained in variant (b) identifying the individual aggressor as a person who simply 'order[ed] or participate[d]' actively in the criminalised act. The precursors of 'intentionally and knowingly' from the 2002 Paper had been dropped in line with the majority opinion. For a detailed discussion of the differentiated versus monistic approach, see Kress and von Holtzendorff (2010) (n. 5), 1184.

⁵⁶² See n. 482. ⁵⁶³ For Article 25 (3) and Article 25 (3) *bis*, see n. 481-2.

⁵⁶⁴ See the Discussion Papers in the *Travaux Preparatoires* (n. 317) and Resolution 6 (n. 2).

⁵⁶⁵ See the above discussion in section 3.1.

was (a) a principal or (b) was an accessory to the commission of any such crime or ordered or abetted the same or (c) took a consenting part therein or (d) was connected with plans or enterprises involving its commission or (e) was a member of any organization or group connected with the commission of any such crime or (f) with reference to paragraph 1 (a) if he held a high political, civil or military (including General Staff) position in Germany or in one of its Allies, cobelligerents or satellites or held high position in the financial, industrial or economic life of any such country. ⁵⁶⁶

This paragraph demonstrates that accessory liability was indeed far wider in the Nuremberg precedent than in the modern definition of aggression.

3.6 The accessory liability doctrines of Conspiracy and Joint Criminal Enterprise

It may be possible to apply the liability doctrine of Joint Criminal Enterprise to the Crime of Aggression, as a way of 'getting around' the narrow conduct clause and the control or direct requirement in the 2010 definition, thereby widening jurisdiction. The doctrine of conspiracy, by comparison, is usually invoked after the event, and this would be certain to raise legality concerns on account of creating liability retrospectively. Furthermore, conspiracy is not recognized by most civil law jurisdictions, and international criminal

⁵⁶⁶ Article II (2) of Control Council Law No. 10 (n. 121).

⁵⁶⁷ See, for instance, the discussion in Weisbord (2009) (n. 495), at 55-9, who views the Joint Criminal Enterprise doctrine as 'the most appropriate liability doctrine for the definition of aggression', 58. For the purpose of the ICC's jurisdiction over the crimes in Article 5 of the Rome Statute, Cassese contends that it is 'joint control' over the criminal act and potential 'indirect perpetration' through another person, which constitutes the relevant basis of accessory liability. See Cassese (2013) (n. 12), 163-179, in particular 178.

⁵⁶⁸ See Robert Cryer et al. *An Introduction to International Criminal Law and Procedure* (CUP 2010), 228.

law currently recognises conspiracy only in relation to the crime of genocide.⁵⁶⁹ Whilst the Nuremburg IMT attempted to prosecute conspiracy to commit crimes against peace, this crime was a separate offence rather than a connection by which group culpability could be transferred to an individual.⁵⁷⁰ It is therefore doubtful that conspiracy could be easily applied to the modern Crime of Aggression.⁵⁷¹

The doctrine of Joint Criminal Enterprise, on the other hand, may fare better as a liability doctrine for the Crime of Aggression, since it is 'a common plan, design or purpose which amounts to ... the commission of a crime'⁵⁷². In the *Tadic* case the ICTY read JCE into its statute,⁵⁷³ and in that and subsequent ICTY cases 3 forms of enterprise participation were identified. The first one consisted of a 'shared intent to bring about a certain offence', the second of 'organized systems of repression and ill-treatment', and the third of 'criminal acts beyond the common design, but a natural and foreseeable consequence of effecting it'.⁵⁷⁴

Joint Criminal Enterprise is also recognized generally in the Rome Statute in Article 25 (3) (d).⁵⁷⁵ Reading it into the Crime of Aggression therefore would make perfect sense,

⁵⁶⁹ In Article III of the 1948 Genocide Convention, Convention on the Prevention and Punishment of the Crime of Genocide, 9 December 1948. However, Article III is only partially included in Article 6 of the Rome Statute, unlike in the Statutes of the ICTY and ICTR.

⁵⁷⁰ See the discussion above.

⁵⁷¹ See also Cassese (2013) (n. 12 and 567 above).

⁵⁷² Prosecutor v Kupreskic, Case No IT-95-16-T, Judgment (Jan 14, 2000), para 772.

⁵⁷³ For *Tadic*, (IT-94-1-A) (n. 115), Appeals Chamber Decision 15 July 1999, paras 193, 200.

⁵⁷⁴ See Weisbord, (2009) (n. 495), 58, citing A.M. Danner &J.S. Martinez, 'Guilty Associations: Joint Criminal Enterprise, Command Responsibility, and the Development of International Criminal Law', (2005) 9 CAL. L. Rev. 75, 120.

⁵⁷⁵ Article 25 (3) (d) of the Rome Statute provides for criminal liability of a person who '[i]n any other way contributes to the commission or attempted commission of such a crime by a group of persons acting with a common purpose. Such contribution shall be intentional or shall either (i) [b]e made with the aim of furthering the criminal activity or criminal purpose of the group, where such activity or purpose involves the commission of a crime within the jurisdiction of the Court; or (ii) [b]e made in the knowledge of the intention of the group to commit the crime.'

allowing the ICC and its prosecutors to locate individual responsibility within a wide range of organizations. Roger Clark, for instance, asserts that the drafters of the Aggression amendment considered the application of Article 25 (3) (d) to the crime of aggression, and JCE therefore ought to be a culpable form of participation in aggression.⁵⁷⁶ He also derives support for the application of JCE to the crime of aggression from the Nuremberg precedent, citing directly:

A plan in the execution of which a number of persons participate is still a plan, even though it is conceived by only one of them; and those who execute the plan do not avoid responsibility by showing that they acted under the direction of the man who conceived it. Hitler could not make aggressive war by himself. He had to have the co-operation of statesmen, military leaders, diplomats and businessmen. When they, with knowledge of his aims, gave him their cooperation, they made themselves parties to the plan he had initiated.⁵⁷⁷

Whilst Roger Clark's views have received fervent support, and have been developed even further to insist that the entire future of individual liability for the crime of aggression lies within JCE,⁵⁷⁸ they have also encountered strong opposition. Thus Robert Heinsch, for instance, interprets the narrow leadership clause of Article 8 *bis*, reiterated by Article 25 (3) *bis*, as determinatively limiting accessory liability.⁵⁷⁹ Following this argument, it should not be possible to prosecute perpetrators in a less responsible position simply for the crime of aggression through the 'backdoor' of accessory responsibility. A different, if familiar, argument against the inclusion of JCE has been presented by Mark Osiel,

⁵⁷⁶ See Roger S. Clark, 'Individual Conduct', in Kress and Barriga (eds.) (2017) (n. 246), 568.

⁵⁷⁷ Clark, *ibid*, citing from the Nuremberg Judgment (n. 3), pp. 223, 226.

⁵⁷⁸ See for instance Noah Weisbord, (2009) (n. 495), 58.

⁵⁷⁹ See Heinsch, (2010) (n. 376), 734.

criticising the possible application of the doctrine in criminal aggression for violating the legality principle, since the defendant cannot know in advance how inclusively the 'common purpose' will be interpreted.⁵⁸⁰ Consequently, Osiel argues, the application of JCE to the Crime of aggression is 'dangerously illiberal'.

This view can, however, be countered by the fact that the high standard of a leadership qualifier and the narrow conduct clause would prevent prosecution on the basis of JCE, where such 'common purpose' is not blatantly obvious. Fears of abuse and illegality should therefore be unnecessary. Interpretations by the judiciary through close analogy may be necessary and ought to be permitted.⁵⁸¹ The development of precedents may be required for effectiveness of the overly narrow conduct and leadership clause, and consistent interpretations will hopefully create a predictable concept of aggression 'consonant with the differing dimensions of mass atrocities,'⁵⁸² to use Osiel's own words.

3.7 An overall evaluation of the strict leadership requirement

In evaluation, doubts arise whether the stringent leadership requirement in the Kampala definition of aggression is capable of dealing with modern challenges. Even more traditional forms of warfare now exclude potential culpability of perpetrators of acts of

⁵⁸⁰ M. Osiel, 'The Banality of Good: Aligning Incentives against Mass Atrocity', (2005) 105 Colum. L. Rev. 1751, 1770. For the familiar argument of *nullem crimen sine lege*, see the discussion above, and section 4.3 in Chapter 1.

⁵⁸¹ On this point, it is acknowledged that Article 22 (2) of the Rome Statute (n. 477) limits wide analogy, insisting on strict construction of the Statute. Article 31 (1) of the VCLT also reinforces strict application of treaty provisions, and this view has been strongly defended, for instance, by Dapo Akande (n. 459). The point made here is that without wider interpretation, the definition is simply not functional.

aggression at less than the highest political or military level. Mid to upper level support, which is essential to classic aggressive war-making is now not captured, because of the exclusion of accessories in Article 25 (3) *bis*.⁵⁸³

The provision therefore makes prosecutions of aides and accessories to the crime of aggression highly unlikely, despite the fact that, without their efforts, the aggressive act might never have succeeded. The resulting likelihood of prosecutions, against top leaders only, can be criticised as taking place 'in a vacuum' in the absence of jurisdiction over these aides and accomplices. This protection afforded to perpetrators at only a slightly lower level is contrary to the Nuremberg and Tokyo precedents, and effectively means bestowing collective innocence. On such aides and accomplices without whom an act of aggression could not be executed.

Furthermore, the above discussion on the requirement to 'control or direct' versus 'shape or influence', and reference to political or military leaders only, has demonstrated that it is now virtually inconceivable that any non-governmental or non-military actor could be prosecuted for the Crime of Aggression. Consequently, the application of individual criminal responsibility to private economic actors is now extremely unlikely. The potential prosecution of industrial or financial leaders of sufficient influence could have sent a message to bankers, weapons manufacturers and traders, financiers, and ideological spin doctors that they would not be immune from prosecution for aggression for their participation at instrumental level. This avenue now seems barred.

⁵⁸³ See n. 482.

⁵⁸⁴ See M. Drumbl, 'The Push to Criminalize Aggression: Something Lost Amid the Gains?' (2009), 41 (2) Case W. Res. J. Int'l L. 291, 315.

⁵⁸⁵ Heller (2007) (n. 244), see also Drumbl, *ibid*.

⁵⁸⁶ Drumbl, *ibid*, 316.

Similarly, the leadership requirement would now also render it extremely difficult to attribute liability to complicit third party states' officials, a situation which may arise in a proxy war. The leadership standard has been set so high, that a political or military leader would have to be in a position to control the political and military actions of both their own State as well as those of the third party aggressor state, operating as a puppet. The only possible example of such a high level of control may be 'a superpower's use of a client state to fight a proxy war – a situation that certainly existed during the Cold War.'587 The post-War history demonstrates third party states' frequent recourse to blatant support of such acts of aggression.⁵⁸⁸ Due to the development of modern warfare, ⁵⁸⁹ and the ability to conduct acts of force with minimal exposure of the aggressor state, such proxy wars have become a far less common occurrence.⁵⁹⁰

It is now extremely doubtful that 'aiding or assisting' by third party state officials would demonstrate sufficient control to satisfy the stringent leadership requirement for individual criminal responsibility for aggression, even though, paradoxically, a third party country could incur state responsibility.⁵⁹¹ This is because, under Resolution 3314, the Security Council is entitled to consider other circumstances,⁵⁹² and the list of aggressive acts is not exhaustive.⁵⁹³ This situation sits uncomfortably with the effort throughout the

⁵⁸⁷ See Kahn, *Nuclear Weapons and the Rule of Law*, 31 NYU J Int'l L & Pol (1999) 349, at 402, cited in Heller (2007) (n. 244), 493 at footnote 103. See also the discussion above on pp. 92-3 and fn. 376.

⁵⁸⁸ See for instance China's return of 50,000 Korean soldiers to North Korea just prior to the Korean War, which has been cited as 'aiding and assisting' another State's commission of aggression, see C. Jian, *China's Road to the Korean War* (Columbia University Press 1994), 110-111. See also William Schabas, 'The Unfinished Work of Defining Aggression: How Many Times Must the Cannonballs Fly, Before They Are Forever Banned', in D. McGoldrick et al. (eds), *The Permanent International Criminal Court: Legal and Policy Issues* (2004), at 137.

⁵⁸⁹ See the discussion in Chapter 3.

⁵⁹⁰ See, however, the situation in the Yemen, and Reeve's account (n. 376).

⁵⁹¹ Under Resolution 3314 (n. 14) providing a wider definition than the crime (see n. 347).

⁵⁹² *Ibid*, Article 2, which has been excluded from Article 8 *bis* Rome Statute.

⁵⁹³ *Ibid*, Article 4, also excluded from Article 8 *bis*.

20th century to eradicate the crime of aggression and develop international criminal law to hold individuals liable. In the words of the IMT at Nuremberg '[C]rimes against international law are committed by men, not by abstract entities, and only by punishing individuals who commit such crimes can the provisions of international law be enforced.'594 The lack of a possibility to prosecute third party officials is as unfortunate as the fact that private economic actors, who are essential to an aggressive state act, will escape criminal liability under the current provisions because of the stringent leadership requirement. Whilst the application of accessory liability to the crime of aggression was pursued at Nuremberg and was considered as a jurisdictional path for the modern crime, this route may now be barred because of the incorporation of Article 25 (3) *bis* and objections on the ground of legality concerns. It will be interesting to look to the Court's own jurisprudence as deciding the question whether accessory liability has been definitively and finally ruled out, or whether it may survive as a mode of participation for the crime of aggression.

⁵⁹⁴ See Nuremberg Judgment (n. 3).

CHAPTER 3: THE CRIME'S STATE-CENTRICITY AND THE LIMITED LIST OF AGGRESSIVE ACTS

1. Introduction

The previous chapter considered the elevated gravity threshold of a 'manifest violation', and the stringent leadership requirement in Article 8 *bis* (1), which together set a very high bar for the crime and consequently limit jurisdiction of the Court extensively. This chapter considers the issues arising, in particular, from the second paragraph of the definition of the crime. The first of these examines the characteristic state centricity of the crime, which was also apparent in the first paragraph of Article 8 *bis*. This stipulates that individual criminal liability can only arise for a person in a position 'effectively to exercise control over or to direct the political or military action of *a State*', ⁵⁹⁵ and this is greatly reinforced in Article 8 *bis* (2). Thus, the crime of aggression can only be committed by one State, through its military and political leaders, against another State. The first sentence of Article 8 *bis* (2) provides:

For the purpose of paragraph 1, 'act of aggression' means the use of armed force by a State against the sovereignty, territorial integrity or political independence

⁵⁹⁵ Article 8 *bis* (1) Rome Statute, see Chapter 2 (p. 85).

of *another State*, or in any other manner inconsistent with the Charter of the United Nations.⁵⁹⁶

This state centricity is a fundamental feature of the crime of aggression, since any individual criminal liability is based on the act of the State. Consequently, the definition in Article 8 *bis* concerns itself solely with state-on-state aggression, raising the question whether this, in reality, provides a sufficiently modern interpretation of aggression scenarios, which are increasingly committed by non-state groups. Statehood itself has traditionally been defined by a set number of criteria, ⁵⁹⁷ however it will be considered whether the term could be stretched in order to apply to state-like entities and other aggressive groups in order to potentially avoid over-narrowness and extend jurisdiction.

The second issue arising from Article 8 *bis* (2) is the limited list of specific state acts which are capable of amounting to an act of aggression. This is derived directly from the 1974 General Assembly Resolution 3314 and consequently reflects the type of aggressive warfare prevalent at that time.⁵⁹⁸ Thus Article 8 *bis* (2) continues in its second sentence:

Any of the following acts, regardless of a declaration of war, shall, in accordance with United Nations General Assembly resolution 3314 (XXIX) of 14 December 1974, qualify as an act of aggression:

(a) The invasion or attack by the armed forces of a State of the territory of another State, or any military occupation, however temporary, resulting

⁵⁹⁶ Article 8 bis (2), emphasis added.

⁵⁹⁷ For the Montevideo criteria of statehood, see n. 608 below.

⁵⁹⁸ See Article 3 of Resolution 3314 (n. 14).

- from such invasion or attack, or any annexation by the use of force of the territory of another State or part thereof;
- (b) Bombardment by the armed forces of a State against the territory of another State or the use of any weapons by a State against the territory of another State;
- (c) The blockade of the ports or coasts of a State by the armed forces of another State;
- (d) An attack by the armed forces of a State on the land, sea or air forces, or marine and air fleets of another State;
- (e) The use of armed forces of one State which are within the territory of another State with the agreement of the receiving State, in contravention of the conditions provided for in the agreement or any extension of their presence in such territory beyond the termination of the agreement;
- (f) The action of a State in allowing its territory, which it has placed at the disposal of another State, to be used by that other State for perpetrating an act of aggression against a third State;
- (g) The sending by or on behalf of a State of armed bands, groups, irregulars or mercenaries, which carry out acts of armed force against another State of such gravity as to amount to the acts listed above, or its substantial involvement therein.⁵⁹⁹

Only Articles 1 and 3 of Resolution 3314 have been incorporated into Article 8 *bis* of the Rome Statute, which is therefore silent on other evidence and circumstances, ⁶⁰⁰ the non-

⁵⁹⁹ Article 8 *bis* (2) Rome Statute.

⁶⁰⁰ Resolution 3314 (n. 14), Article 2.

exhaustiveness of the list of acts of aggression, ⁶⁰¹ and the non-availability of political, economic or military justifications for an act of aggression. ⁶⁰² Consequently, the use of force other than armed seems to be precluded, and the list of acts appears to be closed, as the amendments to the Rome Statute currently do not allow for an extension to other acts of aggression. ⁶⁰³ Here the question arises yet again whether this list of acts is adequate and appropriate in creating efficient jurisdiction over modern types of conflict, and whether any wider interpretation or extension might be possible. Before contemplating this question, let us consider the limitation of culpable acts to state-on-state aggression, which pervades the definition of the crime in Article 8 *bis*.

2. State versus State aggression

Article 8 *bis* (1) and the first sentence of Article 8 *bis* (2) insist on the application of the definition of aggression to state-on-state aggression only, at a time when most modern threats originate from non-state actors.⁶⁰⁴ Thus, Article 8 *bis* (1) describes the perpetrators of aggression as controlling or directing the political or military action *of a State*. Article 8 *bis* (2) adds to the emphasis on the State, by describing aggression as the use of a State's armed force against the sovereignty, territorial integrity or political independence *of*

٠,

⁶⁰¹ Ibid, Article 4.

⁶⁰² *Ibid*, Article 5.

⁶⁰³ See the discussion below.

⁶⁰⁴ See for instance John Robb, *Brave New War: The Next State of Terrorism and the End of Globalisation* (Wiley 2007) 7.

another State. 605 All of the examples in paragraph 2 (a)-(g) reiterate the requirement for state-on-state acts of aggression, 606 as do the elements of the crime. 607

The traditional criteria of statehood were established in Montevideo in 1933 and have been widely accepted as 1) a permanent population, 2) a defined territory, 3) the existence of some form of government, and 4) the capacity to enter into relations with other States. 608 In addition, it may now be possible that the recognition by a significant number of other States may serve as an additional criterion. 609 The emphasis on statehood as a qualifier for the Crime of Aggression is derived from the general prohibition of the use of force by States in Article 2 (4) of the UN Charter⁶¹⁰ and the definition of the state act in Resolution 3314.⁶¹¹ The statehood qualifier now appears in Article 8 *bis* as a double prerequisite, applying to both the perpetrator and the victim.

Nevertheless, the insistence on state-on-state aggression was not entirely inevitable. A comparison with the other three crimes under the Rome Statute reveals that they do not require such a state act as a prerequisite to individual criminal liability. 612 Furthermore, the International Law Commission had, as early as 1991 considered individual criminal responsibility for acts of aggression committed by non-state actors, in the context of mercenaries. 613 The need for a shift of focus towards the acts of individuals rather than a

⁶⁰⁵ Article 8 *bis* (2), emphasis added.

⁶⁰⁶ This is unsurprising, as they are taken directly from Resolution 3314 (n. 14), which defines the state act of aggression.

⁶⁰⁷ See 'Elements of the crime', Annex I to Resolution 6 (n. 2).

⁶⁰⁸ Montevideo criteria of statehood, as established in 1933, see Hersch Lauterpacht, Recognition in *International Law* (1947), 52-8).

⁶⁰⁹ See Rosalyn Higgins, *Problems & Process – International Law and How We Use It*, (OUP 1994), 42.

⁶¹⁰ Article 2 (4) UN Charter (n. 195).

⁶¹¹ Resolution 3314 (n. 14).

⁶¹² Genocide, Article 6, Crimes against humanity, Article 7 and War Crimes Article 8 of the Rome Statute

⁶¹³ In Chapter III A 3 e of the 1991 Draft Code (n. 298).

State has also gained growing support in the academic world.⁶¹⁴ Thus, Weisbord, for instance, criticised the state element in the 2009 Draft definition, as 'riddled with anachronistic concepts that undermine its relevance today',⁶¹⁵ due to its focus on only conventional state-on-state war and state-sponsored insurgency.

Despite the developing recognition of the increased involvement of non-state actors in situations of aggression, it is a surprising fact that the state element was neither discussed in detail nor disputed during the negotiations of the definition of the crime of aggression. This anomaly leads to the conclusion that the majority of States in the lead-up to Kampala felt that the inclusion of non-state actors would have complicated agreement on a definition even more, and endangered consensus. Notwithstanding the apparent certainty of the SWGCA in applying the crime of aggression to States only rather than including non-state actors, the question arises whether this narrow perception of state aggressor and victim state is still valid in modern times. Proxy wars between two States may engage non-state groups only, on the territory of a third State, making attribution of aggressive acts to a specific State very difficult, if not increasingly impossible. Alternatively, a proxy war could take place under the guise of a civil war between an externally supported non-state group and the State government, which is supported by another State. Aggressive acts of non-state groups under control of a State could also be used directly against another State, muddying clear attribution.

⁶¹⁴ See for instance Antonio Cassese, 'On Some Problematical Aspects of the Crime of Aggression' (2007), 20 Leiden J. Int'l L., 841, 847. See also Claus Kress, 'The State Conduct Element' (n. 460) in Kress and Barriga (eds) (2017) (n. 246).

⁶¹⁵ Weisbord (2009) (n. 495), 22.

⁶¹⁶ See Astrid Reisinger-Coracini's and Pal Wrange, 'The Specificity of the Crime of Aggression', in Kress and Barriga (2017) (n. 246) 313-4.

⁶¹⁷ See for instance the war in Yemen discussed in Chapter 2 (n. 389).

⁶¹⁸ Mark S. Stein, 'The Security Council, the International Criminal Court, and the Crime of Aggression: How Exclusive is the Security Council's Power to Determine Aggression?' (2005), 16 Indiana Int'l. & Comp. L. Rev., 4.

situation may arise where non-state actors act independently of a State, but employ traditional methods of aggressive warfare reminiscent of those utilised by States. In any of the above situations, it may be extremely difficult to apply the current aggression provisions to either the non-state group or the State which may instruct or sponsor the non-state group, unless it could be shown that the State exercised a sufficient level of control over the actions of the non-state group in order to attribute them to the State in question itself.

By comparison, customary international law appears to be wider, although in order to establish effective control, attribution to a State is still essential. Furthermore, the State behind the act of aggression would have to provide more than merely logistical support to the non-state group for their actions to be attributable to the State. Thus both *Nicaragua* and the *Armed Activities* case confirmed that the actions of non-state groups could be attributable to a State, where they were employed by that State and it exercised sufficient control over their actions. ⁶¹⁹ The *Armed Activities* case specifically established an explicit link between the activities of a non-state armed group and an instructing State, based on Article 8 of the ILC's Draft Articles on State Responsibility. ⁶²⁰ This article provides:

Article 8

The conduct of a person or group of persons shall be considered an act of a **State** under **international** law if the person or group of persons is in fact acting on the

⁶¹⁹ For *Nicaragua* (1986), see n. 37, paras 75-125. For *Armed Activities* (2005), see n. 37, Judgment of 19 December 2005, para 345.

Article 8 of the ILC's *Draft Articles on Responsibility of States for Internationally Wrongful Acts* (November 2001) Supplement No. 10 (A/56/10), Chp. IV.E.1.

instructions of, or under the direction or control of that **State** in carrying out the conduct.⁶²¹

Following these precedents, there should be no reason why responsibility should not be attributed to another State for the actions of a non-state group, at least at customary law, if these essential links and the required level of control could be proved, and their exclusion in the 2010 Kampala definition remains striking.

Moreover, the possibility of applying the aggression provisions to non-state actors would be warranted simply in order to maintain validity of the new criminal aggression provisions in the face of contemporary phenomena. Thus, military experts join academics in pointing to the emerging irrelevance of traditional state-on-state combat and the rise of terrorist and guerrilla warfare. The US counterterrorism expert John Robb, for instance, advises that '[w]ars between States are now, for all intents and purposes, obsolete.' He continues:

The real threat, as seen in the rapid rise in global terrorism over the past five years, is that this threat isn't another State but rather the super-empowered group $[\ldots]$ – with the ability of one man to declare war on the world and win. 623

Similarly, T.X. Hammes points to a trend '... downward from nation-states using huge, uniformed armies to small groups of like-minded people with no formal organization who simply choose to fight.' Philip Bobbitt adds an interesting twist to this discussion by

.

⁶²¹ *Ibid*.

⁶²² Robb (n. 604), 7.

⁶²³ Robb, *ibid*, at 7-8.

⁶²⁴ T.X. Hammes, 'Fourth Generation Warfare Evolves, Fifth Emerges' (2007) Mil. Rev., 14, 20.

contemplating the possibility of reinterpretation of 'State' to include non-state groups. Whilst conceding that 'asymmetric warfare will become the norm when great powers are confronted', he also points out that '[t]he state has undergone many transformations in the constitutional order - the basis for the state's legitimacy...Could Al Quaeda be an example of this new form?' Consequently, traditional notions of statehood may require reconceptualisation in order to include powerful non-state groups who are the main actors in modern warfare. Denying the existence of this development, and excluding non-state actors from the crime of aggression, defies logic and reason.

The above statements by military experts also demonstrate that decentralisation of war is now commonplace, and States no longer have a monopoly on maintaining armed forces, and effective control over a territory and its people can unquestionably be held by organisations other than a State. At a time when independent non-state groups have been 'identified as the most important emerging threat to global security', ⁶²⁶ a strict and narrow interpretation of the definition of the crime of aggression failing to capture 'weapons other than the state (as currently conceived) and aggressors besides the statesman..., ⁶²⁷ is as unhelpful as it is inappropriate.

A timely and innovative reinterpretation of the definition of the crime to include 'groups' or 'state-like entities, on the other hand, might be able to capture 'political-military organizations', 629 which attack States but are not strictly classed as States themselves.

⁶²⁵ Philip Bobbit, *Terror and Consent: The Battle for the Twenty-First Century* (Knopf 2008), 126. More recently, the non-state group ISIS (Islamic State of Iraq and Syria) have been responsible for many acts of aggression and atrocities. The name is misleading, since application of the Montevideo criteria (p. 148) would preclude the group from being a State.

⁶²⁶ Weisbord (2009) (n. 495), 27.

⁶²⁷ *Ibid*, 24.

⁶²⁸ This view is also shared by May (2008) (n. 438), 319-41.

⁶²⁹ Weisbord (2009) (n. 495), 30.

This could potentially widen the Court's jurisdiction by extending liability through a link to aggressive organisations other than the traditional State. Determination of such state-like entities would have to be left to the judges of the ICC, which admittedly could lead to accusations of judicial activism. Nevertheless, this thesis argues that judicial determination would be preferable to the effect of a lacuna in the legislation, leading to irrelevance of the definition and ineffectiveness of jurisdiction.

Moreover, organisations and groups already make an appearance in the Rome Statute in the provisions on Crimes against Humanity⁶³⁰, and War Crimes.⁶³¹ Extending this understanding to the Crime of Aggression should not be too outlandish an idea. However, the SWGCA's documents throughout the discussions on the Crime of Aggression demonstrate negotiating parties' tenacious insistence on a strictly limited definition, as part of an ongoing struggle to achieve any consensus at all. Any radical suggestion of change of wording may have overthrown all efforts and any hope of ratification. Nevertheless, without such a wider interpretation of 'State' as a 'group' or 'state-like entity', and a leader who is 'in a position effectively to exercise control over or to direct the political or military action of a State [or group],' the definition is at great risk of being irrelevant from the outset.

⁶³⁰ Article 7 (2) (a), Rome Statute (n. 1), referring to an '[a]ttack directed against any civilian population, pursuant to or in furtherance of a State *or organizational policy* to commit such attack'.

⁶³¹ Article 8 (2) (f), Rome Statute, describing armed conflicts not of an international character, and referring to 'armed conflict between governmental authorities and organized armed groups or between such groups'.

3. The List of acts deemed to be aggressive

The final major issue raised by the definition of the crime of aggression is whether the limited list of acts in Article 8 *bis* (2), describing outmoded forms of state aggression as perceived in 1974,⁶³² is suitable for the prosecution of modern day acts of aggression. The list of aggressive acts in the Kampala Definition includes invasion, bombardment, blockade, an attack by armed forces or use of those armed forces within another State's territory contrary to agreed conditions, a State's permission to allow its territory to be used for an act of aggression against a third State, and the sending or directing of proxy forces into another State.⁶³³ The list of acts is taken directly from Article 3 of Resolution 3314,⁶³⁴ but without the reference to possible determination of other acts, which was also contained in Article 4 of that resolution.⁶³⁵ Consequently, the actual list of acts in Article 8 *bis* reflects only modes of armed conflict experienced up to the time of the drafting of the 1974 Resolution. Additionally, the definition of aggressive acts focuses on bureaucratic organizational forces with permanent infrastructures and institutions, i.e. standing armies, and mainly considers traditional state-on-state combat. To a lesser degree state-sponsored insurgencies are also recognised.⁶³⁶

The resort to the limited list of traditional acts of aggression in the final version of Article 8 *bis*, to the exclusion of other provisions, was a perplexing choice by the SWGCA, particularly in light of the prominent emergence of new forms of warfare during the

⁶³² Resolution 3314 (n. 14).

⁶³³ Article 8 bis (2) (a)-(g), see pp. 145-6.

⁶³⁴ Resolution 3314 (n. 14), Article 3 (a)-(g).

⁶³⁵ *Ibid*. Article 4.

⁶³⁶ Article 8 *bis* (2) (g), see p. 146.

prolonged negotiating phase.⁶³⁷ This feature will be explored in greater detail below, however the salient fact worth mentioning at this point is that a great number of new forms of aggressive threats have, over the last two decades, been recognised.⁶³⁸ Many of these are capable of great devastation of a foreign State's systems and processes without any of the physical territorial transgression insisted on in Article 8 *bis* (2).

An examination of the negotiations shows that the failure of Article 8 *bis* to reflect modern types of warfare was neither uncontroversial nor entirely predictable. There were, in fact, a number of proposals ranging from even greater restriction to much wider jurisdiction. At one end of the spectrum of proposals considered by the SWGCA, lay suggestions of a result-based definition instead of a limited list, suggesting that, in order to achieve 'legal precision, clarity and certainty, the criminal standard adopted should be higher than for the political compromise achieved in Resolution 3314.⁶³⁹ This view would have insisted on the armed attack resulting in effective occupation or annexation in order to qualify as aggression. In evaluation, a definition along those lines would be even more restrictive than the outmoded list in Article 8 *bis* (2). Thus, even the traditional acts of aggression which have been included do not, necessarily, require physical occupation or annexation of territory and aggression may take many other forms than traditional usurping and occupation of physical territory. The inclusion of the narrow prerequisite of occupation or annexation would have resulted in even greater limitation.

The second option was for the SWGCA to adopt a limited list along the lines of Resolution 3314. This choice was based on the different, but equally restrictive view that

⁶³⁷ See section 3.1 below.

⁶³⁸ *Ibid*

^{639 &#}x27;1997 German Proposal' (n. 395).

the greater the independent determination powers to be granted to the new Court, the greater the need for such a specific list in order to 'achieve full respect for the principle of legality'. This particular argument, however, also did not take into account the danger to a country's territorial integrity, sovereignty and political independence if subjected to an attack not covered by that restrictive list. Consequently, a third option was introduced, which suggested replacing the list with a general, generic definition, 42 or, if a list was unavoidable, using this as mere examples of aggressive acts, non-exclusive to other perceivable forms of aggression.

In 2006, the SWGCA considered a compromise solution to the impasse between the 'specific list' and generic approaches.⁶⁴⁴ This compromise suggested combining the generic and specific options by expressly providing for a *non*-exhaustive list which was to be illustrative, as well as a general 'chapeau' similar to the Crimes against Humanity provisions in Article 7.⁶⁴⁵ Legality could thus be safeguarded, because the determination of other acts was subject to the restrictive chapeau, but equally some flexibility was allowed for, because the list was only illustrative. The SWGCA also recommended that a reference should be included to *all* the relevant documents, including the Nuremberg Charter, Resolution 3314 and the 1996 Draft Code.⁶⁴⁶

⁶⁴⁰ See for instance 'Suggestion made orally by Italy on 13 March 2000 with regard to a structure for discussion on the crime of aggression', PCNICC/2000/WGCA/DP.3

⁶⁴¹ As pointed out by the Colombian delegation. 'Comments by Colombia on the Italian proposal made orally in the Working Group on 13 March 2000', PCNICC/2000/WGCA/DP.2. ⁶⁴² *Ibid*.

⁶⁴³ See the 2000 'Proposal submitted by Greece and Portugal', 28 November 2000, PCNICC/2000/WGCA/DP.5.

^{644 2006} Princeton Report (n. 552).

⁶⁴⁵ See n. 612.

⁶⁴⁶ 2006 *Princeton Report* (n. 552), para 13. This approach gained support, particularly, by Ferencz, see Benjamin Ferencz, 'Enabling the International Criminal Court to Punish Aggression' (2007) 6 Wash. U. Global Stud. L. Rev. 551, pt. I B.

Despite the solid grounds expressed by the SWGCA in 2006 for the compromise of a non-exhaustive list reined in by the chapeau, which could have satisfied both camps, States' disagreements on the subject continued. Thus, the 2007 Princeton Report reflects the ongoing discussion and ambiguity, by reporting on the unwavering view expressed by some States that the list should be open and non-exhaustive,⁶⁴⁷ as well as the uncompromising argument by others for a closed list in order to satisfy the legality principle. Yet other States, without specifying which ones, could not decide on either stance, viewing the list as somewhere in between, i.e. 'semi-closed', or 'semi-open.⁶⁴⁸ However, consensus had now been reached that, whatever the nature of the list, an act of aggression would have to fulfil the requirements of the chapeau.⁶⁴⁹

The text of the final definition in Article 8 *bis* (2) now contains this chapeau, and then moves on to list the acts amounting to aggression, thus far reflecting the SWGCA's 2006 recommendations. However, the discussion has revealed that Article 8 *bis* does not provide for the determination of acts other than the restrictive list, nor does it refer to other relevant documents, for instance the Nuremberg Charter or the 1996 ILC Draft Code, or the missing articles from Resolution 3314. It also does not state that the list of acts is merely illustrative, which would have expressly allowed for flexibility. Therefore, it falls far short off the 2006 compromise solution.

⁶⁴⁷ 2007 *Princeton Report* (n. 416), paras. 46-53. See also the December 2007 *SWGCA Report* (n. 422), para. 21. The reports are not specific as to which States favoured which stance, reflecting opinions generically.

⁶⁴⁸ 2007 *Princeton Report* (n. 416), para 46. See also the discussion in Stefan Barriga, 'Negotiating the Amendments on the crime of Aggression', in Barriga and Kress (eds) (2012), *Travaux Preparatoires*, (n. 317), 28.

^{649 2007} Princeton Report (n. 416), para. 48.

3.1 The relevance of the list of aggressive acts in a modern context

As a result of the narrow list of acts, the description of forms of war in Article 8 *bis* (2) appears to be greatly outdated, and does not reflect advances in modern warfare, which is constantly developing. The above discussion has already considered this phenomenon with regards to the fact that the concept of 'State' no longer accurately reflects the modern aggressor (or victim).⁶⁵⁰ The same problem arises with regards to the limited list of acts. Thus, over the last 20 years, in particular, modern war-making organizations have been re-structuring into 'hundreds of mercurial cells that make up organizations like Alquaeda',⁶⁵¹ and, more recently, Boko Haram, and ISIS.⁶⁵² Non-state sponsored forms of aggression like terrorism and guerrilla warfare are increasingly common, employing new methods. Unfortunately, none of these considerations are currently reflected in the acts listed in Article 8 *bis* (2).

In addition, a new threat of systems disruption has been emerging, which involves no use of arms whatsoever. For example, a State's vital systems are nowadays capable of being disrupted remotely. Such a thoroughly modern threat is currently not included in the list of aggressive acts, and yet it may result in the annihilation of a country's defence mechanism, or paralyse transport, communications or emergency response systems, despite a lack of physical territorial transgression. Examples are the sabotage of critical systems and cyber-warfare, which are capable of causing a vast amount of damage using widely affordable, untraditional means of warfare, and are able to hit the modern world's

⁶⁵⁰ See the discussion in section 2, particularly at 150-2.

⁶⁵¹ Weisbord (2009) (n. 495), 16.

⁶⁵² See the discussion at pp. 150-3.

⁶⁵³ This phenomenon was also briefly considered in section 3 of Chapter 2.

Achilles heel of increasing reliance on interdependent networks. This type of sabotage could, for example, severely impede the supply of electricity, telecoms, gas and water, transport, banking, national health services, government operations, as well as election processes.

This emerging threat of sabotaging another country's systems and processes has been widely recognised, for instance in the National Security Strategy and Strategic Defence and Security Review 2015, setting out the UK's Strategy until 2020, and the first annual report in 2016.⁶⁵⁴ It was stated:

The range of cyber threats and cyber actors threatening the UK has grown significantly – both from state and non-state actors. The UK increasingly relies on networked technology in all areas of society, business and government. This means that we could be vulnerable to attacks on parts of networks that are essential for the day-to-day running of the country and the economy. 655

A similarly prominent recognition of the problem featured in a report by the United States Senate Select Committee on Intelligence.⁶⁵⁶ This report acknowledged developing cyber threats to US security, potentially affecting 'critical infrastructure, public health and safety, economic prosperity, and stability'.⁶⁵⁷ Additionally, the deployment of anti-

National Security Strategy and Strategic Defence and Security Review 2015 (23 November 2015), setting out the UK's Strategy until 2020, and the first annual report in 2016 (7 December 2016), http://www.gov.uk/government/publications/national-security-strategy-and-strategic-defence-and-security-review-2015-annual-report-2016, accessed 30th October 2020. The latest review is now expected early in 2021.

⁶⁵⁵ *Ibid*, para 1.8.

⁶⁵⁶ National Security Strategy of the US in 2019,

http://www.dni.gov/files/ODN/documents/National_Intelligence_Strategy_2019.pdf, accessed 30 October 2020

⁶⁵⁷ *Ibid*. 4-5.

satellite weapons in order to 'reduce US military effectiveness and overall security' were also recognised as an emerging threat.

Furthermore, the General Assembly has recently mandated two contemporaneous projects. The purpose of the first of these, undertaken by an Open-Ended Working Group (OEWG), is to determine the application of international law to cyber warfare. The purpose of the second, a Group of Governmental Experts (GGE), is to assess how cyber threats impact on international peace and security generally. Additionally, the Permanent Mission of Liechtenstein, with other partners, has created a Council of Advisers on the Application of the Rome Statute to Cyberwarfare. This project, aims to develop an understanding of and identify 21st Century challenges and their alignment with existing international criminal law. Its report is still being awaited, however it is expected to propose sensible modification that will update the law for the coming age in order to ensure its relevance in the cyber domain. These ongoing projects clearly demonstrate a recognition within the international community as a whole, and by experts in the field, that international law and practices ought to apply to modern threats to peace and security, but may need to be modified.

⁶⁵⁸ The Open-Ended Working Group on Developments in the Field of Information and Telecommunications (OEWG) in the Context of International Security (ICTs) was established in 2018 by UNGA A/RES/73/27 (5 December 2018). This group is currently (20th January 2021), preparing its Zero Draft and will meet for a final session in March 2021.

⁶⁵⁹ Group of Governmental Experts on Advancing Responsible State Behaviour in Cyber Space in the Context of International Security (GGE), established by UNGA A/RES/73/266 (22 December 2018). Their final session should take place in May this year and their report will be submitted to the 2021 session of the General Assembly.

⁶⁶⁰ See 'The Council of Advisers on the Application of the Rome Statute to Cyberwarfare – Project Background and Goals', 29 October 2019, The Global Campaign for Ratification and Implementation of the Kampala Amendments on the Crime of Aggression, available at http://crimeofaggression.info/the-campaign/the-council-of-advisers-on-the-application-of-the-rome-statute-to-cyberwarfare/, accessed 30 October 2020.

⁶⁶¹ *Ibid*.

The effect of systems disruptions, cyber-attacks or even mere misinformation may be devastating and provoke military responses, for instance in anticipation of an imminent attack. At other times, the consequences may not be immediately observable or directly result in the physical disabling of a State's systems. Thus, information can be manipulated through the spreading of 'fake news', or democratic elections can be deliberately interfered with. These acts are clearly capable of constituting a threat to the territorial or political sovereignty of a State, ⁶⁶² but are currently not included in the crime of aggression provisions. One such example is the alleged Russian Interference in the 2016 elections in the United States, as a result of which 12 Russian military officers were indicted for their role in the act of interfering with the democratic process. ⁶⁶³ Similarly, in November 2017, the UK's then Prime Minister Theresa May accused Russia of 'meddling in elections and planting "fake news" in an attempt to "weaponise information" and sow discord in the West,' by hacking the Danish Ministry of Defence and the German Bundestag. 664 Concerns were raised about Chinese interference prior to, and continue even after, the recent Taiwanese presidential elections⁶⁶⁵ and China's cyber-attack on Australia's parliament and political parties. 666

Despite the fact that these emerging threats first began to come to prominence during the negotiations for a definition of the crime, no provision for their consideration exists.

Germanic See for instance Disinformation and 'fake news': Interim Report: Government Response to the Committee's Fifth Report of Session 2017-19, (HC 363), published 23 October 2018, available at www.publications.parliament.uk/pa/cm201719/cmselect/cmcumeds/1630/163002.htm, accessed 30 October 2020.

⁶⁶³ For details of this indictment, see the website of the FBI, http://www.fbi.gov/wanted/cyber/russian-interference-in-2016-u-s-elections, accessed 30 October 2020.

⁶⁶⁴ See Recommendation 41 of the House of Commons *Interim Report* (n. 662).

⁶⁶⁵ Washington Post, 13 January, 2020, 'Opinion: Despite election result, China won't give up on Taiwan', by John Pomfret, https://www.washingtonpost.com/opinions/2020/01/13/despite-election-result-china-wont-give-up-taiwan/.

⁶⁶⁶ According to the Australian Signals Directorate. See 'Exclusive: Australia concluded China was behind hack on parliament', September 16, 2019, available at http://www.reuters.com/article/us-australia-china-cyber-exclusive-idUSKBN1W00VF, accessed 30 October 2020.

Consequently, the failure to provide an expressly illustrative list in line with the 2006 compromise suggestion by the SWGCA, renders the definition of aggression subject to the danger of being mostly irrelevant in a modern context, unless the Court adopts wider interpretation powers. Even an addendum to the effect that it may be possible to determine other acts of aggression, similar to Article 4 of Resolution 3314, would have improved this situation. In the meantime, the academic debate on the (non-)exhaustiveness of the list of acts in Article 8 *bis* (2) continues.

3.2 The (non-)exhaustiveness of the list - a continuing academic debate

Notwithstanding the decision made at Kampala to include only certain acts of aggression expressly, academic opinion is still greatly divided as to whether this list is now firmly closed or, in fact, non-exhaustive and whether the ICC judiciary are able to determine other acts of aggression. Thus, one view argues that the list of acts is definitively closed. As Sergey Sayapin states: 'As a conventional provision of international criminal law, Article 8 *bis* (2) of the Rome Statute is exhaustive and must be interpreted restrictively.'667 This assessment is based on the view that a non-exhaustive list would, by its nature, offend the principles of certainty at law and retrospectivity.⁶⁶⁸

Other scholars critique the existing definition of the crime of aggression, including even the limited list of acts in Article 8 *bis* (2), as overly broad and vague on account of its generally uncertain language taken from Resolution 3314. Thus Michael Glennon, for

162

 $^{^{667}}$ S. Sayapin, The Crime of Aggression in International Criminal Law, (Springer 2014), 272. 668 n. 227.

example, regards *any* consideration of Resolution 3314 as entirely inappropriate for application in criminal prosecutions, as it is insufficiently specific by allowing for the determination of future acts of aggression in the original version of 1974. ⁶⁶⁹ Following this view, judicial determination of other acts would be at least as, if not more, precarious than determinations of other acts by the Security Council provided for in Resolution 3314, ⁶⁷⁰ by offending the principle of *nullum crimen sine lege*, the principle in Article 31 (1) VCLT, ⁶⁷¹ as well as Article 22 (2) of the Rome Statute. ⁶⁷² Glennon's blanket dismissal of Resolution 3314 as an inappropriate basis for the crime of aggression definition, has, however, been persuasively countered, ⁶⁷³ not least by the SWGCA's own declaration that they considered Resolution 3314 to codify existing customary international law. ⁶⁷⁴

Notwithstanding the SWGCA's defence of their reliance on the list derived from Resolution 3314, Glennon's criticism is shared by other academics. Thus, Harold Hongju Koh's statement at the Kampala conference also demonstrates the view that the definition in Article 8 *bis* is too uncertain to serve as a justiciable provision of customary international law, despite approving of the list's limitations.⁶⁷⁵ However, in a later statement, Koh and his co-writer Buchwald admit that the list is not only 'over inclusive', but also 'under inclusive'.⁶⁷⁶ Thus Koh and Buchwald regard the list as, on the one hand, too vague and too concerned with traditional forms of warfare, however, on the other

⁶⁶⁹ Glennon (2010) (n. 8) 71.

⁶⁷⁰ Article 4, Resolution 3314 (n. 14).

⁶⁷¹ n. 458.

⁶⁷² Article 22 (2) (n. 477). See also the academic debate on legality issues (n. 459 and related text).

⁶⁷³ See Dinstein's statement that the inclusion of Article 3 (a)-(g) of Resolution 3314 'implies that each of the seven clauses has been deemed by the Kampala Conference to be declaratory of customary international law.' Dinstein (2017) (n. 154), para 395.

⁶⁷⁴ See the 2007 *Princeton* Report (n. 416), para 46.

⁶⁷⁵ Statement of Harold H. Koh, Legal Adviser for the US Department of State at Kampala, June 4, 2010, available at https://2009-2017.state.gov/s/l/releases/remarks/142665.htm.

⁶⁷⁶ Harold H. Koh and Todd F. Buchwald, 'The Crime of Aggression: The United States Perspective' (2015) 109 AJIL. 267.

hand, they also acknowledge that it is remarkably outdated with respect to the modern threat posed by international terrorism, non-state actors, and cyber-warfare. This latter critique, therefore, represents at least a partial change of stance, acknowledging that the list, as it stands, necessitates expansion in order to apply to modern forms of warfare and achieve the purpose of outlawing contemporary acts of aggression.

Similar to Koh's and Buchwald's arguments, Robert Heinsch submits arguments for both a limited and a non-exhaustive list. Thus, he admits that the determination of other acts of aggression, from a criminal law perspective, constitutes a problem. He states that '[t]he principle of legality, at least in national law, would require that the list of punishable actions is clearly defined and not open to interpretation (or amendment) by the judges. Nevertheless, Heinsch also asserts that the legality principle is not as fundamental in international law as it is in national law, citing as an example the possibility of judicial determination power of other acts in Article 7 (1) (k) of the Rome Statute on crimes against humanity which refers to '[o]ther inhumane acts of a similar character...'. Thus, Heinsch does not rule out the possibility of determination of other acts entirely, although he acknowledges the validity of the legality and specificity requirements.

Other academics defend the argument that the list is open and serves merely as an illustration, allowing for the determination of other acts. Thus, Claus Kress considers the list in Article 8 *bis* (2) to be non-exhaustive, although he admits that 'the nature of the list was the subject of controversy, [and] no consensus emerged on the exhaustiveness of

⁶⁷⁷ Heinsch, (2010) (n. 376), 724.

⁶⁷⁸ Article 7 Rome Statute (n. 612, 630).

⁶⁷⁹ Heinsch, (2010) (n. 376), 725.

⁶⁸⁰ Kress, in Kress and Barriga (eds) (2017) (n. 460), 435.

the list'.⁶⁸¹ Kress contends that the silence on articles other than 1 and 3 of Resolution 3314 merely restricts the determination of other acts and the consideration of other grounds by the Security Council.⁶⁸² The Court, Kress argues by comparison, is not restricted by the lack of an express reference to all the provisions in Resolution 3314, in particular Articles 2 and 4.⁶⁸³ Consequently, the judges of the ICC are entitled to draw on the entirety of the resolution, allowing for the determination of other acts of aggression, as long as there is no contradiction with the text and spirit of the Kampala Amendment.⁶⁸⁴

Following this argument, the list in Article 8 *bis* (2) could be indeed interpreted as non-exclusive, despite the appearance to the opposite.⁶⁸⁵ Moreover, Kress reinforces his argument with the assertion that even the requirement to abide by the legality principle need not be offended, stating:

In the absence of genuine ambiguity, the second sentence of article 22 (2) of the Rome Statute also does not require a reading of the list as exhaustive.⁶⁸⁶ The same is true for the principle of legality, as the first sentence of article 8 *bis* (2) of the Rome Statute ensures that the definition is not unduly vague.⁶⁸⁷

⁶⁸¹ *Ibid*, 436.

⁶⁸² Kress, *ibid*, citing Barriga (2012) in *Travaux Preparatoires* (n. 317), 26-7. See also C. McDougall, *The Crime of Aggression under the Rome Statute of the International Criminal Court*, (CUP 2013), 97-100.

⁶⁸³ See n. 600-1.

⁶⁸⁴ Kress (2017) (n. 680), 436. This view is also shared by McDougall (2013) (n. 682), 99.

⁶⁸⁵ Kress, *ibid*, 451.

⁶⁸⁶ For Article 22 (2) of the Rome Statute, see n. 477.

⁶⁸⁷ Kress (2017) (n. 680), 435-6. See also Roger. S. Clark, 'Amendments to the Rome Statute of the International Criminal Court' (2010), 2 Go. J. Int'l L., 696; and McDougall (2013) (n. 682), 103-5. For different views, see Kai Ambos, 'The Crime of Aggression after Kampala' (2010) 53 GYIL, 487; and Marina Mancini, 'A Brand New Definition for the Crime of Aggression: The Kampala Outcome' (2012) 81 NordJIL, 227, 235.

Support for the inclusion of other acts can also be derived from Keith Petty.⁶⁸⁸ Thus, he argues that the words in Article 8 *bis* (2) 'in any other manner inconsistent with the Charter', not only reinforce Article 2 (4) of the UN Charter, but also indirectly allow for acts other than the ones listed. Similarly, Jennifer Trahan would view the list as being 'semi-open' or 'semi-closed', using the words in the 2007 Princeton Report,⁶⁸⁹ as long as any new form of aggression fulfils the manifest qualifier.⁶⁹⁰

In evaluation, the latter arguments allowing for the determination of other acts would undoubtedly alleviate some of the difficulties of applying the definition to emerging threats and contemporary acts of aggression. If the list of aggressive acts could be expanded, whilst still observing the legality requirement, the Court's jurisdiction may be widened sufficiently to incorporate devastating new forms of aggression, which are currently not included, rather than focus solely on 1974-style combat situations.

An approach along these lines would also widen the Court's overall jurisdiction over the crime, aiding application to modern aggression scenarios and furthering deterrence. From this particular point of view, an interpretation of the list as non-exhaustive, allowing for the possibility of the judicial determination of other acts, would serve that purpose better than the current limited list. However, care must be taken not to offend the legality principle. Whilst acknowledging this problem, it is suggested that legality concerns could be addressed by reference to the chapeau contained in the first sentence of Article 8 *bis* (2), which ensures that the definition is not unduly vague.⁶⁹¹

⁶⁸⁸ Petty (2009) (n. 429).

⁶⁸⁹ n. 416.

⁶⁹⁰ Trahan (2011) (n. 5), 59-60.

⁶⁹¹ See Kress (2017), 435-6, (n. 687 above and related text). See also Weisbord (2009) (n. 495), arguing for a combination of a non-exhaustive list together with a chapeau, 22, 39.

Great concerns have been expressed with regards to wider determination powers by the ICC judiciary over the crime of aggression, based on the underlying fact that the determination of acts of aggression is essentially political.⁶⁹² In response to these views, it is contended that this perception is overly restrictive and the concerns are unwarranted. The Court, as an international criminal tribunal, is concerned with the determination of individual criminal liability rather than interference in relations between States per se. The determination of the state act of aggression is required solely for the establishment of that individual criminal liability, not for political condemnations of States. Other international courts and tribunals have, on occasion, had to resort to wider determinations, ⁶⁹³ and, in order to fulfil their duty, the judiciary of the ICC has to be entrusted with the power to apply any required analogy for the purpose of essential determinations. Such judicial determinations need not necessarily offend the principle of legality, as long as the analogy is close enough. Other, developing forms of aggression not envisaged at the time of the definition could and should, therefore, be included. A restrictive view of the list in Article 8 bis (2) as being exhaustive, without acceptance of other acts or reference to the entirety of the provisions in Resolution 3314, which would have allowed for other considerations, endangers the definition as being unnecessarily narrow and is likely to render it unsuitable to a modern context, hampering effective jurisdiction.⁶⁹⁴

⁶⁹² See, for instance, the views expressed post Kampala in the *Travaux Preparatoires* (n. 317), in the statements of position by a number of States, including the P5 (from p. 810).

⁶⁹³ See for instance, the *Nuremberg* precedent itself (n. 3), see also the ICJ's statement that the Court had 'never shied away from a case brought before it merely because it had political implications'. (*Nicaragua* case (1986) (n. 37), para 96, citing the *Corfu Channel* case (n. 369)). For a discussion, see for instance Christine Gray, 'The Use and Abuse of the International Court of Justice: Cases concerning the Use of Force after *Nicaragua*' (2003) 14 (5) EJIL, 867.

⁶⁹⁴ See the discussion above, and throughout section 3.2.

Nevertheless, the debate on the (non-)exclusivity of the list remains, at this time, still unresolved. It will be enlightening to see how the Court views its own determination powers in cases involving modern acts of aggression currently not listed. Future judicial declarations by the ICC on the determination of other acts will, ultimately, shed a light on the dilemma.

The discussion of the limitations of the crime to state-on-state aggression only, and inclusion of the highly restrictive list of acts concludes the consideration of the narrow definition of the crime of aggression, limiting the Court's jurisdiction indirectly but significantly through the definition itself. The next two chapters will consider the jurisdictional restrictions arising directly from the conditions placed on the Court's jurisdiction in Articles 15 *bis* and *ter*.

4. An overall evaluation of the definition of the crime of aggression

The definition of the crime of aggression, examined in this and the previous chapter, contains a high threshold requiring a manifest violation of the UN Charter, and additionally insists on political or military leadership and a requirement to control or direct the actions of a State. It also necessitates attribution of an aggressive act to a State against another State, and includes a list of acts which currently takes into account traditional forms of warfare only. The combination of these elements results in severe restriction of the jurisdiction of the Court. Whilst the SWGCA's documents demonstrate

that this effect was deliberately and intentionally sought by some States, ⁶⁹⁵ others would have preferred even greater restrictions. ⁶⁹⁶ An entirely different perspective was taken by a number of other States and academics seeking a wider, more flexible definition in order to allow the Court greater jurisdictional powers over modern aggressors.

In evaluation of the provisions contained in Article 8 *bis*, serious doubts can be voiced that the definition is sufficiently wide to capture 21st century acts of aggression and their perpetrators. The limitations of Article 8 *bis* represent a substantial retreat from the precedents set at Nuremberg and Tokyo.⁶⁹⁷ Furthermore, the crime of aggression is currently limited to the same situations of conflict contained within the 1974 Resolution 3314, narrowing the scope of the crime even more because of the exclusion of provisions for other interpretations and determinations.⁶⁹⁸

The 'supreme international crime' has now been equipped with a supreme threshold, ⁶⁹⁹ rendering effective prosecution for modern acts of aggression potentially impossible, unless construed widely. The intended effect of putting leaders of nation states on notice that their aggressive acts may have criminal consequences for them as individuals, has been weakened substantially. Any deterrence may now be ineffective, allowing States to continue criminally aggressive acts without fear of being called to account. The credibility of the Court itself may be called into question, since its own effectiveness will most certainly be assessed by the level of compliance and cooperation. Judicial extension by

⁶⁹⁵ See the 2006 and 2007 *Princeton Reports* (n. 552 and 416) and the December 2007 *SWGCA Report* (n. 422)

⁶⁹⁶ See for instance the German Proposal of 2000 (n. 399), adding the object or result of an occupation or annexation. See also the Italian stance (n. 640), and concerns expressed by some observer states, for instance the United States (n. 675).

⁶⁹⁷ See the discussion in Chapter 1 and 2.

⁶⁹⁸ See n. 600-2 and related discussion.

⁶⁹⁹ See Robert Jackson's opening statement (n. 135).

analogy is, if strictly interpreted, ruled out by Article 22 (2) of the Rome Statute, and Article 31 (1) of the VCLT,⁷⁰⁰ however without such extension, the criminalisation of aggression may be largely ineffective.

Nevertheless, judicial analogy has been commonplace in the history of international tribunals and courts. If the analogy is close enough, provisions against retrospectivity of criminal law need not be offended. The existing filters of Security Council involvement and Pre-Trial Division authorisation are more than capable of preventing what might be perceived as frivolously instigated or politically motivated actions brought before the Court. Alternatively, an argument can be presented for re-writing the definition entirely, allowing for other modern acts of aggression and aggressors of influence at a less controlling level, or simply including a provision for the judicial determination of other acts of aggression over other perpetrators of sufficient influence.

Naturally, these suggestions in particular are likely to incur criticism by sceptics, citing legality concerns. As summed up by Noah Weisbord:

'It is no small step to acknowledge that the ICC judges exercise discretion as they interpret the law. The danger is that a spell will be broken, and with it, the law's claim to authority. But the law's authority can be built – and squandered – in various ways. Perhaps the most treacherous is for judges to deliberately remain blind to context as they make decisions that undermine rather than advance the law's purpose.'702

⁷⁰⁰ Article 22 (2) of the Rome Statute (n. 477), and Article 31 (1) VCLT (n. 458).

⁷⁰¹ See the discussion in Chapter 4.

⁷⁰² Weisbord (2011) (n. 430), 168.

Without extending the definition and thereby jurisdiction of the Court, either incrementally through analogy or explicitly through a modernised re-write, the crime of aggression is in great danger of being irrelevant at the point when jurisdiction has only just arisen.

CHAPTER 4: CONSENT OF THE POWERFUL – THE ROLE OF THE SECURITY COUNCIL IN AGGRESSION PROCEEDINGS

1. Introducing Articles 15 bis and ter

Article 15 *bis* and *ter* contain the jurisdictional conditions imposed on the prosecution of the crime of aggression which, the examination has already revealed, proved to be a matter of the greatest controversy throughout the negotiations. Finding any agreement on the jurisdictional conditions continued to be particularly difficult, due to the controversial and political nature of the crime and the relative novelty of the condemnation and prosecution of aggression. At the root of the disagreement was the prevalent reluctance of many States to relinquish, even partially, their sovereignty and grant their consent to jurisdiction over a novel crime by a new and untested permanent international court. We was also widely felt that the Security Council ought to retain control over aggression proceedings as a matter of peace and security. Its permanent members, in particular, felt that aggression ought to be dealt with solely in the international political sphere, rather than the criminal justice system, thus seeking a monopoly. The main 'bone of contention' during the entire negotiations was therefore the role of the Security Council

⁷⁰³ See the discussion in section 6.2, Chapter 1.

⁷⁰⁴ See the account in Kress and von Holtzendorff (2010) (n. 5), 1179, 1194.

⁷⁰⁵ *Ibid*

and whether its decisions ought to be binding on the Court.⁷⁰⁶ As Kress and von Holtzendorff observed:

To agree on a solution that bridged the wide gap between the monopoly claim of the five permanent council members on the one end and the rejection of any role for the Council beyond the one already recognized in Article 16 of the ICC Statute on the other end of the spectrum posed a formidable challenge.⁷⁰⁷

Despite the remarkable efforts of the SWGCA to resolve these issues, negotiations on the jurisdictional conditions nearly resulted in failure, however eventually a last-minute compromise was agreed. Instrumental for this compromise was the so-called ABCS non-paper, which was highly influential in the closing stages of the negotiations and is largely responsible for the wording of Articles 15 *bis* and *ter*. This paper provided for a) automatic jurisdiction where a Security Council referral exists, b) in the absence of such a Security Council referral, non-jurisdiction of the Court over non-signatories, and c) again in the absence of such a Security Council referral, jurisdiction over signatories limited by a choice of those members to opt out. The resulting compromise involved separating the provisions into two substantially different articles depending on the source of referral. The differences reflect the pervading difficulties incurred during the negotiations. Notwithstanding the substantial differences, Articles 15 *bis* ans *ter* contain identical delay provisions.

⁷⁰⁶ See Cassese (2013) (n. 12), 138.

⁷⁰⁷ Kress and von Holtzendorff (2010) (n. 5), 1194.

⁷⁰⁸ *Ibid*, 1203-4, citing the unpublished *Declaration (Draft of 9 June 2010, 16h00)*.

The final Article 15 *bis* confers highly limited jurisdiction to the Court, where a state party has referred a case to the Prosecutor for investigation, ⁷⁰⁹ or the Prosecutor decides to investigate a case under their own powers (*proprio motu*). ⁷¹⁰ This article, at least *prima facie*, provides an alternative to absolute Security Council monopoly over the crime of aggression, allowing for limited jurisdiction of the Court in situations where the Security Council cannot agree on a referral or chooses not to actively authorise investigation and prosecution. Article 15 *bis* provides:

Exercise of jurisdiction over the crime of aggression (State referral, *proprio motu*)

- 1. The Court may exercise jurisdiction over the crime of aggression in accordance with article 13, paragraphs (a) and (c), subject to the provisions of this article.
- 2. The Court may exercise jurisdiction only with respect to crimes of aggression committed one year after the ratification or acceptance of the amendments by thirty States Parties.
- 3. The Court shall exercise jurisdiction over the crime of aggression in accordance with this article, subject to a decision to be taken after 1 January 2017 by the same majority of States Parties as is required for the adoption of an amendment to the Statute.
- 4. The Court may, in accordance with article 12, exercise jurisdiction over a crime of aggression, arising from an act of aggression committed by a State Party, unless that State Party has previously declared that it does not accept such jurisdiction by lodging a declaration with the Registrar. The withdrawal of such a declaration may be effected at any time and shall be considered by the State Party within three years.
- 5. In respect of a State that is not a party to this Statute, the Court shall not exercise its jurisdiction over the crime of aggression when committed by that State's nationals or on its territory.
- 6. Where the Prosecutor concludes that there is a reasonable basis to proceed with an investigation in respect of a crime of aggression, he or

State Party in accordance with article 14'.

710 Derived from Article 13 (c), granting jurisdiction where '[t]he Prosecutor has initiated an investigation in respect of such a crime in accordance with article 15'.

⁷⁰⁹ Derived from Article 13 (a) of the Rome Statute (n. 1), which provides for jurisdiction in '[a] situation in which one or more of such crimes appears to have been committed is referred to the Prosecutor by a State Party in accordance with article 14'.

she shall first ascertain whether the Security Council has made a determination of an act of aggression committed by the State concerned. The Prosecutor shall notify the Secretary-General of the United Nations of the situation before the Court, including any relevant information and documents.

- 7. Where the Security Council has made such a determination, the Prosecutor may proceed with the investigation in respect of a crime of aggression.
- 8. Where no such determination is made within six months after the date of notification, the Prosecutor may proceed with the investigation in respect of a crime of aggression, provided that the Pre-Trial division has authorized the commencement of the investigation in respect of a crime of aggression in accordance with the procedure contained in article 15, and the Security Council has not decided otherwise in accordance with article 16.
- 9. A determination of an act of aggression by an organ outside the Court shall be without prejudice to the Court's own findings under this Statute.
- 10. This article is without prejudice to the provisions relating to the exercise of jurisdiction with respect to other crimes referred to in article 5.

Article 15 *ter*, by comparison, confers automatic and unlimited jurisdiction to the Court, where the Security Council refers a case. ⁷¹¹ Article 15 *ter* provides:

Exercise of jurisdiction over the crime of aggression (Security Council referral)

- 1. The Court may exercise jurisdiction over the crime of aggression in accordance with article 13, paragraph (b), subject to the provisions of this article.
- 2. The Court may exercise jurisdiction only with respect to crimes of aggression committed one year after the ratification or acceptance of the amendments by thirty States Parties.
- 3. The Court shall exercise jurisdiction over the crime of aggression in accordance with this article, subject to a decision to be taken after 1 January 2017 by the same majority of States Parties as is required for the adoption of an amendment to the Statute.

175

⁷¹¹ See Article 13 (b) Rome Statute, granting jurisdiction in '[a] situation in which one or more of such crimes appears to have been committed is referred to the Prosecutor by the Security Council acting under Chapter VII of the Charter of the United Nations'.

- 4. A determination of an act of aggression by an organ outside the Court shall be without prejudice to the Court's own findings under this Statute.
- 5. This article is without prejudice to the provisions relating to the exercise of jurisdiction with respect to other crimes referred to in article 5.

Before embarking on an examination of the focal point of this chapter, the Security Council's role in aggression proceedings, let us briefly comment on the special jurisdictional conditions contained within Articles 15 *bis* and *ter*, building the foundation for the discussion in this chapter as well as the next.

1.1 The unusual procedural and temporal delays common to Articles 15 bis and ter

Articles 15 *bis* and *ter* jointly establish a special procedure for the crime of aggression. Additionally, both articles provide for identical procedural and temporal delays and hurdles, before jurisdiction of the Court comes into force. Common paragraph 2 of Articles 15 *bis* and *ter* stipulates a minimum of 30 ratifications in addition to a delay of one year thereafter. Common paragraph 3 insists on a further 'decision to be taken after 1 January 2017 by the same majority of state parties as is required for the adoption of an amendment to the Statute, i.e. 2/3 of member states as stipulated in Article 121 (3).

⁷¹² Article 15 *bis* (2), and Article 15 *ter* (2) Rome Statute (n. 1).

⁷¹³ Article 15 *bis* (3), and Article 15 *ter* (3).

⁷¹⁴ Article 121 Rome Statute sets out the procedure for amendments to the Statute, and para 3 of that article provides that '[the adoption of an amendment at a meeting of the Assembly of States Parties or at a Review Conference on which consensus cannot be reached shall require a two-thirds majority of States Parties.' This figure applies to the percentage of *all* member states, not just those present at an Assembly of States Parties. The relationship between Article 15 *bis* and Article 121 will be examined in detail in the next chapter.

By comparison, these special provisions can be distinguished from the coming into force conditions for the other crimes under the Statute, which provided for no such delays.⁷¹⁵ The delay provisions in paragraphs 2 and 3 were cumulative,⁷¹⁶ signifying the intent of state parties to delay jurisdiction of the Court for at least 7 years after Kampala. The decision to postpone may have been the result of 'practical discords that [had] not yet been put to rest'.⁷¹⁷ Some of these concerned the question of state consent, even by member states, to jurisdiction of the Court.⁷¹⁸ This divisive issue is the central subject of the next chapter.

Another reason for the unusual delay, directly related to the main topic of this chapter, was the unwavering conviction, particularly by the P5 members, that the Security Council ought to have a monopoly over any aggression proceedings.⁷¹⁹ As Antonio Cassese observes:

These limitations arose, in part, because of intense debate over whether the UN Security Council would have the exclusive power to refer a situation of alleged aggression to the ICC. Ultimately, the amendments did not give the UN Security Council a monopoly on referrals. Because of controversy over the matter, the vesting of jurisdiction was delayed pending state party vote and ratifications.⁷²⁰

⁷¹⁵ See Article 6, Article 7, and Article 8 and simple jurisdiction through any of the mechanisms in Article 13 Rome Statute.

⁷¹⁶ Third Understanding included in Annex III to Resolution 6 (n. 2). For commentary, see Cassese (2013) (n. 12), 138, and McDougall (2013) (n. 682), 286. See also Dinstein, (2017) (n. 154), 139-40, paras 371-3. ⁷¹⁷ Dinstein, *ibid*, 141, para 378.

⁷¹⁸ Ibid

⁷¹⁹ Cassese (2013) (n. 12), 138. See also the 'Explanations of Position', particularly by the P5, in *Review Conference Official Records*, RC/11, Annexes VII-IX, 122, and *Travaux Preparatoires* (2012) (n. 317), from 810.

⁷²⁰ Cassese, ibid.

The postponement allowed for another opportunity to revisit the issue before activation and lodge objections. It therefore appeased the concerns of the P5 sufficiently not to object to the adoption of the Kampala amendment.⁷²¹ The United Sates, for instance, considered it a 'wise decision to delay implementation of the crime of aggression to permit examination of the practical implication of the two methods [in Articles 15 *bis* and *ter*] being proposed for the operationalization of this crime'.⁷²² The other P5 members took a similar stance in their post-Kampala statements, and confirmed their continuing conviction that the Security Council ought to have a monopoly over aggression proceedings.⁷²³

It is also conceivable that the delay provisions may have been incorporated in order to grant the Court sufficient 'breathing room', before facing difficult and complicated political and legal situations concerning the crime of aggression. The postponement of jurisdiction thus allowed the Court to develop sufficient competence, preventing it from being overburdened with decisions on 'major controversies about contemporary international security law through the backdoor of ... international criminal justice'. Additionally, the delays effectively enabled States to amend their policies and government and military operations, before jurisdiction arose. Consequently, the postponement of jurisdiction allowed States time to withdraw from questionable and

⁷²¹ See Stefan Barriga and Niels Blokker, 'Entry into Force and Conditions for the Exercise of Jurisdiction', in Kress and Barriga (eds), (2017) (n. 246), 634.

^{722 &#}x27;Statement by the United States of America', in 'Explanations of Position', *supra*, n. 719, at 816.

⁷²³ *Ibid.* 'Statement by France', 811, 'Statement by the United Kingdom', 813, 'Statement by China', 814, and 'Statement by the Russian Federation', 816.

⁷²⁴ See Stefan Barriga and Leena Grover, 'A Historic Breakthrough on the Crime of Aggression' (2011) 105 (3) AJIL, 517, 530, and Barriga and Blokker (n. 721), at 633, 634, and 636.

⁷²⁵ Kress and von Holtzendorff (2010) (n. 5), 1211.

Weisbord (2011) (n. 430), 99. See also Heinsch (2010) (n. 376), 734, arguing that ICL is still in its infancy, and the delays allow 'the young ICC time to establish itself, and states [...] time to adjust their national legislation to the new crime'.

potentially aggressive interventions in other countries.⁷²⁷ The encouragement of physical withdrawal from illegal interventions and a change of state policy to non-aggression was most certainly one of the underlying considerations of the negotiators in their choice to delay jurisdiction.⁷²⁸

The procedural conditions have now been satisfied, following the 30th ratification by Palestine on 26th June 2016⁷²⁹ and the decision taken at the 2017 ASP to activate the Court's jurisdiction as of 17 July 2018.⁷³⁰ Nevertheless, the imposition of the delay, after 12 years of additional negotiating time, represents a stark reminder of the scepticism surrounding the Court's jurisdiction over the crime of aggression in the first place.⁷³¹ This perception is also reflected by the other special jurisdictional conditions.

⁷²⁷ In particular the illegal Iraq War, see section 2.2. in Chapter 2 and n. 376.

⁷²⁸ See Heinsch, (2010) (n. 376), 734. UK forces had already withdrawn from Iraq by the end of April 2009. See BBC News, *UK combat operations end in Iraq*, 30 April 2009, http://news.bbc.co.uk/1/hi/8026136.stm. US forces completed their withdrawal in December 2011. See Reuters, *Timeline: Invasion, surge, withdrawal; US forces in Iraq*, https://www.uk.reuters.com/article/iraqusa-pullout-idINDEE7BH05520111218, December 18th 2011. Both accessed 14 December 2020.

⁷²⁹ See Coalition for the ICC, 'Palestine', https://www.coalitionfortheicc.org/country/palestine, accessed 12 July 2020.

⁷³⁰ See Coalition for the ICC, 'Historic activation of the Crime of Aggression', https://www.coalitionfortheicc.org/explore/icc-crimes/crime-aggression, accessed 12 July 2020.

⁷³¹ See Cassese's assessment (2013) (n. 12), 138.

1.2 The special jurisdictional conditions in Article 15 bis in comparison with

Article 15 ter

Where a State has referred the situation to the Court, or the Prosecutor has initiated an investigation *proprio motu*, Article 15 *bis* (4) allows for the possibility of a declaration of non-acceptance of jurisdiction by a member state. Thus paragraph 4 effectively provides for the pre-condition of consent (which is given by failing to take advantage of the right to opt out of the acceptance of jurisdiction) of a member state before one of its political or military leaders can be investigated or prosecuted for the crime of aggression.⁷³² This provision is fundamentally different to the other jurisdictional conditions and exceptional within the Rome Statute, since neither prosecution for the crime of aggression following a Security Council referral under Article 15 *ter*, nor for any of the other core crimes in Article 5, require a member's consent to jurisdiction.⁷³³ The consent requirement represents a critical handicap to the Court's jurisdiction and will be considered in detail in the next chapter.⁷³⁴

Paragraph 5 of Article 15 *bis* stipulates non-jurisdiction of the Court over non-member states. This constitutes a total exclusion of non-signatories from jurisdiction under Article 15 *bis*, even where they are the victim of an act of aggression by a member state. The requirement for both victim and aggressor to be signatories, before jurisdiction arises

⁷³² Article 15 bis (4) Rome Statute 1998, for text and discussion see Chapter 5.

⁷³³ See Article 12 (1) Rome Statute 1998, providing that '[a] State which becomes a Party to this Statute thereby accepts the jurisdiction of the Court with respect to the crimes referred to in article 5.'

⁷³⁴ See Chapter 5.

through state referral or *proprio motu*, is a further substantial difference from the other core crimes under the Court's jurisdiction.⁷³⁵ Thus it represents another considerable handicap to prosecution of the crime, which will be discussed in detail in the next chapter.

Article 15 *bis* (6) - (8) relate specifically to the main topic of this chapter, the emphasis on the role of the Security Council in, and *inter alia* the Council's consent to, aggression proceedings. Whilst Article 15 *bis* does operate as an alternative mechanism of initiation of proceedings to that of a Security Council referral, prosecutions as a result of state referral or *proprio motu* investigations are still dependent on some form of Security Council involvement. Thus, paragraph 6 creates an obligation for the Prosecutor to ascertain if the Security Council has made a determination of an act of aggression and submit all relevant information and documents to the Secretary General, prior to an investigation. The such a Security Council determination of an act of aggression exists, the Prosecutor is free to commence an investigation (paragraph 7). The however, no such determination exists, the Prosecutor has to wait for a period of 6 months and submit the request of investigation to the Pre-Trial Division for authorisation before he or she may begin to investigate or prosecute (paragraph 8). Paragraph 8 also expressly reserves the right of the Security Council under Article 16 to shelve any investigation for periods of 12 months at a time, and potentially for good.

⁷³⁵ Article 12(2) Rome Statute 1998 provides that '[i]n the case of article 13, paragraph (a) [i.e. state referral] or (c) [i.e. *proprio motu*], the Court may exercise its jurisdiction if one or more of the following States are Parties to this Statute or have accepted the jurisdiction of the Court in accordance with paragraph 3:

⁽a) The State on the territory of which the conduct in question occurred or, if the crime was committed on board a vessel or aircraft, the State of registration of that vessel or aircraft;

⁽b) The State of which the person accused of the crime is a national.'

⁷³⁶ Article 15 bis (6) Rome Statute.

⁷³⁷ Article 15 *bis* (7).

⁷³⁸ Article 15 *bis* (8).

⁷³⁹ Article 16 of the Rome Statute provides that '[n]o investigation or prosecution may be commenced or proceeded with under this Statute for a period of 12 months after the Security Council, in a resolution

It would not be unreasonable to view these provisions as simply acknowledging the authority of the Security Council over aggression proceedings as a matter of peace and security. The security of the Security Council over aggression proceedings as a matter of peace and security. The security of the security of the security of the problem arising from the insistence on this high level of Security Council involvement is that, in order to initiate proceedings, effectively the state consent of these five particular permanent members has to be sought, as a result of the internal voting structure of the Council. This applies even where the alleged aggressor and victim are signatories to the Rome Statute, have potentially ratified the amendments and not opted out of jurisdiction. The underlying question here, central to the discussion in this chapter, is whether this is appropriate, and whether it serves to further deterrence from commission of aggression, the Court's reach over the crime and its own reputation as an independent criminal court.

Notwithstanding the contentious emphasis on Security Council involvement established in the previous paragraphs, Paragraph 9 of Article 15 *bis* declares that an existing determination of an act of aggression is without prejudice to the Court's own findings.⁷⁴⁴ This provision represents an attempt to rebalance the potentially prejudicial effects of determinations by an external organ, and provide reassurance of a fair trial.⁷⁴⁵ Para 10

adopted under Chapter VII of the Charter of the United Nations, has requested the Court to that effect; that request may be renewed by the Council under the same conditions'.

⁷⁴⁰ See Article 24 and 39 of the UN Charter (n. 134) and detailed discussion below in section 2.

⁷⁴¹ See for Kress and von Holtzendorff (2010) (n. 5), 1186. See also the post-Kampala statements by the P5 (n. 722-3).

⁷⁴² See Chapter 1, section 4.2, and the discussion below particularly in section 3.3.

⁷⁴³ As a result of combining Article 15 *bis* (4) with Article 121 (5), and its negative interpretation. See Chapter 5, section 2 for detail.

⁷⁴⁴ Article 15 bis (9) Rome Statute.

⁷⁴⁵ Nevertheless, it is contended that there is a real danger that a prior determination of aggression by an external organ may, in reality, be prejudicial to the finding that the *actus reus* has been committed, since the state act itself forms part of the crime. See Chapter 3, section 2.

expressly preserves the existing jurisdictional conditions of the other core crimes in Article 5,⁷⁴⁶ as established in Article 12. Thus, paragraph 10 of Article 15 *bis* effectively serves to highlight the different treatment of the crime of aggression to that of the other core crimes.⁷⁴⁷

The provisions of Article 15 *ter*, by comparison with Article 15 *bis*, contain few surprises. The temporal restrictions mirror those of Article 15 *bis*, also delaying jurisdiction of the Court. The without prejudice provision and preservation of conditions for the other crimes in Article 5 are also identical to Article 15 *bis*. Except for these provisos, no further limitations apply in cases of Security Council referral, allowing for automatic and unrestricted jurisdiction of the Court.

1.3 The issues to be considered

As this introduction to the jurisdictional conditions of the crime of aggression demonstrates, Article 15 *bis*, in particular, raises two substantive issues, firstly, the amplified requirement of States' consent to jurisdiction in addition to membership, 750 and, secondly, the great emphasis on Security Council scrutiny. The insistence on multiple consent by member states and non-application to non-members will be examined in detail in the next chapter. This chapter will continue to investigate the insistence on substantial

⁷⁴⁶ Article 15 bis (10) Rome Statute.

⁷⁴⁷ See the brief introduction to this problem at p. 180 above, and the discussion in Chapter 5 of the interaction of Article 12, 15 *bis* (4) and 121 (5).

⁷⁴⁸ See Article 15 ter (2) and (3) Rome Statute.

⁷⁴⁹ See Article 15 *ter* (4) and (5).

⁷⁵⁰ See Chapter 5.

involvement of the Security Council in any aggression proceedings regardless of the source of the referral to the Court. Because of the internal voting structure within the Council, its resolutions and any referrals to the Court essentially require acquiescence by its permanent members. This is based on the fact that Article 27 (3) of the UN Charter stipulates that substantive decisions of the Security Council require a majority of nine out of 15 votes, including the unanimous votes of the five permanent members. A negative vote by one of the P5, accordingly amounts to a veto. If one of these permanent members uses their veto power, thus expressing lack of consent, a resolution will fail and no referral to the Court will ensue. This effectively amounts to an implicit requirement of consent by the powerful, before the Court is granted jurisdiction over an aggression case under Article 15 ter.

By the same token, and in spite of the alternative sources of referral eventually agreed on and incorporated into Article 15 *bis*, these investigations and prosecutions also require some form of consent to jurisdiction by the Council. Thus they have to be 'run past' the Council first, which then has the power to halt investigations or prosecutions temporarily or even permanently through application of Article 16.⁷⁵³ This requirement of effective Security Council approval prior to investigation or prosecution is unusual within the provisions of the Rome Statute,⁷⁵⁴ and amounts to a considerable limitation on the Court's jurisdiction.

⁷⁵¹ Article 27 (3) UN Charter (n. 210), see also the discussion in section 4.2 of Chapter 1.

⁷⁵² Ibid

⁷⁵³ By virtue of Article 16 of the Rome Statute.

⁷⁵⁴ See n. 715 above.

Related problems include, for instance, the likelihood of frustrations of proceedings due to Security Council inactivity and legality problems raised by these external retrospective determinations. Is such a high level of control by the political organ of the Security Council appropriate in criminal prosecutions, considering such decisions are subject to the individual consent of its members, particularly the P5? Would alternatives to substantial Security Council involvement have been feasible? Should the Court have been granted wider jurisdiction instead, independent of the political forum of the Council? Is the insistence on Council involvement beneficial, by potentially providing jurisdiction through a referral where the Court would otherwise not have such jurisdiction because of the requirements of membership and consent? Or are those benefits, perhaps, outweighed by the fact that the Security Council itself is dependent on the consent of its powerful permanent members? A good starting point in this evaluation is the examination of the role of the Security Council as envisaged in the UN Charter.

2. The role of the Security Council – exclusivity or mere primacy?

2.1 Security Council responsibility according to the UN Charter

Any attempt to evaluate the appropriateness of Security Council involvement in the Court's jurisdiction over aggression ought to begin by considering the instrument which establishes the Security Council's power, the UN Charter. Article 24 of the UN Charter provides in paragraph 1:

In order to ensure prompt and effective action by the United Nations, its Members confer on the Security Council primary responsibility for the maintenance of international peace and security, and agree that in carrying out its duties under this responsibility the Security Council acts on their behalf.⁷⁵⁵

This article refers to 'primary responsibility' or primacy, suggesting that other types or levels of responsibility co-exist with that of the Security Council. Article 39 elaborates on the powers of the Security Council and provides:

The Security Council shall determine the existence of any threat to the peace, breach of the peace, or act of aggression and shall make recommendations, or decide what measures shall be taken in accordance with Articles 41 and 42, to maintain or restore international peace and security.⁷⁵⁶

This article does not provide for a hierarchy of determination power, instead referring simply to the powers of the Council. The combination of these two articles consequently raises the question whether the determination power of the Security Council, based on the responsibility allocated to them in Article 24, is merely primary, i.e. it can be executed by and shared with other organs, or whether it is in fact exclusive. In the former case, the UN Charter provisions would grant primary competence to the Council relating to the

⁷⁵⁵ Article 24 (1) of the UN Charter (n. 134).

⁷⁵⁶ *Ibid*, Article 39 (Chapter VII).

power to take action and maintain peace and security, but not the sole power to authorize judicial action to the exclusion of any other organ. If Security Council authority, on the other hand, could be perceived as exclusive, it would follow that aggression, as a matter of peace and security, ought to be entirely within the Council's prerogative. Each of these two positions was fiercely defended by negotiating States even before the inception of the International Criminal Court and up to and even beyond Kampala, polarising States and academics alike.

2.2 The stance of the ILC and SWGCA on the contentious question of Security

Council monopoly

The discussion in Chapter 1 revealed that a large number of States historically considered the Security Council's determination powers over acts of aggression as absolute, one of the possible interpretations derived from a literal reading of Article 39 on its own. In fact, some States' insistence on a literal interpretation of Security Council exclusivity went as far as opposing *any* jurisdiction of the Court over the crime of aggression even in the leadup to the Rome Conference. Nevertheless, the International Law Commission included the crime of aggression as one of the core crimes under the future Statute in the 1994 Draft Statute for the creation of an International Criminal Court.

⁷⁵⁷ See the discussion below, and arguments for shared competency, in section 2.2.

⁷⁵⁸ See Bassiouni and Ferencz (1999) and their description of opposition to the inclusion of aggression in the Rome Statute by the United States and some of its allies, including the United Kingdom (n. 306), 346. ⁷⁵⁹ 1994 ILC Draft Statute (n. 309), Article 20.

Although the ILC promoted inclusion of aggression in the jurisdiction of the Court, they addressed the contentious question of Security Council monopoly by suggesting that a determination of an act of aggression by the Security Council was indeed an essential prerequisite to any potential prosecution of the crime. Thus, Draft Article 23 (2) provided that '[a] complaint of or directly related to an act of aggression may not be brought under this Statute unless the Security Council has first determined that a State has committed the act of aggression which is the subject of the complaint'. ⁷⁶⁰

Consequently, because of the voting structure within the Security Council, any determination of an act of aggression would be dependent on the consent of the P5 members. If any of these members were to reject such a determination, no prosecution could take place. The ILC's suggestion to make prosecutions of the new Court over the Crime of Aggression entirely dependent on Security Council approval or scrutiny can be seen as appearing the prevalent, although not unchallenged, view that jurisdiction over the political crime of aggression ought, ultimately, to be regulated by a political organ. The ILC included the provision for Security Council exclusivity, despite acknowledging some criticism from within that it 'would introduce into the statute a substantial inequality between State members of the Security Council and those that were not members, especially between the Permanent Members of the Security Council and other States'.

⁷⁶⁰ *Ibid*, Article 23 (2).

⁷⁶¹ See Barriga (2012), in Barriga and Kress (eds.), *Travaux Preparatoires* (n. 317), 6. See also 1998 Rome Summary Records (18 June through 8 July), *ibid*, 255-314, for the vastly diverging views of States on the inclusion of the crime within the Court's jurisdiction and the level of independence of the Court. These documents demonstrate that most Arab and developing countries favoured independent jurisdiction of the Court, whereas the P5 and many Western States defended Security Council control, with some opposing *any* jurisdiction of the Court.

⁷⁶² 1994 ILC Draft Statute (n. 309) with commentaries, para 15 of commentary on Draft Article 23.

The ILC's view of an absolute requirement for Security Council determinations ignored existing evidence for shared, rather than exclusive, determination power. Such evidence can be derived from the occasional application of the General Assembly of its Uniting for Peace Declaration, 763 and other findings and condemnations of 'aggressive acts', 'acts of aggression', or simply 'aggression', where the Security Council has remained silent on violations of the Charter prohibition. 764 Additional support can be found in the *Certain Expenses* case, in which the ICJ confirmed that the Security Council had primary but not exclusive responsibility, permitting the General Assembly to make recommendations where the Security Council remained silent. 765 The contention that the General Assembly ought to be able to provide alternative authorisation of investigation and prosecution of the crime, with or without additional involvement of the ICJ, was considered in some States' proposals. 766 This suggestion of alternative authorisation was also reflected in a number of discussion papers and reports by the SWGCA. 767

⁷⁶³ *Uniting for Peace* Resolution (U4P) (n. 253). For the ten Emergency Special Sessions of the General Assembly invoked either the Assembly itself or request of the Council, and resolutions passed under U4P, see n. 258.

⁷⁶⁴ See the discussion in Blokker, 'The Crime of Aggression and the United Nations Security Council' (2007), 20 Leiden JIL, 881.

⁷⁶⁵ Certain Expenses of the United Nations (Article 17 (2) of the Charter), (Advisory Opinion) [1962], ICJ Reports, 151, page 163.

⁷⁶⁶ See for instance 'Proposal submitted by Bosnia and Herzegovina, New Zealand and Romania on jurisdictional conditions', 23 February 2001, PCNICC/2001/WGCA/DP.1. See also 'Proposal submitted by Samoa', 2 June 2002, PCNICC/WGCA/D.P.2, which emphasised the importance of an external determination of a state act without specifying by which particular organ. Australia submitted the argument that the Security Council should have 'the first bite of the cherry, but not necessarily the last,' citing Article 24 of the UN Charter, as cited in the 2006 *Princeton Report* (n. 552). Similarly, see 'Proposal presented by Belgium on the Question of Jurisdiction of the Court with Respect to the Crime of Aggression', 29 January 2007, ICC-ASP/5/SWGCA/W.P.1., which introduced the idea of the Pre-Trial Division acting as a filter. ⁷⁶⁷ See the 'Discussion Paper Proposed by the Coordinator', Conditions, Option 1, Variant 2 (9 December 1999), UN Doc. PCNICC/1999/WGCA/RT.1; the Coordinator's Discussion Paper of April 2002, (n. 413) para 4, Option 3; the Coordinator's Discussion Paper of July 2002 (n. 316) para 5, Option 4; 'Discussion Paper on the Crime of Aggression Proposed by the Chairman', (2008 Chairman's Paper), Article 15 *bis*, Alternative 2, in *Report of the Special Working Group on the Crime of Aggression*, in ASP Official Records, ICC-ASP/6/20/Add.1, Annex II, 9. See also the 2007 *Princeton Report* (n. 416), para 31, and the December 2007 *SWGCA Report* (n. 422), para 39.

There is little room for questioning the ICJ's proficiency. ⁷⁶⁸ The assumed competence of the ICJ gains support from Article 36 of its constitutive Statute, providing for jurisdiction over its own jurisdictional matters, and therefore autonomy. Additionally, the above discussion of Article 24 of the UN Charter itself has shown that this provision can be interpreted to the effect of granting merely primary competence to the Council, not exclusivity. 770 Examples of past ICJ's condemnations of aggression include the judgments in Nicaragua,771 the so called Nuclear Weapons advisory opinion772 and Armed Activities case. 773 Confirmation of the ICJ's competence to determine an act of aggression, can be derived specifically from the separate opinion of Judge Simma in the Armed Activities case, who admitted:

The Council will have had its own – political reasons for refraining from such a determination [of aggression]. But the Court, as the principal judicial organ of the United Nations, does not have to follow that course. Its very raison d'etre is to arrive at decisions based on law and nothing but the law, keeping the political context of the cases before it in mind, of course, but not desisting from stating what is manifest out of regard for such non-legal considerations.⁷⁷⁴

⁷⁶⁸ For support, see Blokker (2007) (n. 764) 886.

⁷⁶⁹ Article 36 (6) of the ICJ Statute (n. 35) provides: 'In the event of a dispute as to whether the Court has jurisdiction, the matter shall be settled by the decision of the Court.'

⁷⁷⁰ Article 24 of the UN Charter (n. 134).

⁷⁷¹ *Nicaragua* (1986) (n. 37), paras 32-35.

⁷⁷² Nuclear Weapons (1996) (n. 370), Decision on the Admissibility of the legal question before the Court on the use of force, 8 July 1996, para 13.

⁷⁷³ Armed Activities case (2005) (n. 37), Joint Separate Opinion of Judges Higgins, Koojimans, Elaraby, Owada, and Simma, paragraph 23.

⁷⁷⁴ *Ibid*, separate opinion of Judge Simma, para 3 (page 335 of the Judgment).

Despite the clear competence of either the General Assembly or the ICJ, it is an interesting observation that the SWGCA's own stance grew to support independence of the Court from either Security Council scrutiny or that of any other organ, ⁷⁷⁵ in contrast to the ILC and the insistence by some States on external authorisation. The SWGCA recognised that the right of the accused to a due process within the international criminal justice system relies on external determinations not being binding on the Court. Therefore, it was felt that the Court itself, in order to fulfil its role as a criminal court, was the appropriate organ to exclusively determine whether the element of the state act of the Crime of Aggression had been fulfilled, negating any requirement for external involvement. ⁷⁷⁶ The Court and the Council were deemed to have 'autonomous, but complementary roles'.777 The 'without prejudice' provision, now contained in Articles 15 bis (9) and 15 ter (4) originally suggested that any external determination of aggression 'shall be without prejudice to the Court's determination of an act of aggression', 778 and was later changed to 'the Court's own findings'. The view of the SWGCA, which considered all the nuances of the various views equally and with apparent impartiality, provides great support for the argument for independent determination powers of the Court, despite the resistance displayed particularly by powerful States and the view of the ILC.

Notwithstanding the growing consensus within the SWGCA that the Security Council's role was, within the framework of international criminal justice, merely procedural, ⁷⁸¹ it

.

⁷⁷⁵ See Weisbord's assessment (2008) (n. 176), 206, whilst serving as a delegate to the SWGCA. See also the 2006 *Princeton Report* (n. 552), paras 70-2, and 2007 *Princeton Report* (n. 416), para 54.

⁷⁷⁶ 2006 and 2007 Princeton Reports, ibid.

⁷⁷⁷ See the December 2007 SWGCA Report (n. 422), para 24.

⁷⁷⁸ 2008 *SWGCA Report* (n. 766), para 26 (emphasis added).

⁷⁷⁹ Report of the Special Working Group on the Crime of Aggression (2009) ICC-ASP/7/SWGCA/2, para

⁷⁸⁰ Weisbord (2008) (n. 176), 206.

⁷⁸¹ See the 2006 and 2007 *Princeton Reports* (n. 552 and 416 respectively).

was still felt necessary to attempt placation of the powerful minority of States insisting on some form of external authorisation before permitting exercise of jurisdiction. For this reason, and in the attempt to avoid absolute Security Council monopoly, the SWGCA continued to include the suggestion of alternative authorisation by the external organs of the General Assembly and the ICJ.⁷⁸²

In an attempt to assess the benefits of such alternative authorisation, it is undeniable that a General Assembly declaration and referral would have had the benefit of carrying the weight of a condemnation of the state act of a large number of States, ⁷⁸³ although the individual's participation would be a separate matter for the Court to decide. An ICJ opinion, on the other hand, would be likely to raise legal arguments which could subsequently assist the ICC in establishing the elements of the crime, ⁷⁸⁴ although the ICJ, as a civil court for the adjudication between States, would necessarily apply different evidentiary standards to those of the ICC. A determination by either the General Assembly or the ICJ would circumnavigate Security Council deadlock and forced inactivity of the ICC as a result. If dependence of the Court on external determination was indeed inevitable, either of these options or a combination of both may have been preferable to total dependence on Security Council authorisation alone.

Notwithstanding these potential benefits, the question remained whether the Court's aggression proceedings ought to be dependent on determination by any external organ at all, be it the political organs of the General Assembly, the Security Council, or the ICJ in its capacity of adjudicating on civil matters between States. Should the Court enjoy

⁷⁸² See the reports and papers in n. 766.

⁷⁸³ Weisbord (2008) (n. 176), 211.

⁷⁸⁴ *Ibid*. 211-2.

independent jurisdiction, entirely free of external determinations, instead? In order to attempt to answer these questions, it may be helpful to engage in a deeper examination of the cardinal reasons for the powerful resistance to independent jurisdiction over the crime of aggression.

3. Why insist on dependence on the Security Council in the first place?

3.1 Protection of the Security Council's prerogative and the fear of conflicting

determinations

Following the inclusion of the still undefined crime of aggression in the Rome Statute 1998, States in favour of absolute Security Council control over aggression proceedings submitted proposals to that effect. Protection of the Security Council's prerogative over aggression was sought particularly by its permanent members and their supporters. Kress and von Holtzendorff confirm that Security Council exclusivity over the issue of aggression continued to be defended strongly by the permanent members of the Council throughout and even beyond the negotiations. The following statute of the council throughout and even beyond the negotiations.

An early proposal by Russia exemplifies the attempt to retain Security Council monopoly, insisting on sole determination power of an act of aggression as a prerequisite to any jurisdiction of the Court.⁷⁸⁷ The observer delegation also suggested that no time limits

⁷⁸⁵ See Cassese's assessment (2013), 138, cited in Chapter 1, p. 79.

⁷⁸⁶ See Kress and von Holtzendorff (2010) (n. 5), 1187.

⁷⁸⁷ Proposal by the Russian Federation, 1999, PCNICC/1999/DP.12.

should apply, leaving the matter entirely in the hands of the Security Council in perpetuity. This proposal therefore sought total dependence of any criminal aggression proceedings on the contingencies of agreement within the Security Council and non-application of the veto by any of the P5 members. Otherwise prosecutions would be barred entirely. Similar to Russia, China had already expressed their general distrust of the new Court *per se* and made little effort to participate in the discussions, apart from insisting on Security Council monopoly. The United States, the United Kingdom, France and a number of their allies also opposed the new Court's jurisdiction over the crime of aggression throughout the negotiations, despite their participation in the process.

France and the UK, as signatories, did eventually abandon their resistance to the adoption of the Kampala aggression amendment, if only in the final negotiation stages. Nevertheless, immediately after the adoption of the definition, France and the UK distanced themselves from the provisions, again siding with their fellow P5 members. Thus they expressed disapproval of the Kampala amendment as restricting Security Council powers in contravention of the UN Charter. This perceived flaw was, in their view, only slightly mitigated by the stringent conditions placed on the exercise of the Court's jurisdiction, in particular the continued extensive involvement of the Security Council. The Statement by the United Kingdom referred expressly to Article 39 and insisted that the 'text adopted cannot derogate from the primacy of the United Nations

⁷⁸⁸ *Ibid*.

⁷⁸⁹ See Bassiouni and Ferencz (1999) (n. 306), at p. 346. For China's and Russia's unchanged stance, see their post-Kampala Statements (n. 723).

⁷⁹⁰ See Bassiouni and Ferencz, *ibid*, and Statements by France, the UK (n. 723) and the US (n. 722).

⁷⁹¹ Statements by France and the UK, *ibid*.

⁷⁹² *Ibid*.

Security Council'. 793 France's Statement expressed regret that Article 15 bis (8) 'restricts the role of the United Nations Security Council and contravenes the Charter of the United Nations under the terms of which the Security Council alone shall determine the existence of an act of aggression.'794

The other P5 members, non-signatories China, Russia and the United States, also confirmed in their post-Kampala statements that they continued to hold onto their perception of Security Council monopoly over aggression, equating primary determination power with exclusivity. Thus, the Statement by China expressed disagreement with both Articles 15 *bis* and *ter* for failing to reflect that a Security Council determination was an absolute prerequisite before any jurisdiction of the Court could arise. Similarly, the Statement by the Russian Federation criticised the consensus definition for not reflecting the 'Security Council prerogatives in defining the existence of an act of aggression'. The United States 'associate[d] itself with the critical point of principle' asserted by France and the United Kingdom 'regarding the primacy of the Security Council under Article 39', claiming monopoly over matters of peace and security.

Searching for an explanation for the insistence on Security Council monopoly, it has been posited that these powerful States were simply attempting to preserve their privileges and

⁷⁹³ See n. 723 and *Travaux Preparatoires* (n. 317), 813.

⁷⁹⁴ *Ibid*, 811.

⁷⁹⁵ *Ibid*, 814.

⁷⁹⁶ *Ibid*, 816.

⁷⁹⁷ *Ibid*.

power status due to their special role within the Council, ⁷⁹⁸ and this reasoning is unsurprising. Retaining these special powers would enable a P5 member to veto a determination of aggression made against itself or one of its allies, therefore effectively gaining immunity from prosecution. ⁷⁹⁹ This point, addressing the imbalance of power between P5 members and other States, resulting potentially in unequal application of criminal justice, had been highlighted as a problem as early as 1994 in the commentary on the ILC's Draft Statute. ⁸⁰⁰ Looking back at the past record of the Security Council's determinations of aggression, ⁸⁰¹ the argument of self-interest appears hard to refute particularly in situations where the interests of one of the P5 or one of their allies are involved, and the record of the use of veto powers bears its own witness. ⁸⁰²

In an interesting twist, this argument of a prevailing and natural self-interest has been applied to further promote Security Council exclusivity and abandon the Court's jurisdiction over the crime of aggression entirely. Michael Glennon, for instance, claims that individual strategic self-interest of States, reflecting their security concerns and national interests, will always trump any attempt to regulate criminal aggression

⁷⁹⁸ See for instance Bassiouni and Ferencz (1999) (n. 306). See also Kress and von Holtzendorff (2010) (n. 5), 1194.

⁷⁹⁹ This situation has arisen recently, for instance, in the case of Russia's annexation of the Crimea, and aggressive incursions in Eastern Ukraine. In the face of Security Council paralysis, the General Assembly adopted Resolution 68, 'Territorial Integrity of Ukraine', A/RES/68/262 on 27 March 2014, condemning the aggressive annexation of the Crimea.

⁸⁰⁰ Commentary to the 1994 ILC Draft Statute, see n. 309, 759 and related text.

⁸⁰¹ See the discussion in Chapter 1, section 5.

⁸⁰² See the discussion of the veto in Chapter 1, section 4.2. For an example of recent prolific use of the veto, see the Security Council's record concerning Syria. Since 2011, 16 Draft Resolutions concerning the humanitarian crisis, violence, and use of chemical weapons in Syria have been vetoed (by Russia, an historic ally of the Syrian government, and sometimes China), most recently Draft Resolution 667, UN Doc. S/2020/667 (10 July 2020).

⁸⁰³ Glennon (2010) (n. 8), 111-2.

through international law. 804 For that reason, 805 he suggests, the Court's jurisdiction over aggression should be abandoned entirely. 806 Additionally, this argument defends a narrow interpretation of Article 39 of the UN Charter that monopoly over aggression is plainly and exclusively vested in the Security Council.⁸⁰⁷ Additional support for exclusivity is sought from Article 12 of the UN Charter which suspends the power of the General Assembly to make recommendations where the Security Council is considering the same situation. 808 A similar 'pre-emptive power', Glennon argues, exists by analogy with regards to suspending the work of the judicial organ of the Court, despite the fact that the Court has jurisdiction over criminal matters and was not even considered in or established by the Charter. 809 The alternative of allowing the Court concurrent or 'without prejudice' determination powers, would, the argument continues, lead to inevitable conflict by allowing different determinations with regard to state behaviour. Furthermore, it could result in the possible imposition of contradictory obligations on members of both the Security Council and the Court. 810 Glennon admits that the Council's own determinations might amount to a 'political roulette', since they are generally dispensed in an ad hoc, arbitrary and inconsistent fashion, based on policy decisions. 811 This, he argues, is a

⁸⁰⁴ *Ibid*, 72-3.

Reflections on the Judicialization of the Crime of Aggression', https://digitalcommons.law.yale.edu/fss-papers/4926, accessed 14 March 2020.

⁸⁰⁶ Glennon (n.8), 106-9.

⁸⁰⁷ *Ibid*.

⁸⁰⁸ Article 12 of the UN Charter (n. 134).

⁸⁰⁹ Glennon (n. 8), 108. For similar arguments of Security Council exclusivity, see also Saeid Yengejeh, 'Reflections on the Role of the Security Council in Determining Aggression' in Mauro Politi & Guiseppe Nesi (eds) *The International Criminal Court and the Crime of Aggression*, (Routledge 2004) 125, and Theodor Meron, 'Defining Aggression for the International Criminal Court' (2001), 25 Suffolk Transnat'l L. Rev. 1, 13. For a different view see for instance Weisbord (2008) (n. 176), 197-200, and Heinsch (2010) (n. 376), 742, dismissing the legality problem as minor, since it is not as strictly applied at international law as at domestic law, see the discussion below.

⁸¹⁰ See Glennon (n. 8), 106.

Ibid, 73. For other inconsistency arguments, see also Mark Stein, 'The Security Council, the International Criminal Court, and the Crime of Aggression: How exclusive is the Security Council's Power to determine aggression?' (2005) 16 Ind. Int'l L. & Comp. L. Rev. 1, 3, discussed below. For an account of Security Council inconsistency see Chapter 1, and n. 802 above. See also 'Historical Review of Developments Relating to Aggression, Prepared by the Secretariat', at 115-121,

necessary and unavoidable evil, concluding that the Council should have sole or absolute determination power over aggression to the exclusion of any other organ, regardless of whether they come to a finding or not. However, retroactive prosecutions for the crime of aggression as a result of the Council's broad discretion and *ex post facto* referrals to the Court, also raise legality issues and would result in 'an impossible compromise'. 812 Consequently, Glennon argues for the summary abandonment of the attempt to subject acts of aggression to any criminal prosecutions. 813

Michael Glennon is not alone in his view of inevitable conflict between the political organ of the Security Council and the judicial one of the International Criminal Court. The perception of antagonism between the two organs has led to the development of some radical and expansionist visions of Security Council control. In particular, this argument has been utilized to advocate extension of the Council's already substantial powers to curb jurisdiction of the Court over the Crime of Aggression, if political situations require it, notwithstanding the Council's own history of inconsistency. Mark Stein, for instance, would insist on absolute dependence of the Court on the Council in all situations, recommending a strict requirement to seek express Security Council approval in order to proceed with a trial for the crime of aggression. This is surprising, since he does not deem the Security Council's power to determine an act of aggression as exclusive. Any trial would be barred without such unequivocal consent.

Doc.PCNICC/2002/WGCA/L.1

(2002),

available

at

http://www.un.org/law/icc/documents/aggression/aggressiondocs.htm, accessed 12 October 2020.

⁸¹² Glennon (n. 8), 109.

⁸¹³ *Ibid*, 73, 114.

⁸¹⁴ See Stein (2005) (n. 811).

⁸¹⁵ *Ibid*, 2. For an account of Security Council inconsistency, see Chapter 1 and n. 802.

⁸¹⁶ Stein, *ibid*, 13.

⁸¹⁷ *Ibid.* 5.

Council chose to invoke Chapter VII powers or to make its own determination under Article 39, no other body (including the judiciary of the ICJ or the ICC itself) would, following Stein's recommendations, be allowed to make any independent determination. Stein would also extend Article 16 of the Rome Statute to provide for immediate permanent halting of ICC proceedings at the behest of the Security Council, thereby allowing them to retain ultimate control over the prosecution of the crime of aggression. Yet more controversially, Stein's vision would grant the Security Council quasi-judicial powers to 'vacate charges' and 'expunge convictions'. Stein's National Stein's Stein's National Stein's Stein's National S

Glennon's and Stein's arguments for absolute Security Council control over aggression proceedings are refutable, and will be answered in turn. Glennon's view of supremacy of national interests and security concerns over the applicability of international criminal law ignores and denies the general validity of the prohibition of the use of force under the UN Charter, which has been firmly upheld under customary international law, for instance by the ICJ in the case of *Nicaragua*. Aggressive use of force is prohibited under international law, unless authorised by the Security Council, or in very limited situations of self-defence, and any national interests. It also seems a little harsh to deny absolutely any well-meaning intentions of States generally, in their attempt to criminalise aggression. If this statement were accurate, consensus on even limited jurisdiction of the

⁸¹⁸ *Ibid*.

⁸¹⁹ *Ibid*, 32.

⁸²⁰ *Ibid*.

⁸²¹ Article 2 (4) UN Charter (n. 195).

⁸²² *Nicaragua* case (1986) (n. 37), paras 187-201. See also *Armed Activities* (2005) (n. 37), Judgment of 19 December 2005, paras 148-165 and *Oil Platforms* (2003) (n. 368), paras 27-30, 78.

⁸²³ UN Charter, Article 42.

⁸²⁴ Ibid. Article 51.

Court would never have been established at all, and no State would ever ratify the amendments, submitting themselves to the Court's jurisdiction.⁸²⁵

Countering Stein's suggestions, this thesis argues that his drastic measures are excessive in even the most delicate political situations. Furthermore, and on principle, they amount to an unacceptable level of political interference in the judicial process. Even if one did not perceive such interference of politics in matters of criminal law to be utterly objectionable, the suggested measure of extending Article 16 would be unnecessary, since orders for deferral can already be made consecutively and in effect permanently. Stein's proposal for severe limitations on the Court's ability to function independently and extensive political interference in judicial proceedings is highly questionable. The ICJ asserted in *Nicaragua*:

The Council has functions of a political nature assigned to it, whereas the Court exercises purely judicial functions. Both organs can therefore perform their separate but complementary functions with respect to the same events. 827

This declaration by the ICJ on the separation of judicial and political functions, and the independence and competence of its own jurisdiction, therefore provides, by analogy, as valid a reply to Stein as it does to Glennon. Furthermore, conflicting determinations by

^{825 39} member states have, as of 25 September 2019 ratified the aggression amendments, see www.crimeofaggressioninfo.com, accessed last 20 January 2021.

⁸²⁶ See Article 16 Rome Statute (n. 753).

⁸²⁷ *Nicaragua*, Decision on Jurisdiction of the Court and Admissibility of the Application, Judgment of 26 November 1984, para 95.

the Security Council and the Court need not necessarily be a foregone conclusion, nor would uncertainty as a result of inconsistencies be necessarily a fatal flaw. 828 As Professor Gaja asserts:

One cannot assume that the absence of a finding by the Security Council that aggression occurred implies that in the Security Council's view there is no aggression and that therefore a conflict would arise with a positive finding by the ICC that an individual has committed a crime of aggression.⁸²⁹

Applying this argument, there is no need for the findings by Security Council and the Court to concur, and therefore no need for dependence of the Court on determinations by the Council, relying essentially on the consent of the P5. Even Glennon concedes that concurrent determination powers of the Court and Council would allow for a wider choice of sanctions, i.e. concurrent judicial and political measures to deal with acts of aggression, ⁸³⁰ a statement which, in actual fact, unwittingly provides strong support for independent determination power by the Court and even operational partnership of the two organs. ⁸³¹

⁸²⁸ Ian Hurd, 'How not to argue against the Crime of Aggression: A Response to Michael Glennon' (2010) http://buffett.northwestern.edu/documents/working-papers/Buffett 10-001 Hurd.pdf, accessed 15 March 2020. See also Heinsch (2010) (n. 376).

⁸²⁹ G. Gaja, 'The Respective Roles of the ICC and the Security Council in Determining the Existence of an Act of Aggression' (2004) in Politi and Nesi (eds) *The International Criminal Court and the Crime of Aggression* (n. 809), 124.

⁸³⁰ Glennon (2010) (n. 8), 104.

⁸³¹ Benjamin Ferencz, in 'Speaking Frankly about the Crime of Aggression: Reconciliating Legitimate Concerns and Removing the Lock from the Courthouse Door', (2008), http://www.benferencz.org/arts/93b.html, accessed 19 March 2020. See also Cassese (2007) (n. 614).

In addition to these direct responses to Glennon and Stein, other arguments can be presented in favour of the Court's jurisdictional autonomy over aggression. Article 8 bis relies on determinations of a state act of aggression by the Court as an essential element of establishing the actus reus, 832 inextricably linking the state act and criminal responsibility of the individual. These determinations are therefore merely tangential to the assessment that the state act element of the crime was actually committed. They are not made for the purpose of political statements or condemnations, unlike determinations by the Security Council. Consequently, a determination by the Court does not venture into the political realm of the Security Council, and, equally, the Security Council has no business establishing individual criminal responsibility of leaders. This point was made, particularly, by Antonio Cassese, asserting that the Security Council has no jurisdiction over individuals, whilst conceding that Article 39 of the UN Charter has the effect of granting the Council 'primary and exclusive jurisdiction' over *States*' breaches of the prohibition of the use of force contained in Article 2 (4).833 Therefore, he argues, the Court should rightly enjoy free and independent jurisdiction over the Crime of Aggression.834

The independence and impartiality of the International Criminal Court ought to be preserved even where the Security Council has made an Article 39 declaration. Article 39 has the effect of granting primary competence to the Security Council to take action to maintain peace and security, in conjunction with Article 24, but not to authorise judicial action to the exclusion of any other institution.⁸³⁵ Both the General Assembly and the ICJ

⁸³² See Chapter 3, section 2.

⁸³³ Cassese (2007) (n. 614), 846, emphasis added.

⁸³⁴ *Ibid*, 846-7.

⁸³⁵ For Articles 24 and 39 UN Charter (n. 134), see the discussion in section 2.1 above. Support can be gained from the *Certain Expenses Case* (1962) (n. 765, page 163) and the argument for inherent competence

have in the past taken it upon themselves to come to independent findings of aggression, 836 and a reasonable argument can be made that the ICC ought to benefit from equal independence, particularly since the court necessarily requires and applies its own criminal standard of proof. Whilst the Rome Statute contains no express provision for jurisdictional autonomy akin to Article 36 (6) of the ICJ Statute, 837 the general statement in *Tadic* that a tribunal's 'jurisdiction to determine its own jurisdiction' was 'a necessary component in the exercise of the judicial function and does not need to be expressly provided for in the constitutive documents', 838 could be applied equally to the ICC. Deciding on their inherent competence, the tribunal decided that it was 'duty-bound' to exercise its jurisdiction 'regardless of the political background or the other political facets of the issue'. 839 Additional authority can be gained from the Armed Activities case, which asserts that '...it is clear that the practice of the International Court reflects this trend for tribunals and courts themselves to pronounce on compatibility with object and purpose, when the need arises'. 840 This opinion thus acknowledges that courts and tribunals other than the ICJ can also pronounce themselves on their own competence. Academic support has been provided, for instance by Dov Jacobs, 841 who argues that the Court ought to be presumed as inherently competent, applying 'mutatis mutandis' the authority of Article 36 (6) of the ICJ Statute, 842 and the logic adopted in the *Tadic* 843 and *Reparations* 844 cases that where an international organ has been instructed with a certain task, it ought to be

of the ICC by Jacobs (2010) (n. 556), 136, Weisbord, (2008) (n. 176), 197-200, and Heinsch (2010) (n. 376), 742-3.

⁸³⁶ See Chapter 1, section 5.1, and Chapter 2, section 2.1, and pp. 189-91 of this Chapter.

⁸³⁷ Article 36 (6) (n. 769) of the ICJ Statute (n. 35).

⁸³⁸ See *Tadic* (IT-94-1), Decision on the Defence Motion for Interlocutory Appeal on Jurisdiction, 2 October 1995 (para 18, 24), in which the ICTY cited the *Certain Expenses* case (1962) at p. 163 (n. 765). ⁸³⁹ *Tadic*, *ibid*, para 24.

⁸⁴⁰ Armed Activities case (2006) (n. 773), Joint Separate Opinion, para 23.

⁸⁴¹ Jacobs (n. 217 Chapter 2), 135-9. For similar arguments, see also Weisbord (2008) (n. 176), 197-200.

⁸⁴² ICJ Statute, Art 36 (6) (n. 769).

⁸⁴³ Tadic (n. 838).

⁸⁴⁴ Reparations for Injuries Suffered in the Service of the United Nations (hereinafter Reparations Case), ICJ, Opinion, 11 April 1949, p. 179.

regarded as having been 'clothed [...] with the competence required to enable those functions to be efficiently discharged'.⁸⁴⁵

The Court and the Council operate in very different hemispheres, and therefore it is unnecessary to consider different determinations as conflicting. Precedents for a judicial organ's perceived competence in even politically sensitive and complicated issues were set by the Nuremberg and Tokyo tribunals. Furthermore, the ICJ's above declaration in *Nicaragua* provides a convincing argument for the separation of judicial and political functions and independent exercise of jurisdiction by the ICC through analogy. Importantly, the ICJ also stated that the Court had 'never shied away from a case brought before it merely because it had political implications. The *Armed Activities* case, specifically the separate opinion of Judge Simma, provides once again an additional profound argument for independent judicial determinations.

Moreover, the Court is more likely to apply a legalistic and consistent definition in order to comply with the general requirement for certainty at law, and is also bound to consider the gravity threshold in Article 8 *bis* in their determinations, whereas the Security Council would have complete discretion due to their Chapter VII powers. Past examples of Security Council (non-)determinations demonstrate inconsistencies, ⁸⁴⁹ because the Council is bound to take into account political rather than legal considerations. These inconsistent political determinations are also applied retrospectively, and that

⁸⁴⁵ *Ibid*.

⁸⁴⁶ Nicaragua, (1984), see n. 827 and p. 200 above.

⁸⁴⁷ *Ibid*, para 96.

⁸⁴⁸ Armed Activities (2005), Joint separate opinion, para 23 (see n. 773 and related text).

⁸⁴⁹ See Chapter 1 and n. 802 above.

legality concerns.⁸⁵⁰ Inconsistent, combination raises serious retrospective condemnations are also not likely to inspire deterrence from future acts of aggression, whereas likely determinations by the Court for judicial purposes and resulting prosecutions might. It would follow that the Court is better equipped, and therefore the appropriate organ, to consistently determine criminal aggression independently of political considerations and inhibitions. 851 Many other crimes require similar judicial determinations, and the Rome Statute is restricted to the 'most serious crimes of concern to the international community as a whole', 852 intrinsically limiting the scope for excessive judicial expansion. Additionally, the crime of aggression has long been a crime at customary international law, even before a definition was achieved, and to deny the first permanent international criminal court independent jurisdiction is nonsensical. As the UK House of Lords stated in R v Jones:

The core elements of the crime of aggression have been understood, at least since 1945, with sufficient clarity to permit the lawful trial (and, on conviction, punishment) of those accused of this most serious crime. It is unhistorical to suppose that the elements of the crime were clear in 1945, but have since become in any way obscure. 853

The argument that the Court cannot independently try aggression for legality reasons is, accordingly, unpersuasive, and unfettered judicial determination and interpretation

 $^{^{850}}$ See Glennon's arguments above, n. 812 and related text.

⁸⁵¹ See Ferencz (2009) (n. 4), 290. See also Jacobs (n. 556), 138-40, and Weisbord (2011) (n. 430), 88, 120-2, and the discussion above, particularly at pp. 189-91.

⁸⁵² See Preamble of the Rome Statute (n. 85).

⁸⁵³ Lord Bingham in *R v Jones* [2006] UKHL 16 (n. 37), para19.

powers, irrespective of political considerations and consent by member states of the Security Council, are both necessary and appropriate. Security Consequently, an argument can be made for extensive jurisdictional determination powers and wide ranging discretion of the Court, and it may even be conceivable that such judicial interpretations will lead towards a redefining of the international legal order, promoting 'peace as they do justice'. Second Sec

Because the criminalization of aggression requires the ICC to intervene in the domain of fundamental national security, an area of discretion warily guarded by states, the stakes will be especially high. Judicial interpretation of the crime of aggression [including the determination of an act of aggression] will not only determine the outcome of particular cases, it will shape the international legal order.⁸⁵⁶

An example of such a need for judicial determination could arise, for instance, where a Security Council determination and condemnation of aggression had failed due to use of the veto. In this case the Court should have the discretion to interpret this failed resolution as mere silence rather than a determination that no act of aggression had occurred, continuing with its investigation and prosecution if the evidence of the case supports it.

⁸⁵⁴ See also Weisbord (2011) (n. 430) 87-8, advocating extensive jurisdictional determination powers and wide ranging discretion of the Court, in polar opposition to the restrictive view supported for instance by Glennon (n. 8 and 812), Weisbord expresses hope that judicial interpretation will result in redefining 'the international legal order [by intervening] in the domain of fundamental national security, and area of discretion warily guarded by states...' This argument is, however, likely to be countered with accusations of judicial policy-making and undermining of the Security Council's powers.

⁸⁵⁵ Weisbord, ibid, 88.

⁸⁵⁶ *Ibid*.

Such judicial discretion would allow for case by case solutions, incrementally developing the infant crime of aggression, whilst establishing guiding principles. At the same time, criminal prosecutions could be kept separate from political considerations by external organs, permitting the Court to focus on objective evaluations of whether an act of aggression has actually occurred as a precursor to a decision on the defendant's potential criminal liability.

Furthermore, it has been suggested that whilst the Court should not be bound by the findings of the Security Council, conversely international political institutions including the Council should take account of judicial findings. Ideally, there would thus be a working relationship between the Court and the Council, operating in tandem, without encroaching on each other's area of expertise. Such a relationship had been envisaged in a 1998 Proposal by Cameroon. Interestingly, that particular proposal also foresaw the possibility of an enforcement mechanism in the relationship between the Court and the Security Council, suggesting that the Court could ask the Security Council for assistance in investigations, by providing arrest warrants and enforcement of ICC judgements under its Chapter VII powers. In a similar vein, and notwithstanding concerns that aggression investigations might be too 'resource-consuming' for the Court, Norway stated:

In cases of manifest acts of aggression, we consider that it is incumbent on the Security Council of the United Nations and its individual members to fully take

⁸⁵⁷ Weisbord, (2008) (n. 176), 208-10.

⁸⁵⁸ Cassese (2007) n. 614, 846.

⁸⁵⁹ UN Doc. A/CONF.183/C.I./L.39, Suggested Article 10.

⁸⁶⁰ *Ibid*, although, somewhat contradictorily, the proposal also refers to pre-emptory determination power by the Council (10 (1)) prior to proceedings.

into account the possibility of referring such situations to the Court in accordance with the Charter. 861

Consequently, these views credit the Court with independence and a perception of impartiality, although the court's own decisions may be politically influential, and even compel the Council to take action. This perceived onus to cooperate with the Court and consider judicial outcomes is highly interesting, as it could potentially entail a requirement for justification, where a P5 member continuously blocks a resolution pertaining to an act of aggression and the Court has made an autonomous determination of a state act. Disagreements between the two organs may, therefore, even have the positive impact of resolving a deadlock by placing an onus or duty on member states to cooperate with the Court, or at least justify their decisions of objection.

This concept of a duty of cooperation is reminiscent of some of the theoretical arguments encountered in the Introduction, for instance, as presented by Ronald Dworkin. 862 Following this approach, States were perceived to have a sovereign duty to promote a peaceful, effective international order for the benefit of their own citizens, pursue international stability, as well as cooperate internationally towards the prevention of human rights abuses, war crimes, genocide etc. It takes only a very small imaginary step to include the prevention and prosecution of the supreme international crime of aggression in this list. This envisaged onus or duty of cooperation would also assist in

^{861 &#}x27;Statement by Norway', in 'Explanations of Positions', Travaux Preparatoires (n. 317), 813.

⁸⁶² See Dworkin (2013) (n. 43) and pp. 17-20. See also the concept by early liberal internationalists of international cooperation out of necessity to achieve a 'higher unity of mankind', Collins, citing J. Bluntschli (n. 78 and related text).

promoting the 'minimum public world order' devoid of aggression, as envisaged by McDougal and Feliciano. Ref Furthermore, the promotion of such a political order based on States' cooperation was the aim behind the creation of the United Nations in the first place, of which the Security Council is merely the executive organ. The withholding of individual state consent, through use of the veto by any of the P5, ought to at least require justification in referrals of aggression cases to the Court. Accordingly, the perception of an onus placed on the Security Council to cooperate with the Court and take notice of judicial findings is, this thesis argues, persuasive.

On balance, therefore, the above arguments provide compelling evidence for non-exclusivity of the Security Council's powers of determination, but, perhaps even more importantly, independent jurisdiction is also justified by the paramount necessity to observe the rights of defendants to due process before the ICC as a criminal court. Ref For the politically motivated Council to be granted Stein's envisaged extensive powers over judicial proceedings is against all notions of such judicial independence and due process of law. Prosecutions and trials for the crime of aggression would, under this vision, be simply a foregone conclusion without any observance of defendants' rights, or general principles of criminal law based on the political and often arbitrary will of the Security Council, in particular the P5.

Ultimately, the argument presented by advocates of Security Council monopoly of inevitable conflict between the judicial institution of the ICC and the Security Council

⁸⁶³ See McDougal and Feliciano (1958), 812 (88).

⁸⁶⁴ For support, see Cassese, (n. 614) and the suggestions by Cameroon (n. 859) and Norway (n. 861).

⁸⁶⁵ This point is also made by Jacobs (2010) (n. 556), 137,

can be refuted by the counter-argument that the Court's responsibility is to determine the criminal liability of individuals rather than to politically stigmatise States by applying sanctions. Some overlap may be incidental because of the prerequisite of identifying the state act of aggression in order to prosecute an individual,⁸⁶⁶ but this thesis submits that this is preferable to a Court whose jurisdiction is severely hamstrung.

Having attempted to refute the arguments for Security Council monopoly of determinations based on prerogative and concerns of conflicting outcomes, the following section considers a number of arguments against judicial independence from a protectionist standpoint.

3.2 Protecting the Court from a political quagmire

An entirely different argument in favour of Security Council monopoly is based on the well-meaning intention to protect the court's judicial integrity, rather than a defence of the Council's prerogative, or a fear of inconsistencies and conflict. R67 Following this view, the argument has been presented that it is simply not 'good for the Court in the long run to be faced with the politically charged Crime of Aggression', because international criminal law is still in its infancy and slow to develop. Some negotiating States

⁸⁶⁶ See Article 8 bis, and paragraph 3 of the Elements section (n. 607) and the discussion below.

⁸⁶⁷ See the arguments presented above, particularly by Glennon (2010) (n. 8 and 812) and Stein (2005) (n. 811)

⁸⁶⁸ Heinsch (2010) (n. 376), 713, 742.

appeared to share this view. Thus, for instance, the post-Kampala statement by Norway

expressed the sentiment that aggression investigations could be 'extremely resource-

consuming', detracting from the exercise of jurisdiction over the other crimes in Article

5.869 Consequently, the Court ought to be protected from having to make independent

determinations of aggression.

Similarly, the view has been expressed that the Court is simply the wrong forum for

aggression scenarios, and should be saved from having to become engulfed in political

controversies between States.⁸⁷⁰ As Thomas Meron states:

Such an immersion would endanger the Court's judicial role and image. Imagine

the immense difficulties the ICC, as a court of individual criminal responsibility,

would face in dealing even with relatively simple matters of aggression. Is it

equipped to consider such matters as historical claims to territory, maritime

boundaries, legitimate self-defense under Article 51, or legitimate reprisals? Do

we want to expose the ICC to the inevitable accusations of politicization? And is

the competence of the ICC, in any event, not limited to jurisdiction over natural

persons? We must not turn the ICC into a political forum for discussing the

legality of use of force by states.⁸⁷¹

The intention to protect the Court from having to make political determinations and

consequently be subjected to accusations of politicisation may be well-meant, but is,

869 'Statement by Norway' (n. 861).

⁸⁷⁰ See Meron (2001) (n. 809), 4.

⁸⁷¹ *Ibid*.

211

arguably, overly protective for the following reasons. Firstly, the Court has now been tasked with making a determination of a state act of aggression in any case, as this is part of the *actus reus* established in Article 8 *bis* and the elements of the crime, and is therefore essential for an assessment of individual criminal liability.⁸⁷² Secondly, support for the ICC's independence can be derived by analogy from the ICJ's self-assessment of having 'separate but complementary functions' to those of the Security Council, and there is no reason why this approach should not apply to the ICC.⁸⁷³ It therefore provides as convincing a response to protectionist concerns, as it does to arguments for absolute political control of the Council over the ICC's jurisdiction.

Moreover, the ICJ's statement in *Nicaragua* provides authority that political implications are no reason for a court to shun a case.⁸⁷⁴ Admittedly, the ICJ's perceived capacity derives support from Article 36 of its own constitutive Statute,⁸⁷⁵ and no equivalent provision has been expressly included in the Rome Statute. Nevertheless, the ICC should be credited with 'jurisdiction to determine its own jurisdiction', as an essential requirement to exercise their function as a criminal court,⁸⁷⁶ deriving support from the ICTY's decision on a similar lacuna in favour of its own jurisdiction in the case of *Tadic*.⁸⁷⁷ There is, therefore, sufficient authority for crediting the ICC with the ability to rely on its own expertise and functionality. The Court ought to be seen as competent, both

⁸⁷² See n. 607 and 866.

⁸⁷³ Nicaragua (1984) (n. 827), at para 95.

⁸⁷⁴ *Ibid*, para 96. The original precedents of the Nuremberg and Tokyo tribunals had already set an early example by having to consider exceedingly complicated political issues, and they were also deemed competent to do so. See Chapter 1.

⁸⁷⁵ Article 36 (6) (n. 769) of the ICJ Statute (n. 35).

⁸⁷⁶ The argument for competence of the ICC has been developed particularly by Dov Jacobs (2010) (n. 556), 135-9. See also Weisbord (2008) (n. 176), 197-200.

⁸⁷⁷ *Tadic*, (1995 Decision on Jurisdiction), n. 838, and related discussion above. See also the *Reparations* case (1949) (n. 844).

out of necessity and principle, on account of being the organ which has been charged with the prosecution of the most serious international crimes regardless of their political nature.

A related and equally protectionist argument for exclusive Security Council determination power is based on the aim to discourage some States from gaining political advantages, or to prevent frivolous referrals to the Court. The delegation of Norway, in their post-Kampala statement, also expressed concern that States' accusations of aggression may be unfounded and politically motivated. The potential for abuse of the Crime of Aggression provisions. An additional source of consternation for Scheffer is that the optional and patchy jurisdiction of the Court, due to the membership and consent requirement in Article 15 *bis*, may lead to inequality and inequitability. Following this view, Security Council determination powers serve to circumvent the strict membership and consent requirements of Article 15 *bis*.

These arguments, however, rely on a view of the Security Council as an entirely impartial arbiter, unencumbered by political bias, and accredited with extensive determination powers, at the cost of independent decisions by the Court. Furthermore, wide Security Council powers over criminal court proceedings are being advocated in order to capture

⁸⁷⁸ See for instance, David Scheffer, 'The Complex Crime of Aggression' (2011), 43 Stud. Transnat'l Legal Pol'y 173, 178.

^{879 &#}x27;Statement by Norway' (n. 861).

⁸⁸⁰ Scheffer (2011) (n. 878).

⁸⁸¹ *Ibid*, 186-8.

⁸⁸² *Ibid.* For more on the argument that jurisdiction is actually expanded through Security Council involvement, see the discussion below.

non-member states or non-consenting members, assuming that such Security Council determinations will be straightforward and readily forthcoming. Member states' own referrals, on the other hand, are viewed with great scepticism as to their motives and merit.

These views, arguably, appear to be overly naive and excessively cynical at the same time, and none of the assumptions are necessarily inevitable. For example, the fact is ignored that the Security Council is often polarised and not immune to political selfinterest of powerful States. The veto power can and, as recent history has shown, is also likely to be used to prevent Security Council determinations and block resolutions, if directed against one of the P5 or one of their allies.⁸⁸³ Equally, the argument that all referrals submitted by States are frivolous or meritless, and brought solely for political reasons rather than seeking the prosecution of actual criminal aggressors, is overly simplistic. Thus, Article 15 bis has been equipped with a variety of safeguards to prevent abuse, particularly in view of the Security Council's deferral powers in Article 16.884 In addition, prior to an investigation, the Prosecutor is now required to ascertain whether the Security Council has made a determination, and is obliged to notify the Secretary-General of the UN of the situation.⁸⁸⁵ If no determination exists, the Prosecutor then has to delay the investigation for 6 months while he or she waits for the Security Council to consider the matter, and then still has to wait for authorization of proceedings by the Pre-Trial Division. 886 All of these provisions for delays and the requirement for reflection by various organs render the jurisdictional procedure in Article 15 bis more than sufficiently

⁸⁸³ See for instance the vetoed Security Council draft resolutions on the Ukraine and annexation of the Crimea (n. 799) and the situation in Syria (n. 802).

⁸⁸⁴ Article 16 of the Rome Statute (n. 739), providing for the power of the Council to halt an investigation or prosecution for a period of 12 months at a time.

⁸⁸⁵ Article 15 *bis* (6).

⁸⁸⁶ Article 15 bis (8).

equipped (in fact, arguably, excessively restricted) to provide a safeguard against the frivolous abuse by States. The added suggestion that the Pre-Trial Division should in all cases and in addition to the Security Council's involvement determine that an act of aggression has occurred, prior to authorising investigations, 887 is therefore unnecessary, and would complicate the procedure even more.

The following subsection examines, in more detail, an additional contention, which was already encountered in David Scheffer's above proposal. This argument for exclusive Security Council determination power suggests that such extensive involvement of the Council actually has the effect of expanding the Court's jurisdiction over non-members and non-consenting members, and is therefore to be seen in a positive light.

3.3 The argument that Security Council involvement expands the Court's reach

The above discussion considered and attempted to refute arguments for Security Council control over aggression proceedings, based on prerogative, fear of conflicting determinations and the wish to protect the Court from becoming entangled in political turmoil. An alternative view propounds that Security Council determinations are essential for the reason of expanding the Court's jurisdiction to situations in which it would

.

⁸⁸⁷ Scheffer (2011) (n. 878), 181.

normally have none. 888 This argument is based on the assumption that international and, in particular, treaty law makes it impossible for the Court to have broad jurisdiction over non-signatories. Consequently, according to Dapo Akande, the Security Council's involvement should be regarded in a positive light as expanding the Court's jurisdiction. Furthermore, the Court should be barred from exercising greater independence, as this would amount to an untenable judicial review of Security Council decisions. 890

These arguments, as defended by Professor Akande, operate from a completely different angle to the by now familiar arguments of Security Council prerogative and States' national interests, ⁸⁹¹ as well as protectionism of the integrity of the Court. ⁸⁹² They also represent a paradox. On the one hand, the argument defends active restriction of the Court's independent jurisdiction. ⁸⁹³ At the same time, it is being reasoned that the Court's jurisdiction would be expanded through this effective inhibition of independent jurisdiction and the insistence on Security Council determinations. ⁸⁹⁴ The exercise of Security Council power is, following Professor Akande's view, also necessary due to the requirement for individual state consent to jurisdiction. This individual state consent manifests itself threefold: firstly through membership, secondly through ratification of the aggression amendments, and thirdly through non-application of the additional choice to opt out. ⁸⁹⁵ This argument is based on the strict application of the VCLT, providing that

⁸⁸⁸ Dapo Akande, 'Prosecuting Aggression: The Consent Problem and the Role of the Security Council', (2011) Oxford Legal Studies Research Paper No. 10/2011, 37.

⁸⁸⁹ *Ibid*, 37.

⁸⁹⁰ *Ibid*.

⁸⁹¹ See the arguments of Glennon (2010) (8 and 803 above) and Stein (2005) (n. 811).

⁸⁹² See Meron (2001) (n. 809), and Scheffer (2011) (n. 878).

⁸⁹³ Akande (2011) (n. 888), 37-8.

⁸⁹⁴ *Ibid*.

⁸⁹⁵ See Chapter 5.

treaty provisions can only bind their signatories, excluding third parties. ⁸⁹⁶ Consent to be bound by a treaty can be expressed by signature, exchange of the treaty, ratification, acceptance, approval or accession. ⁸⁹⁷ The crime of aggression provisions, however, take the consent requirement one step further requiring even signatories to the Rome Statute to have individually consented to jurisdiction by accepting the amendments and failing to opt out. ⁸⁹⁸ In addition to this unusual extra consent layer, and differing from the other crimes under the Statute, jurisdiction only arises, where *both* victim and aggressor have consented to jurisdiction, by virtue of the choice to apply Article 121 (5) in its strictest interpretation. ⁸⁹⁹

Security Council determinations and referrals are perceived to be the only way to circumvent this strictly interpreted membership and consent requirement. On sequently, extensive control by the Council, Akande argues, is not only essential, it is also advantageous, since it extends the Court's jurisdiction to non-consenting parties. As a result of this potential application of involuntary jurisdiction to non-consenting States through a Security Council referral, Professor Akande observes:

There has not been a warm embrace by States who at the very least pay lip service to, and at most worship at the altar of independence, equality and consent. An amended ICC Statute allowing adjudication of aggression by non-consenting

⁸⁹⁶ VCLT (n. 32), Article 34.

⁸⁹⁷ *Ibid*, Article 11.

⁸⁹⁸ Article 15 *bis* (4) Rome Statute (n. 1).

⁸⁹⁹ Article 121 (5) Rome Statute, for an in-depth discussion, see Chapter 5.

⁹⁰⁰ Article 15 bis (4) Rome Statute.

⁹⁰¹ Akande (2011) (n. 888), 40.

States [through Security Council referral] might well be the first explicit embrace by States (some States) of non-voluntarist principles in a treaty. 902

The accuracy of this logic, insisting on strict application of the VCLT, appears persuasive, by suggesting Security Council determinations as a solution to the otherwise patchy jurisdiction of the Court. It is undeniable that Security Council involvement may occasionally be effective in overcoming the problem of necessary state consent to prosecutions, thus potentially enabling jurisdiction over an unpopular non-consenting or non-member state.

However, the argument of linking extensive Security Council control over criminal aggression proceedings to the positive result of wider jurisdiction through the possibility of extension to non-consenting parties, may be overly optimistic. In reality such extensive control is unlikely to ease the restrictiveness of the Court's jurisdiction. The past history of Security Council determinations of aggression has shown that they have almost without exception addressed 'specific geographic controversies rather than broad global threats'. States targeted by Security Council condemnations generally enjoy little support by, and are not usually allied to, one of the P5 members. As Andrew Guzman observes, '[i]t is one thing to establish a peacekeeping force in Cote d'Ivoire. It is

⁹⁰² *Ibid.* For evidence of a lack of a 'warm embrace', see the statement of positions post-Kampala, in particular by the P5 (n. 722-3). For a deeper discussion of Professor Akande's view of strict application of treaty law generally, see Dapo Akande and Antonios Tzanakopoulos, 'Treaty Law and ICC Jurisdiction Over the Crime of Aggression' (2018), 29 EJIL, 939.

⁹⁰³ Andrea Bianchi, 'Security Council's Anti-Terror Resolutions and their Implementation by Member States' (2006) 4 J Int'l Crim. Just. 1044, at 1046-47. For an account of Security Council (non-) determinations of acts of aggression, see Chapter 1.

⁹⁰⁵ Citing Security Council Resolution 1528, S/RES/1528 (2004), which authorised ECOWAS and French deployments in Cote d'Ivoire.

altogether more difficult to get a resolution attempting to resolve global security threats that implicate important interests for the P 5. '906

As a result, Security Council determinations of aggression against a non-member state or non-consenting state are only likely to be made, and therefore able to overcome the consent problem, where they have no implications for the interests of the P5. The considerable disadvantage, arguably outweighing the benefit of comprehensive Security Council involvement, is that it is potentially capable of preventing jurisdiction of the Court on far more occasions than it may enable it, because of the Council's very own internal voting structure and the ability to veto a determination.

In reality, prosecutions under Article 15 ter are unlikely to take place if there is the slightest objection by a P5 member, due to protection of their own national interests or those of an ally, to a referral to the Court by the Security Council. Consequently, jurisdiction is potentially restricted, via the back door of the requirement for state consent, where one of the P5 fails to grant what effectively amounts to such individual consent, even where the majority of the Security Council would deem an aggression situation to be of sufficient concern to refer it to the Court.

Prosecutions under Article 15 bis are also subject to the substantial powers of the Security Council, even where a State is a member. This is because of the requirement for the Prosecutor to first ascertain whether the Council has made a determination, 907 followed by a considerable time delay caused by leaving the matter with the Council, 908 and the

⁹⁰⁶ Andrew Guzman, (2011) (n. 25), 43.

⁹⁰⁷ Article 15 bis (6) of the Rome Statute.

⁹⁰⁸ Article 15 *bis* (8).

Council's subsequent ability to defer (or permanently halt) any investigation. 909 Additionally, these prosecutions under Article 15 *bis* require acceptance through the member state's choice not to opt out. 910 It is, however, unlikely that a State which has intentionally perpetrated an act of aggression consents to the prosecution of its officials, unless there has been a change of regime, or policy, or both. These existing loopholes are, therefore, prone to be taken advantage of by States pursuing acts of aggression.

Impunity as a result of optional state consent to jurisdiction is not conducive to overall deterrence. For these reasons, this thesis raises the question whether consent to prosecution for aggression is an appropriate concept at all, regardless of whether this manifests itself through a requirement of Security Council approval, reliant on consent by the powerful, or – as discussed in the next chapter - the multiple consent sought from member states.

4. An evaluation of Security Council involvement

The above discussion considers and aims to refute a variety of arguments defending Security Council monopoly or overall control. These have ranged widely from a claim of exclusive prerogative to deal with any aggression scenarios to the preservation of the existing power structure and sovereign interests, and from the wish to protect the Court

⁹⁰⁹ Under Article 16 (n. 739).

⁹¹⁰ Article 15 *bis* (4).

from accusations of politicisation and States' abuse to the requirement for strict observation of treaty law and consent at international law.

Counterarguments have been presented for wide and independent determination powers by the Court. These have ranged from an inherent assumption of competence, akin to that of the ICJ, provided for in Article 36 of its constitutive statute, ⁹¹¹ and confirmed in a number of cases, ⁹¹² to the factual need to arrive at these determinations as an integral element of the crime. ⁹¹³ Such determinations by the Court are also necessary to satisfy legality requirements, and in order to achieve this, the Court requires flexibility and independent determination powers. If the judiciary are expected to fulfil their purpose, their expertise, competence and impartiality have to be trusted. The requirement for legality is also better served by judicial case by case development than *ex post facto* determinations by a political organ, ⁹¹⁴ whose history with regards to such determinations is, at best, unreliable. ⁹¹⁵

Security Council paralysis results in a lack of a balanced political solution as well as unjust effective impunity for perpetrators of aggression. This leads to scenarios, where some States with no allies amongst the P5 members of the Council might be subject to determinations and referrals, whilst other aggressors might enjoy effective immunity from prosecution simply because of their relationship with one of the powers capable of

.

⁹¹¹ For Article 36 ICJ Statute, see n. 769 and related discussion.

⁹¹² See for instance *Nicaragua* (1984) (n. 827), *Armed Activities* case (2005), n. 37. See also *Tadic*, 1995 Decision on Jurisdiction, (n. 838), and the much earlier *Reparations* case (1949) (n. 844).

⁹¹³ See the discussion on pp. 201-3, and the fact that the ICC is concerned with individual criminal liability rather than making determinations for political purposes.

⁹¹⁴ For strong support for this contention, see the SWGCA's own stance, despite their need to placate the majority view (section 2.2).

⁹¹⁵ See the discussion in Chapter 1.

vetoing a Security Council resolution.⁹¹⁶ Inertia within the Security Council and failure to agree to refer a case to the Court should not be allowed to block independent investigation and possible prosecution, particularly since the Security Council, in any case, retains the power to defer a particular investigation potentially forever under Article 16.⁹¹⁷

A greater focus on independent determinations and prosecutions of acts of aggression would aid deterrence and prevent impunity, particularly if the Court and the Security Council were to cooperate as operational partners. Thus, Benjamin Ferencz, for instance, envisaged such cooperation through the provision of arrest warrants, enforcement of judgments and generally taking note of judicial decisions. Granting ICC judges independent determination powers could indeed be seen to be a positive development, creating the envisaged chance of nuanced case by case solutions and the development of guiding principles, whilst avoiding politicization of decisions. Conversely, if an independent Security Council determination of aggression exists, the Court would have to take care not to inadvertently reverse the burden of proof due to giving this too much weight, bearing in mind that the individual *actus reus* would still have to be proven in each case. The rights of a defendant in a criminal trial must be observed, and this requirement counters an argument by critics of the Court that findings by other organs, specifically the Security Council, should be prejudicial to the Court's own findings in order to create certainty and uniformity. Even well-meaning attempts to preserve the

⁹¹⁶ See for instance enforced Security Council inactivity in the face of blatant aggression, in the Ukraine (n. 799), and serious human rights violations in Syria (n. 802).

⁹¹⁷ See n. 739.

⁹¹⁸ See Ferencz (2008) (n. 831) and Cassese (2007) (n. 614).

⁹¹⁹ See Articles 66 and 67 of the Rome Statute (n. 1).

⁹²⁰ See for instance Glennon (2010) (n. 8 and 803), and Stein (2005) (n. 811), and related discussion. For a provision against prejudicial findings see Articles 15 *bis* (9) and *ter* (4).

Court's integrity as a non-political organ as grounds for insistence on external determinations, ⁹²¹ are strongly refutable, since the Court is well-equipped to venture into the political arena on account of its independence and the stringent safeguards placed on its prosecutions of aggression. ⁹²²

The arguments presented also counter those of commentators, who view independent determinations by the ICC as an untenable review of Security Council decisions. 923 Instead, this thesis submits that the Court should be entitled to discretionary judicial interpretations. The Court should be allowed to conduct independent investigations and prosecutions, even if these are deemed by some to constitute an unwarranted intervention in areas of international politics traditionally regarded as the sole preserve of the Security Council. These wider judicial interpretations and independent determinations are also necessary, in light of a definition which may be too narrow to apply to modern versions of aggression, on account of an extremely limited list of acts, and application to state actors and military and political leaders only. 924 As touched upon in the Chapters on the definition, warfare evolves constantly with modern developments and changing geopolitics. In order for the new Crime of Aggression to become and remain relevant, a certain amount of interpretation in judicial determinations is essential and ought to be permissible, as long as the analogy is close enough and the underlying prohibited behaviour is still recognisable as such. The existing high threshold of the crime would help to prevent frivolous abuse of aggression proceedings. 925

⁹²¹ For the protectionist views in favour of Security Council monopoly, see particularly Meron (2001) (n. 809), Scheffer (2011) (n. 878) and section 3.2 above.

⁹²² For instance the limitation to the worst international crimes only, see Preamble (n. 85).

⁹²³ See for instance Dapo Akande (2011) (n. 888) and arguments in section 3.3.

⁹²⁴ See the discussion in Chapters 2 and 3.

⁹²⁵ *Ibid*.

The arguments for independent determination powers are also highly persuasive in consideration of the overarching purpose of the Rome Statute itself, the endeavour to outlaw the worst international crimes, and eradicate the 'supreme international crime' of aggression. However, in making such determinations, the Court would have to be careful not to offend Article 22 (2) of the Rome Statute, ensuring that the analogy applied was sufficiently close not to offend general legality principles.

The debate is certain to continue, however it is difficult not to agree with the great Benjamin Ferencz that '[f]ailure to allow the ICC to punish aggression is a repudiation of Nuremberg and a step backward in the development of international criminal law... Prohibiting the ICC from exercising its jurisdiction is an indefinite guarantee of continuing immunity for future aggressors.'928 Granting effective and politically unencumbered jurisdiction over aggression, independent of extensive Security Council involvement, and potentially subject to the concept of universality of such jurisdiction — to be explored in the next chapter - would, on the other hand, prevent 'a renewed licence to wage illegal wars with impunity', representing a 'triumph of Reason over Power.'929

⁹²⁶ See Robert Jackson's opening statement (n. 135) at the Nuremberg Trials (n. 3) and the Final Judgment of the Tribunal (n. 235), at page 16.

⁹²⁷ Article 22 (2) provides for strict construction of the definition of a crime, and, in case of ambiguity, for interpretation in favour of the defendant (n. 477).

⁹²⁸ Ferencz (2009) (n. 4), 290.

⁹²⁹ *Ibid*.

CHAPTER 5: THE STRICT REQUIREMENTS OF MULTIPLE CONSENT AND MEMBERSHIP OF STATES

1. Introduction

The previous chapter introduced the special jurisdictional conditions imposed by Articles 15 *bis* and *ter*. The chapter examined in particular the role of the Security Council in aggression proceedings and the attempted justifications for this extensive involvement. The discussion considered the fact that the Court's dependency on the Security Council raises problems of legitimacy in criminal aggression proceedings, on account of the possibility of retrospective determinations of acts of aggression by an external organ. Furthermore, the independence and credibility of the Court's decisions may be compromised by its reliance on the Council's approval. Onversely, the Court's hands may be tied by the decision of the Council to postpone or halt an investigation under Article 16.932 An additional concern of great importance is that the Council's decisions are reliant on consensus of its powerful permanent members. Despite the fact that the Kampala compromise allowed for alternative referrals to the Court through state referral

⁹³⁰ See section 3.1 of Chapter 4 and the discussion on pp. 205-7.

⁹³¹ *Ibid*.

⁹³² See n. 739.

or *proprio motu* investigations, the Court is effectively dependent on the approval of and consent to prosecution by the Security Council and, in particular, its powerful contingent with permanent seats.⁹³³

This chapter focuses on another facet of the requirement of consent to jurisdiction of the Court, arising specifically from Article 15 *bis*. ⁹³⁴ This article, at least *prima facie*, provides an alternative to absolute Security Council monopoly over the crime of aggression, allowing for limited jurisdiction of the Court in situations where the Security Council cannot agree on a referral or chooses not to actively authorise investigation and prosecution. Nevertheless, Article 15 *bis* contains a number of severe limitations on the jurisdiction of the Court.

Most strikingly, Article 15 *bis* (4) insists on fulfilment of very stringent consent conditions, by allowing for the possibility of a declaration of non-acceptance of jurisdiction by a State. As introduced in the previous Chapter, the paragraph provides:

The Court may, in accordance with article 12, exercise jurisdiction over a crime of aggression, arising from an act of aggression committed by a State Party, unless that State Party has previously declared that it does not accept such jurisdiction by lodging a declaration with the Registrar. The withdrawal of such a declaration

⁹³³ See Article 27 (3) of the UN Charter (n. 210).

⁹³⁴ And its combination with Article 121 (5), see section 2.2 below.

may be effected at any time, and shall be considered by the State Party within three years. 935

Thus, a State can opt out of jurisdiction, notwithstanding the fact that they are signatories, and even if they have previously ratified the aggression amendments. ⁹³⁶ This amplified consent requirement in state referral or *proprio motu* investigations constitutes a fundamental difference to the conditions which have been established for the other core crimes in Article 5 (1). ⁹³⁷ Thus Article 12 (1) of the Rome Statute establishes the jurisdictional provisions for the crimes under the Rome Statute generally, stating that '[a] State which becomes a Party to this Statute thereby accepts the jurisdiction of the Court with respect to the crimes referred to in article 5.'938 The opt-out provision in Article 15 *bis* (4) is therefore exceptional within the Rome Statute, and entails a great restriction on the Court's jurisdiction. Still more perplexing, the discussion will show that this unusual consent requirement has been additionally reinforced by the choice of the 2017 ASP to apply the narrowest of interpretations of entry into force conditions contained in Article 121 (5).⁹³⁹

A further limitation on the Court's jurisdiction has been established by Article 15 *bis* (5). The article provides:

⁹³⁵ Article 15 bis (4) Rome Statute.

⁹³⁶ Although the actual wording of Article 15 *bis* (4) does not require prior ratification, see the discussion below in section 2.1.

⁹³⁷ Article 5 (1) Rome Statute lists genocide, crimes against humanity, war crimes, as well as aggression (n. 9).

⁹³⁸ Article 12 (1) of the Rome Statute (n. 1).

⁹³⁹ For the text of Article 121 (5) see p. 231 and for discussion see section 2.2 below. For a discussion of the 2017 ASP's decision, see sections 2.5 and 2.6 below.

In respect of a State that is not a party to this Statute, the Court shall not exercise its jurisdiction over the crime of aggression when committed by that State's nationals or on its territory. 940

This provision excludes non-members in any capacity, even where they are the victim of aggression by a ratifying signatory of the Rome Statute. It is also exceptional compared with the other core crimes, jurisdiction over which may exist where *either* the perpetrator *or* the victim are signatories to the Court.⁹⁴¹

As a result of the combination of these conditions, jurisdiction of the Court under Article 15 *bis* only arises where both victim and aggressor a) are members, b) have individually ratified the amendments, and c) have not taken advantage of the extraordinary possibility to opt out of jurisdiction. The following discussion examines the effect of these exceptional provisions, requiring multiple consent in addition to ratification by all parties.

⁹⁴⁰ Article 15 *bis* (5) of the Rome Statute 1998 (n. 1).

⁹⁴¹ See Article 12 (2), which stipulates that '[i]n the case of article 13, paragraph (a) [i.e. state referral] or (c) [i.e. *proprio motu*], the Court may exercise its jurisdiction if *one or more* of the following States are Parties to this Statute or have accepted the jurisdiction of the Court in accordance with paragraph 3:

 ⁽a) The State on the territory of which the conduct in question occurred or, if the crime was committed on board a vessel or aircraft, the State of registration of that vessel or aircraft;
 (b) The State of which the person accused of the crime is a national.

The existing jurisdictional conditions for the other core crimes of Article 5 (1) are expressly preserved by Article 15 *bis* (10), reinforcing the different treatment of the crime of aggression. Above emphasis added.

2. Combining Article 15 bis (4) with Article 121 (5) – The multiple consent requirement

2.1 Article 15 *bis* (4)

Article 15 *bis* (4) of the Kampala Amendment effectively establishes a pre-condition of consent of even a member state, which is given by failing to take advantage of the right to opt out of the acceptance of jurisdiction. Only if the member state impliedly grants this consent by not opting out, can one of its political or military leaders be investigated or prosecuted for the crime of aggression, in situations of state referral or on the initiative of the Prosecutor.⁹⁴²

This provision represents a fundamental difference to other jurisdictional conditions and is exceptional within the Rome Statute. Thus Article 12, which lays down the jurisdictional preconditions for the four core crimes under the Statute, 943 simply provides for general acceptance of member states on signature, regardless of the source of the referral to the Court. No mention is made in this article of the possibility of non-acceptance of jurisdiction by a member state, and none of the other crimes in Article 5 require a member's individual acceptance of (or consent to) jurisdiction. 944 Article 15 *bis* (4), consequently, creates an exception for the Crime of Aggression in *proprio motu* or

⁹⁴² Unless there is a Security Council referral, in which case the Court's jurisdiction arises automatically (Article 15 *ter*), see Chapter 4.

⁹⁴³ Article 12 (1) p. 227.

⁹⁴⁴ *Ibid*.

state referral situations, requiring consent in addition to membership. ⁹⁴⁵ Importantly for this discussion, the wording of Article 15 *bis* (4) suggests, however, that denial of consent demands physical action of a State, by actively opting out of jurisdiction. Otherwise consent may be presumed to exist. ⁹⁴⁶

In addition to the inclusion of this exceptional consent requirement in Article 15 *bis* (4), jurisdiction of the Court over the crime of aggression has been limited even further. Thus, the restrictions are exacerbated by the final decision of the Kampala negotiators to combine Article 15 *bis* (4) with Article 121 (5).⁹⁴⁷

2.2. The choice to apply Article 121 (5) to the crime of aggression

Article 121 generally considers amendment procedures, and was nominated in the now extinct Article 5 (2) as the procedure to be followed once a definition of the crime of aggression had been achieved.⁹⁴⁸ Article 5(2) did however not state which paragraph of Article 121 ought to apply, and the negotiators therefore had a choice, settling on Article 121 (5) in the implementing Resolution 6.⁹⁴⁹ Article 121 (5) provides:

⁹⁴⁵ For a critique, see Heinsch (2010) (n. 376), 743, stating that the opt-out provisions merely add to the uncertainty surrounding aggression.

⁹⁴⁶ Stefan Barriga and Leena Grover (2011) (n. 724 at 517, 526, 532) and Barriga, 'The Scope of ICC Jurisdiction Over The Crime of Aggression: A Different Perspective', (EJIL:Talk! 29 March 2017), https://www.ejiltalk.org/the-scope-of-icc-jurisdiction-over-the-crime-of-aggression-a-different-perspective/, accessed 20 October 2019. See also Nikolas Stuerchler, 'The Activation of the Crime of Aggression in Perspective/, (EJIL:Talk! 26 January 2018), https://www.ejiltalk.org/the-activation-of-the-crime-of-aggression-in-perspective/, and Noah Weisbord's presentation before the General Assembly, 'Activation of the Crime of Aggression', 2 June 2017, available at https://sls.sbg.ac.at/data/Weisbord.pdf, both accessed 20 October 2019.

⁹⁴⁷ See section 2.4 and 2.5 below.

⁹⁴⁸ For Article 5 (2) see n. 10.

⁹⁴⁹ Resolution 6 (n. 2).

Any amendment to articles 5, 6, 7 and 8 of this Statute shall enter into force for those States Parties which have accepted the amendment one year after the deposit of their instrument of ratification or acceptance. In respect of a State Party which has not accepted the amendment, the Court shall not exercise its jurisdiction regarding a crime covered by the amendment when committed by that State Party's nationals or on its territory. 950

The decision made at the Review Conference in Kampala to apply Article 121 (5) arguably does not sit easily. The original Article 5 (2) speaks of 'adoption' of the aggression amendments, whereas Article 121 (5) considers 'entry into force' for individual member states who have ratified the amendments. This particular anomaly was a substantial point of criticism by the Japanese delegation, as expressed in their statement after the Kampala Conference:

Article 5, paragraph 2, is invoked as the basis with respect to 'amendment', whereas article 121, paragraph 5, is invoked as the basis with respect to 'entry into force'. This is a typical 'cherry picking' from the relevant provisions related to the amendment, that is, in Japan's view, very difficult to justify.... The upshot is a highly accentuated complication in the legal relation after the amendment

⁹⁵⁰ Article 121 (5) of the Rome Statute (n. 1).

⁹⁵¹ See sections 2.2 and 2.3 below.

⁹⁵² See n. 10.

⁹⁵³ See above (n. 950).

between States Parties, as well as the relation between States Parties and non-States Parties, which is extremely unclear and hard to understand.⁹⁵⁴

In addition to the contradiction between the term 'adoption', derived from Article 5 (2), and 'entry into force', used in Article 121 (5), other arguments against the application of the provision can be raised. Thus, aggression already forms part of the core crimes under the Rome Statute, albeit undefined, by virtue of its inclusion in Article 5.955 It is therefore possible to argue that it was not even necessary to view the aggression provisions as an amendment *per se*.956 The aggression provisions also do not affect the existing Articles 5, 6, 7 and 8, whereas Article 121 (5) refers to amendments of these articles.957 Following this view, the original reference to Article 121 in the original Article 5 (2) itself was unnecessary, 958 casting doubt on the logic applied. Furthermore, even if the crime of aggression could be considered as amending Article 5, it by far surpasses a simple amendment of that article, dealing with jurisdictional issues as well,959 and, in fact creating its own special jurisdictional procedure in Articles 15 *bis* and *ter*.960

In consideration of these arguments, the Kampala decision to apply paragraph 5 of Article 121 appears contrived and illogical, and would have been avoidable. Attempting to

⁹⁵⁴ See Barriga and Kress (eds) (2012), *Travaux Preparatoires*, (n. 317), 'Statement by Japan', at p. 812.

⁹⁵⁵ These points have been made, particularly, by Marko Milanovic (2012) (n. 479), 165, 179.

⁹⁵⁶ *Ibid*.

⁹⁵⁷ *Ibid*.

⁹⁵⁸ See Article 5 (2) of the Rome Statute (n. 10).

⁹⁵⁹ Milanovic (2012) (n. 479). See also Barriga (2017) (n. 946). For a different interpretation, see Dapo Akande, 'What Exactly was Agreed in Kampala on the Crime of Aggression?' 21 June 2010, available at http://www.ejiltalk.org/what-exactly-was-agreed-in-kampala-on-the-crime-of-aggression/, accessed 20 October 2019.

⁹⁶⁰ See Astrid Reisinger-Coracini, 'The Kampala Amendments on the Crime of Aggression Before Activation' (OpinioJuris, 29 September 2017 and 2 October 2017) http://opiniojuris.org/2017/09/29/the-kampala-amendments-on-the-crime-of-aggression-before-activation-evaluating-the-legal-freamework-of-a-political-compromise/ accessed 19 October 2019.

explain the contradiction, Roger Clark has pointed out that Articles 5 and 121 were drawn up by separate drafting teams at different times in the lead-up to the Rome Conference, and the ambiguities resulting from the insistence on applying Article 121 can be seen to be a natural consequence. Avoiding this entire controversy, it should, instead, have been possible to seek simple application of jurisdiction over the Crime of Aggression to member states, in line with the other crimes under the Rome Statute according to Article 12.962

Notwithstanding all of the above arguments against the choice of Article 121 (5), this article was nominated as the procedure to be followed at the Review Conference, and only another amendment could now change that. Considering the extent of disagreement at the time, it is, perhaps, a fair assessment that this prospect is highly unlikely. Still, the question remains: Why was there such strong insistence on the application of Article 121 (5) rather than any other paragraph or even simple adoption?

2.3 Why insist on Article 121 (5) instead of (3) or (4)?

The reasons for the controversial decision to apply Article 121 (5) become clearer when one looks at the result achieved, as well as the fierce defence of a strict consent

⁹⁶¹ See Roger S. Clark, 'Ambiguities in Articles 5 (2), 121 and 123 of the Rome Statute', (2009) 41 Case W. Res. J. Int'l L., 413, 421-5.

⁹⁶² Milanovic (2012) (n. 479), and Astrid Reisinger-Coracini, 'The International Criminal Court's Exercise of Jurisdiction Over the Crime of Aggression – at Last...in Reach...Over Some' (2010) 2 GoJIL, 745, 763-9. For a different view that 'the text and spirit of Article 121 suggests that it is paragraph 5 that ought to apply', see Dapo Akande, 'Prosecuting Aggression' (2011) (n. 888), 11, dismissing the argument that the Crime of Aggression amends articles other than Articles 5-8 of the Rome Statute.

requirement, which continued to be insisted on by a number of powerful States. ⁹⁶³ The direct consequence of a combination of Article 15 *bis* (4) with the whole of Article 121 (5) is the creation of an amplified consent requirement, which has to be fulfilled before the Court can exercise its jurisdiction over the crime of aggression. ⁹⁶⁴ This is as unusual as it is restrictive. Thus, a State is now considered to grant consent to prosecutions only if they a) become or remain a signatory, ⁹⁶⁵ b) additionally ratify the aggression amendments, ⁹⁶⁶ and c) fail to opt out of jurisdiction in accordance with Article 15 *bis* (4). ⁹⁶⁷

Revisiting the situation the SWGCA found itself in prior to Kampala also sheds some light on the reasoning behind the creation of this reinforced consent criterion and application of Article 121 (5). The previous discussion demonstrated that States continued to remain sceptic of the Court's jurisdiction prior to the Rome Conference, on account of a perception of Security Council monopoly over aggression. This scepticism did not subside during the subsequent negotiations. Even during the preparation of the final proposals for the Kampala Review Conference and beyond, the SWGCA were faced with continuing resistance by powerful States, particularly the P5, to anything short of Security Council monopoly. If such an absolute monopoly had to be diluted at all, by allowing state referrals or *proprio motu* investigations subject to scrutiny by the Council, an additional filter of individual consent to jurisdiction by member states was sought to

⁹⁶³ See for instance the 2010 Explanations of Positions, particularly of the P5 members (n. 722-3).

⁹⁶⁴ Unless there is a Security Council referral (Article 15 ter), rendering consent irrelevant.

⁹⁶⁵ Article 15 *bis* (5) of the Rome Statute (n. 940).

⁹⁶⁶ Applying Article 121 (5) as a whole (n. 950).

⁹⁶⁷ Article 15 *bis* (4) (n. 935).

⁹⁶⁸ See Cassese (2013) (n. 12), 138. See also Chapter 1.

⁹⁶⁹ See the statements by the P5 (n. 722-3) and Chapter 4, section 2.

be applied.⁹⁷⁰ Consequently, after developing a compromise between Security Council referrals and alternative initiations of investigation through either a member state's referral or the Prosecutor's own powers, the SWGCA focused on this additional consent issue, and which particular sub-paragraph of Article 121 was to apply to the exercise of jurisdiction of the Court. Several possibilities arose with different levels of consent required.

The first of the alternatives the SWGCA were faced with, the so-called 'Adoption Model', was to simply apply Article 121 (3) of the Rome Statute, in line with the arguments presented above. ⁹⁷¹ Thus, Article 121 (3) provides:

The adoption of an amendment at a meeting of the Assembly of States Parties or at a Review Conference on which consensus cannot be reached shall require a two-thirds majority of States Parties.'972

Once a two-thirds majority had been achieved at a meeting of the ASP or a Review Conference, the aggression amendments would, under this model, have come into force. Individual ratification of the aggression provisions would not have been required. Despite the logic behind it, such a simple adoption procedure would, however, have ignored States' insistence on consent through individual ratification in cases where a referral originated from a source other than the Security Council. ⁹⁷³ The powerful States, in

⁹⁷⁰ Barriga and Grover (2011) (n. 724), 527-9, 532, and Reisinger-Coracini (2010) (n. 962), 763-9.

⁹⁷¹ Milanovic (2012) (n. 479), 179, and Reisinger-Coracini (2010) (n. 962), 764.

⁹⁷² Article 121 (3)

⁹⁷³ See the explanation of states Positions in the *Travaux Preparatoires* (n. 317) from p. 810, particularly the P5 (n. 722-3).

particular, felt that if complete Security Council monopoly over aggression had to be

given up, and the Court was to be allowed some form of independent jurisdiction, at the

very least individual state consent had to be given.⁹⁷⁴ In light of this resistance to

independent jurisdiction, the potential possibility of a simple adoption might have been

perceived as 'an astonishing consequence, given the political importance of the matter,'975

and the unwavering insistence on additional consent. Consequently, the SWGCA felt that

following the 'Adoption Model' might have endangered the entire negotiations. 976

The second possibility was to apply Article 121 (4) to the aggression provisions. This

would equate 'adoption' with 'entry into force'. 977 Article 121 (4) provides:

Except as provided in paragraph 5, an amendment shall enter into force for all

States Parties one year after instruments of ratification or acceptance have been

deposited with the Secretary-General of the United Nations by seven-eighths of

them.'978

Under Article 121 (4), the provisions would enter into force for all members one year

after 7/8 of them had ratified them. This model could, eventually, have achieved

jurisdiction over all member states, in line with the aim of ending impunity as expressed

974 *Ibid*, see also Kress and von Holtzendorff (n. 5), 1196.

⁹⁷⁵ *Ibid*

976 **Ibi**d

⁹⁷⁷ Article 121 (4) Rome Statute (n. 1).

⁹⁷⁸ *Ibid*

236

in the Preamble of the Rome Statute. ⁹⁷⁹ Nonetheless, similar to Article 121 (3), it also did not provide for a requirement of individual consent of every single State before general jurisdiction over members arose. Furthermore, the SWGCA was faced with the argument that applying Article 121 (4) would require downplaying the aggression provisions as a 'mere filling in' of the already existing Article 5 rather than as an actual amendment. ⁹⁸⁰ Consequently, the SWGCA felt that applying Article 121 (4) was also likely to be met with resistance by member states who wished to retain the choice of individual acceptance of jurisdiction. ⁹⁸¹ An additional consideration may have been that the threshold of achieving a 7/8 ratifications of ICC member states was likely to delay jurisdiction for a very long time. This particular point is admittedly merely an assumption, but is based on the fact that even 10 years after Kampala only 39 States out of 123 ICC members have ratified the amendments. ⁹⁸²

From this examination, it is apparent that neither of these two options would have been acceptable to States insisting on consent. Consequently, the SWGCA decided to apply Article 121 (5) to the crime of aggression, effectively adding a third layer of consent. Jurisdiction over the crime of aggression was now subject to insistence on membership, additional ratification, and failing to take advantage of the opting out provision in Article 15 *bis* (4).

⁹⁷⁹ See the Preamble to the Rome Statute (n. 85), '[a]ffirming that the most serious crimes of concern to the international community as a whole must not go unpunished... and [d]etermined to put an end to impunity for the perpetrators of these crimes and thus to contribute to the prevention of such crimes.'

⁹⁸⁰ See Kress and von Holtzendorff (n. 5), 1196. For a view that this interpretation would not have been unreasonable, despite States' resistance, see Barriga (2017) (n. 946), Milanovic (2012), (n. 479), 179.

⁹⁸¹ See for instance the 2009 *Princeton Report*, 'Informal Intersessional Meeting on the Crime of Aggression', hosted by the Liechtenstein Institute on Self-Determination, Woodrow Wilson School, at the Princeton Club, New York, from 8-10 June 2009, ICC-ASP/8/INF.2, (hereinafter the 2009 *Princeton Report*) para 38, (not published in the *ASP Official Records*), p. 703 of the *Travaux Preparatoires* (n. 317). ⁹⁸² As of 21 Oct 2019, see www.crimeofaggression.info, accessed 24th August 2020.

Article 121 (5) could, however, be interpreted in two different ways, and therefore gave rise to an additional problem for the SWGCA to consider in their deliberations. Was just one of the States concerned, most likely the victim, required to have ratified the amendment in line with the normal procedure under the Rome Statute, 984 or should each State, i.e. victim and aggressor, be required to do so? These two different interpretations of Article 121 (5) have become known as the 'negative' and 'positive' understanding, depending on whether both sentences of Article 121 (5) ought to be read into the crime of aggression, or merely the first. The difference is more than semantics and, as the following discussion shows, is of the greatest significance.

2.4 Consent by whom exactly? – Two different interpretations of Article 121 (5)

Having settled on Article 121 (5) as the procedure to be followed, and thus acquiescing to the insistence on additional consent of member states to jurisdiction, 986 the SWGCA failed to resolve the question of a 'positive' or 'negative' understanding at the time. This led to considerable debate post-Kampala on the appropriate reading of the article in combination with the new Article 15 *bis* (4). 987

⁹⁸³ See item No. 9, paragraph 25 of the 2010 WGCA Report, submitted in preparation for the Kampala Review Conference. Report of the Working Group on the Crime of Aggression, RC/5 in Review Conference Official Records, RC/11, Part II, Annex III, 45.

⁹⁸⁴ Under Article 12 (2) (n. 941), providing for jurisdiction based on *either* nationality *or* territory, see fn. 11.

⁹⁸⁵ For the two sentences of Article 121 (5) see p. 231.

⁹⁸⁶ In addition to the consent expressed through membership itself, and failure to take advantage of the optout provision of Article 15 *bis* (4).

⁹⁸⁷ For the two polarised stances, see, *infra*, n. 989 and 993 respectively, and related discussion.

Under the 'negative' interpretation, Article 121 (5) would be read strictly as a whole, precluding jurisdiction of the Court '[i]n respect of [any] State Party which has not accepted the amendment... when committed by that State Party's nationals or on its territory.'988 Consequently, this negative understanding insisted on individual acceptance by both aggressor *and* victim. No jurisdiction would arise if only one of the member states in question, most likely the victim of an act of aggression, were to ratify the aggression amendment.⁹⁸⁹ This interpretation clashes, however, with the provisions in Article 12 (2), which allows for the Court's adjudication where *either* the perpetrator or the victim have accepted jurisdiction.⁹⁹⁰ The conflict between Article 12 (2) and the second sentence of Article 121 (5) has been conceded as problematic even by unwavering supporters of the negative interpretation.⁹⁹¹ Thus Dapo Akande, for instance, admitted in the run-up to the Kampala Conference:

The lack of clarity in the second sentence of Article 121 (5) is unfortunate. The problems generated by that provision are in all probability attributable to sloppy drafting. It is doubtful that the intention behind the second sentence of Article 121 (5) was (i) to change the normal jurisdictional rules for the ICC; and/or (ii) to

as

⁹⁸⁸ Article 121 (5) (p. 231).

⁹⁸⁹ For a prominent defence of this 'negative' interpretation, see Akande 'What Exactly was Agreed in Kampala on the Crime of Aggression' (2010) (n. 959) and Akande, 'The ICC Assembly of States Parties Prepares to Activate the ICC's Jurisdiction over the Crime of Aggression: But Who Will be Covered by that Jurisdiction', (EJIL:Talk! 26 June 2017), https://www.ejiltalk.org/the-icc-assembly-of-states-parties-prepares-to-activate-the-iccs-jurisdiction-over-the-crime-of-aggression-but-who-will-be-covered-by-that-jurisdiction/, accessed 23 November 2019, and Akande and Tzanakopoulos (2018) (n. 902). Continuing support for this position has also been provided by Kevin Heller, see 'The Draft Resolution's Curious Paragraph 3', (OpinioJuris, 15 December 2017), https://opiniojuris.org/2017/12/15/the-curious-paragraph-3/, accessed 3 January 2018.

⁹⁹⁰ Article 12 (2) (n. 941).

⁹⁹¹ See Professor Akande's continuing support for the restrictive interpretation based on a strict application of the VCLT, discussed below in section 2.6, pp. 249-50. See also Akande and Tzanakopoulos (2018) (n. 902) and Heller (n. 989).

make a distinction between the effect of the Statute on nationals of non-party

States and nationals of State parties that do not accept the amendments. 992

This acknowledgement of the problematic wording in the second sentence of Article 121 (5), serves to emphasise its incompatibility with the general jurisdictional provisions in Article 12 of the Rome Statute and the negotiated compromise in Article 15 *bis* (4) of a choice to opt out. Additionally, insisting on the negative interpretation also has the effect of raising the already augmented consent requirement to an entirely new level.

Applying the 'positive understanding', on the other hand, would leave the door open for jurisdiction of the Court where *either* the aggressor *or* the victim had previously ratified the amendment, but the other had not, in line with the general jurisdictional provisions of the Rome Statute established in Article 12 (2).⁹⁹³ Following this interpretation, states parties which had not taken any active steps to either accept the amendments, or officially opt out under Article 15 *bis* (4), could be subject to the Court's jurisdiction, as long as the victim state had previously ratified the amendment.⁹⁹⁴ This interpretation made sense from both a logical and legalistic point of view, taking into account the provisions and aims of the Rome Statute as a whole. This particular option was considered in the pre-

⁹⁹² Akande, Prosecuting Aggression (2011) (n. 888).

⁹⁹³ Defending this position is, for instance, Astrid Reisinger-Coracini 'More Thoughts on "What Exactly was Agreed In Kampala on the Crime of Aggression", (EJIL:Talk! 2 July 2010), https://www.ejiltalk.org/more-thoughts-on-what-exactly-was-agreed-in-kampala-on-the-crime-of-aggression/, accessed 10 May 2020; Reisinger-Coracini (2010) (n. 962) and (2017) (n. 960). See also Jennifer Trahan, 'The Rome Statute's Amendment on the Crime of Aggression: Negotiations at the Kampala Review Conference' (2011), Int'l Crim. L. Rev., 49; Trahan, 'One Step Forward for International Criminal Law; One Step Backwards for Jurisdiction', (OpinioJuris, 16 December 2017), http://opiniojuris.org/2017/12/16/one-step-forward-for-international-criminal-law-one-step-backwards-for-jurisdiction-the-perspective-of-someone-present-at-the-un-during-negotiations/ accessed 7 January 2018, and 'Activation of the International Criminal Court's Jurisdiction over the Crime of Aggression and Challenges ahead' (OpinioJuris, 18 July 2018), https://opiniojuris.org/2018/07/18/33604, accessed 6 August 2018. See also Barriga (2017) (n. 946), Stuerchler (n. 946), Weisbord (n. 946), and Claus Kress, 'On the Activation of ICC Jurisdiction over the Crime of Aggression', (2018) 16 (1) JICJ.

⁹⁹⁴ Ibid.

Kampala discussions on the interplay between Article 121 (5) and Article 15 *bis* (4), as summarised in the 2009 Princeton Report:⁹⁹⁵

Some participants expressed interest in the idea of an opt-out declaration, combined with a system that would otherwise not require that the alleged aggressor State have accepted the amendment on aggression. Such an approach would strongly reduce the number of States who were beyond the Court's jurisdictional reach, as it would exclude only those States who took an active step to that effect. A system that required potential aggressor States to accept the amendment would not be effective: It was unlikely that such States would move to take such a step. An opt-out declaration, however, reversed that default situation and provided an incentive for States to reflect on the amendment and to come to a decision as to whether they could live with the amendment or not. 996

This description demonstrates that the very purpose of the inclusion of Article 15 *bis* (4), allowing for the possibility to opt out of the aggression amendment, was to replace the need for additional individual ratification. The intention expressed was the prevention of wide-scale non-application of the aggression provisions to signatories of the Rome Statute, leading the SWGCA to consciously adopt the specific wording they chose. The Crime of Aggression entails special entry into force conditions, contained in Article 15 *bis* (4), making a literal reading of Article 121 (5), which applies to amendments without special entry into force conditions in general, unnecessary. ⁹⁹⁷ Thus, Article 15 *bis* (4) deliberately does *not* require individual member states' acceptance before jurisdiction

⁹⁹⁵ 2009 *Princeton Report* (n. 981).

⁹⁹⁶ *Ibid*, paragraph 41. Emphasis added.

⁹⁹⁷ See Noah Weisbord's presentation to the United Nations General Assembly (n. 946), in preparation for the activation decision to be taken by States, prior to the 2017 ASP.

arises, but instead insists on active 'opting out' as the only way to escape jurisdiction. The article simply states that jurisdiction over an aggressive state party exists, 'in accordance with article 12', *unless* that state party has previously declared that it does not accept jurisdiction. ⁹⁹⁸ Consequently, jurisdiction of the Court can, by virtue of Article 15 *bis* (4), only be avoided by actively opting out, and prior individual acceptance should not be required. ⁹⁹⁹

Applying this literal reading of Article 15 *bis* (4), and taking the view that it supersedes Article 121 (5), ratifications by member states ought to be immaterial. This assessment is also supported by the first operative paragraph of the Kampala amendment resolution, in which '[the Review Conference] *notes* that any State Party may lodge a declaration referred to in article 15 *bis* prior to ratification or acceptance.' Furthermore, the first preambular paragraph of Kampala's aggression amendment expressly recalled Article 12 (1) and the general assumption of states parties' acceptance of jurisdiction over the four crimes in Article 5, including the crime of aggression. 1001

2.5 Combining Article 15 bis (4) with the negative interpretation of Article 121 (5)

Article 15 *bis*, in its entirety, establishes a special jurisdiction procedure, arrived at after exceedingly lengthy and convoluted negotiations. Its deliberate restrictions on the Courts'

242

⁹⁹⁸ See Article15 bis (4) (n. 935).

⁹⁹⁹ Barriga and Grover (2011) (n. 724), 531. This view is also strongly supported, for instance, by Astrid Reisinger Coracini (n. 962 at 764) and n. 993.

¹⁰⁰⁰ Resolution 6 (n. 2).

¹⁰⁰¹ *Ibid*.

jurisdiction are both unusual and stringent, ¹⁰⁰² evidenced by the procedural delays, ¹⁰⁰³ the opt-out provisions, ¹⁰⁰⁴ blanket non-application to non-state parties, ¹⁰⁰⁵ as well as Security Council scrutiny. ¹⁰⁰⁶ A strict application of the negative interpretation of Article 121 (5), allowing non-jurisdiction through mere passive non-acceptance, would render that very special procedure empty of meaning and pointless, in particular the exceptional provision in Article 15 *bis* (4). ¹⁰⁰⁷ In the words of Jennifer Trahan, 'if exercise of jurisdiction were *never* possible without ratification of the amendment by the aggressor State Party, there would be no need for a pre-ratification opt out.' ¹⁰⁰⁸ Similarly, Roger Clark, states that, 'if all States Parties are not to be bound, the opt-out option makes no sense.' ¹⁰⁰⁹ It follows that ratification by the aggressor should not be required, particularly where they have not filed an opt-out declaration and the victim has accepted jurisdiction.

Additionally, the valid critique has been levelled that Article 15 *bis* (4) on its own already creates the 'perverse incentive' for States to reject jurisdiction of the Court. Thus the provision potentially protects them from prosecutions for their own aggressive acts committed post opt-out, whilst simultaneously shielding them against aggression by another State which has not taken advantage of opting out. ¹⁰¹⁰ Applying the negative understanding of Article 121 (5) makes this situation even more 'perverse', by preventing jurisdiction, even if the aggressor has not done anything, i.e. not ratified but also not opted

¹⁰⁰² See Milanovic (n. 479), 178-182, arguing that the application of Article 121 (5) is in direct contradiction to Article 15 *bis* (4). See also Weisbord (n. 946).

¹⁰⁰³ Article 15 bis (2) and (3).

¹⁰⁰⁴ Article 15 bis (4).

¹⁰⁰⁵ Article 15 bis (5).

¹⁰⁰⁶ Article 15 *bis* (6)-(8).

¹⁰⁰⁷ Weisbord (n. 946).

¹⁰⁰⁸ Trahan (2011) (n. 993), 64. For a different view, see Akande (2010) (n. 959) and (2017) (n. 989), Akande and Tzanakopoulos (2018) (n. 902), and Heller (2017) (n. 989).

¹⁰⁰⁹ Clark (2010) (n. 687), 704.

¹⁰¹⁰ Milanovic (2012) (n. 479), 180.

out, despite actual ratification of the victim state. This effectively provides no protection from aggression even for member states, unless both states individually have ratified and failed to opt out, an unlikely scenario in an aggression case by its very nature.

The perception of perversity is, in fact, reinforced when one looks at the practical effects of the two different interpretations: Under the negative understanding the Court only has jurisdiction over aggression committed by any of the 39 ratifying states parties on the territory of another one of those same 39,¹⁰¹¹ whereas under the positive version the Court would have jurisdiction over aggression committed by any of the 39 on the territory of any of the 123 member states.¹⁰¹² Notably, the difference between the two versions is substantial.

Following these compelling arguments for a positive understanding, it seems reasonable to assume that the wording in Article 15 *bis* (4), specifically written for the crime of aggression, was indeed intended to supersede the negative interpretation insisting on individual ratification of both victim and aggressor states. Moreover, support for following Article 15 *bis* (4) over a strict reading of Article 121 (5) can also be gained from emerging state practice. Thus Kenya and Guatemala, the first two member states to actively opt out of the aggression provisions, ¹⁰¹³ had not previously ratified the amendments. They therefore appear to have followed the procedure in Article 15 *bis* (4)

¹⁰¹¹ See <u>www.crimeofaggression.info/the-role-of-states/status-of-ratification-and-implementation/</u> [last accessed 20th January 2021].

¹⁰¹² These figures apply in prosecutions brought under 15 *bis*. In cases of Security Council referral, Article 15 *ter* confers automatic jurisdiction.

¹⁰¹³ Kenya declared non-acceptance of jurisdiction of the Court over the crime of aggression in 2015, see MFA.1NT.8/14AVOL.X (86) 30th November, 2015. Guatemala declared non-acceptance on 16 January 2018. See www.icc-cpi.int [last accessed 24 November 2020].

directly. Nonetheless, the contradiction between the different articles remains unhelpful, as it adds to the uncertainty surrounding the new jurisdictional authority of the Court. The above considerations make the ASP's recent decision to resolve the matter in favour of the more restrictive position, all the more perplexing. Thus, on the 14th December 2017, the Assembly voted to activate the Court's jurisdiction over the crime of aggression as of 17th July 2018, subject to the negative interpretation, deciding that both sentences of Article 121 (5) ought to apply. ¹⁰¹⁴ The ASP's central purpose was to decide on activation of the Court's jurisdiction, rather than resolve the question of interpretation of Article 121 (5), which could have been left for the Court itself to settle. Instead, the controversy has been amplified by the choice of a negative interpretation of Article 121 (5), in direct contrast to Article 12 (2), ¹⁰¹⁵ as well as Article 15 *bis* (4) itself. The ASP's decision to apply the restrictive approach has been described as indicating

[...] an intention on the part of a small but powerful minority of States Parties to erode the finely-tuned compromise agreement negotiated in Kampala. If a State which fails to ratify simply cannot be subjected to the Court's Article 15 *bis* jurisdiction at all, then Art 15 *bis* (4) loses its meaning, and the 'political cost' of avoiding the Court's 15 *bis* jurisdiction no longer involves paying the price of having to formally opt out. ¹⁰¹⁶

The following discussion examines some potential reasons for this divisive decision.

¹⁰¹⁴ Resolution ICC-ASP/16/Res.5, 16th Assembly of States Parties, 14th December 2017 (Resolution 5, 2017).

¹⁰¹⁵ See the discussion in section 2.4. For support, see Clark (2009) (n. 961). See also Dapo Akande's comments, p. 239.

¹⁰¹⁶ Donald Ferencz, 'Aggression Is No Longer a Crime in Limbo', FICHL Policy Brief Series No. 88 (2018), 3, http://www.toaep.org/pbs-pdf/88-ferencz/, accessed 25 October 2020.

2.6 Yet another 'layer' of consent? – The reasons for the ASP's decision

In December 2017, the ASP provided in Resolution 5 for activation of the jurisdiction of the Court. In operative paragraph 2 of that Resolution, the ASP stipulated that Article 121 (5) ought to apply subject to the negative understanding requiring both victim and aggressor state to have accepted the amendments. This paragraph therefore

[c] onfirms that, in accordance with the Rome Statute, the amendments to the Statute regarding the crime of aggression adopted at the Kampala Review Conference enter into force for those States Parties which have accepted the amendments one year after the deposit of their instruments of ratification or acceptance and that in the case of a State referral or proprio motu investigation the Court shall not exercise its jurisdiction regarding a crime of aggression when committed by a national or on the territory of a State Party that has not ratified or accepted these amendments; 1017

The ASP was not required to make a decision on the understanding of Article 121 (5), and could have proceeded with simple activation of the Courts' jurisdiction as of July 2018. The dilemma could thus have been left for the Court to decide, as a matter of interpretation of the 2010 aggression amendment itself. In coming to its controversial decision, the ASP succumbed to pressure from France and the UK, supported by Japan, Canada, Norway, and Colombia, 1018 a group of States which has also been referred to as

¹⁰¹⁷ Resolution 5, 2017 (n. 1014), para 2.

¹⁰¹⁸ See Dapo Akande, 'The International Criminal Court Gets Jurisdiction Over the Crime of Aggression' (EJIL:Talk! 15 December 2017), https://www.ejiltalk.org/the-international-criminal-court-gets-jurisdiction-over-the-crime-of-aggression/, accessed 20 February 2018, defending the ASP's decision to

'camp consent'. ¹⁰¹⁹ 'Camp consent', to be distinguished from 'camp protection', refers to States insisting on consent by the aggressor state by expressly granting jurisdiction to the Court. ¹⁰²⁰ The opposing 'camp protection', by comparison, refers to States seeking protection from states parties through jurisdiction without a requirement for individual ratification. ¹⁰²¹ Many states parties present at the 2017 ASP expressed their continuing support for the positive and wider interpretation of Article 121 (5), which had featured strongly since the lead-up to the Kampala Review Conference. ¹⁰²² Consequently, and despite the fact that 'camp consent' included a number of powerful States, for instance the P5 and most European States, ¹⁰²³ the actual majority of states parties present and taking part in the actual debate, preferred either the positive interpretation or favoured a middle ground, rather than the restrictive view. ¹⁰²⁴

Instrumental in developing such a middle ground in the prelude to the December 2017 ASP, was Dapo Akande's and Claus Kress' attempt to build a 'bridge' between the negative and positive interpretations. The 'bridge' between the two opposing

apply the restrictive view. For a different view, see for instance Stuerchler (2017) (n. 946) and Donald Ferencz (2018) (n. 1016).

¹⁰¹⁹ See Barriga and Grover (2011) (n. 724), citing an informal 'roll call' in March 2010, by Prince Zeid Ra'ad Al Hussein in March 2010 (at p. 524 and n. 23 in Barriga and Grover).

¹⁰²⁰ For the distinct division of views between the two camps prior to the 2017 ASP, see *Report on the facilitation on the activation of the jurisdiction of the International Criminal Court over the crime of aggression*, ICC-ASP/16/24, 27 November 2017, paras. 11-22. Annex II A of that report contains a paper submitted in March 2017 by Canada, Colombia, France, Japan, Norway, and the UK representing the negative interpretation. Annex II B of that report contains the response by Liechtenstein, and Annex II C summarises the response by Argentina, Botswana, Samoa, Slovenia, and Switzerland, defending the positive interpretation, but suggesting simple activation without a decision by the ASP on the point. See also Trahan (2017 and 2018) (n. 993).

¹⁰²¹ Barriga and Grover (2011) (n. 724), 524, describe 'camp protection' as including all of the African States, members of the Non-aligned Movement and most Latin American and Caribbean countries.

¹⁰²² See Barriga and Grover, *ibid*, Stuerchler (2017) (n. 946), as well as Kress (2018) (n. 993). For an account of pre-Kampala support for the wider view see the 2009 *Princeton Report* (n. 981), at para 37.

¹⁰²³ Barriga and Grover, *ibid*, citing Prince Zeid's 2010 headcount (n. 724 and 1019).

¹⁰²⁴ Stuerchler (n. 946) counted 37 out of 46 state parties taking the floor against the restrictive view, in favour of the positive view or a middle ground.

¹⁰²⁵ See Kress, (2018) (n. 993) and Akande (15 December 2017) (n. 1018).

understandings suggested treating a mere statement by a States Party, advocating the negative interpretation, as a declaration which would effectively negate the Court's jurisdiction without the need to expressly opt out. This was a major concession to the restrictive position, but still incurred remarkably strong opposition. Consequently, this effort at achieving a compromise was ignored due to the pressure applied by a small, but nonetheless powerful, number of States, in particular France, the UK, Japan, Canada, Norway and Colombia. 1026 As stated by Kress, the opposition to this middle ground is particularly incomprehensible, since it would have granted those countries even greater certainty of immunity from prosecution, if a simple record of their views would have counted as a declaration of non-acceptance of jurisdiction. 1027

The decision by the ASP to apply the negative understanding of the second sentence of Article 121 (5) in the activating Resolution 5 has resulted in a variety of strong reactions. One such view, in the aftermath of the 2017 ASP, insisted that this double requirement of acceptance by *both* victim and aggressor, in addition to refraining from opting out of the aggression amendments, had now become binding on the Court. This argument, asserted particularly by Dapo Akande, considered the ASP's decision to confirm the narrow view as amounting to a decisive 'subsequent agreement' under Article 31 (3) (a) of the Vienna Convention on the Law of Treaties. Professor Akande viewed the ASP's narrow interpretation as conclusive, despite the contentiousness of the subject and the reluctance of many States to accept this restrictive view. According to this view, the ASP's decision was also deemed to be incontrovertible, notwithstanding the addition

¹⁰²⁶ See n. 1026.

¹⁰²⁷ Kress (2018), n. 993, 9-11.

¹⁰²⁸ Resolution 5 (n. 1014), para. 2.

¹⁰²⁹ Akande (December 2017) (n. 1018).

¹⁰³⁰ VCLT (n. 32), Article 31 (3) (n. 458).

of a 3^{rd} paragraph to the activating resolution, 1031 which reaffirmed the Court's independence and its ability to decide on disputes concerning the judicial functions of the ICC. 1032

Despite continuing to defend the principle of the narrow interpretation, ¹⁰³³ Professor Akande himself has more recently retreated, at least partially, from the certainty that the ASP's decision was absolutely decisive. ¹⁰³⁴ Thus, in a co-authored article of 2018, Akande now expresses some doubt that paragraph 2 of the activating resolution definitely amounts to a 'subsequent agreement', legally obliging the Court to follow this interpretation. He submits that, whilst a consensus resolution may or may not be seen as evidence of a decisive subsequent agreement, at the very least, it *adds considerable weight* to the view that the narrow view is to be regarded as an 'authentic interpretation of the [Rome] treaty'. ¹⁰³⁵

The argument that the ASP's decision in favour of the restrictive interpretation represents a 'subsequent agreement' under Article 31 (3) of the VCLT, can also be countered with the response that this reading did 'not reflect the 'actual' view of most, let alone all, States Parties to the Rome Statute, and was arrived at due to the procedural hurdles imposed on

¹⁰³¹ Resolution 5, 2017 (n. 1014). For para 3, see below p. 255.

¹⁰³² See Akande (December 2017) (n. 1018), and Akande and Tzanakopoulos (2018) (n. 902). See also Heller (n. 989), who criticises the ASP's decision to include paragraph 3, reasserting the Court's independence and ability to settle disputes related to the Court's own judicial functions.

following a strict application of Article 40 (4) of the VCLT (n. 32), stipulating that a treaty cannot bind a non-accepting State, and a state party to an original treaty has a choice whether or not to accept an amendment to that treaty. This general principle was also confirmed in the *Monetary Gold* case (*Italy v France, United Kingdom & United States*), (1954) ICJ Rep. 19. See Akande, (June 2017) (n. 989), and Akande (December 2017) (n. 1018).

¹⁰³⁴ Akande and Tzanakopoulos (2018) (n. 902).

¹⁰³⁵ *Ibid*, 945-6. The reasoning follows the ICJ's statement in the so called 'Whaling Case', *Whaling in the Antarctic (Australia v Japan: New Zealand intervening), Judgment*, ICJ Reports 2014, 226, para. 46.

activation.'1036 The ASP was attended by a reduced complement of states parties, many of whom left before the close of proceedings. Considering the fact that a two-thirds majority out the less than 100 states present was required to activate the Court's jurisdiction, a potential decision by the ASP in favour of the positive instead of negative interpretation may have caused failure of the resolution. In addition to this concern, there existed a strong desire to achieve activation by consensus rather than a 2/3 majority decision, in order to provide the Court's jurisdiction over the new crime with greater credibility in the face of prevailing scepticism and prevent the appearance of 'divided commitment'. 1037 Nikolas Stuerchler's account, in particular, describes the pressure applied by France, the UK and 6 supporting States, which may have resulted in 10 middle ground States wavering towards the restrictive view. 1038 As a result of this pressure, there was a fair chance that, without the acceptance of the restrictive interpretation of the Court's jurisdiction, a consensus was out of reach. This would have resulted in yet another considerable postponement of the jurisdiction of the Court with no clear indication of the length of the delay. 1039 This potential postponement would, as Stuerchler observed, 'have buried activation'. 1040

This description of the situation bears witness to the fact that the restrictive interpretation was accepted reluctantly, due to sheer pressure, rather than as a voluntary 'subsequent agreement' of States. Moreover, and notwithstanding the ASP's negative application of Article 121 (5), States widely expressed views of disagreement with this restrictive interpretation after the 'consensus' vote, allowing room for the argument that state

¹⁰³⁶ Stuerchler (n. 946).

¹⁰³⁷ See Trahan (2018) (n. 993), 2. See also Stuerchler's observations (n. 946).

¹⁰³⁸ Stuerchler, *ibid*.

¹⁰³⁹ *Ibid*.

¹⁰⁴⁰ *Ibid*.

practice and *opinio juris* do not confirm the ASP's negative interpretation. ¹⁰⁴¹ In addition, the International Law Commission's view also casts doubt on the 'subsequent agreement' argument generally, conceding that consensus does not 'necessarily indicate[...] any particular degree of agreement on substance'. In consideration of all of these factors, an argument can be brought that the ASP's consensus decision should not necessarily be afforded too much weight. ¹⁰⁴²

Additionally, persuasive doubt has also been cast on the legal validity of paragraph 2 of the 2017 ASP's activating resolution confirming the restrictive interpretation, since it contradicts Article 15 *bis* (4) of the Kampala agreement and Article 12, both of which are component parts of the Rome Statute itself. Paragraph 2 of the ASP's resolution cannot be said to be *amending* Article 15 *bis* (4), since the required amendment procedure was not applied, therefore it can only be interpreted as an attempt to clarify Article 121 (5), with a contradictory result. Consequently, the only absolutely certain way for a State to exempt its officials from prosecution for aggression, remains to file an opt-out declaration in accordance with Article 15 *bis* (4). Paragraph 2 of the ASP's resolution and the procedure was not applied, therefore it can only be interpreted as an attempt to clarify Article 121 (5), with a contradictory result.

Notwithstanding these powerful arguments, the ASP's decision to weigh in in favour of the restrictive view has, as it stands, narrowed jurisdictional reach over the crime of aggression even further, through the emphasis on dual consent by both aggressor and

¹⁰⁴¹ *Ibid*, see also Kress (2018) (n. 993), and Barriga (2017) (n. 946).

¹⁰⁴² See Conclusion 11(3), *ILC Draft Conclusions on Subsequent agreements and subsequent practice in relation to the interpretation of treaties, adopted on second reading in May 2018*, UN Doc. A/CN.4/L.907 (2018).

¹⁰⁴³ Stuerchler (n. 946). This point is also made by Kress (2018) (n. 993), Trahan (2017 and 2018) (n. 993), Barriga (2017) (n. 946) and Donald Ferencz (n. 1016). See also Weisbord's UN presentation (n. 946), and section 2.4 of this chapter.

¹⁰⁴⁴ Stuerchler, *ibid*.

victim state. The insistence on double ratification, coupled with the ability to subsequently opt out of jurisdiction under Article 15 *bis* (4), enables a State to be protected against aggression committed by another ratifying member state, but can also protect that first State against jurisdiction over its own acts of aggression. Non-accepting states parties thus enjoy total immunity from prosecution for aggression committed by them unless there is a Security Council referral, however they may paradoxically be entitled to redress if aggression is committed on their territory by an accepting states party. As a result, the victim's state of ratification is effectively immaterial, and jurisdiction is based on the necessary acceptance of the aggressor state. This particular feature has the problematic effect of favouring non-accepting states parties over non-states parties, by affording them exceptional protection from prosecution. The resulting inequality and privilege of non-consenting states parties serves to highlight this reinforced consent problem, which has been exacerbated by the negative interpretation. A defence of this feature as an incentive to join the ICC in the first place, is, this thesis argues, unconvincing. 1046

Aggression, by its very nature and definition, is rarely committed accidentally or unintentionally, and States with intentions of aggression are therefore unlikely to voluntarily accept jurisdiction. Additionally, States in general are reluctant to relinquish any part of their sovereignty. The narrow interpretation of Article 121 (5) has had the unfortunate effect of discouraging even member states from ratifying the aggression amendments. Refusing to ratify, or ratifying and then opting out of jurisdiction, would

¹⁰⁴⁵ *Ibid.* See also Barriga (n. 946).

¹⁰⁴⁶ See the discussion below at p. 254.

¹⁰⁴⁷ See the discussion in Chapter 1.

now grant member states protection from aggression by other ratifying member states but shield them from prosecution for their own acts.

The insistence on this extremely high level of consent, expressed through ratification and failure to opt out by both victim and aggressor state, promotes a perception of impunity for the crime. It also ignores entirely the special jurisdictional conditions arrived at only a few years earlier in Kampala, which already established an exceptional consent requirement in Article 15 *bis* (4). Expressing similar sentiments, Ambassador Wenaweser's statement for Liechtenstein sums up the consequences of the restrictive interpretation in the activating resolution:

We are disappointed that a few States conditioned such activation on a decision that reflects a legal interpretation on the applicable jurisdictional regime over the crime of aggression that departs from the letter and spirit of the Kampala compromise, and which aims to severely restrict the jurisdiction of the Court and curtail judicial protection for States Parties. ¹⁰⁵⁰

The result is the sacrifice of individual accountability for aggression at the altar of state consent. Ultimately, the choice to apply the restrictive interpretation amounts to 'close to an annihilation of the Court's jurisdiction over the Crime of Aggression arising from an act of aggression by a non-State Party and a State party that has lodged a declaration of non-acceptance'. ¹⁰⁵¹

¹⁰⁴⁸ See, for instance, Barriga (n. 946) and Trahan (n. 993).

¹⁰⁴⁹ See section 2.4.

¹⁰⁵⁰ See Christian Wenaweser's post-ASP statement, as recorded in Kress (2018, at fn. 47) (n. 993).

¹⁰⁵¹ Reisinger-Coracini (2010) (n. 962), 773.

It has been suggested that the excessive concession to 'camp consent' and the resulting limitations on jurisdiction could unwittingly serve as an actual incentive for membership of the Court, on account of the perverse protection which non-ratifying or non-consenting member states stand to enjoy. 1052 This thesis submits, however, that allowing such double standards to be applied in new admissions to the ICC would be in complete contradiction to the purposes and aims of the Rome Statute. Its authors expressed their clear intention to eradicate the most serious international crimes, including aggression, universally, ¹⁰⁵³ rather than aggression by a select few States who are willing to commit themselves to the jurisdiction of the Court and consequently are probably the least likely to commit the crime in the first place. Additionally, Article 120 of the Statute had aimed to prevent such situations of selective jurisdiction from the outset. 1054 This article stipulates that membership of the ICC does not allow for reservations, suggesting that every signatory to the statute ought to be seen as accepting jurisdiction of the ICC generally. Moreover, admissions of States seeking to gain protection from prosecution for their own acts of aggression would not only be counterintuitive to promoting credibility of the Court, but also simply inequitable, although this view admittedly carries moralistic undertones.

The opinion has been expressed that it may be possible for the Court to interpret a State's declaration of non-acceptance of jurisdiction, where it is followed by a deliberate act of aggression, as part of the process of 'planning, preparation, initiation or execution, ¹⁰⁵⁵ thus invalidating the declaration. ¹⁰⁵⁶ This may provide an interesting possible solution, allowing for wider interpretation by the Court and even disregard of a State's lack of

¹⁰⁵² Akande and Tzanakopoulos (2018) (n. 902), 958.

¹⁰⁵³ Preamble of the Rome Statute (n. 85).

¹⁰⁵⁴ Article 120 of the Rome Statute (n. 1).

¹⁰⁵⁵ Article 8 *bis* (1).

¹⁰⁵⁶ Reisinger-Coracini (2010) (n. 962), 773.

consent to jurisdiction. Its mere suggestion would, however, be likely to be met with severe criticism by advocates of 'camp consent', on account of ignoring the insistence and emphasis on state consent so fiercely defended in the first place. Severe accusations of judicial activism may also be levelled against the Court.

On the other hand, support for the Court's discretion in such situations can be gleaned from the fact that the ASP, despite their controversial decision in favour of the restrictive interpretation, added an extra paragraph to that effect to the activating resolution. Thus Paragraph 3:

Reaffirms paragraph 1 of article 40 and paragraph 1 of article 119 of the Rome Statute in relation to the judicial independence of the judges of the Court. ¹⁰⁵⁷

Article 119, first paragraph, stipulates that any dispute concerning the judicial functions of the Court shall be settled by the decision of the Court. This article, and the reference to it in the activating resolution, may therefore provide support for such interpretative powers of the Court. It may even supply the Court with the opportunity to decide that the positive interpretation of Article 121 (5) should be applied after all, in spite of paragraph 2 of the activating resolution, and contrary to the restrictive view. Such a decision could be based on the fact that the ASP's decision does not itself amount to an amendment of Article 15 *bis* (4), contradicting it on so many levels. Strong criticism for judicial

1

¹⁰⁵⁷ Resolution 5, 2017 (n. 1014), para 3.

¹⁰⁵⁸ Article 119 (1) Rome Statute (n. 1).

¹⁰⁵⁹ Resolution 5 (n. 1014), para. 2.

¹⁰⁶⁰ See the discussion above, and section 2.4.

activism would, admittedly, be likely to ensue. 1061 Nevertheless, the matter may not be as firmly closed as it appears, and the ASP's decision to apply the restrictive view may not be as final as it seems at first glance.

2.7 Assessing the effect of the emphasis on multiple consent by member states

The restrictive interpretation by the ASP, as a result of the pressure applied by a few powerful States, is in direct contradiction to Article 15 bis (4). There is no formal link to a legal basis for such a determination by the ASP, ¹⁰⁶² nor does subsequent practice confirm the view expressed as opinio juris. 1063 The multiple consent requirement for jurisdiction over aggression is also particularly unprecedented compared with the other crimes under the Rome Statute, and it contradicts Article 12. Furthermore, it cannot be classed as an 'amendment to the [Kampala] amendment', 1064 and its legal effect is therefore uncertain. 1065

Since there was neither a legal reason to interpret the provisions this strictly nor decide on the point at that particular time, the ASP's interpretation gives the impression of a

¹⁰⁶⁴ *Ibid*, 8.

¹⁰⁶¹ Heller (n. 989), who, similarly to Professor Akande, has consistently defended the legal correctness of the negative view, despite lamenting the restrictive result of impunity of a non-accepting state party. See Aggression' (OpinioJuris, 'The Sadly Neutered Crime of 13 http://www.opiniojuris.org/2010/06/13, accessed 7 January 2018.

¹⁰⁶² Kress (2018) (n. 993), 8. See also Stuerchler and Barriga (both n. 946).

¹⁰⁶³ Kress, *ibid*.

¹⁰⁶⁵ For a different view, see Heller (n. 989), Akande (n. 959 and 888), and Akande and Tzanakopoulos (n. 902).

deliberate policy decision, having succumbed to the influence exerted by 'camp consent'. It is possible that the insistence on double ratifications in addition to an opt-out possibility might, cynically, serve as a 'sweetener' for States to sacrifice some level of sovereignty by signing up to the International Criminal Court. But the question remains whether this potential enticement to membership came at too high a cost, allowing for largely elective impunity. It also may, unfortunately, result in effective emasculation of the first permanent International Criminal Court at a time when it has only just achieved jurisdiction over the 'supreme international crime' after decades of deliberation.

Protection from acts of aggression may now be mostly ineffective even for signatories of the Rome Statute, since that protection exists only for the 39 ratifying States against another of these same 39 States, who have taken a clear stand against the commission of aggression. It is tempting to view this excessive limitation of jurisdiction as an indictment of generally prevailing scepticism by powerful States, evidenced by historic reticence, ¹⁰⁶⁶ as well as a recent resurgence of sovereign supremacy, ¹⁰⁶⁷ over international cooperation to further the eradication of aggression. ¹⁰⁶⁸

The possibility remains that the Court itself may, especially after the inclusion of paragraph 3 in Resolution 5,¹⁰⁶⁹ decide differently to the ASP and apply the positive

¹⁰⁶⁶ See particularly Chapter 1.

withdrawal from UNESCO, 2017, UNHRC, 2018, proposed withdrawal from the WHO, as well as threats to withdraw from the WTO and NATO under the Trump administration). See Oona Hathaway 'Reengaging on Treaties and Other International Agreements (Part I), President Donald Trump's Rejection of International Law', (2 October 2020), https://www.justsecurity.org/72690/reengaging-on-treaties-and-other-international-agreements-part-ii-a-path-forward, accessed 1 December, 2020. See also the UK exit from the European Union.

¹⁰⁶⁸ See Reisinger-Coracini's assessment (2010) (n. 962), 773.

¹⁰⁶⁹ See n. 1057 and related text, p. 255-6.

interpretation. Thus it may award itself wider jurisdictional powers, at least where the crime has been committed on the territory of one of the ratifying signatories and the aggressor member state has not previously declared to opt out. In doing so, it may feel encouraged by the assimilation of jurisdictional powers of other international courts and tribunals. 1070 Paragraph 3 may therefore yet provide some hope for proponents of greater deterrence from the crime of aggression, 1071 as well as member states seeking protection from acts of aggression, i.e. 'camp protection'. Such assumption of wider jurisdictional powers is, admittedly, likely to be met with strong resistance from critics of judicial activism as well as the so-called 'camp consent'. 1072 It will be interesting to see if, despite the likely resistance, the Court will award itself wider jurisdiction, finding support in paragraph 3 of Resolution 5 and applying the positive understanding. If the Court chooses not to do so, the only certain way to increase their jurisdictional reach is through the exertion of political pressure on member states to ratify the aggression amendments and the persuasion of non-members to become signatories without reservations. Taking into consideration the relatively low number of only 39 ratifications in the ten years since Kampala, as well as the virtual stagnation of new membership to the Court, that is a significant task. 1073

¹⁰⁷⁰ See the discussion in Chapter 4, particularly the assumption of competence in *Nicaragua* (1984) (n. 827), *Armed Activities* (2005) (n. 37), *Tadic* (1995 Decision on Jurisdiction) (n. 838), and the *Reparations* case (1949) (n. 844).

¹⁰⁷¹ See n. 1057 and related discussion, pp. 255-6.

¹⁰⁷² See above discussion in section 2.6.

¹⁰⁷³ In 1998 the ICC boasted 120 members. Since then only five States joined and 2 have withdrawn (Burundi (2017), and the Philippines (2019)), leaving 123 members.

3. Article 15 bis (5) and the double membership requirement

The above section revealed that the jurisdictional limitations in Article 15 *bis* are extraordinary, even over the statute's own signatories. In light of the strong emphasis on state consent to jurisdiction, it may not come as a surprise that Article 15 *bis* excludes non-member states entirely from the jurisdiction of the Court, in cases of state referrals or *proprio motu* prosecutions. It may be recalled that Article 15 *bis* (5) provides:

In respect of a State that is not a party to this Statute, the Court shall not exercise its jurisdiction over the crime of aggression when committed by that State's nations or on its territory.¹⁰⁷⁴

This provision establishes yet another substantial difference for jurisdiction over the crime of aggression compared with the other three core crimes, to which Article 12 (2) applies generally without exception. Thus, Article 12 (2) provides a jurisdictional link for genocide, crimes against humanity and war crimes to nationals of non-signatories of the Rome Statute by virtue of either the territory or nationality of a member state. This means in practical terms that if nationals of any of the 123 member states commit one of these other three crimes on a non-member state's territory, or a national of a non-member state commits said crimes on the territory of a member state, jurisdiction arises by virtue of Article 12 (2). This link to non-member states is, however, expressly precluded in aggression cases through Article 15 *bis* (5), unless there is a Security Council referral, in

¹⁰⁷⁴ Article 15 bis (5) of the Rome Statute (n. 940).

¹⁰⁷⁵ See Article 12 (2) (n. 941).

which case jurisdiction arises automatically over any State, member or not. 1076 Consequently, the Court has, under Article 15 *bis*, no jurisdiction over crimes of aggression involving non-states parties in any capacity, and non-states parties cannot be prosecuted for the crime of aggression, nor are they protected as a victim of aggression, even when committed by a state party.

This deliberate deviation from other provisions in the Rome Statute presents a paradox, since, as discussed above, non-accepting states parties are granted greater protection, leading to the 'perverse incentive' to join, ratify, and then opt out. Thus, they are shielded against aggression committed by an accepting states party, whereas non-member states are not. The exclusion of non-member states in Article 15 *bis* (5) also has the effect of limiting the overall reach of the Court's jurisdiction even further. These considerations led the Japanese delegation to voice their objections prior to the adoption of the aggression amendments:

As we have pointed out in various informal settings, we have a serious problem with the new article 15 *bis* 1 *quater*. ¹⁰⁷⁹ For example, the government of a state party surrounded by non-states parties will have a difficulty in selling to its parliament an amendment which unjustifiably solidifies blanket and automatic impunity of nationals of non-States Parties: a clear departure from the basic tenet of article 12 of the Statute. ¹⁰⁸⁰

¹⁰⁷⁶ See Article 15 ter and Chapter 4.

¹⁰⁷⁷ See the arguments on pp. 243-4 and 254.

¹⁰⁷⁸ As a result of Article 15 *bis* (5) (n. 940).

¹⁰⁷⁹ Article 15 bis 1 quater has become Article 15 bis 5 in the Kampala Amendment, Resolution 6 (n. 2).

¹⁰⁸⁰ Statement by Japan (n. 954), in explanation of their position before the adoption of the aggression amendments, see *Travaux Preparatoires*, (n. 317), 812.

The exclusion of non-states parties from the jurisdiction of the Court, even as victims of aggression committed by a state party, is likely to lead to unfairness and 'asymmetric results'. Nevertheless, some commentators perceived this limitation as inevitable and necessary to achieve the Kampala Compromise. As Barriga and Blokker point out:

[I]t could be argued that fighting impunity is more important than maintaining such symmetry. Fair or not, the exclusion of crimes [of aggression] involving Non-States Parties was a crucial element for the compromise reached in Kampala.

This statement demonstrates that the decision of blanket non-application to non-member states was based on the desire to achieve any degree of jurisdiction at all, other than through Security Council referral, in the face of continuing resistance by States wishing to protect their sovereignty. The observation, therefore, reinforces the perception of prevailing insistence on individual state consent, encountered throughout the examination in this thesis and in particular in the previous section. The insistence on membership of the Court as a precondition for any jurisdiction to arise in the first place, represents yet another layer of consent, which the Kampala amendments insist on even before the other limiting conditions come into play, as a result of Article 15 *bis* (4) and its combination with Article 121 (5). This consideration leads us to investigate if there is any possibility to circumvent the multiple consent requirement at all in order to develop wider jurisdiction of the Court.

¹⁰⁸¹ See Blokker and Barriga, 'Exercise of Jurisdiction: State Referrals and *Proprio Motu* Investigations', in Kress and Barriga (eds.) (2017), (n. 721), 659. See also Barriga and Grover (n. 724), 526: 'To agree to a consent-based regime, which excluded non-states parties entirely, was a massive compromise for the many delegations that had favoured a protective regime.'

¹⁰⁸² *Ibid.*

4. A way forward? - A teleological interpretation, the *jus cogens* nature of the crime and the principle of universal jurisdiction

In the discussion presented so far, it has been argued that the Court may have to award itself wider jurisdictional powers, ¹⁰⁸³ in order to extend its reach to situations where only one of two member states has consented through ratification of the amendments. This section will investigate whether it may be possible to circumvent the consent requirement altogether, regardless of States' membership or ratification status, on account of the higher normativity of the crime and through application of the principle of universal jurisdiction. It will also examine ICJ and ICC decisions, as well as state practice, to discuss whether a trend is now discernible towards a de-emphasis of state consent and a derogation from related procedural rules on immunity with regards to *jus cogens* crimes. This analysis, it is argued, supports the contention that such consent to jurisdiction, irrespective of States' insistence on its amplification in the Kampala Amendments, may not be as immutable as it appears.

It may be recalled from the Introduction Chapter that the principle of universality confers jurisdiction over widely deplored crimes to any State, entitling them to legitimately prosecute such an offence in their domestic courts under customary international law. The perpetrators are deemed to be 'hostis humanis generis' or enemies of all mankind. The crime itself is regarded as offending all common principles of humanity, and is therefore perceived to be of concern to every single State. States are deemed to have 'the

¹⁰⁸³ Applying para. 3 of Resolution 5 (n. 1014), ignoring the negative interpretation applied by the ASP, see the discussion above.

¹⁰⁸⁴ See Williams (2000) (n. 108), 544.

¹⁰⁸⁵ *Ibid*.

legal competence and jurisdictional competence to define and punish particular offences, regardless of whether that State had any direct connection with the specific offences at issue.'1086 Consequently, the territory on which the crime occurred and nationality of the perpetrator or victim are immaterial. 1087 The exercise of universal jurisdiction is therefore not conditional on consent, nor does it require the traditional jurisdictional links based on territory or nationality. 1088 There is no absolute definitive list of crimes subject to the principle, however the prohibition of these crimes is based on their peremptory or jus cogens nature. As such, their general or universal prohibition cannot be derogated from, on the basis that they constitute fundamental principles of international law which cannot be set aside. 1089 These crimes include piracy, 1090 slavery, 1091 genocide, crimes against humanity, war crimes, 1092 as well as torture, 1093 and are commonly deemed to be subject to universal jurisdiction. Crimes against the peace, the forerunner to aggression, were also prosecuted at Nuremberg arguably under the universality principle, based on the fact that the signatories vested their domestic jurisdiction in the tribunals. 1094 Consequently, the Nuremberg Tribunal can be interpreted as the exercise of collective universal jurisdiction by a treaty-based international court. 1095 Acts of aggression committed by leaders were

¹⁰⁸⁶ Joyner (1996) 153. See also Lauterpacht (1944) 61, (both n. 110).

¹⁰⁸⁷ Randall (1988) 788, Scharf (2012) 358 (both n. 109).

¹⁰⁸⁸ For a discussion of the traditional links of prescriptive jurisdiction based on territory, active or passive nationality and protection, see Cassese (2013) (n. 12) 272-4, see also Vaughan Lowe and Christopher Staker, 'Jurisdiction', in Evans (2010) (n. 181) 318-326.

¹⁰⁸⁹ See the *ILC Report on Peremptory Norms of General International Law (jus cogens)* (n. 39), Art 53 VCLT (n. 32). See also the discussion below.

¹⁰⁹⁰ See US v Smith, US v Klintock, and Dahir, (n. 113), see p. 27.

¹⁰⁹¹ Green (1984) (n. 114), 179.

¹⁰⁹² See Paulet (2017) (n. 115), and *Tadic* (n. 115, at 20) for support that genocide, war crimes and crimes against humanity are crimes subject to universal jurisdiction.

¹⁰⁹³ See *Furundzija* (n. 115) 153, and Lord Browne Wilkinson's summary in *ex parte Pinochet*, stating that torture has developed into a *jus cogens* norm subject to universal jurisdiction (also n. 115), at 589.

¹⁰⁹⁴ See the London Charter, Article 6 (a) (n. 117). See also Control Council Law No. 10 (n. 121). The 19 signatories to Nuremberg are deemed to have conferred their entitlement to the exercise of universal jurisdiction over the heinous international crimes committed during World War II to the international tribunals. See Clark (1997) (n. 123) at 177 and Williams (n. 108) at 545-6 and 559.

¹⁰⁹⁵ Michael Scharf, 'The ICC's Jurisdiction over the Nationals of Non-Party States: A Critique of the US Position', 64 Law and Contemporary Problems (2001), 67, 103-106. See also Randall (n. 109), 804-6.

prosecuted under this delegated universal jurisdiction, or an 'international rule of law', independent of state consent. 1096 The Nuremberg tribunal therefore, conceivably, constitutes a precedent for the exercise of universal jurisdiction by the ICC, derived from delegation by its members. 1097

Prior to the Rome Conference, an interesting proposal was submitted suggesting universal jurisdiction over all of the crimes under the Statute including aggression, although this jurisdiction was intended to be limited to member states only. This proposal rejected a state consent requirement beyond ratification as detracting from the effectiveness of the Court and even its rationale and philosophical underpinnings. In the negotiations leading up to the Kampala Conference, the possibility of crediting the Court with universal jurisdiction was considered again, this time even over non-member states. Thus, the Chairman's Non-Paper, presented in preparation for the 8th Assembly of States Parties in 2009, acknowledged:

Some delegations went further [during the consideration of the previously debated conditions for the exercise of jurisdiction] by proposing that the Court should have universal jurisdiction even over non-member States in the interest of "[e]nding impunity for the most serious crimes of international concern". 1100

.

¹⁰⁹⁶ Goldsmith (n. 120). For a partially differing view, see Akande ('Prosecuting Aggression' (2011)) (n. 888), 30-2.

¹⁰⁹⁷ See Scharf (2001) (n. 1095) 103-6, and Randall (1988) (n. 109) 804-6.

¹⁰⁹⁸ The 'German Proposal' (1998), contained in the Draft Statute for an International Criminal Court, UN Doc. A/Conf. 183/2/Add.1 (n. 126). See also Williams (2000) (n. 108), 544.

¹⁰⁹⁹ Williams, *ibid*, 545. However, this proposal failed to attract sufficient support for the express blanket inclusion of universal jurisdiction over all of the crimes and all members in the final text of the Rome Statute, resulting instead in the insertion of Article 12.

¹¹⁰⁰ In the 2009 *Princeton Report*, ICC-ASP/8/INF.2 (n. 981), 8 (italics in the original). The individual States advocating this view remain unidentified.

The minority proposal considered here goes even beyond the German Proposal of 1998 and envisages universal jurisdiction, specifically over the crime of aggression, irrespective of consent or the existence of a traditionally required territorial link or a connection through the nationality of the perpetrator. This new proposal, therefore, entails jurisdiction over the universally abhorrent 'supreme' crime of aggression, ¹¹⁰¹ regardless of the status of a State's membership or ratification. Jurisdiction would be exercisable, at first instance, in domestic courts by any State over citizens of any other State, but would potentially also be transferrable to an international tribunal, following the precedent of Nuremberg. ¹¹⁰² This proposition of transferability of jurisdiction to the Court is further strengthened through the provision for complementarity of jurisdiction, which runs through the Rome Statute. ¹¹⁰³

Notwithstanding the support provided by these arguments for the applicability of universal jurisdiction over the crime of aggression, a number of counterarguments also require consideration. Firstly, it is hard if not impossible to ignore that the most influential negotiating States and their allies insisted on the deliberate limitations placed on the Court's jurisdiction and their incorporation into the aggression amendment. These powerful States, although not in the majority, continued to demand and consequently achieved the highest consent threshold, including the blanket non-application to non-member states unless a Security Council referral exists. ¹¹⁰⁴ If taken as an absolute

¹¹⁰¹ See Robert Jackson's statement (n. 135) in the Introduction Chapter. See also the final Nuremberg judgment itself (n. 3 and 235, at p. 16): 'To initiate a war of aggression, therefore, is not only an international crime; it is the supreme international crime differing only from other war crimes in that it contains within itself the accumulated evil of the whole.'

¹¹⁰² See Joyner (1996) (n. 110) 167, Clark (1997) (n. 123) at 177 and Williams (n. 108) at 545-6 and 559. See also Scharf (2001) (n. 1095) 103-6, and Randall (1988) (n. 109) 804-6.

¹¹⁰³ See Art 1 and Art 17 (1) (a) of the Rome Statute (n. 1), which provides for jurisdiction of the Court where states with jurisdiction are 'unwilling or unable genuinely to carry out the investigation or prosecution.'

¹¹⁰⁴ See for instance the so-called ABCS non-paper, and the related account of the negotiating history in Kress and von Holtzendorff (2010) (n. 5) 1193-4.

determinant, the strict observation of the intentions of the negotiating parties and the emphasis placed on state consent to jurisdiction indeed appears to provide a powerful rebuttal of the proposed application of universality of the aggression provisions. ¹¹⁰⁵

Additionally, and on the insistence of the observer state of the United States, an express understanding was attached to the Kampala amendments, that it 'shall not be interpreted as creating the right or obligation to exercise domestic jurisdiction with respect to an act of aggression committed by another State.' This understanding represents a deliberate attempt to exclude universal jurisdiction over the crime of aggression, as exercisable by individual States domestically. Nonetheless, the understanding does not contemplate the transfer of such jurisdiction to the ICC as supported by the complementarity principle enshrined in the Rome Statute, nor was it included in the actual amendments. Additionally, it is doubtful that the understanding is capable of having the effect of limiting the availability of universal jurisdiction of States under customary international law. This latter aspect, discussed in detail below, refers to the fact that the crimes under the Rome Statute are of a *jus cogens* nature, from which derogation by treaty provision is not possible. Consequently, the legal effect of the understanding remains in doubt.

The discussion, nevertheless, merits consideration of a second potential counterargument to the proposed availability of universal jurisdiction. Thus, it has to be acknowledged that international organisations generally, including the ICC, are designed to perform specific

¹¹⁰⁵ See for instance Akande (2011) (n. 888, 17-8) and for a continued defence of strict application of Arts 31 (1) and 2 (a) of the VCLT in particular, see Akande's blog entries of 2013 (n. 459) and 2017 (n. 989). For a differing view of non-application of the VCLT on the grounds of legality considerations, see Jacobs (2013) (n. 459).

¹¹⁰⁶ Understanding No. 4 bis, Annex III, attached to Resolution 6 (n. 2).

¹¹⁰⁷ See Art 53 VCLT (n. 32) and the discussion below.

¹¹⁰⁸ See *Handbook for the Ratification and Implementation of the Kampala Amendments*, p. 11, available at http://www.crimeofaggression.info, accessed 28 January 2021.

tasks and are therefore perceived to be subject to the principle of speciality. As such, they are 'invested by the States which create them with powers, the limits of which are a function of the common interests whose promotion those States entrust to them.' Consequently, a narrow view of the ICC's speciality is likely to focus on the deliberate limitations on the Court's jurisdiction, and its dependence on amplified state consent to that jurisdiction, the pervading subject matter of this Chapter.

In response to this particular counterargument, it can, however, be posited that the principle of speciality also takes into account special functions and objectives, relative to the international body in question. To that effect, Article 31 of the VCLT states that a treaty ought to be interpreted 'in the light of its object and purpose'. This purpose or *telos* of the institution may be implied rather than explicitly provided for. As a general but wider rule of interpretation, Art 31 (3) (c) provides for taking into account 'together with the context... any relevant rules of international law applicable in the relations between the parties.' Applying this teleological approach, the special purpose of the Rome Statute in its Preamble will be remembered as:

¹¹⁰⁹ As stated by the ICJ in the *Nuclear Weapons Opinion* (1996) (n. 370), para 25.

Art 31 (1) VCLT (n. 32), although that article also refers to interpretation in accordance with the 'ordinary meaning to be given to the terms of the treaty in their context'. This article therefore does not envisage a conflict between the ordinary or literal meaning of the treaty terms and its object and purpose. In case of the aggression provisions, this thesis has argued that there exists a conflict between the treaty terms on account of their limitations and restrictiveness of jurisdiction and the object and purpose of the Rome Statute as stated in the Preamble (n. 85). On this conflict, the VCLT provides, it is argued little guidance. For a critique of strict application of the VCLT to the Rome Statute see Dov Jacobs (2013) (n. 459), for a counter-argument see Dapo Akande's blog entries of 2013 (n. 459) and 2017 (n. 989).

¹¹¹¹ This point has arisen, particular in cases concerning the European Convention of Human Rights (Convention for the Protection of Human Rights and Fundamental Freedoms), Nov. 4, 1950, 213 UNTS 221 (entered into force Sept 3, 1953).

¹¹¹² Art 31 (3) (c) VCLT, n. 32. See also the reliance of the European Court of Human Rights (ECtHR) on this principle in *Loizidou v Turkey* (1996) VI Eur.Ct.H.R. 2216, 2231, para 43. For a discussion of the relevance of a teleological interpretation, particularly within Human Rights Law, see Vassilis P. Tzevelekos, 'The Use of Article 31 (3) (c) in the Case Law of the ECtHR: An Effective Anti-Fragmentation Tool or a Selective Loophole for the Reinforcement of Human Rights Teleology – Between Evolution and Systemic Integration (2010), 31 Mich. J. Int'l L. 621, 623.

Affirming that the most serious crimes of concern to the international community as a whole must not go unpunished and that their effective prosecution must be ensured...; and

Determined to put an end to impunity for the perpetrators of these crimes and thus to contribute to the prevention of such crimes.¹¹¹³

The authors of the Rome Statute thus envisaged the primary purpose of the Court to be the universal eradication of the most heinous international crimes through wide jurisdiction. Although the Court was formed technically as an international treaty subject to voluntary membership and agreed limitations, its speciality could be viewed more widely as an institution with the higher normative purpose of delimiting criminal state and individual behaviour through the codification, investigation and prosecution of the universally objectionable international crimes under its Statute. The special character of the International Criminal Court, applying this teleological interpretation, is derived from the combination of its multiple functions. Thus the ICC can be perceived to be a specialised institution within the international system, which gains its competence only partly from its express subscription by its membership but, more significantly, from its special purpose of universal eradication of the worst international crimes and the pursuit of impunity for their commission. This interpretation, in fact, was the very point of the German Proposal of 1998, which considered the Court's effectiveness as well as 'rationale and philosophical underpinnings'. 1114

¹¹¹³ Preamble, Rome Statute 1998 (n. 85).

¹¹¹⁴ German Proposal of 1998 (n. 126 and n. 1098)

The specificity of the Court's *telos* is also reinforced by the special normative quality of the prescription of the crimes within its jurisdiction, derived from their inherent peremptory nature. As the following discussion develops, this interpretation of the special normative nature of the Court's jurisdiction may, in fact, permit its expansion beyond the confines of ordinary treaty law and absolute insistence on voluntarist principles, propounding and promoting universality. It may also provide a rebuttal to the first counterargument above, which focused on a narrow observance of the treaty parties' intentions, by recognising the prohibition of aggression as a *jus cogens* norm, from which it is impossible to derogate.¹¹¹⁵ As addressed in Article 53 of the VCLT:

A treaty is void if, at the time of its conclusion, it conflicts with a peremptory norm of general international law. For the purposes of the present Convention, a peremptory norm of general international law is a norm accepted and recognized by the international community of States as a whole as a norm from which no derogation is permitted and which can be modified only by a subsequent norm of general international law having the same character. ¹¹¹⁶

Jus Cogens or peremptory norms are owed erga omnes partes, i.e. towards all.¹¹¹⁷ These generally applicable legal norms are deemed to be compelling and 'non-negotiable', ¹¹¹⁸ and thus potentially capable of superseding the requirement for state consent to their

¹¹¹⁵ See Report of the International Law Commission, A/74/10, 16th May, 2019, *Chapter V: Peremptory Norms of General International Law (jus cogens)* (n. 39) and the discussion below. For a critique of *jus cogens* and *erga omnes* obligations as eroding traditional international law, see for instance Prosper Weil, 'Towards Relative Normativity in International Law' (1983), 77 AJIL, 413.

¹¹¹⁶ Article 53 VCLT (n. 32).

¹¹¹⁷ See the ICJ's obiter dicta in Case concerning the Barcelona Traction Light and Power Company Ltd. (Belgium v Spain) (Second Phase) [1970] ICJ Rep 3, para 34. However, not all erga omnes obligations qualify as jus cogens norms, see ILC, Final Report of the Study Group on Fragmentation of International Law, UN Doc. A/CN.4/L.682 (April 13, 2006), para 408, emphasising that all jus cogens norms have 'necessarily an erga omnes scope, whereas not all erga omnes obligations have weight as jus cogens.' 1118 Dworkin (2013) (n. 43) 6.

application. Since, under international law, an obligation to abide by a rule of *jus cogens* normativity is owed to all States, all States are allowed to react to the violation of a peremptory norm. In 2019, the International Law Commission adopted a number of draft conclusions on peremptory norms. The second draft conclusion defined a *jus cogens* norm identically to Article 53 of the VCLT:

Conclusion 2

Definition of a peremptory norm of general international law (jus cogens)

A peremptory norm of general international law (*jus cogens*) is a norm accepted and recognized by the international community of States as a whole as a norm from which no derogation is permitted and which can be modified only by a subsequent norm of general international law having the same character.¹¹²¹

The third conclusion confirms that *jus cogens* rules rank higher than ordinary treaty law provisions:

Conclusion 3

General nature of peremptory norms of general international law (jus cogens)

Peremptory norms of general international law (*jus cogens*) reflect and protect fundamental values of the international community, are hierarchically superior to other rules of international law and are universally applicable. ¹¹²²

¹¹¹⁹ Art 48 (1) (b) ILC Articles on the Responsibility of States (n. 620).

¹¹²⁰ See the 2019 ILC Report on Peremptory Norms (n. 39).

¹¹²¹ *Ibid*, 142.

¹¹²² *Ibid*.

The prohibition of aggression is the first peremptory norm specifically identified in a non-exhaustive list contained in the annex to this part of the ILC's Report, ¹¹²³ and has, therefore, clearly been recognised as such a rule. Consequently, a conceptual link can be created between this special, higher normativity of the prohibition of the crime of aggression as a *jus cogens* rule, the ability of States to invoke universal jurisdiction over the crime and, *inter alia*, the transfer of that jurisdiction to the Court where States are unwilling or unable to exercise their jurisdiction.

It is therefore the special character of the Court, as derived from its inherent *telos* and the specific peremptory nature of the crimes under its jurisdiction, combined with the provision for complementarity between its own and States' domestic jurisdiction, which may enable the exercise of universal jurisdiction despite the attempts to exclude it, rendering these ineffective. States traditionally have 'the first bite at the cherry' but, where unable or unwilling to investigate and/or prosecute, their own universal jurisdiction may be transferrable to the Court on behalf of the international community as a whole, 1124 on the grounds of complementarity within the Statute and historical precedent.

This interpretation derives support from the argument that *jus cogens* rules are not affected by the treaty-based limitations placed on the jurisdiction of the Court. They prevail on account of their higher status in the normative hierarchy and the fact that they cannot be contracted out of. Thus it is possible to contend that the *jus cogens* nature

__

¹¹²³ *Ibid*, 205.

¹¹²⁴ See the commentary in n. 1094 and the discussion of the NMT in Chapter 1, section 4.3.

¹¹²⁵ See Art 53 VCLT (n. 32).

¹¹²⁶ This phenomenon itself and its interpretation can be regarded conceptually in one of two ways, as either part of a fragmentation process (see Martii Koskenniemi & Paivi Leino, 'Fragmentation of International Law? Postmodern Anxieties' (2002) 15 Leiden J. Int'l L. 553, or as part of a greater evolution of international law towards the acceptance of wider responsibilities as a result of the socio-normative

of the Crime of Aggression prohibition pushes aside the notion of entirely optional and voluntary acceptance of jurisdiction, rendering it a universally applicable international obligation, towards the *telos* of the eradication of international crimes and impunity for their commission.

Universal Jurisdiction over aggression may, as a result, be able to survive as a type of extra-treaty based jurisdiction on account of the higher normativity of the prohibition of the crime, in which case the statutory restrictions would not be applicable. The ordinarily required bases for jurisdiction, i.e. territoriality and nationality, 1127 would be sidestepped. The right and responsibility to exercise universal jurisdiction would, at first instance, continue to lie with individual States in their domestic jurisdiction, perceived to have 'the legal competence and jurisdictional competence to define and punish particular offences, regardless of whether that State had any direct connection with the specific offences at issue.' Jurisdiction over the individual would be derived from jurisdiction over the offence itself, 1129 and, following the Nuremberg precedent and applying a purposive approach, would be transferrable to the Court in situations of inability or unwillingness of the State. 1130

This discussion raises the valid question whether this purposive and admittedly not uncontroversial approach, effectively devaluing the importance of state consent in international criminal law, can derive support from judicial decisions and state practice.

prohibition of aggression and a related and arguably necessary de-valuation of state consent and sovereignty. See the discussion in the Introduction Chapter, pp. 16-26.

¹¹²⁷ See n. 1088 and related text.

¹¹²⁸ Joyner (1996) (n. 110) 153, 167. Jurisdiction over the person appears to be merged with jurisdiction over the offence.

¹¹²⁹ See Lauterpacht, (1944) (n. 110), 61.

¹¹³⁰ See the discussion above, and in particular n. 1094 and 1103 and related text.

To that effect, the ICJ has in the past had the opportunity to consider the domestic exercise of universal jurisdiction. Thus, in the *Arrest Warrant* Case, the Democratic Republic of the Congo complained that Belgium had, under universal jurisdiction, issued an arrest warrant for of the DRC's incumbent Foreign Minister for the commission of war crimes. Although the ICJ's majority opinion supported the DRC's complaint on the basis of immunity of incumbent Foreign Ministers in regard to national jurisdictions, 1132 the separate and dissenting opinions express support for the availability of universal jurisdiction for international crimes generally, and war crimes specifically. The Joint Separate Opinion of Judges Higgins, Kooijmans and Buergenthal, for instance, stated that universal jurisdiction for crimes of a heinous nature was available and lawful where these crimes are detrimental to the interests of humanity, 1133 and where there existed a duty under international criminal treaties to deliver or prosecute the suspect (*aut dedere aut prosequi*). 1134 As the judges asserted:

[T]he international consensus that the perpetrators of international crimes should not go unpunished is being advanced by a flexible strategy, in which newly established international criminal tribunals, treaty obligations and national courts all have their part to play.¹¹³⁵

Judges Higgins, Koojimans and Buergenthal also confirmed that the availability of universal jurisdiction continues to reflect current customary international law, applicable

11

¹¹³¹ Arrest Warrant of 11 April 2000 (Democratic Republic of the Congo v Belgium), Judgment, 14 February 2002, ICJ Reports 2002 [hereinafter Arrest Warrant Case].

¹¹³² *Ibid*, para 58.

¹¹³³ Arrest Warrant Case, Joint Separate Opinion of Judges Higgins, Koojimans and Buergenthal, para 61. ¹¹³⁴ Ibid, para 46.

¹¹³⁵ *Ibid*, para 51.

regardless of the links of territoriality or nationality,¹¹³⁶ on account of the higher normativity of the prohibition of and jurisdiction over such heinous crimes. This phenomenon was perceived to be part of an ongoing evolution,¹¹³⁷ a 'slow but steady shifting to a more extensive application of extra-territorial jurisdiction by States reflect[ing] the emergence of values which enjoy an ever-increasing recognition in international society.'¹¹³⁸

The dissenting opinions in the Arrest Warrant case also provide support for a perception of an inherent peremptory nature of international crimes, from which derogation is not permissible, 1139 reinforcing the argument for the availability of universal jurisdiction. Thus, the dissenting opinion of Judge Al-Khasawneh asserted that the 'effective combating of grave crimes has arguably assumed a *jus cogens* character reflecting recognition by the international community of the vital community interests and values it seeks to protect and enhance', rendering the prohibition and prosecution of these crimes a 'hierarchically higher norm'. 1140

Judge van den Wyngaert, in his dissenting opinion, underwrote this interpretation of a higher normativity of international crimes, 1141 criticising the majority judgment for ignoring the development of modern international criminal law towards greater individual

 $^{^{1136}}$ Ibid, see also R v Bow Street Metropolitan Stipendiary, ex parte Pinochet Ugarte (Amnesty International Intervening) (no. 3) [1997] UKHL 17.

¹¹³⁷ Arrest Warrant case, Joint separate opinion (n. 1144), paras 62 and 65. For a similar view of an evolution of international criminal law, see also (n. 1162 below) Judge Trinidade's dissenting opinion in the case of Jurisdictional Immunities of the State (Germany v Italy: Greece intervening) [2012] ICJ, Judgment of 3 February 2012, hereinafter Germany v Italy.

¹¹³⁸ Joint separate opinion, Arrest Warrant case (ibid) (n. 1144), para 73. As the judges expanded on, 'a trend is discernible that in a world which increasingly rejects impunity for the most repugnant offences, the attribution of responsibility and accountability is becoming firmer, the possibility for the assertion of jurisdiction wider and the availability of immunity as a shield more limited.' (Para 75).

¹¹³⁹ See Art 53 VCLT (n. 32).

¹¹⁴⁰ Arrest Warrant Case (n. 1142), Dissenting opinion by Judge Al-Khasawneh, para 7.

¹¹⁴¹ Arrest Warrant Case (n. 1142), Dissenting opinion by Judge van den Wyngaert, para 26.

accountability for the commission of core crimes. ¹¹⁴² These core crimes were customary international law crimes of a *jus cogens* nature, and their prohibition ought to be considered as a general principle of law in the sense of Art 38 of the ICJ's Statute. ¹¹⁴³ Universal jurisdiction was available, based on wide international condemnation of these serious international crimes and in order to avoid impunity. ¹¹⁴⁴ This jurisdiction, in the opinion of Judge van den Wyngaert, remained illimitable by either the national or territorial principles of jurisdiction or even by the statutory jurisdictional conditions of the ICC itself. ¹¹⁴⁵ This dissenting opinion expressly emphasises the prevalence of a teleological interpretation of the Rome Statute as derived from its Preamble, ¹¹⁴⁶ and derives support for a higher normativity of individual criminal liability, impunity and the responsibility of all States to punish core crimes from the *opinio juris* of UN resolutions. ¹¹⁴⁷

The question of a hierarchy of *jus cogens* violations and procedural rules arose again in the ICJ Judgment in *Germany v Italy*. ¹¹⁴⁸ Although this case does not consider the exercise of universal jurisdiction directly, it also raises the question of whether procedural impediments should be able to bar jurisdiction where *jus cogens* violations are alleged.

¹¹⁴² *Ibid*, para 27.

¹¹⁴³ *Ibid*, para 28, citing M. C. Bassiouni in 'International Crimes: Jus Cogens and Obligatio Erga Omnes' (1996), 59 Law and Contemporary Problems (4), 63-74 and *Crimes against Humanity in International Criminal Law* (Nijhoff 1999) 210-217, Randall (n. 109) 829-832, the opinion by Senior Counsel C.J.R. Dugard in *Re Bouterse* (2001), para. 4.5.5, available at: http://www.icj.org/objectives/opinion.htm, accessed 27 August 2021, and the ICTY judgment in *Furundzija* (n. 115), para 153.

¹¹⁴⁴ Judge van den Wyngaert's dissenting opinion (n. 1152), para 46.

¹¹⁴⁵ *Ibid*, para 64.

¹¹⁴⁶ *Ibid*.

¹¹⁴⁷ *Ibid*, referring to, for instance, Sub-Commission on Human Rights, Res. 2000/24, 'Role of Universal or Extraterritorial Competence in Preventive Action against Impunity', 18 August 2000, E/CN.4/SUB.2/RES/2000/24; Commission on Human Rights, Res. 2000/68, 'Impunity', 26 April 2000, E/CN.4/RES/2000/68; Commission on Human Rights, Res. 2000/70, 'Impunity', 25 April 2001, E/CN.4/RES/2000/70.

¹¹⁴⁸ Germany v Italy case, n. 1148.

In a policy decision, ¹¹⁴⁹ the Court held that there was no conflict between the status of peremptory norms and rules on immunity, on account of the fact that the latter had to be decided on procedural grounds *before* a substantive decision could be considered on the former. ¹¹⁵⁰ *Prima facie*, the majority judgment in the case of *Germany v Italy*, therefore, appears to devalue the status of *jus cogens* rules, as taking second place after preliminary procedural considerations.

Judge Trinidade's dissenting opinion, however, voiced strong disagreement with the majority opinion, criticising it on account of leading to 'a groundless deconstruction of *jus cogens*, depriving this latter of its effects and legal consequences.' Thus he asserted:

Grave breaches of human rights and of international humanitarian law, amounting to international crimes, are not at all acts *jure imperii*. They are anti-juridical acts, they are breaches of *jus cogens* that cannot simply be removed or thrown into oblivion by reliance on State immunity. This would lock the access to justice and impose impunity. It is, in fact, the opposite that should take place: breaches of *jus*

¹¹⁴⁹ In a majority decision, the Court decided not to reopen the debate on reparations and the compensation of individuals, payable by the German State for gross human rights violations and war crimes committed during the Second World War. For a critique, see for instance Kimberley N. Trapp and Alex Mills, 'Smooth Runs the Water where the Brook is Deep: The Obscured Complexities of Germany v Italy' (2012), 1 Cambridge J Int'l Comp L (1), 153, 154.

formany v Italy (n. 1148), para 82. The Court referred to both the Armed Activities case (n. 37, paras 64 and 125) and Arrest Warrant case (n. 1142, paras 58 and 78) in its ruling to confirm that jus cogens does not bear on either the ICJ's jurisdiction or on a state's prerogative arising from immunity. For a critique of the Court's ruling and circular logic applied as overly simplistic and avoiding 'conducting more delicate balancing exercises between conflicting values' (citing the Joint separate opinion in the Arrest Warrant case, n. 1144), see for instance Andrea Bianchi, Gazing at the Crystal Ball (again): State Immunity and Jus Cogens beyond Germany v Italy (2013), 4 J Int'l Dispute Settlement (3), 457, 461. See also Alexander Orakhhelashvili, 'Jurisdictional Immunities of the State (Germany v Italy: Greece Intervening)' (2012) AJIL 106, 609, 615.

¹¹⁵¹ Germany v Italy, Dissenting Opinion of Judge Trinidade, para 297.

cogens bring about the removal of claims of State immunity, so that justice can be done. 1152

Notably, even the conservative ruling of the majority explicitly emphasised that the judgment concerned the jurisdictional immunity of *States*, distinct from the availability of *individual* liability for international crimes in the context of immunity, ¹¹⁵³ which it declined to address. ¹¹⁵⁴ Consequently, it may be possible to distinguish this restrictive majority opinion and argue that any potential preliminary procedural limitations, like immunity or state consent are, at best, capable of providing a limitation of state liability only. ¹¹⁵⁵ It is therefore arguable that the majority judgment in the *Germany v Italy* case fails to dismantle the hierarchy established by *jus cogens* norms, and their status, as confirmed in Art 53 of the VCLT, ¹¹⁵⁶ would remain immutable in relation to the establishment of individual criminal liability. ¹¹⁵⁷ Support for this contention can, again, be derived from Judge Trinidade's dissenting judgment, denying the availability of procedural prerogatives which permit sheltering behind the shield of immunity or state

¹¹⁵² *Ibid*, para 129. Judge Trinidade supported Greece's argument that 'a procedural rule cannot take precedence over the substantive *jus cogens* rule' (para 123, citing Greece's argument (in Doc. CR 2011/19, p. 36, para 102). In his opinion, Judge Trinidade relied extensively on the conclusions of, amongst other authorities, the *Institut de Droit International* (n. 1162, paras 44-49).

¹¹⁵³ *Germany v Italy* Judgment, Majority Opinion, n. 1148, para 87. Note here that the ICJ distinguishes the case of *ex parte Pinochet* (No. 3) (n. 1147) on the grounds that the *Pinochet* case concerned the personal immunity of a former Head of State from individual criminal liability rather than the liability of the Chilean State.

¹¹⁵⁴ *Ibid*, para 91. For a critique of the ICJ applying one of the 'conflict avoidance techniques' in order to circumnavigate the discussion of a normative hierarchy, see for instance P. Webb, 'Human Rights and the Immunities of State Officials', in E. de Wet and j. Vidma (eds) *Hierarchy in International Law: The Place of Human Rights* (OUP 2012), 114, 147. Support for the higher normativity of *jus cogens* rules is provided generally by Alexander Orakhelashvili, 'Peremptory Norms as an Aspect of Constitutionalisation in the International Legal System', in M. Frishman and S. Muller (eds), *The Dynamics of Constitutionalism in the Age of Globalisation* (Hague Academic Press 2010) 153, 165. For a different outlook, critiquing the concept of a normative hierarchy, see L. M. Caplan, 'State Immunity, Human Rights, and *Jus Cogens:* A critique of the Normative Hierarchy Theory (2003) 97 AJIL 741, 771-2. See also Hazel Fox, *The Law of State Immunity* (3rd edn. OUP 2008) 151, for a dismissal of *jus cogens* as ineffective in procedural immunity decisions.

¹¹⁵⁵ Applying the reasoning in paras 87 and 91 in *Germany v Italy* (n. 1148).

¹¹⁵⁶ Art 53 VCLT (n. 32).

¹¹⁵⁷ See Alexander Orakhelashvili (2012) (n. 1161) 615.

consent, on account of the fact that international crimes amount to 'grave breaches of absolute prohibitions of *jus cogens*, for which there can be no immunities. 1158

If the procedural prerogative of state immunity is capable of being suspended in criminal prosecutions for *jus cogens* violations, as asserted by Judge Trinidade and conceded even by the conservative majority, ¹¹⁵⁹ the same logic ought to apply to the availability of the shield of state consent, a sovereign right of a procedural nature very closely related to and of the same effect as immunity. As Andrea Bianchi points out:

The argument is one of logic. International law cannot grant immunity from prosecution in relation to acts which the same international law condemns as criminal and as an attack on the interests of the international community as a whole. Nor can the principle of sovereignty, of which immunity is clearly a derivative, be persuasively set forth to defeat a claim based on an egregious violation of human rights.¹¹⁶⁰

The limitation of the right of States to insist on their consent to prosecution of their leaders for international atrocity crimes can be directly assimilated to the limitation of the reliance

¹¹⁵⁸ *Germany v Italy*, Dissenting opinion by Judge Trinidade (n. 1162), para 297. He continues (at para 299): 'Jus cogens stands above the prerogative or privilege of State immunity, with all the consequences that ensue therefrom, thus avoiding denial of justice and impunity.' See also Art 27 of the Rome Statute, providing for the unavailability of an immunity defence for individual criminal responsibility for the crimes under the Statute, which are of a jus cogens nature (see the 2019 ILC Draft conclusions (n. 39)). This stance was also adopted by the ICC in its pursuit of an arrest warrant for the Sudanese President Al Bashir (n. 1210 below and related text), although the issue of the arrest warrant was also based on the referral to the Court by the Security Council (Appeal Chamber's decision, n. 1212 below, para 7).

¹¹⁵⁹ See the concession by the majority in the *Germany v Italy* case that the *jus cogens* nature of international crimes remains unaffected by procedural issues such as immunity (n. 1148, para 87).

Andrea Bianchi, 'Immunity versus Human Rights: The Pinochet Case' (1999) EJIL 10, 217, 260-1, citing the *dicta* of the NMT in 22 Trial of the Major War Criminals Before the International Military Tribunal (1949), 466. 'The principle of international law, which under certain circumstances, protects the representative of the state, cannot be applied to acts which are condemned as criminal by international law. The authors of these acts cannot shelter themselves behind their official position in order to be freed from punishment in appropriate proceedings...'

on immunity for the commission of such crimes. They are, in fact, part of the same coin. If immunity from prosecution is unavailable for the commission of international crimes of a *jus cogens* nature, including the crime of aggression, so too should be sheltering behind the procedural prerogative of state consent to such prosecution.

The protection and promotion of the interests of the international community and the eradication of and impunity for the worst international atrocity crimes, i.e. the *telos* of the Rome Statute, ¹¹⁶¹ justify setting aside an insistence on state consent and voluntarism as an absolute. Consequently, the crime of aggression and the other international crimes of such a peremptory nature, arguably justify a potential limit to democratic state consent on account of their 'imperative norms that may not be derogated from'. ¹¹⁶² Individual States gain the right and, as contended, are under a compelling obligation to exercise universal jurisdiction over these widely deplored crimes of a peremptory nature, which are committed against the international community as a whole, regardless of territorial boundaries or nationality and irrespective of consent. ¹¹⁶³ A State which captures an indicted perpetrator of such *jus cogens* crimes 'either may extradite him to the state where the offense was committed or to an international tribunal established to try such offenders, or retain him for trial under its own legal processes', ¹¹⁶⁴ applying the principle of universality. Jurisdiction over the peremptory crimes under the Rome Statute, including the crime of aggression, could therefore be available under customary international law,

¹¹⁶¹ As stated in the Preamble to the Rome Statute, n. 85.

¹¹⁶² See Art 53 VCLT, n. 32. See also Samantha Besson, 'State Consent and Disagreement in International Law-Making' (2016), 29 Leiden Journal of International Law 289, stating that 'democratic state consent only amounts to an exception to the obligations arising out of international treaties on other grounds and an exception whose justification is not necessarily granted, especially in the context of imperative and core duties...' (at p. 314). See also Andrew Guzman, 'Against Consent' (2012), 52 Virginia Journal of International Law, 747.

¹¹⁶³ See Joyner (1996) (n. 110) 168-9.

¹¹⁶⁴ *Ibid*. 170.

even where state consent to prosecution, in form of membership and/or ratification of the aggression amendment, is non-existent. As Dapo Akande observed in 2003:

The rule permitting delegation of universal jurisdiction to international courts can be regarded as a 'structural' rule of international law that does not require the positive consent of states but rather is deduced from other clearly-established rules. The structural rule established above is supported by pre-and post-ICC delegations of criminal jurisdiction by states to international tribunals, including the delegation of jurisdiction over nationals of states not party to the relevant treaty.' 1166

The most prominent precedent for such delegation is the Nuremberg Tribunal itself. 1167
The International Criminal Tribunals for the Former Yugoslavia 1168 and the International
Criminal Tribunal for Rwanda, 1169 may differ on account of being established by the
exercise of Chapter VII powers of the Security Council, however both statutes expressly
stipulated that no Head of State immunity existed before an international criminal

of universal jurisdiction and its transferability to the ICC was also recognised by Dapo Akande in a 2003 article, 'The Jurisdiction of the ICC over Nationals of Non-Parties: Legal Basis and Limits', JICJ 1 (2003), 618, 626. He states that 'it would be extraordinary and incoherent if the rule permitting prosecution of crimes against the collective interest by individual states – acting as agents of the community – simultaneously prevented those states from acting collectively in the prosecution of these crimes...', and 'where states are acting individually to protect collective interests and values, they are not prohibited and should rather be encouraged, to take collective action for the protection of those collective interests' (p. 626). Notably however, Professor Akande's views in later writings appear to shift towards an acceptance of a greater authoritative weight of state consent as expressed through agreed treaty provisions and membership. (Akande (2011) n. 888, (2013) n. 459, (2017) n. 989). For a sceptical view of the ICC's exercise of jurisdiction over non-consenting parties, and non-members in particular, see also Madeleine Morris (2001) (n. 119), 27.

¹¹⁶⁶ See Dapo Akande (2003), *ibid*, at 626.

¹¹⁶⁷ See n. 1094 and related text.

¹¹⁶⁸ Established by UN Security Council Resolution 827 (1993), S/RES/827, adopted 25 May 1993. The relevant article suspending Head of State immunity is Art 7 (2) ITCY Statute.

¹¹⁶⁹ Established by UN Security Council Resolution 955 (1994), S/RES/955, adopted 8 November 1994, the relevant article being Art 6 (2) of its Statute.

tribunal, and state consent to prosecution of its nationals was therefore not required. These tribunals can, therefore, also be interpreted as operating under a delegation of jurisdiction by the international community at large through the organ of the Security Council of the United Nations, on account of the fact that consent of the countries, whose nationals were being prosecuted, was not essential.¹¹⁷⁰

The Special Court for Sierra Leone, on the other hand, was created in form of a treaty between the UN and a consenting Sierra Leone, but without limitation of jurisdiction to nationals of that State. In 2012, the Special Court convicted Charles Taylor, the former Head of State of Liberia, a non-party to the treaty, in a precedent for jurisdiction over international crimes irrespective of Head of State immunity. Other examples of a form of delegated criminal jurisdiction, irrespective of state consent, are that of the European Court of Justice (ECJ), on account of the fact that Article 234 EC Treaty provides for the referral of questions on EC law to the ECJ, Irra including those regarding non-consensual criminal proceedings, and the jurisdiction of the Caribbean Court of Justice. Irra All of these examples provide support for the availability of delegated criminal jurisdiction, whether consensual or not, and exercisable by international courts and tribunals.

¹¹⁷⁰ See Scharf (2001) (n. 1095), 103-6.

¹¹⁷¹ Agreement between the United Nations and the Government of Sierra Leone on the Establishment of a Special Court for Sierra Leone 2002, and its annexed Statute, available at http://www.sierra-leone.org/specialcourtagreement.html.

The Court's Appeals Chamber upheld his 50 year sentence for crimes against humanity, war crimes and other serious violations of international humanitarian law in 2013. See Lansana Gberie, 'The Special Court for Sierra Leone rests - for good', (April 2014) Africa Renewal, available at https://www.un.org/africarenewal/magazine/april-2014/special-court-sierra-leone-rest---good, accessed 3rd September 2021

¹¹⁷³ European Union, Treaty Establishing the European Community (Consolidated Version), Rome Treaty, 25 March 1957, Art 234.

Agreement Establishing the Caribbean Court of Justice, 14 February 2001, available at http://www.caricom.org/ccj-index.htm.

The ICC's own recent practice also has, on occasion, provided evidence for the availability of its jurisdiction over international crimes irrespective of a lack of state consent or membership. It is in the case of Sudanese President Al Bashir, the ICC made a number of decisions, following the issue of arrest warrants in 2009 and 2010, It which considered the immunity of Heads of State of non-state parties under customary international law vis-à-vis the ICC. Thus, in 2011, the Pre-Trial Chambers addressed the failure of Malawi to arrest and surrender Al-Bashir, whilst present within their territory, and stated that the President was not entitled to immunity before the ICC, regardless of the fact that Sudan was not a member state. UN Security Council Resolution 1593 (2005) conferred jurisdiction onto the Court in any event, and therefore Art 27 (2) of the Rome Statute had the effect of suspending any such immunity even of a Head of State of a non-member state.

On 6 May 2019, the highest Chamber of the ICC unanimously confirmed the Pre-Trial Chambers' decisions in the Jordan Referral that the Court had jurisdiction as a result of the Security Council referral. The Appeals Chamber held that this permitted stripping Mr. Al Bashir of his Head of State immunity in accordance with Art 27 (2). Significantly

-

¹¹⁷⁵ Arguably at least where there is an additional basis for jurisdiction. The least controversial of these decisions rely on a pre-existing Security Council referral, see the discussion which follows in the main text. ¹¹⁷⁶ *The Prosecutor v Omar Hassan Ahmad Al Bashir*, Arrest Warrant of 4 March 2009, ICC-02/05-01/09-1; Arrest Warrant of 12 July 2010, ICC-02/05-01/09-95.

¹¹⁷⁷ Pre-Trial Chamber I, Corrigendum to the Decision Pursuant to Article 87 (7) of the Rome Statute on the Failure by the Republic of Malawi to Comply with the Cooperation Requests Issued by the Court with Respect to the Arrest and Surrender of Omar Hassan Ahmad Al Bashir (ICC-02/05-01/09), 13 December 2011, available at https://www.icc-cpi.int/CourtRecords/CR2011 21750.PDF. Pre-Trial Chamber II, Decision under article 87 (7) of the Rome Statute on the non-compliance by South Africa with the request by the Court for the arrest and surrender of Omar Al-Bashir, (ICC-02/05-01/09), 6 July 2017, available at https://www.icc-cpi.int/CourtRecords/CR2017_04402.PDF . Pre-Trial Chamber II, Decision under article 87 (7) on the non-compliance by Jordan with the request by the Court for the arrest and surrender of Omar December Al-Bashir, (ICC-02/05-01/09), 2017, available https://www.icc-11 cpi.int/CourtRecords/CR2017 07156.PDF, all accessed 3rd September 2021.

¹¹⁷⁸ Appeals Chamber, *Judgment in the Jordan Referral re Al-Bashir Appeal* (ICC-02/05-01/09) OA2, 6 May 2019, available at https://www.icc-cpi.int/CourtRecords/CR2019 02856.PDF, para 7, accessed 3rd September 2021.

and also unanimously, the Chamber added an additional pillar to their justification for non-consensual jurisdiction over an individual of a non-member country, asserting that, in general and regardless of consent, under customary international law no immunity applied for Heads of State before the ICC. Thus, the Appeals Chamber declared that neither State practice nor *opinio juris* supports the existence of Head of State immunity vis-à-vis an international court, ¹¹⁷⁹ or even horizontally (i.e. at national level) between States, once an international court has jurisdiction and has requested arrest and surrender of an individual. ¹¹⁸⁰ Following this decision, the ICC has refuted calls to seek an advisory opinion from the ICJ, ¹¹⁸¹ relying on the general international law principle that a Court is competent to rule on its own competence, as well as Article 119 (1) of the Rome Statute, stipulating that disputes concerning the judicial functions of the Court are to be settled by the Court's own decisions. ¹¹⁸²

Other recent assumptions of investigation and potential prosecuting powers by the ICC against non-member States can be seen as confirming this trend. This practice of the Court, on occasion, demonstrates an expansion of the reach of jurisdiction over international crimes regardless of consent, even where the Court has not been able to rely on Security Council referrals as a basis for jurisdiction (and is unlikely to in the future). The relevant situations include, for example, the assumption of investigation powers regarding the forced deportation of the Rohingya from Myanmar, 1183 on the basis that the

1

¹¹⁷⁹ *Ibid*, para 1.

¹¹⁸⁰ *Ibid*, para 2. This latter statement, in particular, potentially provides at least an indirect challenge to the restrictive interpretation of the majority of the ICJ in the *Germany v Italy* case, although the ICC's decision concerns immunity in the context of individual criminal liability, whereas the *Germany v Italy* case was decided in the context of state responsibility, immunity and compensation (see the discussion above).

¹¹⁸¹ See ICC, *The Prosecutor v Omar Hassan Ahmad Al Bashir*, Questions and Answers, available at https://www.icc-cpi.int/itemsDocuments/190515-al-bashir-qu-eng.pdf, accessed 3rd September 2021.

¹¹⁸² Art 119 (1) Rome Statute (n. 1).

¹¹⁸³ ICC Pre-Trial Chamber III, Decision Pursuant to Article 15 of the Rome Statute on the Authorisation of an Investigation into the Situation in the People's Republic of Bangladesh/Republic of the Union of

alleged crime began in the latter non-member state but continued into Bangladesh, which is a member state. 1184

In a different investigation, the Prosecutor's Office 2020 Report concluded that, after preliminary examination, there were reasonable grounds to believe that war crimes and crimes against humanity were committed on the territory of Ukraine and the Crimea, ¹¹⁸⁵ the latter having been unlawfully annexed by the Russian Federation in early 2014. ¹¹⁸⁶ Whilst the Ukraine had accepted jurisdiction of the Court for events from 21 November 2013, the Court will assume jurisdiction over these crimes on territory claimed by non-member and non-consenting state Russia, ¹¹⁸⁷ if the Pre-trial Chamber authorises prosecution. Consequently, this investigation and potential future prosecution is another example of the expansion of the Court's jurisdiction without strict adherence to consent and/or membership requirements.

Myanmar, ICC-01/19, 14 November 2019 [the *Myanmar* decision]. Note here the purposive approach of the PTC, taking into account the gravity (paras 34-7) and the interests of justice (paras 38-9).

The ICC again expanded its territorial jurisdiction to non-consenting and non-member states, relying on the 'continuing *actus reus* theory' in customary international law (*Myanmar decision*, para 61-2). For a critique, see for instance Douglas Guilfoyle, 'Is the International Criminal Court destined to pick fights with non-state parties? EJIL:Talk!, 14 July 2020, available at https://www.ejiltalk.org/is-the-international-criminal-court-destined-to-pick-fights-with-non-state-parties/, accessed 5th September 2021. Note, however, that so far the Prosecutor's Office has declined to seek authorisation of investigation of alleged forced deportations of Uyghur Muslims into the non-member state China from Tajikistan and Cambodia (both member states). See the 'Prosecutor's Report on Preliminary Examination Activities 2020', 14 December 2020, available at https://www.icc-cpi.int/itemsDocuments/2020-PE/2020-pe-report-eng.pdf [hereinafter the 2020 Prosecutor's Report], accessed 5th September 2021. The report here states that the alleged criminal conduct on the territory of the States Parties did not fulfil the material elements of the crime of deportation (para 76), and it had no jurisdiction over alleged crimes committed solely within the non-member state of China (paras 73-4).

¹¹⁸⁵ See 2020 Prosecutor's Report, *ibid*, para 267-90.

¹¹⁸⁶ *Ibid*, para 274. The United Nations General Assembly condemned the illegal annexation of Crimea in Resolution A/RES/68/262 (n. 799), and has reiterated the condemnation on a number of occasions, most recently in A/RES 75/29, 7 December 2020.

¹¹⁸⁷ For a discussion, see for instance Iryna Marchuk, Aloka Wanigasuriya, 'The ICC concludes its preliminary examination in Crimea and Donbas: What's next for the situation in Ukraine?' EJIL:Talk! 16 December 2020, available at https://www.ejiltalk.org/the-icc-concludes-its-preliminary-examination-in-crimea-and-donbas-whats-next-for-the-situation-in-ukraine/, accessed 3rd September 2021.

A further example and landmark decision on non-consensual jurisdiction, is the Appeals Chamber's reversal in March 2020 of the Pre-trial Chamber's decision to stop the Prosecutor's investigation into claims of war crimes, crimes against humanity and torture committed by US military personnel and members of the CIA in Afghanistan, an ICC state party since 2003. Importantly, the Pre-trial Chamber was held to have unduly taken into consideration the non-membership and non-consent of the United States, as well as political and financial considerations and chances of success. Additionally, the Pre-trial Chamber had expressed doubts as to the Prosecutors assessment of the investigation serving the interests of justice, Il89 and the Appeals Chamber rebuked the Pre-trial Chamber's policy decision sharply as inappropriate.

Another example of the ICC's proactive approach in situations of potentially controversial jurisdiction in situations of lack of consent, non-membership and disputed territories, is the assumption of jurisdiction and investigation into the situation in Palestine, in particular the areas of the West Bank, East Jerusalem and the Gaza Strip. Whilst Palestine is both a member state of the ICC and has also expressly accepted the Court's jurisdiction, Israel remains a non-member state. Consequently, continuing controversy exists over the ICC's assumption of jurisdiction over crimes committed on the disputed and/or occupied territories, in addition to the fact that Palestine's statehood itself (and therefore its capacity to accept jurisdiction) is not universally recognised.

-

¹¹⁸⁸ ICC Appeal's Chamber, Judgment on the appeal against the decision on the authorisation of an investigation into the situation in the Islamic Republic of Afghanistan, 5 March 2020, ICC-02/17 OA4, available at https://www.icc-cpi.int/CourtRecords/CR2020 00828.PDF, accessed 3rd September 2021.

¹¹⁸⁹ *Ibid*, paras 5, 20, 25, 45 and 46.

¹¹⁹⁰ See Pre-Trial Chamber I, Decision on the 'Prosecution request pursuant to article 19 (3) for a ruling on the Court's territorial jurisdiction in Palestine', 5 February 2021, ICC-01/18, available at https://www.icc.cpi.int/CourtRecords/CR2021_01165.PDF accessed 3rd September 2021.

Palestine accepted jurisdiction of the ICC on 1 January 2015 with regards to crimes committed since 13 June 2014 and acceded to the Rome Statute on 2 January 2015.

¹¹⁹² 22 out of 42 *amicus curiae* who submitted their views to the Pre-Trial Chamber argued that the conditions for the exercise of the Court's jurisdiction had, at the time of the decision (n. 1201), not been

These situations demonstrate that the ICC itself is prepared to challenge not only immunity, but even pierce the shield provided by the traditional jurisdictional parameters of state membership and consent to its own treaty, where international atrocity crimes of a *jus cogens* nature have been committed. They also confirm the expanding 'evolution' of individual accountability in international criminal law, as perceived by the dissenting judges in the ICJ's *Arrest Warrant* and *Germany v Italy* cases.¹¹⁹³

General state practice also corroborates this development of greater individual accountability for *jus cogens* crimes through the revival of universal jurisdiction at customary international law. The number of States claiming such universal jurisdiction over international crimes is ever growing. According to Amnesty International's 2012 report, 163 out of 193 UN member states acknowledge the potential exercise of universal jurisdiction over one or more crimes under international law. An example is the recent recommendation by the International Fact-Finding Mission on Myanmar set up by the Human Rights Council for individual States to resort to universal jurisdiction in order 'to investigate and prosecute alleged perpetrators of serious crimes under international law committed in Myanmar.' A first case was brought in Argentina in 2020, which is, at the time of writing, before the Court of Appeal in Buenos Aires. In a recent landmark

fulfilled (*ibid*, paras 51-52). For a critique of the PTC's decision, see for instance Kai Ambos, "Solid jurisdictional basis"? The ICC's fragile jurisdiction for crimes allegedly committed in Palestine, EJIL:Talk! 2 March 2021, https://www.ejiltalk.org/solid-jurisdictional-basis-the-iccs-fragile-jurisdiction-for-crimes-allegedly-committed-in-palestine/, accessed 3rd September 2021.

¹¹⁹³ See Judge van den Wyngaert's dissenting opinion in the *Arrest Warrant* case, n. 1152, and Judge Trinidade's dissenting opinion in the *Germany v Italy* case, n. 1162.

¹¹⁹⁴ Amnesty International Report on universal jurisdiction (n. 131). See also International Justice Resource Center, *Universal Jurisdiction* (n. 131).

¹¹⁹⁵ 'Compilation of all recommendations made by the Independent International Fact-Finding Mission on Myanmar', para 102, A/HRC/42/CRP.6, 16 September 2019, available at https://www.ohchr.org/Documents/HRBodies/HRCouncil/FFM-

Myanmar/201090916/A HRC 42 CRP.6.pdf, accessed 3rd September 2021.

1196 Tun Khin, 'Universal Jurisdiction, the International Criminal Court, and the Rohingyan Genocide',

Tun Khin, 'Universal Jurisdiction, the International Criminal Court, and the Rohingyan Genocide', OpinioJuris, 23 October 2020, available at http://opiniojuris.org/2020/10/23/universal-jurisdiction-the-international-criminal-court-and-the-rohingya-genocide/, accessed 5th September 2021.

decision, the High Regional Court of Koblenz, Germany, convicted a former member of the Syrian intelligence services at the beginning of this year for complicity in crimes against humanity and torture, exercising universal jurisdiction. Other cases against members of the Syrian regime have been brought in France and Sweden.

This evolution towards a de-emphasis of state consent and related procedural rules on immunity with regards to *jus cogens* violations strengthens the argument for wide jurisdiction of the ICC over these crimes, as uninhibited by procedural consent restrictions as the jurisdiction of domestic States under the principle of universality. The peremptory nature of the crimes, the teleological interpretation of the Rome Statute and its principle of complementarity reinforce this contention. Additionally, the argument has been presented that acts of aggression committed by leaders were prosecuted at Nuremberg under such delegated universal jurisdiction, ¹¹⁹⁹ subject to an 'international rule of law' and independent of state consent. ¹²⁰⁰ This precedent, therefore, lends extra support to the exercise of collective universal jurisdiction by a treaty-based international court, ¹²⁰¹ particularly over the crime of aggression.

Such universal jurisdiction over the crime of aggression, exercisable by the ICC where its member states are unwilling or unable to prosecute, ¹²⁰² may be capable of overriding

http://www.bbc.co.uk/news/world-europe-56160486, accessed 27 February 2021. See also Jennifer Triscone, 'Universal Jurisdiction, the only hope for prosecuting international crimes committed in Syria', 6 September 2021, Trial International, available at https://trialinternational.org/latest-post/universal-jurisdiction-the-only -hope-for-prosecuting-international-crimes-committed-in-syria/, accessed 8th September 2021.

¹¹⁹⁸ *Ibid*.

¹¹⁹⁹ See n. 1094 and related commentary.

¹²⁰⁰ Ibid

¹²⁰¹ See Scharf (2001) (n. 1095) 103-6, and Randall (n. 109), 804-6.

¹²⁰² To date 39 member states have ratified the aggression amendments, and a number of these have, since the Kampala Conference, incorporated domestic legislation, see www.crimeofaggression.info. Even as early as 2010, some countries had already enacted domestic legislation of universal jurisdiction over the

the greatly reinforced consent requirement within Article 15 *bis*, rendering the aggression provisions applicable to all member states despite the loopholes created. This possibility follows from the argument that the peremptory prohibition of aggression cannot be derogated from by treaty. ¹²⁰³ If this admittedly contentious supposition was successful, the exercise of universal jurisdiction by individual States could, therefore, revert back to the ICC, just as it reverted back to the Nuremberg tribunals and allow for jurisdiction over non-consenting states parties. It would also constitute an alternative to the requirement for a Security Council referral, where the strict double membership requirement has not been satisfied, and the effect would be to de-emphasise the currently great dependence of the Court on the Council. Universal jurisdiction under customary international law might, therefore, allow the ICC to adopt jurisdiction where it is otherwise restricted by its very own Statute insisting on membership and consent.

crime of aggression. See UN Secretary-General, Report of the Secretary-General Prepared on the Basis of Comments and Observations of Governments: The Scope and Application of the Principle of Universal Jurisdiction, at 29, UN Doc. A/65/181 (July 29, 2010).

¹²⁰³ See Article 53 VCLT (n. 32) and Conclusion 2 of the 2019 ILC Report (n. 39).

Conclusion

The fact is not denied that any exercise of universal jurisdiction over aggression is, considering all the effort expended at creating a highly limiting multiple consent requirement for the crime, likely to encounter fierce objections. States prosecuting domestically, as well as the Court itself, would be subject to accusations of attempting to circumvent or undermine the specific state consent provisos, which were deliberately sought by a powerful minority of States and incorporated in the amendment. Critics may also argue that strict interpretation in accordance with the ordinary meaning of a treaty provision, as per Article 31 (1) VCLT, 1204 and the exclusion of non-states parties, as per Art 34, 1205 counters the possibility of universal jurisdiction over the crime of aggression. Attempts to prosecute non-member states or non-consenting member states without a prior Security Council referral and under assumption of universal jurisdiction is, furthermore, likely to attract condemnation by defenders of extensive Security Council control. 1206

Nevertheless, it is contended that the prohibition of aggression is a peremptory norm under Art 53 VCLT, which supersedes the other treaty provisions, including insistence on state consent. Norms prohibiting international atrocity crimes of a *jus cogens* nature supersede other provisions and are accorded greater weight in the hierarchy of norms.¹²⁰⁷ The availability of universal jurisdiction over such crimes therefore survives regardless

. .

¹²⁰⁴ VCLT ((n. 32), Article 31 (1).)

¹²⁰⁵ Ibid, Article 34.

¹²⁰⁶ See the arguments considered in section 3 of Chapter 4.

¹²⁰⁷ See the dissenting opinions in the *Arrest Warrant* case, n. 1151 and n. 1152, and the *Germany v Italy* case, n. 1148.

of the attempts of exclusion.¹²⁰⁸ As has been argued above, this phenomenon can be interpreted as part of an evolution of modern international law,¹²⁰⁹ towards the greater normativity of *jus cogens* violations.¹²¹⁰ This development, in the words of Judge Trinidade, 'gradually resolves the tension between state immunity and access to justice rightly in favour of the latter, particularly in cases of international crimes,'¹²¹¹ leading to a more universal recognition of individual liability within international criminal law.¹²¹²

Evidence for such a trend can be seen all around, from the quest for impunity pursued by the first permanent international criminal court, declared as its *telos* in the Preamble and reflected in the recent decisions of this fledgling court, to the judicial practice of other international tribunals since Nuremberg and growing state recognition. Despite the likely objections, the potential application of universal jurisdiction offers at least a partial solution to the tightly restricted, marginalised jurisdiction of the Court as accepted at Kampala. Without this possibility of escaping the straightjacket of statutory restrictions, prosecutions for aggression involving non-member states in any capacity would effectively be barred, unless there is a Security Council referral. As the discussion showed, non-members, as a result of Article 15 *bis* (5), do not even enjoy protection

¹²⁰⁸ For support, see the fact that even the joint separate opinion in the *Arrest Warrant* hinted at this possibility (n. 1144), and to some degree also the conservative majority in the *Germany v Italy* case (n. 1148).

¹²⁰⁹ See the Joint Separate opinion in the *Arrest Warrant* case (n. 1144), and the dissenting opinion of Judge van den Wyngaert (n. 1152).

¹²¹⁰ For an elaboration on this evolution, evidenced by domestic jurisdictions but ignored by the ICJ in its consideration of case law in *Germany v Italy*, see Andrea Bianchi, n. 1184 at 474. He cites in particular a case by the High Court of Justice, Queen's Bench Division, Administrative Court, *Khurts Bat v The Investigating Judges of the German Federal Court*, Judgment of 29 July 2011 (2011) EWHC, 2029 (Admin) (denying immunity to foreign state officials for criminal charges in respect of acts committed in the territory of a third State while in post) and by the Swiss Federal Criminal tribunal in *A v Ministere Public de la Confederation*, 25 July 2012 (denying immunity *ratione materiae* to a former foreign state's defence minister for war crimes and acts of torture allegedly committed while he was in post), available at http://www.trialch.org/fileadmin/user_upload/documents/affaires/algeria/BB.2011.140.pdf, accessed 27 August 2021.

¹²¹¹ Dissenting opinion by Judge Trinidade, n. 1162, para 303.

¹²¹² *Ibid*.

¹²¹³ See the discussion presented above.

against aggression committed by signatories. 1214 The argument has been presented that this latter fact, on its own, may inadvertently operate as an incentive to join the Rome Statute, 1215 depending on the strength or vulnerability of a State relative to its neighbours. However, once the State in question has signed up to the treaty, the greatest protection is derived from non-acceptance of the amendment and/or opting out. The new member state would be immune from prosecution, under Article 15 *bis*, over their own acts of aggression whilst enjoying protection from aggression by other signatories who have ratified the amendments. This combination acts as a great disincentive for acceptance of jurisdiction, as a direct result of the insistence on multiple consent layers. It also counteracts any real chance of wide membership of the Court without reservations and the eradication of the crime of aggression.

Divorcing individual criminal liability from state responsibility by allowing States or their leaders to hide behind the shield of state consent to jurisdiction, or immunity, is contrary to the *telos* of the Rome Statute and its express aim of impunity. ¹²¹⁶ It also ignores the *jus cogens* nature of the Crime of Aggression. Elective impunity of States is likely to result in greatly emasculated jurisdiction rather than effective deterrence. The first permanent International Criminal Court has only very recently obtained jurisdiction, after nearly a century of attempts to eradicate the crime. To curtail its jurisdiction in quite such a comprehensive way, through reinforced insistence on state consent, is counterintuitive to the aims of the Rome Statute, and indeed the imperative need of the global community at large to eliminate the worst international crimes. A parallel can be drawn with the remarks

-

¹²¹⁴ See section 3 of this Chapter.

¹²¹⁵ Akande and Tzanakopoulos (2018) (n. 902), 958.

¹²¹⁶ See the Preamble, n. 85.

of the International Court of Justice in the *Armed Activities Case*, ¹²¹⁷ commenting on claims that its own jurisdiction over acts of genocide was limited due to lack of state consent. As stated in the separate joint opinion:

It is a matter for serious concern that at the beginning of the twenty-first century it is still for States to choose whether they consent to the Court adjudicating claims that they have committed genocide [or crimes of aggression]. It must be regarded as a very grave matter that a State should be in a position to shield from international justice scrutiny any claim that might be made against it concerning genocide [or aggression]. A State so doing shows the world scant confidence that it would never, ever, commit genocide [or aggression], one of the greatest crimes known. 1218

This statement is, it is argued, as applicable to the 'supreme' international crime of aggression, as it is to the crime of genocide. The combination of the wholesale exclusion of non-signatories contrary to Article 12 and the multiple consent requirement, even for signatories, ¹²¹⁹ make it difficult not to view the restrictions as a deliberate policy decision sacrificing impunity for the perpetuation of state sovereignty and consent. Despite the fact that jurisdiction has now, at least technically, been bestowed on the Court, the question arises whether the compromise leading to this limited jurisdiction has been achieved at too high a cost. The Kampala consensus and subsequent policy decision by the 2017 ASP may have placated sceptics of the Court's independence at least

-

¹²¹⁷ Armed Activities (2005) (n. 37 and 773), Joint separate opinion of Judges Higgins, Koojimans, Elaraby, Owada and Simma (para 25).

¹²¹⁸ *Ibid*. Text in brackets added.

 $^{^{1219}}$ Applying the negative interpretation of Article 121 (5) combined with Article 15 bis (4) (see section 2.5).

temporarily, but at the expense of widely restricting effective jurisdiction over the supreme international crime. In light of these extraordinary and tight restrictions incorporated into the amendments, the assumption of universal jurisdiction may, despite its contentiousness, be one of the very few ways of expanding jurisdictional reach over the crime in practice.

CHAPTER 6: TRANSCENDING THE

JURISDICTIONAL LIMITATIONS

1. Introduction

The evaluation of the jurisdictional limitations placed on the new Crime of Aggression has dissected the detailed provisions contained in Articles 8 *bis* and 15 *bis* and *ter*. Comparisons have been drawn with the historical precedents at Nuremberg, the definition of the state act in Resolution 3314 and the other crimes under the Rome Statute. It has also been attempted to draw conclusions on the insubstantial appetite of the international community since Nuremberg to expressly condemn such acts of aggression, as demonstrated by the political organs of the Security Council and the General Assembly and occasionally also the International Court of Justice. It has examination has shown that the new crime of aggression provisions have been extensively and deliberately restricted, providing for an extremely narrow definition, It great control by the Security Council over Court proceedings, even in ostensibly independent prosecutions, It and the strictest state consent requirements imaginable. So, where does this leave the new jurisdiction of the first permanent International Criminal Court over the Crime of

¹²²⁰ See Chapters 2 and 3.

¹²²¹ See section 5, Chapter 1 and section 2.1, Chapter 2.

¹²²² As argued in Chapters 2 and 3.

¹²²³ As argued in Chapter 4.

¹²²⁴ See Chapter 5.

Aggression? How might these limitations placed on the Court's jurisdiction be surmountable, in order to be at all effective in a modern context? In order to provide suggestions of how to broaden jurisdiction, whether they are entirely realistic or not, let us first briefly sum up the specific problems encountered in this analysis, and the arguments presented.

2. Revisiting the need to reconceptualise the definition

This leads to undesirable delays even before prosecution of a crime can begin, since it is a precondition, which has to be established before the start of proceedings, but also constitutes an element of the crime which has to be proven again during the trial. The intention of negotiators was to exclude humanitarian interventions, however the argument has been presented that the *mens rea* threshold of intent and knowledge to commit an act of aggression, derived from Article 30 of the Rome Statute, is unlikely to have been crossed in such interventions. ¹²²⁶ Additionally, Article 1 (1) of the Rome Statute limits prosecutions to the most serious international crimes only, and an intervention for humanitarian purposes is not likely to be viewed as such by either the Prosecutor, who has to decide whether a prosecution would serve the interests of justice in accordance with Article 53 (1), ¹²²⁷ or the Pre-Trial Division, which serves as an authorisation 'filter', by virtue of Article 15 *bis* (8). ¹²²⁸ Furthermore, the Security Council is entitled to a

¹²²⁵ Article 8 bis (1) of the Rome Statute.

¹²²⁶ See section 2.2 in Chapter 2.

¹²²⁷ See Article 53 (1) Rome Statute (see n. 453 and section 2.4 Chapter 2).

¹²²⁸ Article 15 *bis* (8).

consideration period of six months, 1229 and can halt investigations and prosecutions by the Court. 1230

The argument has also been presented that a genuine humanitarian intervention is not inconsistent with the UN Charter, and there was therefore no need to insist on such a high threshold in the first place. 1231 The Court should be entitled to use their discretion in order to distinguish acts of aggression incrementally, however this view is likely to encounter accusations of politicisation, interference in matters of peace and security, violation of the Vienna Convention on the Law of Treaties, as well as the prohibition of analogy contained in Article 22 (2) of the Rome Statute itself. 1232 Nevertheless, the inherent safeguards within the Rome Statute ought to be sufficiently effective to prevent abuse. Furthermore, incremental clarification, careful interpretation and close analogy are necessary to make sense of the aggression provisions in a modern context, and achieve greater applicability, at a time when the crime of aggression may take on many different guises not envisaged in the definition. 1233 Enforcement of the aggression provisions subject to these considerations would also place renewed emphasis on the general prohibition of the use of force contained in Article 2 (4) of the UN Charter. 1234

The second problem arising from the definition is the limitation to political and military leaders of a State only. 1235 It was noted that this is a substantial retreat from the Nuremberg prosecutions, which also considered the culpability of economic and

¹²²⁹ *Ibid*.

¹²³⁰ Article 16 of the Rome Statute (n. 739).

¹²³¹ Chapter 2.

¹²³² Section 2.4 in Chapter 2.

¹²³⁴ *Ibid.* For Article 2 (4) of the UN Charter see n. 195.

¹²³⁵ Chapter 2.

industrial leaders. Moreover, Nuremberg set the precedent of a wider 'shape or influence' standard. 1236 If a leader was capable of shaping or influencing the policies of the aggressive State, they were deemed capable of attracting individual criminal liability, if their actions led to an act of aggression. 1237 The much higher 'control or direct' test, which has now been adopted for the crime of aggression, was derived from the ICJ's consideration of state responsibility in *Nicaragua*, which did not contemplate individual criminal liability. 1238 The negotiating history demonstrated that the SWGCA was misguided in their interpretation of the test at Nuremberg as one of 'control or direct', despite their own highly elaborate Historical Review, which time after time confirmed the 'shape or influence' standard. 1239 The higher standard was incorporated into the Kampala Amendments, even though the 2006 Princeton Report acknowledged the culpability of non-military and non-political leaders capable of shaping and influencing a State's policies, and despite the inclusion in the 2009 Report of the Special Working Group of 'non-governmental actors, for instance high-level industrialists'. 1240

The practical effect is that culpable influential war mongers, spin doctors, religious leaders and economic leaders profiteering from aggressive wars, are not captured, unless the Court applies wide judicial discretion to decide otherwise. Some support for the inclusion of non-military and non-governmental actors can also be derived from the crime of genocide, which considers culpability of actors in a lesser capacity who incite the commission of the crime. The modern accessory liability doctrine of Joint Criminal

¹²³⁶ See section 4.3, Chapter 1 and section 3.1 and 3.2, Chapter 2.

¹²³⁷ *Ibid*.

¹²³⁸ *Ibid*.

¹²³⁹ n. 534, Chapter 2.

¹²⁴⁰ See Chapter 2, n. 555.

¹²⁴¹ See Article 6 Rome Statute, particularly in combination with Article 25 (3). See also *ICTR v Akayesu* (n. 558) and related discussion in Chapter 2, section 3.4 (p. 134).

Enterprise (JCE), involving a 'common plan,[...] which amounts to the commission of a crime' 1242 as well as a shared intent, may also provide a path to wider application. To that effect, the discussion considered the argument that Article 25 (3) d) Rome Statute includes JCE generally, and consequently JCE ought to be read into the crime of aggression as well. Some views even contended that the entire future of the crime of aggression lies within JCE. All of these suggestions, however, require ignoring the fact that JCE appears to have been deliberately excluded by Article 25 (3) *bis*, and critics also counter the suggestion of including JCE as offending the legality principle. Nevertheless, this thesis argues that if the common purpose is so obvious that the culpability of the defendant's behaviour ought to have been foreseeable, this should provide a sufficient basis for the Court's jurisdiction. Ave been greater 'retreat from Nuremberg'. 1247

A third limitation on the jurisdiction of the Court arises from the fact that the definition currently does not extend to include non-state actors and acts of aggression committed by and against non-state entities. Consequently, the definition may not be equipped to consider modern threats which have, over the past 20 years, often involved non-state actors. Similarly, proxy wars, external interference in civil wars and foreign state support of non-state groups raise problems of clear attribution. Modern warfare is also

1

¹²⁴² See *Tadic* (IT-94-1-A) (n. 115), Appeals Chamber Decision 15 July 1999, paras 193, 200 and related discussion in section 3.4 of Chapter 2.

¹²⁴³ See section 3.6, Chapter 2.

¹²⁴⁴ See particularly Weisbord's arguments, pp. 138-9 and section 3.6, Chapter 2.

¹²⁴⁵ See Heinsch and Osiel, pp. 139-40 (n. 579 and 580).

¹²⁴⁶ See the argument presented in Chapter 2, section 3.4.

¹²⁴⁷ Petty (2008) (n. 531 and related arguments).

¹²⁴⁸ See Chapter 3, section 2.

¹²⁴⁹ *Ibid*.

¹²⁵⁰ *Ibid*.

often asymmetric and decentralised. The state element within the crime of aggression definition therefore requires reinterpretation to include groups or state-like entities, and render it capable of capturing political-military organisations other than States. ¹²⁵¹ The potential extension of liability to such organisations would widen the Court's jurisdiction, however such reinterpretation is also likely to raise the familiar legality concerns which pervade any attempt to extend jurisdiction to situations other than the very limited circumstances in which jurisdiction of the Court was envisaged. ¹²⁵²

Furthermore, the Court's jurisdiction over the crime of aggression is greatly restricted by the limited list of 1974-style traditional warfare scenarios. Whilst Article 4 of Resolution 3314, from which the crime of aggression is largely derived, had provided for the determination of other acts, rendering the list non-exclusive, Article 8 *bis* includes no such provision for extension to other types of aggression. State proposals, which attempted to include a provision that the list was to serve merely as examples of acts and could be extended, were ignored. Even the SWGCA's own suggestion of a compromise between a non-exhaustive list and a chapeau, for the purpose of reigning in excessively wide interpretations whilst still providing some flexibility, were unsuccessful.

As a result, it is now extremely difficult to extend the current definition to include modern and emerging threats of systems disruptions, sabotage, cyber warfare, terrorism, guerrilla warfare, and information warfare including the spreading of fake news etc. This fact leads

¹²⁵¹ *Ibid*.

¹²⁵² *Ibid*.

¹²⁵³ Section 3, Chapter 3.

¹²⁵⁴ Ibid

¹²⁵⁵ *Ibid*, Proposals by Colombia, Greece and Portugal (n. 641, 643) and related text.

¹²⁵⁶ See the 2006-7 *Princeton Reports* (n. 416 and 552).

to the realisation that wider interpretations are required in order to render the crime of aggression fit for prosecutions in a modern context. Support for such wider interpretations can also be gained, for instance, from the fact that a number of new projects have recently been mandated by the General Assembly in order to establish the application of international law to cyberwarfare, ¹²⁵⁷ and attempt to assess how cyber threats impact on international peace and security. ¹²⁵⁸ Additionally, the Permanent Mission of Liechtenstein with other partners, has created a Council of Advisers, specifically in order to investigate the application of the Rome Statute to cyberwarfare. ¹²⁵⁹ These ongoing projects clearly demonstrate the increasing recognition that international law and practices need to be adapted in order to apply to modern threats to peace and security.

The recognition of a requirement to apply existing law to new and developing situations also provides support for wider judicial determinations, despite likely criticism of offending the principle of *nullum crimen sine lege*. Concerns of judicial interference in political matters are, this thesis argues, exaggerated, since the Court is required to determine the existence of an act of aggression in each situation in any case, in order to establish individual criminal liability. Determinations of state acts and the exact type of individual behaviour amounting to aggression, are partial elements of the crime, ¹²⁶⁰ and therefore incidental to the prosecution of aggression without necessarily evidencing politically motivated judicial interference and undue expansion of existing law. Support for a presumption of inherent competence and independent determination powers of the

Open-ended working group (OEWG) on developments in the field of information and telecommunications in the context of international security. See n. 658.

¹²⁵⁸ Group of Governmental Experts (GGE) on Advancing Responsible State Behaviour in Cyberspace in the Context of International Security. See n. 659.

¹²⁵⁹ The Council of Advisers on the Application of the Rome Statute to Cyberwarfare, n. 660.

¹²⁶⁰ As stipulated in Article 8 *bis*.

Court can, the argument was presented, be derived from *Nicaragua*, ¹²⁶¹ and the *Reparations* case, ¹²⁶² as later applied by the ICTY in *Tadic*. ¹²⁶³ Additionally, the treatment of the other crimes in the Rome Statute also provides evidence for permissible judicial interpretation. ¹²⁶⁴

An alternative possibility could be to simply view the list as non-exhaustive, despite the appearance to the opposite. ¹²⁶⁵. This is based on the argument that the exclusion of Articles 2 and 4 of Resolution 3314 merely restricts determination of other acts and consideration of other grounds by the Security Council, not the Court itself. ¹²⁶⁶ Following this argument, the Court can draw on Resolution 3314 as a whole, and neither Article 22 (2) nor the legality principle require a reading of the list as exclusive of other acts. ¹²⁶⁷ Here again arguments can, therefore, be presented for careful, incremental interpretations by the Court, as long as the principle of *nullum crimen sine lege* is not blatantly offended, and the analogy is sufficiently close to have been foreseeable. ¹²⁶⁸ A number of safeguards are in place to prevent overly wide interpretations, for instance the limitation of the Statute to the worst international crimes only, the high threshold of the crime, the filter of the Pre-Trial Division, and the ability of the Security Council to halt investigations and prosecutions under Article 16. ¹²⁶⁹

¹²⁶¹ *Nicaragua* case (1984), (n. 827), para 96, in which the ICJ stated that it had 'never shied away from a case brought before it merely because it had political implications.'

¹²⁶² Reparations case (1949) (n. 844). Here the ICJ held that where an international organisation had been instructed with a certain task, it had also been 'clothed [...] with the competence required to enable those functions to be efficiently discharged.'

¹²⁶³ See the discussion of *Tadic* (n. 115 and 838, at para 18) and section 3.4 of Chapter 2. The ICTY, citing the *Certain Expenses* case (1962) at p. 163 (n. 765), considered itself competent and 'duty-bound' to exercise its jurisdiction 'regardless of the political background or the other political facets of the issue'.

¹²⁶⁴ See for instance Article 7 (1) (k) of crimes against humanity.

¹²⁶⁵ See for instance, Kress (2017), n. 614 in Chapter 3 and related text, as well as section 3.2 and particularly pp. 164-5.

¹²⁶⁶ *Ibid*.

¹²⁶⁷ *Ibid*.

¹²⁶⁸ Ibid.

¹²⁶⁹ See the discussions in Chapters 2 and 3.

Without such incremental extension, the crime of aggression is likely to be irrelevant in a modern context, at the very time when jurisdiction by a permanent international Court has only just been established. The only other option would be to re-write the definition entirely, however the negotiating history has demonstrated that agreement and consensus by States is difficult to achieve. Any amendment would have to be subject to the amendment procedure in Article 121 (3), stipulating acceptance by two thirds of the Court's members. In light of the protracted negotiations and the jurisdictional limitations intentionally sought at the insistence of powerful States, resulting in near failure at Kampala, 1272 the predictions for finding easy agreement on a wider definition and expansive jurisdictional reach are not promising. On the other hand, the developing recognition of the need to match current international criminal law to modern threats like cyberwarfare, 1273 may operate to put some pressure on States to reconsider their position and accept such an extension.

3. Revisiting the stringent jurisdictional restrictions

In addition to the definitional limitations, the examination also revealed that the jurisdictional conditions contained in Articles 15 *bis* and *ter* add substantially to the restrictions on the Court's independent jurisdiction over the crime of aggression. Thus, the level of Security Council control over aggression proceedings in situations where the

-

¹²⁷⁰ Here, a conclusion by the Council of Advisers (n. 660), the OEWG (n. 658) or the GGE n. 659) that the crime of aggression ought to apply to cyberwarfare would be extremely useful, and might even negate the need to rewrite the definition as it stands, as long as the Court feels free to take note of this potential conclusion.

¹²⁷¹ See Article 121 (3) Rome Statute (n. 972).

¹²⁷² See Kress and von Holtzendorff (2010) (n. 5).

¹²⁷³ See for the OEWG, GGE and Councils of Advisers (n. 658-60).

referral emanates from a member state or the Prosecutor's own powers, is remarkable. ¹²⁷⁴ Security Council referrals are subject to consensus by the P5, and if any of those five States withhold what effectively amounts to their consent, no referral ensues. ¹²⁷⁵ The powerful States, in particular, consider the prerogative of the Security Council over aggression proceedings to be exclusive, ¹²⁷⁶ and underlying reasons for continuing insistence of Security Council monopoly are the desire to preserve the existing power structure and protect their sovereign interests. ¹²⁷⁷ Their defence of this view, however, ignores evidence that the determination power of the Security Council is, and always has been, shared, ¹²⁷⁸ and the privilege of the Council over aggression is not exclusive of determinations by other organs. ¹²⁷⁹ This was confirmed for example by the ICJ in *Nicaragua*, ¹²⁸⁰ and *Armed Activities*, ¹²⁸¹ and also a number of General Assembly resolutions. ¹²⁸²

Some delegations of States and academics also expressed strong preference for Security Council scrutiny in order to protect the Court from accusations of politicisation and States' abuse of aggression referrals to the Court. Other arguments defend significant Security Council involvement, because it extends jurisdiction to situations in which the Court would otherwise have none. This view is based on the fact that treaties bind only consenting signatories, and Security Council involvement is regarded as a way to achieve

¹²⁷⁴ See the discussion in Chapter 4.

¹²⁷⁵ *Ibid*

¹²⁷⁶ See the post-Kampala Explanations of Positions in the *Travaux Preparatoires* (2012), Barriga and Kress (eds.) (n. 722-3) from p. 810.

¹²⁷⁷ See Chapter 4, sections 2-3.

¹²⁷⁸ *Ibid*.

¹²⁷⁹ *Ibid*.

¹²⁸⁰ See the determination of acts of aggression by the ICJ in *Nicaragua* (1986) (n. 37), (1984) (n. 827).

¹²⁸¹ See the determination of acts of aggression in *Armed Activities* (2005) (n. 37).

¹²⁸² Through, for instance, application of the U4 P resolution, n. 253. For a discussion, see section 2.2 and pp. 189-90 in Chapter 4.

¹²⁸³ Chapter 4, section 3.2.

jurisdiction over non-signatories or non-consenting state parties. ¹²⁸⁴ Even though this fact is not disputed, such extension of jurisdiction to non-signatories is only likely to arise if a State is unpopular and does not enjoy the support of at least one P5 member, who could prevent a referral by use of their veto power.

The most radical argument for Security Council exclusivity encountered during the research for this thesis asserted that Security Council powers over the crime of aggression ought to be extended even more, by granting the Council the power to vacate charges before the Court and expunge convictions. This thesis strongly refutes this argument. The extension and exercise of such power would amount to unacceptable political interference in criminal proceedings and the judicial process, as well as prevent any future deterrence from committing the crime. Jurisdiction of the Court would be rendered a farce. The contention of inherent competence of the Court, reiterated above in the context of wider judicial determinations of other acts of aggression, has been defended throughout this discussion, and gains support from a similar assumption in Article 36 of the ICJ Statute, as well as case law. Furthermore, it has also been contended that Article 8 bis requires the ICC to make its own determinations of a State act as an element of the crime.

The two organs of Court and Council are separate entities, and decisions by the Security Council should not be prejudicial to the Court's determinations, a principle which has in

¹²⁸⁴ Chapter 4, section 3.3.

¹²⁸⁵ See Stein's arguments (n. 811 and pp. 198-200 in Chapter 4).

¹²⁸⁶ See the counterarguments presented in section 3.1 of Chapter 4, from p. 200.

¹²⁸⁷ Ibid

¹²⁸⁸ *Ibid.* For the ICJ Statute, see n. 35.

¹²⁸⁹ *Ibid.* See also n. 1295-6 above.

¹²⁹⁰ See pp. 201-3, Chapter 4.

fact been expressly affirmed in Articles 15 *bis* (9) and 15 *ter* (4).¹²⁹¹ *Ex post facto* determinations by the Security Council as a political organ, if viewed as binding by the Court, raise serious legality concerns and impede the rights of the defendant in a criminal trial. ¹²⁹² Consequently, it is the very requirement to abide by the legality principle, which requires the Court to make these determinations without prejudice and independently, based on the facts of each case, whilst avoiding politicisation. In order to do this, the Court requires flexibility and independent determination powers.

In addition to Security Council control, the specific state consent requirements incorporated into the crime of aggression provisions constitute yet another fundamental restriction placed on the jurisdiction of the Court. Thus Article 15 bis (4) specifically allows for member states to opt out of jurisdiction. This acknowledgment of a member state's reservations is contrary to Article 120 and the jurisdictional regime for the other crimes under the Statute in Article 12 of the Rome Statute. The argument has been presented that the combination of Article 15 bis (4) with Article 121 (5) was an irrational choice. Article 121 (5) requires ratification as an expression of consent, whereas Article 15 bis (4), specifically adopted for the crime of aggression, requires a choice not to opt-out as the necessary expression of consent to jurisdiction, irrespective of ratification. State practice appears to confirm application of Article 15 bis (4) over that of Article 121 (5).

¹²⁹¹ See section 4 in Chapter 4.

¹²⁹² Ibid.

¹²⁹³ See the discussion in Chapter 5.

¹²⁹⁴ Ibid.

¹²⁹⁵ See Chapter 5, section 2.5. See also Donald Ferencz' assessment (n. 1016 and related text in Chapter 5), and the 'Statement by Japan', accusing the negotiators of 'cherry-picking' (n. 954) and related discussion.

¹²⁹⁶ Chapter 5.

¹²⁹⁷ See Kenya's and Guatemala's choice to opt out without having ratified first (n. 1013).

The combination of Article 121 (5) with Article 15 *bis* (4), exceptional by itself, creates in fact a triple consent requirement. Thus, States express their consent to jurisdiction by being or becoming a signatory of the Rome Statute, additionally ratifying the amendment, due to the decision to apply Article 121 (5), and failing to opt out. ¹²⁹⁸ Since the decision to apply Article 121 (5) on its own was controversial at Kampala, no decision was made at the time as to whether the article was to be interpreted with a negative or a positive meaning, i.e. whether just one of the member states had to have ratified the amendment, or whether both victim and aggressor state had to have ratified the amendment. The 2017 ASP decided in favour of the negative interpretation, which narrows jurisdiction of the Court even further. ¹²⁹⁹ This interpretation is contrary to the first operative paragraph of the Kampala amendment in Resolution 6, ¹³⁰⁰ and also acts as a perverse incentive to join the ICC and ratify, but then opt out of the aggression amendments. This combination gives protection against aggression by another ratifying member state, but shields a State from prosecution for their own acts of aggression under Article 15 *bis*.

The negative interpretation of Article 121 (5) and combination with Article 15 *bis* (4) provides yet another exception to the normal jurisdictional procedure under Article 12 of the Rome Statute. If allowed to stand, the negative application of Article 121 (5) renders the special procedure of Article 15 *bis*, arrived at after difficult and lengthy negotiations, irrelevant. The practical effect is that jurisdiction under Article 15 *bis* arises now only if one of the 39 ratifying States commits aggression against another of those ratifying 39

¹²⁹⁸ Chapter 5.

¹²⁹⁹ Ibid

¹³⁰⁰ Chapter 5, section 2.4. For Resolution 6 see n. 2.

¹³⁰¹ See Ambassador Wenaweser's statement for Liechtenstein (n. 1050) and related text in Chapter 5. See also Coracini, who describes it as 'close to an annihilation' of the negotiated Kampala compromise, n. 1051, at 773.

States. The majority of States' opinion, as expressed after the ASP, does not support this interpretation, ¹³⁰² notwithstanding its insistent pursuit by the small but powerful minority group of 'camp consent'. ¹³⁰³ In addition to the evidence of dissenting state opinion, the ASP's narrow interpretation can also be challenged on the grounds that their decision is not actually capable of amending Article 15 *bis* (4) itself, and its legal validity is therefore questionable. ¹³⁰⁴ These arguments provide support for a future decision by the Court to a different effect. Furthermore, the addition of paragraph 3 of the 2017 activating resolution, which reemphasises the Court's judicial independence, effectively supports the notion that the Court may be free to come to a different decision than the ASP and apply the positive meaning of Article 121 (5) after all. ¹³⁰⁵

The final limitation on the Courts jurisdiction, considered in this examination, is the fact that Article 15 *bis* (5) provides a blanket exemption for non-member states in any capacity. This paragraph also represents a highly unusual exception to the normal jurisdictional procedure in the Rome Statute, and excluding jurisdiction even where a non-member state is the victim of aggression by an accepting states party. This provision creates asymmetric results to the effect that a non-consenting states party is afforded the greatest protection for their acts of aggression. Ultimately, the jurisdictional limitations placed on the crime of aggression, in their entirety, create an overall perception of impunity rather than greater accountability at international criminal law, as a result of the extraordinary emphasis placed on state consent.

¹³⁰² See the discussion in section 2.6, Chapter 5.

¹³⁰³ *Ibid*.

¹³⁰⁴ See Kress (2018) (n. 993), 8.

¹³⁰⁵ See n. 1057 and the discussion in section 2.7 of Chapter 5.

¹³⁰⁶ See section 3, Chapter 5.

¹³⁰⁷ Article 12 (2) of the Rome Statute.

¹³⁰⁸ See sections 3 and 4, and p. 291, Chapter 5.

4. Proposals to overcome the difficulties

4.1 Rewriting aggression, room for analogy and determinative independence of the

Court

This summary of problematic issues has highlighted that, ideally, the definition of aggression and its jurisdictional conditions ought to be re-written, in order to enable it to capture modern acts of aggression and modern types of aggressors. Explicit inclusion of other specific acts, for instance cyberwarfare, would be the only absolutely certain way to avoid the possibility of offending the legality principle. ¹³⁰⁹ Support can be gained from the fact that projects to determine the relevance of existing international law to cyber threats, ¹³¹⁰ and specifically application of the Rome Statute to cyberwarfare, ¹³¹¹ are underway. Nevertheless, it may not be possible to foresee today all of the future ways in which the crime of aggression may be committed, on account of constantly developing technology. Building a certain amount of flexibility into the definition would, therefore, serve the purpose of covering a wider range of acts better. Sceptics will, quite understandably, continue to raise concerns about the need for certainty, in order to satisfy the legality requirements of criminal law. ¹³¹² Nevertheless, this thesis argues that if an act of aggression, by analogy, was that close, or its effects were so devastating that they ought to have been foreseen, the actus reus can be established. Otherwise, the attempt to criminalise aggression is, in a modern context, pointless. The use of analogy is,

¹³⁰⁹ See Chapter 3.

¹³¹⁰ See the work of the OEWG (n. 658) and GGE (n. 659).

¹³¹¹ See the work of the Council of Advisers (n. 660).

¹³¹² See the arguments above, and in Chapter 3.

admittedly, far more likely to be acceptable to States with common law jurisdictions than those subject to civil law, on account of the fact that judicial interpretations are less commonplace in the latter. 1313

An argument can also be presented that it would also be beneficial for certain exceptions to be included explicitly, for instance self-defence, corresponding to Article 51 of the UN Charter, 1314 and genuine humanitarian interventions involving concerted efforts to alleviate suffering, improve humanitarian welfare and rebuild essential infrastructure. 1315 Distinguishable as an illegal use of force, on the other hand, ought to be actions for the political purpose of regime change, undertaken under humanitarian pretext but lacking evidence of substantial humanitarian efforts post-intervention. 1316 In addition to self-defence and genuine humanitarian intervention, it may also be possible to envisage a new defence of necessity for survival, on account of increasing global shortages of essential natural resources, i.e. water or food, as a result of climate change. This issue has been recognised as a matter of serious concern to the United Nations, and it is now reasonably foreseeable that future wars may be conducted over such essential resources. 1317 The ILC has considered the subject of necessity in the context of humanitarian interventions and in its projects on the Responsibility for Internationally Wrongful Acts, 1318 however such a defence is currently not considered in the crime of aggression provisions. In 2001, the

_

http://www.unwater.org/publication/world-water-development-report/2019/. See also the warning issued in *The State of Food and Agriculture* (SOFA) 2020, report, available at

http://fao.org/3/cb1447en/online/cb1447en.html, accessed 21st January 2021.

¹³¹³ See for instance Giacomo Ponzetto and Patricio Fernandez, 'Case Law versus Statute Law: An Evolutionary Comparison' (2008), 37 (2) JLS, University of Chicago Press, 379, 382.

¹³¹⁴ Article 51 of the UN Charter (n. 249).

¹³¹⁵ Chapter 2, section 2.2.

¹³¹⁶ *Ibid*.

¹³¹⁷ See UN World Water Development Report 2019, available at

¹³¹⁸ Report of the International Law Commission on the Work and Its Thirty-Second Session, UN Doc. A/35/10 (1980), and Draft Articles on Responsibility for Internationally Wrongful Acts with Commentaries, UN Doc. A/56/10 (2001) (see also n. 522 and 620), Article 25, 80.

ILC defined necessity as: 1) essential interest; 2) grave peril; 3) no alternative; 4) balancing of interests; 5) no violation of a peremptory norm or treaty; and 6) no fault of the State undertaking the intervention in bringing about the humanitarian crisis. These criteria have been developed specifically in order to distinguish humanitarian interventions from aggression, however the question arises whether similar criteria may, in exceptional circumstances, justify the violation of another State's territory in order to gain access to resources essential for survival. It may be possible to argue that such an ostensible act of aggression is distinguishable by not constituting a manifest violation of the UN Charter, similar to the argument presented with regards to external interventions in Chapter 2. The subject is, however, under-explored and the necessity defence would pose a pertinent and interesting subject for separate research.

These potential exemptions are mere examples, and re-writing the definition to accommodate new acts of aggression, and possibly also new defences, would require extensive research and canvassing of experts in modern technological warfare, international humanitarian lawyers, NGOs and regional organisations, to name just a few. Furthermore, a proviso ought to be added here that achieving consensus within the membership of the Rome Statute on a completely revised definition of the crime and jurisdictional conditions would, in all likelihood, be extremely difficult, considering the negotiating record to date. As mentioned above, the amendment procedure in Article 121 (3) of the Rome Statute requires a minimum of a two thirds majority if consensus cannot be achieved, 1320 but even this threshold may, in reality, be too high.

¹³¹⁹ For a discussion, see for instance Christopher DeNicola, 'A Shield for the "Knights of Humanity": The ICC should adopt a Humanitarian Necessity Defense to the Crime of Aggression', (2009) 30 (2) U. Pa. J. Int'l. L., 641, 674-681.

¹³²⁰ See Article 121 (3) of the Rome Statute.

Leaving these practical difficulties to one side for the time being and continuing the list of desiderata, rewriting the definition of aggression should also entail providing for wider and independent jurisdiction of the Court, in order to avoid elective impunity. ¹³²¹ If non-jurisdiction due to the lack of consent of an individual P5 member, attempting to protect their own interests or those of an ally, is to be avoided, the tight control of the Security Council over aggression proceedings must be relaxed.

The development of a more organic, forward looking definition and jurisdictional approach should also be accompanied by a change of view of the Court as an independent partner of the Security Council, ¹³²² rather than as the usurper of the Council's political privileges, a perception encountered in the post-Kampala statements particularly by the P5. ¹³²³ Following this partnership approach, and notwithstanding the independence of the two organs, judicial decisions may provide persuasive evidence for the Council to take political action against a State.

Such a symbiotic relationship is not unprecedented. Thus, Article 36 (3) of the UN Charter provides for a similar relationship of the Council with the ICJ, enabling the Council to settle international disputes utilizing the ICJ's jurisdiction. Article 96 (1) of the Charter provides for the ability of the Council to seek advisory opinions on legal questions within its work. States fail to comply with ICJ judgments, Article 94 (2)

¹³²¹ See for instance the arguments made by Weisbord in his trilogy of articles (2008 (n. 176), 2009 (n. 495) and 2011 (n. 430)).

¹³²² See the argument made by B. Ferencz (2008) (n. 831) stating that it is imperative to regard the Security Council as the operational partner of the ICC rather than its adversary. See also Cassese's interesting view that political institutions ought to take note of judicial decisions, whereas courts are not bound by political determinations (2007) (n. 614), 25.

¹³²³ See the 'Explanations of Positions' in the *Travaux Preparatoires*, (n. 317) from p. 810.

¹³²⁴ UN Charter (n. 134), Article 36 (3).

¹³²⁵ *Ibid*, Article 96 (1).

of the Charter places a duty on the Council to address any relevant non-compliance. ¹³²⁶ Admittedly, as recently acknowledged in the January 2017 Security Council Report, the tool of cooperation between the Council and the ICJ has, in the past, been underused, ¹³²⁷ however this does not detract from the fact that the Charter envisaged and encouraged such cooperation, and a similar relationship between the Council and the ICC would be highly beneficial, enhancing credibility of both organs. ¹³²⁸

The ICC will have had to apply a high legal threshold in their consideration of the facts establishing the existence of an act of aggression, and their findings may consequently be helpful to the Council. This partnership approach also requires the Security Council to deal with requests by the prosecutor as a matter of priority and endeavour to cooperate with the Court, by enforcing decisions. Unity within the Council would certainly send a strong message to aggressors. As expanded on in the following section, the perception of an onus of the Council to take note of judicial decisions should also require justifications of a P5 member for the use of their veto and denial of consent to a determination of aggression.

¹³²⁶ *Ibid*, Article 94 (2).

December 2016, https://www.securitycouncilreport.org/monthly-forecast/2017-01/in hindsight the security council and the international court of justice.php, accessed 3rd March 2021. Thus, the only time the Council utilized Article 36 (3) UN Charter, was in the early *Corfu Channel* case, through Resolution 22 (UN Doc. S/324) (9 April 1947). An advisory opinion under Article 96 (1) has been sought only once in Resolution 284, concerning South Africa's presence in Namibia (UN Doc. S/RES/284) (29 July 1970). As for Article 94 (2), Nicaragua requested such enforcement of an ICJ judgment against the US in 1986, but, unsurprisingly, the draft resolution (S/18428 of 28 October 1986) was vetoed by the US.

¹³²⁸ See the conclusions of the Security Council Report, *ibid*: 'The potential usefulness of the ICJ to the work of the Council should not be overlooked, and interaction with the Court, as envisioned by the UN Charter, could regularly be considered....A more prominent role for the Court with respect to the Security Council's work, whether on the Council's initiative or otherwise, would likely strengthen and enhance the legitimacy of the Council as an institution.' This statement is, it is contended, equally applicable to the ICC. ¹³²⁹ See the arguments by Ferencz (2008) and Cassese (2007) (n. 1337).

4.2 The duty not to withhold consent

The argument that States' sovereign rights to consent to jurisdiction ought to be viewed as an obligation instead, owes a great debt to Dworkin's reinterpretation of a sovereign right to consent as a sovereign duty, as encountered in the Introduction to this thesis. 1330 Similarly, the right of a P5 member to grant their consent through casting a decisive vote, i.e. the right to veto, could and should be viewed as a duty not to withhold their consent unduly. The strongest support for such a perception of a duty not to withhold consent by mala fide use of the veto is provided by the UN Charter itself. ¹³³¹ Thus, Article 2 (2) poses a duty on member states of the United Nations to 'fulfil in good faith the obligations assumed by them', 1332 and the Security Council exercises their responsibility on behalf of these members. 1333 Article 24 (2) stipulates explicitly that this responsibility of the Council includes acting 'in accordance with the Purposes and Principles of the United Nations', ¹³³⁴ and the primary purpose, as established in Article 1 (1) is the maintenance of international peace and security. 1335 From the combination of these articles, therefore, a clear onus is placed on Security Council members, particularly the P5, to exercise their responsibility in good faith, and that duty includes not withholding their consent through improper use of the veto. 1336 If consent is withheld, solid justification ought to be required. The perception of an onus rather than a right to consent would thus lead to pressure on the Council to come to unanimous and resolute decisions, aid deterrence from acts of aggression, and further eventual eradication of the crime.

¹³³⁰ See Dworkin (2013) (n. 43) and the arguments on pp. 16-9.

¹³³¹ Although Article 27 (3) (n. 210) admittedly does not on its own require bona fide exercise of the right.

¹³³² UN Charter (n. 134), Article 2 (2) (emphasis added).

¹³³³ *Ibid*, Article 24 (1).

¹³³⁴ *Ibid*, Article 24 (2).

¹³³⁵ *Ibid*, Article 1 (1).

¹³³⁶ For support, see Carswell (n. 213), 470.

Furthermore, the ultimate purpose of particularly the Security Council is the maintenance of peace and security for the benefit of the international community. 1337 Their work should therefore aid promotion of a 'minimum world public order', as envisaged by McDougal and Feliciano. 1338 To that effect, the argument encountered in the Introduction may be recalled that the 'economy in the use of force' constitutes a generally applicable rule of coexistence. 1339 Following this interpretation, the prescription of aggression is in the interest of the international community at large, and is consequently, once again, transposed into a duty of States. Interpreting the specific privilege of the P5 States as a specific duty instead, also supports the argument presented for abstention from use of the veto, or a requirement of substantive justifications if the veto is applied. The likely result would be greater cooperation within the Council to react to threats to the peace coherently, decisively, and in a timely manner, including referring a situation to the Court where appropriate.

It is consciously admitted that these ideas for overcoming the limitations of the crime of aggression very much represent a wish list. In reality, past evidence of Security Council reluctance to make determinations of acts of aggression has demonstrated that the Security Council is prone to inactivity and paralysis, particularly if the interests of the P5 or one of their allies are involved. The voting structure within the Council effectively requires the consent of the P5 to a Security Council referral or determination. Failure to grant this consent by use of the veto may block UN action against the P5 member itself

¹³³⁷ Article 24 of the UN Charter, *supra*.

¹³³⁸ See n. 87 and the related discussion from p. 20 Introduction.

¹³³⁹ *Ibid* n 19

¹³⁴⁰ See the Record of the Security Council in relation to determinations of aggression in Chapter 1. For an argument that abolishing the veto would increase the ratio of successful decisions within the Council greatly, see for instance Madeleine O. Hosli, Rebecca Moody, Brian O'Donovan, Serguei Kaniovski and Anna C. H. Little, 'Squaring the circle? Collective and Distributive Effects of United Nations Security Council Reform' (2011) 6 (2) Rev. Int. Organ., 163, 180-1.

or their ally,¹³⁴¹ as well as result in immunity from criminal prosecution. Prolific and unjustifiable use of the veto continues to be unavoidable due to the existing power structure within the Council,¹³⁴² highlighting the need for reform.¹³⁴³

Admittedly, despite the concern expressed above of excessive Security Council control in aggression proceedings, the Council's involvement may, on occasion, extend the reach of the Court to situations in which it would otherwise not have jurisdiction under the current provisions. Thus, a referral to the Court in case of aggression committed by non-member states or non-consenting member states does have the effect of bestowing jurisdiction where the Court cannot otherwise instigate proceedings. ¹³⁴⁴ However, such a situation is only likely where the aggressor state in question is either generally unpopular, or not in favour with one of the P5. This benefit of occasional extension of jurisdiction, through Security Council control over aggression proceedings, therefore struggles to outweigh the problems it causes. As the following discussion shows, the need to reform the voting structure within the Security Council and its working methods has been widely recognised, however the required consensus for change is far from within reach.

¹³⁴¹ Recent examples are Russia's and occasionally China's vetoing of draft resolutions on Syria (S/2014/348, 22 May 2014; S/2016/846, 8 October 2016; S/2016/1026, 5 December 2016; S/2017/172, 28 February 2017; S/2017/315, 12 April 2017, S/2017/962, 16 November 2017, S/2017/970, 17 November 2017, S/2018/321, 10 April 2018, S/2019/756, 19 September 2019, S/2019/961, 20 December 2019, S/2020/654, 7 July, 2020, S/2020/667, 10 July 2020, Russia's use of the veto on a draft resolution concerning the Yemen. S/2018/156, 26 February 2018, a draft resolution condemning the genocide at Srebenica in the year of its 20th anniversary (S/2018/508, 8 July 2015), and a draft resolution concerning Russia's own actions post-annexation of Crimea (S/2014/189, 15 March, 2014) and condemning the downing of Malaysia Airlines flight MH17 in S/2015/562, 29 July, 2015)/ The United States have regularly vetoed draft resolutions concerning Israel and Palestine, see S/2017/1060, 18 December 2017. Most recently, the United States have vetoed a draft resolution reemphasising domestically exercisable jurisdiction over international acts of terrorism, see S/2020/852, 31 August 2020. For a full list of vetos, see Security Council – Veto List, UN Dag Hammarskjold Library (n. 213), available at http://research.un.org/en/docs/sc/quick, last accessed 23 January, 2021.

¹³⁴² See Hosli et al, n. 1242.

¹³⁴³ See 4.3 below. Security Council reform entails an ongoing debate and constitutes an interesting and pertinent subject for further research.

¹³⁴⁴ See Chapter 5, section 3 and the recap above.

4.3 The need for Security Council reform

The UN General Assembly began debating Security Council Reform against the background of Security Council division and inaction in the 1990's, particularly in Rwanda, Somalia and the Balkans. Since then, the development of reform proposals has garnered wide support. The so-called 'Razali Proposal' of 1997, supported by Germany, India, Japan, and Brazil sought to enlarge the Security Council to 24 members, but was unsuccessful. In the build-up to the 2005 World Summit, the then Secretary-General Kofi Annan also pushed for reform and enlargement of the Council. The World Summit Outcome expressed:

We support early reform of the Security Council – an essential element of our overall effort to reform the United Nations – in order to make it more broadly representative, efficient and transparent and thus to further enhance its effectiveness and the legitimacy and implementation of its decisions[...]. 1347

The General Assembly remains supportive of the development of proposals for widening of the membership of the Council. 1348 Intergovernmental negotiations on Security Council reform have been ongoing since the late 1990's, and their mandate was renewed in 2008 in the decision entitled 'Question of equitable representation on and increase in

¹³⁴⁵ See Background on Security Council Reform – Global Policy Forum, http://globalpolicy.org/security-council/security-council-reform.html, accessed 20 January 2021.

¹³⁴⁶ See Dimitris Bourantonis, *The History and Politics of UN Security Council Reform* (Routledge, 2005), 76, 79.

¹³⁴⁷ 2005 World Summit Outcome, Resolution A/Res/60/1, 24 October 2005, para 153.

¹³⁴⁸ For expression of support speeches in the General Assembly, see for instance, UN General Assembly, 23 December 2007, 5 September 2008, 107, A/62/49 (Vol. III). http://undocs.org/A/62/PV.49 (Vol. III), see also UN General Assembly, 12 November 2014, 6, A/69/PV.49, at http://undocs.org/A/69/PV.49, both accessed 20 January 2021.

the membership of the Security Council and related matters'. ¹³⁴⁹ Any proposed solution must garner 'the widest possible political acceptance', or at least two-thirds of UN member states. ¹³⁵⁰ In addition, any proposed solution is subject to the consent (i.e. non-use of the veto) of the P5 themselves, ¹³⁵¹ as stipulated in Article 27 (3) of the UN Charter on substantive decisions, which includes decisions on extension of membership and voting structure within the Council. The P5 may agree to extension of the Council, but are reluctant to voluntarily relinquish their special powers. ¹³⁵²

Agreement on proposals has, to date, proved impossible. Thus the G4, consisting of Germany, Japan, India and Brazil, are seeking additional permanent seats for their own States, plus representation of Africa through a permanent and a non-permanent seat, in addition to four other non-permanent members and reform of the working methods of the Council. A different group, the Uniting for Consensus Group, which includes Italy, Spain, Argentina, Canada, Mexico, South Korea and Pakistan are in favour of abolishing or at least limiting the use of the veto. Their most recent proposal envisages 26 seats in the Council in total, with 9 permanent seats distributed among regional groups and the remaining non-permanent seats held for two years with the potential of re-election. 1354

See Decision 62/557 of 15 September 2008, https://www.securitycouncilreport.org/atf/cf/%7B65BFCF9B-6-D27-4E9C-8CD3-CF6E4FF96FF9%7D/Decision%2062_557.pdf, accessed 20 January 2021.

¹³⁵⁰ Report of the Working Group on the Question of Equitable Representation on and Increase in the Membership of the Security Council and Other Matters Related to the Security Council, A/AC.247/1998/WP.1/Rev.2, adopted by the General Assembly on 24 August 1998. See Press Release GA/9430, https://un.org/press/en/19981980824.ga9430.html, accessed 20 January 2021).

¹³⁵¹ See Article 27 (3) UN Charter (n. 210) on substantive decisions, which includes decisions on extension of membership and voting structure within the Council.

¹³⁵² For support, see the reference to the P5's positions in GA Press release, GA/12288, 16 November 2020, https://www.un.org/press/en/2020/ga12288.doc.htm, accessed 21 January 2021.

¹³⁵³ See G4 Ministerial Joint Press Statement, 23 September 2020, German Federal Foreign Office, accessed 21 January 2021, http://www.mofa.go.jp/files/100095828.pdf.

¹³⁵⁴ See UN GA Press Release, GA/12288 (n. 1367).

The African Group's latest proposal suggests at least two additional permanent seats and two non-permanent seats for African States. The African Group has repeatedly highlighted the 'historic injustice' of less than adequate representation. African membership of the UN consists now of 55 countries out of 193, and at least two thirds of the Security Council's work concentrates on Africa. Nevertheless, no African country has a permanent seat within the Security Council and even the African Union struggles to have a definitive influence on any proposed action on its own continent. Consequently, the Security Council has been accused of failing to take into account the views of key States or regional organisations who are insufficiently represented in the Council, raising questions of the legitimacy of decision-making. The underrepresentation of African States has also been acknowledged by a number of other groups and States. Consequently, the African Group advocates either complete abolishment of the veto, or if the veto power is to persist, extension by two additional permanent seats for themselves.

Yet another group, named the L69, ¹³⁶⁰ includes a number of African States, Small Island States, Latin American States and Caribbean Community members, as well as Brazil and India who are also part of the G4 group. The L69 seek six new permanent seats and six

¹³⁵⁵ *Ibid*.

See for instance UN GA Press Release, GA/12217 of 25 November 2019, https://www.un.org/press/en/ga12217.doc.htm, accessed 21 January 2021.

¹³⁵⁷ See Ville Laettilae and Aleksi. Yloenen, 'United Nations Security Council Reform Revisited: A Proposal' (2019), 30 (1) Dipl. & Statecraft, 164, 166. For example, the Security Council ignored the views of the African Union in its draft resolution against Zimbabwe in 2008, S/2008/447 of 11 July 2008. See also SC Press Release, SC/9396, 11 July 2008, https://www.un.org/press/en/2008/sc9396.doc.htm, accessed 21 January 2021.

¹³⁵⁸ See for instance the L.69 group, 'Security Council Update 2015', https://centerforunreform.org/2015/01/12/security-council-reform-update-2015-similar-scenarios-same-positions-new-outcomes/, accessed 21 January 2021, and UNGA press release GA/12288 (n. 1367). See also the statement by China (press release GA/12288), and the G4 (n. 1368).

¹³⁵⁹ See Press Release GA/12288 (n. 1367).

¹³⁶⁰ *Ibid*.

additional regional non-permanent seats. Similar to the African group, they advocate either abolishment of the veto altogether, or extension to the new permanent seats.¹³⁶¹ The Caribbean Community (CARICOM) on its own is proposing a rotating seat for Small Island States, due to the need for greater representation.¹³⁶² The Arab Group, a group of 22 Arab States, also demands a permanent seat.¹³⁶³ Despite being heavily critical of the veto power, the group presents no alternative proposal.

Finally, a group named ACT, consisting of 21 member states, including Ireland, Switzerland, Peru, Uruguay and Liechtenstein, are working on proposals to improve the operation of the Security Council, including its accountability, coherence, and transparency. Liechtenstein's representative, speaking for the group, recently lamented the Security Council record and suggested enlarging the Council with term extensions to 8-10 years and re-election opportunity. Additional two-year seats should be possible, and no veto power should be extended to any new seats of whatever term length.

These examples of diverging views on the terms of Security Council reform demonstrate that the UN membership itself remains cataclysmically divided on concrete reform proposals, even more than 20 years after intergovernmental negotiations on Security Council reform began. The German Government has recently warned that

-

¹³⁶¹ See 'Security Council Update' (n. 1373), and Press Release GA/12288 (n. 1367).

¹³⁶² *Ibid*. See Grenada's statement in GA/12288.

¹³⁶³ GA/12288, ibid.

¹³⁶⁴ See William Pace, '21 Member States Launch New Initiative to Improve the Working Methods of the Security Council' (21 May 2013), available at http://www.centerforunreform.org/?q=node/541, accessed 21 January 2021.

¹³⁶⁵ GA/12217 (n. 1371).

¹³⁶⁶ *Ibid*.

[i]f it is not adapted to the geopolitical realities of the 21st century, that is, if it does not ensure in particular adequate representation of the global South and major contributors to the UN system, the Security Council's legitimacy and authority are at risk.¹³⁶⁷

Intergovernmental Negotiations on Security Council reform held in February and March 2020 were subsequently adjourned due to COVID-19, although the subject was last debated in the General Assembly on 16 and 17 November 2020 and another round of negotiations has now been scheduled for 2021. In the meantime, G4 Ministers have been calling for an abandonment of debates 'based solely on general statements without actual negotiations' and resumption of intergovernmental negotiations with concrete proposals on Security Council reform to be voted on. Nevertheless, it is unlikely that a dramatic change of the working procedures and veto rights within the Council are, realistically, within reach.

Consequently, if the International Criminal Court is to remain reliant on and constrained by a Security Council, which is, at least for the foreseeable future, subject to a flawed decision-making process, this fact only serves to re-emphasise the arguments presented for finding other ways to overcome the limitations placed on the Court's jurisdiction. These included revision of the aggression provisions, widening the Court's jurisdiction through application of appropriate analogy, greater assumption of independence of the Court from the Council, as well as a reinterpretation of States' right to consent as a duty

¹³⁶⁷ 22nd September 2020, 'Reform of the United Nations Security Council', German Federal Foreign Office, http://www.auswaertiges-amt.de/en/aussenpolitik/internationale-

organisationen/vereintennationen/reformsr/231604, accessed 21 January 2021.

¹³⁶⁸ GA Press Release GA/12289 of 17 November 2020, available at

http://www.un.org/press/en/2020/ga12289.doc.htm, accessed 21 January 2021.

¹³⁶⁹ G4 Statement (n. 1368).

not to withhold that consent unduly. In addition, this thesis has advocated a revival of universal jurisdiction.

4.4 The revival of universal jurisdiction

The argument for an essential revival of universal jurisdiction, presented in section 4 of the previous Chapter, is based on the fact that aggression is perceived as a heinous international crime, the prohibition of which has become a peremptory or *jus cogens* norm under customary international law.¹³⁷⁰ The *telos* of the Rome Statute,¹³⁷¹ the achievement of impunity for and eradication of the worst atrocity crimes known to the international community, supports this interpretation of a hierarchically greater normativity of the prohibition of the crime of aggression.¹³⁷² Additionally, Article 53 VCLT confirms that jurisdiction over crimes of this nature cannot be derogated from by treaty provisions or contracted out of.¹³⁷³

Territoriality and nationality, the traditional bases for jurisdiction, are immaterial where jurisdiction for crimes of this nature arises, as is the requirement for States' consent to jurisdiction. The universality concept therefore allows for jurisdiction regardless of an aggressor state's membership or ratification status. The status of the status

¹³⁷² See also the great support for a hierarchy of norms derived from the separate and dissenting opinions in the *Arrest Warrant* case (n. 1144 and 1152) and Judge Trinidade's dissenting opinion in the *Germany v Italy* case (n. 1162).

¹³⁷⁰ See Art 53 VCLT (n. 32), See also the 2019 *ILC Report and Draft Conclusions on Peremptory norms* (n. 39), which lists aggression as the first crime of a *jus cogens* nature.

¹³⁷¹ See n. 85

¹³⁷³ *Ibid.* See also Article 53 VCLT (n. 32).

¹³⁷⁴ See n. 1088 and 1138 and related text in the previous Chapter.

¹³⁷⁵ See Chapter 5, section 4.

this principle may be capable of overriding the multiple consent requirement, as reinforced by Article 121 (5) in its negative application. 1376 At first instance, universal jurisdiction is exercisable by individual States domestically but member states ought to be able to vest their jurisdiction singly or jointly in the Court, as the appropriate organ to consider all the legal facts. 1377 A precedent can be found within the universal jurisdiction exercised at Nuremberg, delegated to the tribunal by the 19 signatories to the London Charter. 1378 The exercise of jurisdiction by more recent international criminal tribunals also confirms the availability of jurisdiction by such courts in situations where consent to that jurisdiction and/or membership was not necessarily present. 1379 The ICC's recent preliminary rulings on its own jurisdiction demonstrate that the Court does not shrink away from assuming investigation or prosecution powers over non-consenting or nonmember states, even where only a tenuous link to an additional basis for jurisdiction can be established. 1380 Moreover, state practice confirms the popularity of a revival of the concept, since more than three quarters of UN member states acknowledge universal jurisdiction over one or more crimes under international law, ¹³⁸¹ and incidents of cases brought domestically under such jurisdiction are also on the increase. 1382

¹³⁷⁶ See Chapter 5, section 4.

¹³⁷⁷ See the principle of complementarity, Article 1 and 17 (1) (b) Rome Statute (n. 1), and n. 1104.

¹³⁷⁸ Article 6 (a) of the London Charter (n. 117), and Control Council Law No. 10 (n. 121). See also Chapters 1 and 5, section 4 at pp. 263-4 (including n. 1094). See also Scharf (2001) (n. 1095), 103-6, and Randall (n. 109), 804-6, and Goldsmith (n. 120). See also Dapo Akande's 2003 article (n. 1176, at 626) confirming the availability of universal jurisdiction as a 'structural rule', although he appears to retract from this view, at least partially in his 2011 article (n. 888), at 30-2.

¹³⁷⁹ See the discussion in section 4, Chapter 5, and the jurisdiction of the ICTY (n. 1179), the ICTR (n. 1180), the Special Court for Sierra Leone (n. 1182) and recent practice by the ICC itself (pp. 282-7).

¹³⁸⁰ See the recent assumption of investigation powers concerning Myanmar, Russian activities on Ukrainian territory, US and CIA actions in Afghanistan, and actions by (and against Israel) in Palestine, pp. 282-7.

¹³⁸¹ See Amnesty International's 2012 Report (n. 131).

¹³⁸² See section 4, Chapter 5, pp. 286-7.

Notwithstanding this apparent evolution towards wider and non-consensual jurisdiction over heinous international crimes, 1383 it is acknowledged that any attempts to apply universal jurisdiction to the crime of aggression may be seen as consciously defying the provision of Articles 15 bis and ter and their strict consent and membership requirements, as well as Understanding No. 4 bis, which attempted to explicitly exclude domestic (and thus universal) jurisdiction. 1384 Arguments for universal jurisdiction over the crime of aggression, as proposed in this thesis, are therefore likely to incur fierce objections. A proposal along the same lines, made prior to the Rome Conference and suggesting universal jurisdiction as the general basis for prosecutions over member states at least, had been expressly rejected, despite being included in Article 9 of the 1998 Draft Statute for the Court. 1385 Immediately prior to Kampala, another visionary minority proposal suggested applying universal jurisdiction over aggression to any State, regardless of consent or membership, ¹³⁸⁶ but again failed to be adopted, resulting instead in the stringent conditions now deliberately put in place. A strict reading of the Rome Statute alone, it is therefore admitted, does not lend itself to the acceptance universal jurisdiction without controversy. Advocates of a narrow interpretation of treaty provisions, in particular, ¹³⁸⁷ are likely to continue arguing for stringent application of the negotiated elements of Articles 15 bis and ter, excluding non-consenting and non-member states from jurisdiction. 1388

-

¹³⁸³ See the Joint Separate Opinion in the *Arrest Warrant* Case (n. 1144), Judge van den Wyngaert's interpretation in his dissenting opinion in that case (n. 1152), and that of Judge Trinidade in the *Germany v Italy* case (n. 1162).

¹³⁸⁴ As expressly included on the insistence of the United States at Kampala, see Understanding No. 4 (n. 1107).

¹³⁸⁵ See A/CONF.183/2/Add.1 (The German Proposal, n. 126), and the discussion of universal jurisdiction in the previous chapter, section 4.

¹³⁸⁶ See Chairman's Non-Paper, in preparation for the 8th Assembly of States Parties in 2009, ICC-ASP/INF.2 at p. 8, and the discussion in the previous chapter, section 4.

¹³⁸⁷ See Article 31 (1) and 34 VCLT (n. 32).

¹³⁸⁸ See for instance some of Dapo Akande's arguments, defending a literal reading of, for instance Article 121 (5) Akande ('Prosecuting Aggression' (2011), 11 (n. 888)).

And yet it may be possible to counter the insistence on strict application of the VCLT to the crime of aggression provisions. One alternative, for instance is to view the Rome Statute first and foremost as an internal rule book for the Court's judiciary, rather than a classically interpreted treaty, creating an exemption from the requirement of strict application. Defending this point, Dov Jacobs asserts that it is the 'broad and ultimately discretionary rules of interpretation of the VCLT', combined with the specific requirement of legality in international criminal law which demand disapplication of inflexible adherence to the Rome Statute's treaty provisions. This view potentially opens the path to wider application of jurisdiction of the Court. Additionally, a number of other arguments, conceivably even more persuasive, have been presented.

The most powerful contention, as reiterated throughout this thesis, is derived from Article 53 of the VCLT itself, which states that a treaty is rendered void if found to be conflicting with a peremptory norm of general international law, as recognised by the international community as a whole. No derogation from a *jus cogens* norm by treaty is, therefore, permitted. Judicial *dicta*, mostly found in the separate and dissenting opinions of the ICJ's *Arrest Warrant* and *Germany v Italy* cases, ¹³⁹¹ emphasise this hierarchy, as does Chapter V of the 2019 Report of the International Law Commission on peremptory norms. ¹³⁹² Thus, Conclusion 3 of that Report confirms that peremptory norms are 'hierarchically

-

¹³⁸⁹ See Dov Jacob's argument, in 'Why the Vienna Convention should not be applied to the ICC Rome Statute: a plea for respecting the principle of legality' (n. 459).

¹³⁹⁰ *Ibid*

¹³⁹¹ See n. 1144, 1151 and 1152 for the separate and dissenting opinions in the *Arrest Warrant* case and n. 1162 for Judge Trinidade's dissenting opinion in the *Germany v Italy* case. The majority decisions in these cases, whilst not contradicting the status of *jus cogens* crimes generally, decided that issues of immunity, on account of their procedural nature, had to be decided *before* the consideration of substantive questions. See n. 1142 for the majority decision in the *Arrest Warrant* case, n 1148 for the majority decision in the *Germany v Italy* case, and for a discussion of the policy basis for these, see n. 1160 and 1161)

¹³⁹² Conclusion 2, 2019 Report of the ILC on Peremptory Norms (n. 39).

superior to other rules of international law and are universally applicable.' The prohibition of aggression is expressly identified as the first such norm in a non-exhaustive list attached to the ILC's Report. Universal jurisdiction over the crime of aggression at customary international law, may therefore still be capable of overriding the insistence on consent and even membership in Articles 15 *bis* and *ter*.

The evolution of a growing recognition of the principle, as well as recent opening of the floodgates of domestic proceedings based on universal jurisdiction over 'hostis humanis generis', 1395 provide additional evidence of continuing and expanding state support for the concept. The 2017 Review of Universal Jurisdiction by Trial International had reported a then unprecedented number of 47 cases of universal jurisdiction in 13 countries, including 14 charges of torture, 6 charges of genocide, 15 charges of crimes against humanity, and an overall increase of war crimes charges of 55% to previous years. 1396 Eleven of these cases alone were brought for alleged crimes in Syria, a nonmember state of the International Criminal Court, which has long been enjoying military and political support from Russia, which is a P5 member and also non-signatory of the Court. Russia, and its fellow P5 member China, again a non-member of the Rome Statute, have on a number of occasions intentionally prevented Security Council referrals to the Court. 1397 Whilst the most recent 2020 Report of Trial International demonstrates a shift to charges for terrorism offences, 1398 there were a record 146 charges for crimes against

¹³⁹³ *Ibid*, at 142 (n. 1123).

¹³⁹⁴ *Ibid*, 205.

¹³⁹⁵ See the discussion in Chapter 5, pp. 286-7.

¹³⁹⁶ Trial International, Make Way for Justice #3, Universal Jurisdiction Annual Review 2017, https://trialinternational.org/wp-content/uploads/2017/03/UJAR-MEP_A4_012.pdf, accessed 23 January 2021.

¹³⁹⁷ See Security Council – Veto List, UN Dag Hammarskjold Library (n. 213).

¹³⁹⁸ Trial International, Terrorism and international Crimes: Prosecuting Atrocities for what they are, Universal Jurisdiction Annual Review 2020, https://trialinternational.org/latest-post/universal-jurisdiction-annual-review-2020-atrocitites-must-be-prosecuted-soundly-and-rigorously/ accessed 23 January 2021.

humanity, 92 for torture, 21 for genocide and 141 for war crimes brought under universal jurisdiction by States domestically. These figures demonstrate that the appetite by the international community to prosecute the worst international crimes under the principle of universal jurisdiction has grown enormously. If domestic jurisdiction under the principle is transferrable to the International Criminal Court, just as it was at Nuremberg, this would create a potential for much wider jurisdiction over the crime of aggression. Universal jurisdiction, therefore, may yet provide a way to circumvent the tight restriction of application to merely consenting member states.

5. Transcending the limitations? – An overall assessment

Some of the suggestions for widening the jurisdictional limitations placed on the crime of aggression, presented above and throughout this thesis, are, admittedly, more pragmatic than others. The least contentious way to widen jurisdiction of the Court would be to promote universal membership and voluntary acceptance of the aggression provisions without reservations. Signature should equate acceptance without the ability to avoid jurisdiction by opting out, or failing to individually ratify the amendment, in line with Article 120 of the Rome Statute. This point was made, in particular, by the Japanese delegation post Kampala in light of the uncertainty surrounding entry into force for member states.

¹³⁹⁹ *Ibid.* See also the recent domestic cases considered in Chapter 5, Section 4, pp. 286-7.

¹⁴⁰⁰ See Chapter 1 and the arguments in Chapter 5, section 4.

What happens to a non-State Party that desires to accede to the Rome Statute after the adoption of the amendments? How can we be certain that such a newly acceding country will be bound by the amended Rome Statute, while we see no provisions stipulating about the entry into force of the amendments per se? This is an issue that should be squarely addressed, if we are genuinely serious about enhancing the universality of the International Criminal Court.¹⁴⁰¹

Subscription to the Rome Statute entails a State's voluntary subjection to the Court's jurisdiction, and there should be no room for a 'perverse incentive' to become a member and subsequently withhold consent by failing to ratify and/or opt out. The prohibition of aggression constitutes a peremptory norm at customary international law, and opting out through taking advantage of a treaty provision which allows for a loophole should, applying Article 53 of the VCLT, not be permissible.

It may be possible to 'shame' States currently withholding their consent to membership or ratification into granting their consent through political and diplomatic pressure. Furthermore, such pressure could become a tool for compliance, once States have accepted jurisdiction, furthering deterrence. Wide membership and ratifications without reservations would have the additional benefit of enhancing the credibility and reputation of the Court.¹⁴⁰⁴ Wide-scale international cooperation for the promotion of peace and

¹⁴⁰¹ See the 'Statement by Japan' in the *Travaux Preparatoires* (n. 317), p. 812.

¹⁴⁰² See the discussion in Chapter 5.

¹⁴⁰³ See Article 53 VCLT (n. 32), and the 2019 ILC Report on Peremptory Norms (n. 39).

The ICC has often been portrayed as prosecuting nationals of unpopular States only, and has been accused of undermining the sovereignty of and unfairly targeting African States, see for instance 'African Union backs mass withdrawal from ICC', BBC News, 1 February 2017. http://www.bbc.com/news/world-africa-38826073, accessed 18 November 2020.

security also entails a duty to prevent, prosecute and abolish the worst international crimes. Hospital crimes of aggression would bring the world closer to achieving a 'minimum world public order' for the benefit of all people. This concept, as developed by McDougal and Feliciano and encountered at the beginning of this thesis, would aid the development of 'maximum human dignity and minimum destruction of human values', Hospital agoal which is as essential in contemporary times as it was in the era of the Cold War under the threat of potential global destruction through nuclear weapons. The pursuit of such a world public order and the peace and stability it promotes, requires an equilibrium of power between States, complementary humanitarian policies, organisation and centralisation of international authority through their institutions, as well as, importantly, participation in the international legal process. Hospital process.

A reinterpretation of a perceived right of States to grant their consent as a duty instead, as advocated by Ronald Dworkin, is a persuasive starting point. Dworkin's comments on the interpretation of international documents, although made with respect to the UN Charter, could be applied to the Rome Statute and provide justification for the waiving of state consent to aggression prosecutions, the central obstacle to wider jurisdiction:

The correct interpretation of an international document, like the UN Charter [or the Rome Statute], is the interpretation that makes the best sense of the text, given the underlying aim of international law, which is taken to be the creation of an international order that protects political communities from external aggression, protects citizens of those communities from domestic barbarism, facilitates

¹⁴⁰⁵ See McDougal and Feliciano (1961 (n. 87), and 1958 (n. 88)).

¹⁴⁰⁶ See n. 77.

¹⁴⁰⁷ See pp. 20-2.

coordination when this is essential, and provides some measure of participation by people in their own governance across the world. These goals must be interpreted together: they must be understood in such a way as to make them compatible. 1408

If all States are subject to the jurisdiction of the Court for acts of aggression committed by them or on their behalf, the likelihood of such acts will be diminished substantially. All States would enjoy the same protection, and be subject to the same criminal jurisdiction. The existence of such protection, on its own, could be used to promote the benefits of becoming a signatory.

Admittedly, this still leaves the problem of great Security Council control and the existing power to block referrals to the Court. Similar to the argument above, it may be possible to 'shame' the three P5 members, who currently remain non-signatories, into signature of the Rome Treaty. Reinterpreting the right to grant consent as a duty not to withhold it unduly would be essential. The use of the veto should at the very least entail the need for public justifications and ideally be abolished entirely, as part of substantial Security Council reform. The Security Council's role of prevention of threats to the peace ought to be perceived as a shared responsibility, allowing the Court to act independently, but also providing for a partnership between the two organs, which could result in the Council enforcing decisions of the Court. 1409

. . .

¹⁴⁰⁸ Dworkin (2013) (n. 43), 22.

¹⁴⁰⁹ As envisaged by Ferencz (2008) and Cassese (2007), see n. 1337 above.

It has also been argued that it is essential to allow for extension of the definition itself to cover modern acts of aggression. To that effect, a number of promising projects have recently been established for the purpose of examining the relationship between international law and cyber threats and warfare. He is findings are likely to lead to a wider recognition of the need to include acts of aggression, other than the narrow list included in Article 8 *bis* (2), and actors other than the State, in support of the arguments presented in this thesis. Extension to other acts and a wider group of actors would certainly extend jurisdiction of the Court, however attribution, admittedly, may remain a problem in any type of remote warfare. This fact only serves to reinforce the argument for re-writing the crime of aggression provisions in their entirety, reflecting the possibility of including other acts of aggression, committed by other actors, including culpable accessories.

Finally, this thesis also examined the possibility of extending jurisdiction of the Court through the principle of universality. The growing appetite of individual States for the application of universal jurisdiction over the most heinous international crimes demonstrates the increasing recognition of the principle itself. Universal jurisdiction under customary international law is currently the only potential way to extend jurisdiction over aggression to a non-consenting state, unless the State is sufficiently unpopular to attract Security Council condemnation. This fact alone serves to demonstrate that the existing consent-based system of international accountability is deficient, allowing for elective impunity, and requires a substantive rewrite of the crime of aggression in order to overcome the limitations placed on it, as well as far-reaching reform

¹⁴¹⁰ See the efforts by the OEWG (n. 658), GGE (n. 659) and the Council of Advisers (n. 660).

of the Security Council itself and a reconsideration of its relationship with the Court. Only then is there a chance of eradicating the supreme international crime.

CONCLUSION

The Kampala consensus on the crime of aggression was celebrated as a ground breaking innovation and a 'milestone' in the development of international criminal law, heralding the end of an 'old era of impunity' and the beginning of a 'new age of accountability'. As the discussion revealed, the compromise achieved at Kampala is, however, far from perfect, and the limitations placed on the jurisdiction of the Court over the crime are stringent.

The question remains whether this compromise came at too high a price in light of these limitations, and whether the provisions, as they stand, are fit for purpose on account of the fact that the face of war is changing. Has the crime been so severely restricted that it may fail to regulate significant newly emerging forms of aggression? And is the reach of the Court so limited that it is in danger of becoming irrelevant at the very time when jurisdiction over the crime has only just arisen, and deterrence from the crime is as important as it has ever been? Answering these questions closes the circle of this inquiry, the purpose of which was an assessment of the potential effectiveness of the Court's new jurisdiction as a result of the restrictions placed on it.

To that end, a number of important factors were identified, which impact on any prediction of the jurisdictional effectiveness of the Court. Thus, the overly narrow

¹⁴¹¹ See Kress and von Holtzendorff (2010) (n.5), 1179. See also Trahan (2011) (n. 993), 50.

¹⁴¹² Ban Ki-moon, 'An Age of Accountability', Address to the Review Conference on the International Criminal Court, Kampala, 31 May, 2010, <a href="https://www.un.org/africarenewal/web-features/secretary-general's-"-age-accountability"-address-icc-review-conference, accessed 12 April 2018.

definition itself poses several problems, including a high gravity threshold and its restriction to traditional warfare as considered in Resolution 3314. 1413 The Kampala definition, unlike the 1974 resolution, does not explicitly provide for extension and consideration of other acts. As a result, it is in danger of being irrelevant in a modern context at a time when criminal jurisdiction has only just arisen. Modern types of warfare, ranging from the remote disruption of a country's vital systems to cyber threats and the asymmetrical deployment of untraditional weapons, for instance, have not been captured. 1414 And yet, the resort to such methods has, over the last two decades, vastly increased, as is openly recognised by experts, concerned NGO's, and a number of national security agencies. 1415 The reach of the Court is also unlikely to extend to non-state actors or influential participants other than official political or military leaders of a State, on account of the state-centricity of the crime and a restrictively high standard of control or direction of a state act of aggression. 1416 Consequently, the effectiveness of the recently activated jurisdiction of the Court is, in a modern context, impaired on account of the narrow definition of the crime, unless the Court assumes wider powers to determine other acts and to consider a wider range of aggressors or, alternatively, the definition itself is widened. 1417

The potential effectiveness of the Court's jurisdiction has been curbed further by the extraordinary level of Security Council control over criminal aggression proceedings. ¹⁴¹⁸ The restrictions were, as the negotiating history and the post-Kampala statements demonstrate, incorporated deliberately on the insistence of a small number of powerful

-

¹⁴¹³ See Chapters 2 and 3. For Resolution 3314, see n. 14.

¹⁴¹⁴ See particularly section 3.1 of Chapter 3.

¹⁴¹⁵ *Ibid*, n. 654 and 656.

¹⁴¹⁶ As a combination of the different elements of Art 8 *bis*, discussed in Chapters 2 and 3.

¹⁴¹⁷ See section 4, Chapter 3, and Chapter 6.

¹⁴¹⁸ See Chapter 4.

States, who ideally wished to retain absolute Council exclusivity over the political crime of aggression in all situations. ¹⁴¹⁹ In the view of these States, investigations and prosecutions of aggression should always rely on a prior Council referral. If such an absolute monopoly could not be retained, the Court's jurisdiction was sought to be restricted by a number of cumulative consent requirements for both victim and aggressor. Combining Article 15 bis (4) with the negative interpretation of Article 121 (5) delivers on this extreme level of state consent pursued so ardently by the powerful minority. These restrictions have the greatest impact on the potential effectiveness of the Court's jurisdiction and are, it is contended, excessive. ¹⁴²⁰ The combination of the surviving high level of Security Council control over criminal aggression proceedings, albeit short of an absolute monopoly, and the extreme consent requirement results in, at best, 'jurisdiction a la carte' and, at worst, elective impunity.

Admittedly, it is conceivable that a Security Council referral may have the effect of extending jurisdiction over non-signatories or non-consenting states parties, over whom the Court would otherwise have none. However, history shows that the record of the Council taking action in aggression scenarios is unconvincing, and any potential referral is particularly unlikely to be forthcoming if this would affect the interests of one of the P5 members or those of an ally. Extensive Security Council control, as well as retrospective determinations and referrals, may also raise legality concerns and diminish the credibility of the Court as an independent criminal tribunal.

¹⁴¹⁹ See Chapter 4, and 'Explanations of positions', *Travaux Preparatoires*, (n. 317, from 810).

¹⁴²⁰ See Chapter 5.

¹⁴²¹ See Dapo Akande's argument in Chapter 4, section 3.3.

¹⁴²² See the account in Chapter 1.

The overall conclusion arrived at is, therefore, that the likely effectiveness of the new jurisdiction of the first permanent International Criminal Court has, indeed, been substantially impaired by the severe restrictions placed on it. Nevertheless, there may be room for hope that some of the damage can be 'undone'. To that effect, and attempting to address the specific problems perceived, a number of changes have been suggested throughout this thesis, which are hoped to be improvements, although perhaps not perfect solutions.

Thus, it was suggested that the definition itself ought to be widened in order to include political, economic or spiritual puppet masters, for instance, who may exercise sufficient control to be accountable for the commission of aggression. The threshold of a manifest violation of the UN Charter, currently too high to capture modern acts of aggression with the most serious consequences, for instance remote warfare, may have to be lowered to 'violations inconsistent with the Charter'. Similarly, the list of aggressive acts also requires extension, since it does not provide for such modern types of warfare, and is limited to state-on-state aggression only, at a time when aggressive non-state groups have emerged as the most prolific instigators of acts of aggression.

The argument for re-writing the definition of aggression is likely to gain support from a number of ongoing projects investigating the applicability of international law to such modern types of warfare. These are currently undertaken by the Permanent Mission for Liechtenstein, ¹⁴²⁴ a Working Group of a number of UN member states, ¹⁴²⁵ and a group of experts. ¹⁴²⁶ The result of their findings may have an impact on considerations by the next

¹⁴²³ As was the case at Nuremberg, see Chapter 1, section 4.3, and Chapter 2, sections 3.1 and 3.2.

¹⁴²⁴ See n. 660.

¹⁴²⁵ See n. 658.

¹⁴²⁶ See n. 659.

Assembly of States Parties, which is scheduled for 2025. Whether there will be sufficient appetite to amend the existing definition, in reality, remains to be seen. In the meantime, it has been contended, the Court itself may have to widen the definition incrementally, through close analogy and on a case by case basis, in order to give effect to the aggression provisions, although it is acknowledged that this will raise concerns of offending the *nullum crimen sine lege* principle, and Article 22 (2) of the Rome Statute.

Additionally, it was contended that Security Council control needs to be at least relaxed, if not relinquished, particularly on account of the fact that referrals to the Court may currently be prevented through use of the veto by one of the P5. Security Council reform, a subject worthy of extensive further research in its own right, will be required, or at least a change of perception from a P5 member's right to refuse consent, to a duty not to withhold consent unduly. The Security Council and the Court should operate in partnership, without prejudice to the rights of a defendant before the Court.

Suggestions to overcome the consent problem are rather more complicated and not uncontroversial. The strict consent and membership requirements, which must be fulfilled before jurisdiction arises in situations where no Security Council referral is forthcoming, pose severe restrictions on the Court's reach. It is striking that the supreme international crime, often leading to the commission of other humanitarian crimes committed in the context of an illegal war, 1428 requires multiple consent by both aggressor and victim state, whereas this reinforced consent requirement is alien to the other crimes under the Rome

-

¹⁴²⁷ See Dworkin (2013) (n. 43) and the arguments on pp. 15-19 Introduction.

¹⁴²⁸ Particularly crimes against humanity, under Article 7, and war crimes under Article 8 of the Rome Statute (n. 1).

Statute. 1429 This fact, combined with the resistance to the Court's independent jurisdiction by a small number of powerful states, headed by the P5, leads to the cynical conclusion that these States sought and achieved protection from the Court's scrutiny of their own extant and future potential acts of aggression. 1430 Three of the P5 have remained non-signatories and the other two, France and the UK, have not ratified the amendments. Consequently, none of these States can currently be prosecuted for any act of aggression under Article 15 *bis*, even if the victim state has ratified the amendments and has not opted out. If a referral to the Court under Article 15 *ter* was sought, any of the P5 could veto it, and they therefore enjoy total impunity. The problem was addressed by the former UN Secretary-General, Kofi Annan, at Kampala:

[Q]uestions of credibility will persist so long as three of the five permanent members of the Security Council refuse to reconsider their position and join those who have taken the courageous step to become parties to the Statute. The same is valid for countries that aspire to permanent membership. Indeed the problem is not limited to the Security Council. Six of the G20 have not ratified the Rome Statute. The States Parties to this historic Statute must therefore pose the question, "What kind of leadership is this which would absolve the powerful from the rules they apply to the weak?" 1432

1

¹⁴²⁹ See Chapter 5. See also the argument by Kevin Jon Heller in his presentation before the UN General Assembly prior to the activating decision by the ASP in 2017, 'My UN Presentation on the Aggression Amendments' (OpinioJuris, 5 May 2017), http://opiniojuris.org/2017/05/05/my-un-presentation-on-the-aggression-amendments/, accessed 30 October 2020.

¹⁴³⁰ A recent prime example is the aggressive annexation of Crimea, see Chapters 1, 2, 4, 5 and 6.

¹⁴³¹ Indeed in 2021, seven of the G20 States have not ratified the Rome Statute, namely China, India, Indonesia, Russia, Saudi Arabia, Turkey, and the United States, see http://www.g20.org and http://asp.icc-cpi.int.

¹⁴³² Address by Kofi Annan (Kampala, Uganda, 31 May 2010), https://www.kofiannanfoundation.org/speeches/kofi-annan-addresses-the-first-review-conference-of-theassembly-of-states-parties-to-the-rome-statute-of-the-international-criminal-court/, accessed 12 April 2018.

In a similar vein, Claus Kress has pointed out that it is becoming increasingly difficult 'for any victorious power whose judges sat in judgment at Nuremberg and Tokyo to explain why they still do not wish fully to embrace the legacy of their own pioneering course of action after the Second World War.' It can only be hoped that it may be possible to shame the group of powerful, sceptic States, who have not yet expressed their support for the Court's work through signature, into membership and unconditional ratification. Exerting diplomatic pressure to set a positive example, become a signatory, and ratify the aggression amendment is therefore the most practical suggestion.

This particular issue also harks back to the argument presented for a reinterpretation of the right of States to grant their consent as a duty not to withhold it, in order to promote peace and security and in the attempt to create a global 'minimum world public order', as envisaged by McDougal and Feliciano. Additionally, universality of the Court's jurisdiction ought to be pursued and actively promoted, firstly through the concerted encouragement of universal membership of all UN States to the Court, beginning with the permanent members of the Security Council. Ratifications, as stated above, ought to be made without reservations, accepting unconditional jurisdiction of the Court through failing to opt out. Additionally in the permanent members of the Security Council in Ratifications as stated above, ought to

The second, and more controversial way of achieving universality, is the application of the customary international law concept of universal jurisdiction, which is based on the fact that the Crime of Aggression is a peremptory norm, which cannot be derogated from by treaty provisions.¹⁴³⁶ As such, the crime is justiciable by any State regardless of

¹⁴³³ See Kress (2018) (n. 993), 10.

¹⁴³⁴ McDougal and Feliciano (1961) (n. 87).

¹⁴³⁵ As permitted by Article 15 *bis* (4).

¹⁴³⁶ See the discussion in the Introduction and Chapter 5.

territory or nationality and membership of the Court. The member states of the Court could derogate their individually exercisable universal jurisdiction to the Court, following the Nuremberg precedent. Whilst this may provide a convoluted way for the Court to overcome the stringent consent requirements and obtain wider jurisdiction, the application of the concept is likely to incur objections by defenders of Security Council prerogatives, the state consent principle, and advocates of strict application of the VCLT. Notwithstanding these likely objections, recent state practice certainly appears to support a re-emergence of universal jurisdiction generally, in light of potential impunity for international crimes without state consent or Security Council referral. 1438

The price paid for arriving at a consensus on the Court's recently activated jurisdiction is undoubtedly high on account of its limitations. Nevertheless, the very existence of jurisdiction over the crime of aggression 'sends a timely appeal to the conscience of mankind', 1439 thus importantly reinforcing the prohibition of the use of force in Article 2 (4) of the UN Charter. As Ambassador Wenaweser stated after the ASP's activation decision:

The historic significance of the decision we have taken today to activate the Court's jurisdiction over the crime of aggression cannot be overstated. Never has humanity had a permanent international court with the authority to hold

¹⁴³⁷ Although the VCLT itself provides for recognition of the superiority of peremptory norms over treaty application, see Article 53.

¹⁴³⁸ See the *Reports by Trial International*, n. 1435 and 1437. See also the discussion of recent domestic cases under universal jurisdiction, pp. 286-7.

¹⁴³⁹ Ferencz, (2009), 289 (n. 4).

¹⁴⁴⁰ Kress (2018) (n. 993), 1.

individuals accountable for their decision to commit aggression – the worst form of the illegal use of force. Now we do. 1441

Even if the effect of active jurisdiction of the Court over the crime of aggression is merely to put leaders on notice that their actions are now, at least potentially, subject to scrutiny by the first permanent international criminal court, that is an important achievement in its own right. The mere fact that States will now have to consider if their acts could amount to criminal aggression and seek to distinguish them publicly, serves to reinforce the prohibition of aggression. Even imperfect jurisdiction over aggression is better than no jurisdiction at all. As Nikolas Stuerchler eloquently expressed after the 2017 activation decision by the ASP, despite venting disappointment at the restrictive interpretation of that decision: 1442

In all of this, let us not forget that the activation of the crime of aggression is meant to be a contribution to the preservation of peace and the prevention of the most serious crimes of concern to the international community as a whole. More than 70 years after the Nuremberg and Tokyo trials, the ICC has received the historic opportunity to strengthen the prohibition of the use of force as enshrined in the UN Charter and complete the Rome Statute as originally drafted. This is the perspective we should preserve. ¹⁴⁴³

Notwithstanding this achievement of jurisdictional activation and the hopes expressed, an important caveat has to be added that serious deterrence can only be achieved if States

-

¹⁴⁴¹ Statement for Liechtenstein, cited in Kress, *ibid*, at fn. 48.

¹⁴⁴² See sections 2.2.5 and 2.2.6 of Chapter 5.

¹⁴⁴³ See Stuerchler (n. 946).

are actually subject to jurisdiction of the Court without exception. Otherwise, a perception of elective impunity is allowed to persist, as demonstrated, for example, most prominently by Russia's unpunishable aggressive annexation of Crimea, 1444 and credibility of the Court's jurisdiction over the crime of aggression will continue to be seriously impaired. In order for the Court's jurisdiction to be effective, it is therefore essential to promote the widest possible membership with unconditional ratifications of the crime of aggression. Universal jurisdiction of the Court over the crime of aggression furthers the aims of the Rome Statute of eradication of the worst international crimes as well as those of the UN Charter of promoting peace and security at global level.

Additionally, the 20th century definition now incorporated into Article 8 *bis* requires far wider interpretation, in order to respond to 21st century challenges, and allow for prosecutions of modern acts of aggression. Projects are underway to examine the relationship and applicability of international law to modern acts of aggression like cyberwarfare, however they come undeniably late. An opportunity may arise to amend the definition at the 2025 Review Conference, but it remains to be seen whether the international community and the membership of the Rome Statute will have the appetite to reopen the perpetual controversy surrounding aggression and modernise the definition. Even if the member states are willing to consider amending the definition, the jurisdictional limitations of consent and Security Council control, subject to the privileges and questionable internal voting structure within the Council – itself long overdue for revision - pose a continuing incontrovertible problem for jurisdiction of the Court.

¹⁴⁴⁴ Although see the Prosecutor's conclusion in her 2020 Report that there was sufficient grounds to investigate war crimes and crimes against humanity committed on Ukrainian territory, including the illegally annexed territory of Crimea, see the discussion in Chapter 5.4, n. 11196-8 and related text. ¹⁴⁴⁵ See n. 658-60.

The 'lock' has now been 'removed from the courthouse door', ¹⁴⁴⁶ however the Court's jurisdictional powers have been severely curbed. If the Court is to exercise its jurisdiction efficiently in the future, reinforcing the use of force prohibition of the Charter and furthering impunity, the definition has to be extended and the jurisdictional restrictions have to be eased, allowing for wider application. Resorting to the words of Benjamin Ferencz:

Failure to allow the ICC to punish aggression is a repudiation of Nuremberg and a step backward in the development of international criminal law... Prohibiting the ICC from exercising its jurisdiction is an indefinite guarantee of continuing immunity for future aggressors... Every legal step should be taken that might help deter nations from the incredible horrors of armed conflicts. Aggressors should not be granted a renewed license to wage illegal wars with impunity. 1447

Elective impunity should not be an option, and there is no longer a place for the insistence on sovereignty, individual state consent, and consent of the powerful via a flawed determination and referral system which relies on agreement within a highly political forum, subject to individual state interests and unequal voting powers. In an ideal world, States would voluntarily subject themselves to the Court's jurisdiction through universal membership with unconditional ratifications, to be treated equally before the Court, in the knowledge that their acts of aggression will attract individual criminal responsibility. If that is not achievable in the foreseeable future, it may be necessary to retreat from the insistence on a strictly consent-based approach and advance customary universal

. .

¹⁴⁴⁶ See B. Ferencz, n. 831.

¹⁴⁴⁷ B. Ferencz (2009), 290, (n. 4).

jurisdiction regardless of consent or territory, leading to powerful deterrence from aggression and effective prosecutions. The world no longer has a 'choice between peace and justice', 1448 they must go hand in hand.

Suggestions for further research

Important projects are currently underway to determine responsible state behaviour in the context of international security, and the application of the Rome Statute to cyber threats and cyberwarfare. The first two are mandated by the General Assembly of the UN. Thus resolution A/RES/73/266 established a new Group of Governmental Experts (GGE) on Advancing Responsible State Behaviour in Cyberspace in the Context of International Security. The GGE aims to outline the global agenda, examining applicable norms, rules and principles, and to address how international law applies to the digital space. The GGE consists of 25 selected member states of the UN, and has been scheduled to hold 6 consultations with regional organisations with the African Union, the European Union, Organisation of American States, Organisation for Security and Co-operation in Europe and the ASEAN Regional Forum, and 2 with all member states. Its report will be made to the 76th GA Session in 2021. 1450

¹⁴⁴⁸ See Kofi Annan's speech (n. 1448).

¹⁴⁴⁹ See n. 659.

¹⁴⁵⁰ *Ibid.* See also 'UN GGE and OEWG', Geneva Internet Platform Digital Watch observatory for Internet governance and digital policy, http://www.dig.watch/processes/un-gge, last accessed 20 January 2021.

The UN-authorised Open-Ended Working Group (OEWG) on Developments in the Field of Information and Telecommunications in the Context of International Security (ICTs) was established in 2018. ¹⁴⁵¹ The working group involves all interested UN member states and has been holding intersessional meetings with interested stakeholders consisting of business, NGOs, and academic scholars. Its purpose was to further develop and potentially change applicable norms, rules and principles, consider how international law applies to cyberspace, identify existing and potential threats, and establish a regular institutional dialogue within the UN. This group is, at the time of writing preparing its Zero Draft and meet for a final session in March 2021. ¹⁴⁵²

Both groups state (at the time of writing) that there is broad agreement that international law applies to cyberspace. Additionally, there is agreement that jurisdiction over ICT lies with a State on its territory. As far as international law is concerned, there is agreement that States should not conduct internationally wrongful acts nor use proxies for such acts. Open issues for both groups are:

- how humanitarian law should apply to ICT
- how international law applies to cyber-attacks in peacetime (as a form of 'hybrid warfare')
- whether discussions should include control of arms proliferation or a more precise
 use of the law of armed conflict

¹⁴⁵¹ See n. 658.

¹⁴⁵² Geneva Internet Platform, n. 1466 above.

¹⁴⁵³ *Ibid*.

¹⁴⁵⁴ *Ibid*.

¹⁴⁵⁵ *Ibid*.

- how to deal with ICTs which may have dual use, i.e. for peaceful use, but also

capable of endangering peace and security. 1456

Additionally, a Council of Advisers on the Application of the Rome Statute to

Cyberwarfare has been established by the Permanent Mission of Liechtenstein, along with

Argentina, Belgium, Estonia, Luxembourg, Spain, Switzerland and the Global Institute

on the Prevention of Aggression, academics, and civil society members, 1457 discussing

specifically, whether cyberattacks can be understood as 'armed attacks' for the purpose

of Article 8 bis of the Rome Statute, potential jurisdiction over state and non-state actors,

the applicable threshold potentially giving rise to self-defence under Article 51 of the UN

Charter, and the application of international humanitarian law (Articles 7 and 8 of the

Rome Statute) to cyberattacks. 1458 The project is, at the time of writing also still ongoing.

The subject of cyber-aggression, in particular, poses a highly interesting and pertinent

topic for future research.

Finally, it will be interesting to see whether States, having implemented domestic

jurisdiction over the crime of aggression, will in the future assume universal jurisdiction

over the crime, as they have done over the other crimes under the Rome Statute. All of

these projects and campaigns will, in the near future shed light on how jurisdiction of the

Court over the crime of aggression may be widened in practice.

¹⁴⁵⁶ *Ibid*.

¹⁴⁵⁷ See n. 660.

¹⁴⁵⁸ *Ibid*.

345

Cases, International Treaties and other documents

Table of Cases

United States v Klintock, [1820] 18 US (5 Wheat.) 144.

United States v Smith, [1820] 18 US (5 Wheat.) 153.

S.S. 'Lotus' (France v Turkey), [1927] PCIJ Rep Series A No. 10.

Nuremberg Judgment, France and ors v Goering (Hermann) and ors, Judgment and Sentence, [1946] 22 IMT 203. - The Trial of German Major War Criminals: Proceedings of the International Military tribunal sitting at Nuremberg, Germany, Part 22 (22nd August 1946 to 1st October 1946).

Tokyo War Crimes Trials, The International Military Tribunal for the Far East (3 May 1946 to 12 November 1948), established by its Charter, 1946, 14 *DSB* (Article 5 a)) 361, 362 (1946) (Judgment reprinted in reprinted in R. Pritchard (ed); The Tokyo Major War Crimes Trial (1998), at 1190).

United States v Krauch et al., Military Tribunal VI, Judgment 29th July 1948.

United States v Krupp, Von Bohlen und Halbach et al., Nuremberg Military Tribunal (NMT) III, Judgment 31st July 1948.

In re List, at 1241, 11 Trials of War Criminals 759 (1946-49) (US mil. Trib. Nuremberg 1948).

United States v Von Leeb et al., Military Tribunal XII, Case No.12, Judgment 27th October 1948.

United States v Von Weizsaecker et al., Military Tribunal XI, Case No. 11, Judgment 11th April 1949.

The Hadamar Trial, 1 Law Reports of Trials of War Criminals 46 (1949) (US Mil. Commission – Wiesbaden 1945).

Reparations for Injuries Suffered in the Service of the United Nations [1949] ICJ, Opinion, 11 April 1949.

Corfu Channel case (United Kingdom of Great Britain and Northern Ireland v Albania), [1949] ICJ Rep 4.

United States v Otto Ohlendorf, (Einsatzgruppen), IV Trials of War Criminals Before the Nuremberg Military Tribunals Under Control Council Law No. 10 (1950).

The Asylum Case [1950] ICJ Rep. 266.

Monetary Gold case (Italy v France, United Kingdom & United States), (1954) ICJ Rep. 19.

Certain Expenses of the United Nations (Article 17 (2) of the Charter), (Advisory Opinion) [1962], ICJ Reports, 151.

North Sea Continental Shelf cases (Germany v Denmark) [1969] ICJ Rep. 3.

Case concerning the Barcelona Traction Light and Power Company Ltd. (Belgium v Spain) [1970] ICJ Rep 3.

Case Concerning Military and Paramilitary Activities in and against Nicaragua (Nicaragua v United States) (Merits), [1986] ICJ Rep. 14.

The Prosecutor of the Tribunal v Dusan Tadic, (1994) Case No. IT-94-I-T.

Loizidou v Turkey (1996) VI Eur.Ct.H.R. 2216.

R v Bow Street Metropolitan Stipendiary, *ex parte* Pinochet Ugarte (Amnesty International Intervening) (no. 3) [1997] UKHL 17.

Prosecutor v JP Akayesu, Case No. ICTR-96-4-T, Judgment, 2 September 1998.

Prosecutor v Furundzija, Tribunal for Former Yugoslavia, Case No. 17-95-17/1-T (1998).

R v Bow Street Metropolitan Stipendiary, *ex parte* Pinochet Ugarte (Amnesty International Intervening) (no. 3) [1997] UKHL 17.

R v Bartle et al., ex parte Pinochet, [1999] UKHL, 38 ILM.

Prosecutor v Kupreskic, Case No IT-95-16-T, Judgment (Jan 14, 2000).

Re Bouterse, Case No. HR 00749/01 CW2323, NL, 18 September 2001.

Arrest Warrant of 11 April 2000 (Democratic Republic of the Congo v Belgium), Judgment, 14 February 2002, ICJ Reports 2002.

Oil Platforms (Iran v US) [2003] ICJ Rep 161, (Oil Platforms case).

Legal Consequences of the Construction of a Wall in the Occupied Palestinian Territory, [2004] ICJ Rep 136, (*The Wall Opinion*).

Case Concerning Armed Activities on the Territory of the Congo (Congo v Uganda), [2005] ICJ Rep. 168.

R v Jones [2006] UKHL 16.

Application of the Convention on the Prevention and Punishment of the Crime of Genocide (Bosnia & Herzegovina v Serbia) [2007] ICJ General List No. 91.

Republic v Mohamed Ahmed Dahir & Ten (10) [2010] SCSL. 81 (Sey.).

Khurts Bat v The Investigating Judges of the German Federal Court, Judgment of 29 July 2011 (2011) EWHC, 2029

Jurisdictional Immunities of the State (Germany v Italy: Greece intervening) [2012] ICJ, Judgment of 3 February 2012.

A v Ministere Public de la Confederation, 25 July 2012, Swiss Federal Criminal Tribunal, available at http://www.trialch.org/fileadmin/user_upload/documents/affaires/algeria/BB.2011.140.pdf, accessed 27 August 2021.

Whaling in the Antarctic (Australia v Japan: New Zealand intervening) [2014] Judgment, ICJ Reports, 226.

ICC Proceedings

The Prosecutor v Omar Hassan Ahmad Al Bashir, Arrest Warrant of 4 March 2009, ICC-02/05-01/09-1;

The Prosecutor v Omar Hassan Ahmad Al Bashir, Arrest Warrant of 12 July 2010, ICC-02/05-01/09-95.

Pre-Trial Chamber I, Corrigendum to the Decision Pursuant to Article 87 (7) of the Rome Statute on the Failure by the Republic of Malawi to Comply with the Cooperation Requests Issued by the Court with Respect to the Arrest and Surrender of Omar Hassan Ahmad Al Bashir (ICC-02/05-01/09), 13 December 2011, available at https://www.icc-cpi.int/CourtRecords/CR2011 21750.PDF, last accessed 3rd September 2021.

Pre-Trial Chamber II, Decision under article 87 (7) of the Rome Statute on the non-compliance by South Africa with the request by the Court for the arrest and surrender of Omar Al-Bashir, (ICC-02/05-01/09), 6 July 2017, available at https://www.icc-cpi.int/CourtRecords/CR2017_04402.PDF, last accessed 3rd September 2021.

Pre-Trial Chamber II, Decision under article 87 (7) on the non-compliance by Jordan with the request by the Court for the arrest and surrender of Omar Al-Bashir, (ICC-02/05-01/09), 11 December 2017, available at https://www.icc-cpi.int/CourtRecords/CR2017_07156.PDF, last accessed 3rd September 2021.

Appeals Chamber, Judgment in the Jordan Referral re Al-Bashir Appeal (ICC-02/05-01/09) OA2, 6 May 2019, available at https://www.icc-cpi.int/CourtRecords/CR2019_02856.PDF, last accessed 3rd September 2021.

Pre-Trial Chamber III, Decision Pursuant to Article 15 of the Rome Statute on the Authorisation of an Investigation into the Situation in the People's Republic of Bangladesh/Republic of the Union of Myanmar, ICC-01/19, 14 November 2019.

Appeals Chamber, Judgment on the appeal against the decision on the authorisation of an investigation into the situation in the Islamic Republic of Afghanistan, 5 March 2020, ICC-02/17 OA4, available at https://www.icc-cpi.int/CourtRecords/CR2020_00828.PDF, last accessed 3rd September 2021.

ICC Pre-Trial Chamber I, Decision on the 'Prosecution request pursuant to article 19 (3) for a ruling on the Court's territorial jurisdiction in Palestine', 5 February 2021, ICC-01/18, available at https://www.icc.cpi.int/CourtRecords/CR2021_01165.PDF, last accessed 3rd September 2021.

Table of Treaties and other documents

Hague Convention (I) of 1899 for the Pacific Settlement of International Disputes (29 July 1899).

Hague Convention (II) of 1907 for the Pacific Settlement of International Disputes (18 October 1907).

Treaty of Peace between the Allied and Associated Powers and Germany (28 June 1919), UKTS 14, Cmd. 1746.

Covenant of the League of Nations (1919), UKTS 4, Cmd. 153.

Geneva Protocol for the Pacific Settlement of International Disputes, 1924, 2 Int.Leg. 1378.

League of Nations, General Treaty for Renunciation of War as an Instrument of National Policy (Kellogg-Briand Pact), 27 August 1928, LNTS (1929) XCIV, 58.

League of Nations, Conference for the Reduction and Limitation of Armaments, Geneva, 1932-4.

1933 Disarmament Conference, 147 LNTS 67, 148.

London Charter of August 8, 1945, 82 UNTS, 279, Article 6 (a).

United Nations, Charter of the United Nations, 24 October 1945, 1 UNTSXVI.

United Nations, Statute of the International Court of Justice (18 April 1946) 33 UNTS 993 (ICJ Statute).

Convention on the Prevention and Punishment of the Crime of Genocide, 9 December 1948.

Control Council Law No. 10, Punishment of Persons Guilty of War Crimes, Crimes Against Peace and Against Humanity (1945), (reprinted in Telford Taylor, Final Report to the Secretary of the Army on Nuremberg War Crimes Trials, under Control Council Law No. 10, 250-153 (1949)).

European Convention of Human Rights (Convention for the Protection of Human Rights and Fundamental Freedoms), Nov. 4, 1950, 213 UNTS 221.

European Union, Treaty Establishing the European Community (Consolidated Version), Rome Treaty, 25 March 1957.

United Nations, Vienna Convention on the Law of Treaties (23 May 1969, entered into force 27 January 1980) 1155 UNTS 331 (VCLT).

Agreement Establishing the Caribbean Court of Justice, 14 February 2001.

Agreement between the United Nations and the Government of Sierra Leone on the Establishment of a Special Court for Sierra Leone 2002.

Rome Statute of the International Criminal Court 1998 (adopted 17 July 1998, entered into force 1 July 2002) 2187 UNTS 38544.

European Commission on Human Rights Reports

Sub-Commission on Human Rights, Res. 2000/24, 'Role of Universal or Extraterritorial Competence in Preventive Action against Impunity', 18 August 2000, E/CN.4/SUB.2/RES/2000/24.

Commission on Human Rights, Res. 2000/68, 'Impunity', 26 April 2000, E/CN.4/RES/2000/68.

Commission on Human Rights, Res. 2000/70, 'Impunity', 25 April 2001, E/CN.4/RES/2000/70.

United Nations Documents

Security Council Resolutions

22 (1947)	661 (1990)
82 (1950)	662 (1990)
	, ,
83 (1950)	664 (1990)
84 (1950)	665 (1990)
, ,	, ,
85 (1950)	666 (1990)
88 (1950)	667 (1990)
90 (1951)	669 (1990)
	, ,
189 (1964)	670 (1990)
255 (1968)	808 (1993)
284 (1970)	822 (1993)
, , , , ,	, ,
294 (1971)	827 (1993)
326 (1973)	853 (1993)
· · · · · · · · · · · · · · · · · · ·	, ,
328 (1973)	884 (1993)
386 (1976)	955 (1994)
387 (1976)	1234 (1999)
	, ,
405 (1977)	1244 (1999)
411 (1977)	1304 (2000)
	, ,
418 (1977)	1341 (2001)
419 (1977)	1355 (2001)
423 (1978)	1528 (2004)
` '	` ′
424 (1978)	1862 (2009)
428 (1978)	1907 (2009)
445 (1979)	1973 (2011)
	, ,
447 (1979)	2023 (2011)
454 (1979)	2093 (2013)
` '	` ′
455 (1979)	2100 (2013)
475 (1980)	2136 (2014)
496 (1981)	2147 (2014)
	, ,
502 (1982)	2166 (2014)
507 (1982)	2202 (2015)
527 (1982)	2211 (2015)
· · · · · · · · · · · · · · · · · · ·	` ′
535 (1983)	2232 (2015)
546 (1984)	2277 (2016)
· · · · · · · · · · · · · · · · · · ·	` ′
554 (1984)	2293 (2016)
567 (1985)	2297 (2016)
568 (1985)	2348 (2017)
` '	
571 (1985)	2360 (2017)
572 (1985)	2463 (2019)
573 (1985)	2472 (2019)
` '	2472 (2017)
574 (1985)	
577 (1985)	
· · · · · · · · · · · · · · · · · · ·	
580 (1985)	
581 (1986)	
598 (1987)	
· · · · · · · · · · · · · · · · · · ·	
602 (1987)	
611 (1988)	
660 (1990)	
000 (1770)	

Vetoed Draft Resolutions

(For a full list, see Veto List, compiled by the UN Dag Hammerskjold Library)

S/2017/962
S/2017/970
S/2017/1060
S/2018/156
S/2018/321
S/2018/508
S/2019/756
S/2019/961
S/2020/654
S/2020/667
S/2020/852

General Assembly Resolutions

Uniting for Peace (3 November 1950) UN Doc. A/Res/377 (V).

Intervention of the Central People's Government of the People's Republic of China in Korea (5 November 1951), UN Doc. A/Res/498 (V).

Question of defining aggression (20 December 1952) UN Doc. A/RES/688 (VII).

Question of defining aggression (4 December 1954) UN Doc. A/RES/895 (IX).

Question of defining aggression (29 November 1957) UN Doc. A/RES/1181 (XII).

'Need to expedite the drafting of a definition of aggression in the light of the present international situation' (18 December 1967) UN Doc. A/RES/2330 (XXII).

Declaration on Principles of International Law concerning Friendly Relations and Cooperation among States in accordance with the Charter of the United Nations (24 October 1970) UN Doc. A/RES/2625 (XXV).

Definition of Aggression (14 December 1974), XXIX, UN Doc A/RES 3314.

Territorial Integrity of Ukraine (27 March 2014), UN Doc A/RES/68/262.

Open-Ended Working Group on Developments in the Field of Information and Telecommunications in the Context of International Security (5 December 2018) UN Doc A/RES/73/27.

Group of Governmental Experts on Advancing Responsible State Behaviour in Cyber Space in the Context of International Security (22 December 2018) UN Doc A/RES/73/266.

Problem of the militarization of the Autonomous Republic of Crimea and the city of Sevastopol, Ukraine, as well as parts of the Black Sea and the Sea of Azov (14 December 2020), UN Doc A/RES/75/29.

Resolutions on the Suez Crisis in the 1st Emergency Special Session (1956)

A/RES/997 (2 November 1956) A/RES/998 (4 November 1956) A/RES/999 (4 November 1956)

UN Ancillary documents

Statement of the Four Sponsoring Governments on Voting Procedure in the Security Council, UNCIO Document 852, III/1/37(1) [8] UN Doc A/578 (7 June 1945).

Commission III, Security Council, Committee 3, Enforcement Arrangements, Report of Mr. Paul-Boncour, Rapporteur, on Chapter VIII Section B, at The United Nations Conference on International Organization, Doc. 881 III/3/46, 10th June, 1945.

United Nations Commission on International Organisation, Draft Suggestion by Bolivia, United Nations, New York, 1948, Vol. 3, 577.

Lie T, UN Secretary-General's Report on the Nuremberg Trials (1949) UN Doc. A/CN.4/5, UN Sales No. 1949 V.7.

UN Press Release GA/9430, Report of the Working Group on the Question of Equitable Representation on and Increase in the Membership of the Security Council and Other Matters Related to the Security Council, A/AC.247/1998/WP.1/Rev.2, adopted by the General Assembly on 24 August 1998,

https://un.org/press/en/19981980824.ga9430.html, accessed 20 January 2021).

Annan K, Report of the Secretary-General on the work of the Organization, UN GAOR, 54th Sess., 4th plenary meeting at 2, UN Doc. A/54/PV.4 (20th September, 1999), https://documents-dds-ny.un.org/doc/UNDOC/GEN/N99/858/23/PDF/N9985823.pdf?

Historical Review of Developments Relating to Aggression (UN Publications, 2003).

The High-Level Panel Report: 'A More Secure World: Our Shared Responsibility, Report of the High-Level Panel on Threats, Challenges and Change', UN Doc. A/59/565 (2004).

Annan, K, 'In Larger Freedom: Towards Development, Security and Human Rights for All', Report of the Secretary-General (2005), UN Doc. A/59/2005.

2005 World Summit Outcome, GA Res. 60/1 (October 24, 2005).

Security Council Press Release 9396, SC/9396, 11 July 2008, https://www.un.org/press/en/2008/sc9396.doc.htm, accessed 21 January 2021.

Speeches in UN General Assembly, 23 December 2007, 5 September 2008, 107, A/62/49 (Vol. III), http://undocs.org/A/62/PV.49 (Vol. III), accessed 20 January 2021.

Decision 62/557 of 15 September 2008, https://www.securitycouncilreport.org/atf/cf/%7B65BFCF9B-6-D27-4E9C-8CD3-CF6E4FF96FF9%7D/Decision%2062 557.pdf, accessed 20 January 2021.

Ki-Moon, Ban, Report of the Secretary-General Prepared on the Basis of Comments and Observations of Governments: The Scope and Application of the Principle of Universal Jurisdiction, UN Doc. A/65/181 (July 29, 2010).

Speeches in UN General Assembly, 12 November 2014, 6, A/69/PV.49, at http://undocs.org/A/69/PV.49, accessed 20 January 2021.

Research Report on the Veto, 19 October 2015, http://www.securitycouncilreport.org, accessed 12th May 2020.

Security Council Report, In Hindsight: The Security Council and the International Court of Justice, 28 December 2016, https://www.securitycouncilreport.org/monthly-forecast/2017-

01/in hindsight the security council and the international court of justice.php, accessed 3rd March 2021.

Security Council Report on the UN Security Council Working Methods, 'The Veto', 30th October 2017, http://www.securitycouncilreport.org/un-security/council-working-methods/, accessed 12th May 2020.

United Nations, World Water Development Report 2019, available at http://www.unwater.org/publications/world-water-development-report-2019/, accessed 21st January 2021.

Compilation of all recommendations made by the Independent International Fact-Finding Mission on Myanmar', A/HRC/42/CRP.6, 16 September 2019, available at https://www.ohchr.org/Documents/HRBodies/HRCouncil/FFM-Myanmar/201090916/A HRC 42 CRP.6.pdf, last accessed 3rd September 2021.

UN GA Press Release 12217, GA/12217 of 25 November 2019, https://www.un.org/press/en/ga12217.doc.htm, accessed 21 January 2021.

Food and Agriculture Organization of the United Nations, 'The State of Food and Agriculture' (SOFA), 2020 Report, available at http://fao.org/3/cb1447en/online/cb1447en.html, accessed 21st January 2021.

UN GA Press release 12288, GA/12288, 16 November 2020, https://www.un.org/press/en/2020/ga12288.doc.htm, accessed 21 January 2021.

UN GA Press Release 12289, GA/12289 of 17 November 2020, available at http://www.un.org/press/en/2020/ga12289.doc.htm, accessed 21 January 2021.

Practices, Procedures and Working Methods of the Security Council, https://www.un.org/securitycouncil/content/procedures-and-working-methods, accessed 10th May 2020.

UN Documentation Research Guide, Veto List, as compiled by the Dag Hammarskjöld Library, available at http://www.research.un.org/en/docs/sc/quick, last accessed 23rd January 2021.

Documents related to the ICC

Proposal by Germany, 1997, A/AC.249/1997/WG.1/DP.20.

Proposal by Cameroon, 1998, UN Doc. A/CONF.183/C.I./L.39, Suggested Article 10.

'German Proposal of 1998', contained in the Draft Statute for an International Criminal Court, UN Doc. A/Conf. 183/2/Add.1, Articles 6 (b), 7, 9 (further option, and further option for Article 7). Available at https://www.documents-dds-ny.un.org, last accessed 07 December 2020.

Proposal by the Arab States and Iran, 1 July 1998, UN Doc. A/CONF.183/C.1/L.37.

Proposal by Germany, 1999, in PCNICC/1999/INF/2.

Proposal by Egypt and Italy on the definition of aggression', 1999, A/AC.249/1997/WG.1/DP.6, in PCNICC/1999/INF/2.

Proposal by Greece and Portugal, 1999, PCNICC/1999/WGCA/DP.1.

Proposal by the Russian Federation, 1999, PCNICC/1999/DP.12.

Discussion Paper Proposed by the Coordinator (9 December 1999), UN Doc. PCNICC/1999/WGCA/RT.1.

Proposal submitted by Colombia, 1st March 2000, PCNICC/2000/WGCA/DP.1. 'Suggestion made orally by Italy on 13 March 2000 with regard to a structure for discussion on the crime of aggression', PCNICC/2000/WGCA/DP.3

Comments by Colombia on the Italian proposal made orally in the Working Group on 13 March 2000, PCNICC/2000/WGCA/DP.2.

Proposal submitted by Germany: the Crime of Aggression – A Further Informal Discussion Paper, 13 November 2000, in PCNICC/2000/WGCA/DP.4,

Proposal submitted by Bosnia and Herzegovina, New Zealand and Romania on jurisdictional conditions, 23 February 2001, PCNICC/2001/WGCA/DP.1.

Proposal by Guatemala, PCNICC/2001/WGCA/DP.2, 26 September 2001.

Historical Review of Developments Relating to Aggression, Prepared by the Secretariat, Doc.PCNICC/2002/WGCA/L.1 (2002), available at http://www.un.org/law/icc/documents/aggression/aggressiondocs.htm, accessed 12 October 2020.

Definition of the Crime of Aggression and Conditions for the Exercise of Jurisdiction: Discussion Paper Proposed by the Coordinator, 1 April 2002, UN Doc. PCNICC/2002/WGCA/RT.1.

Elements of the Crime of Aggression – Proposal submitted by Samoa, 2nd June 2002, PCNICC/WGCA/D.P.2.

Proposed text on the definition of the crime and act of aggression submitted for the Delegation of Colombia (1 July 2002), PCNICC/2002/WGCA/DP.3.

Proposal submitted by Belgium, Cambodia, Sierra Leone, and Thailand, 8 July 2002, PCNICC/2002/WGCA/DP.5.

Discussion Paper Proposed by the Coordinator, 11 July 2002, UN Doc. PCNICC/2002/WGCA/RT.1/Rev.2 in 'Report of the Preparatory Commission for the International Criminal Court', UN Doc. PCNICC/2002/2/Add.2, Part II.

Preparatory Commission for the International Criminal Court, Working Group on the Crime of Aggression, 'Historical review of developments relating to aggression, prepared by the Secretariat', PCNICC/2002/WGCA/L.1., http://www.un.org/law/icc/documents/aggression/aggressiondocs.htm, accessed 12 October 2020.

Proposal submitted by Cuba, 6th February 2003, Assembly of States Parties, First session, ICC-ASP/1/L.4.

Report of the Special Working Group on the Crime of Aggression, ICC-ASP/4/SWGCA/1 (1 Dec. 2005) Annex II.A.

Informal Intersessional Meeting of the Special Working Group on the Crime of Aggression, held at the Liechtenstein Institute on Self-Determination, Woodrow Wilson School, at Princeton University, New Jersey, United States, from 8 to 11 June 2006, (the 2006 Princeton Report), ICC-ASP/5/SWGCA/INF.1.

Discussion Paper Proposed by the Chairman, 16 January 2007, ICC-ASP/5/SWGCA/2 (not published in ASP Official Records but included in the *Travaux Preparatoires*).

Coalition for the International Criminal Court, 'Observations about the Discussion Paper Proposed by the Chairman', ICC-ASP/5/SWGCA/2 (25 Jan. 2007).

Proposal presented by Belgium on the Question of Jurisdiction of the Court with Respect to the Crime of Aggression, 29 January 2007, ICC-ASP/5/SWGCA/W.P.1.

Informal Intersessional Meeting of the Special Working Group on the Crime of Aggression, Princeton University, 11-14 June 2007, (the 2007 Princeton Report), ICC-ASP/6/SWGCA/INF.1 in ASP Official Records, ICC-ASP/6/20, Annex III, 96.

Report of the Special Working Group on the Crime of Aggression (December 2007) (ICC-ASP/6/SWGCA/1, ASP 6/20, Annex II, 87.

Discussion Paper on the Crime of Aggression Proposed by the Chairman: Revision January 2009, 19 February 2009, ICC/ASP/7/SWGCA/INF.1, in the 'Report of the Special Working Group on the Crime of Aggression' (2009) ICC-ASP/7/SWGCA/2, *ASP Official Records*, ICC-ASP/7/20/Add.1, Annex II.

Informal Intersessional Meeting on the Crime of Aggression, hosted by the Liechtenstein Institute on Self-Determination, Woodrow Wilson School, at the Princeton Club, New York, from 8-10 June 2009, ICC-ASP/8/INF.2 (2009 Princeton Report).

Report of the Working Group on the Crime of Aggression, 2010, RC/5 in Review Conference Official Records, RC/11, Part II, Annex III, 45.

Resolution 6, Review Conference, Assembly of States' Parties, 'The Crime of Aggression', ICC Doc. RC/Res. 6 (11 June 2010).

Explanations of Position, in Review Conference Official Records, RC/11, Annexes VII-IX, 122.

Declaration of Non-Acceptance of Jurisdiction by Kenya, MFA.1NT.8/14AVOL.X (86) 30th November, 2015.

Report on the facilitation on the activation of the jurisdiction of the International Criminal Court over the crime of aggression (2017) ICC-ASP/16/24, 27 November 2017.

Resolution 5, ICC-ASP/16/Res.5, 16th Assembly of States Parties, 14th December 2017.

Declaration of Non-Acceptance of Jurisdiction by Guatemala, 16 January 2018. http://ww.icc-cpi.int, accessed 24 November 2020.

Handbook on the Ratification and Implementation of the Kampala Amendments on the Crime of Aggression, http://crimeofaggression.info/docments/1/handbook.pdf, accessed 19th March 2020.

ILC Documents

First Report on a Draft Code of Offences against the Peace and Security of Mankind, UN Doc. A/CN.4/25, YB ILC (1950), vol. II, 253.

Second Report on a Draft Code of Offences against the Peace and Security of Mankind, including commentary, YB ILC (1951), II 133.

Draft Code of Offences against the Peace and Security of Mankind, (1954), UN Doc. A/CN.4/44.

Report of the International Law Commission on the Work and Its Thirty-Second Session, UN Doc. A/35/10 (1980).

Ninth Report on a Draft Code of Crimes against the Peace and Security of Mankind (1991) UN Doc. A/CN.4/435, and Add. 1 (Feb 8 and Mar 15, 1991).

Yearbook of the ILC (1991), Vol. II (Part Two).

1994 ILC Draft Statute (with Commentary), Report of the International Law Commission, 46th Session, Sept. 1, 1994, UN Doc. A/49/355; GAOR, 49th Sess., Supp. No. 10 (1994).

Draft Code of Crimes against the Peace and Security of Mankind, Report of the International Law Commission on the Work of its 48th Session, May 6-July 26, 1996, UN Doc. A/51/10 (SUPP) (Jan 1, 1996).

Commentary on 1996 ILC Draft Code of Crimes against the Peace and Security of Mankind, Yearbook ILC (1996), Vol. II.

Draft Articles on State Responsibility for Internationally Wrongful Acts (with commentaries), Report of the International Law Commission, 53rd Session, 2 (2) Yearbook of the International Law Commission, 2001, Annex to GA Resolution A/RES/56/83 (12 December 2001).

International Law Commission, Final Report of the Study Group on Fragmentation of International Law, UN Doc. A/CN.4/L.682 (13 April 2006)

International Law Commission, Draft Conclusions on Subsequent agreements and subsequent practice in relation to the interpretation of treaties, adopted on second reading in May 2018, UN Doc. A/CN.4/L.907 (2018).

International Law Commission, Report of the International Law Commission on Peremptory Norms of General International Law (*jus cogens*), Chapter V, (16th May 2019) UN Doc A/74/10.

Bibliography

Books

Aquinas St T, Summa Theologiae (XXXV Blackfriars 1972).

Arend A and Beck R, *International Law and the Use of Force: Beyond the UN Charter Paradigm* (Routledge 1993).

Arend A, Legal Rules and International Society (OUP 1999).

Augustine St, Contra Faustum Manichaeum.

Aust A, Handbook of International Law (CUP 2005).

Bassiouni MC, Crimes against Humanity in International Criminal Law (Nijhoff 1999).

Bassiouni M C, Introduction to International Criminal Law, (Nijhoff, 2nd edn. 2013).

Bernstorff J v, The Public International Law Theory of Hans Kelsen: Believing in Universal Law (CUP 2010).

Bluntschli J, *The Theory of the State*, English translation from the 6th German edn (Clarendon 1885).

Bobbit P, Terror and Consent: The Battle for the Twenty-First Century (Knopf 2008).

Bourantonis D, *The History and Politics of UN Security Council Reform* (Routledge, 2005).

Bowett D, Self-Defence in International Law (Praeger 1958).

Brierly J, *The Law of Nations: An Introduction to the Modern Law of Peace* (Clarendon 1963).

Brownlie I, International Law and the Use of Force by States, OUP (1963).

Brownlie I, Principles of Public International Law (6th edn, OUP 2003).

Cassese A, *International Criminal Law* (3rd edn, OUP 2013).

Crawford J, Brownlie's Principles of Public International Law (9th edn, OUP 2012).

Cryer R et al, An Introduction to International Criminal Law and Procedure (CUP 2010).

Dinstein Y, War, Aggression, and Self-Defence (CUP 5th edn 2011).

Dinstein Y, War, Aggression, and Self-Defence (CUP 6th edn 2017).

Dworkin R, A Matter of Principle (HUP 1985).

Dworkin R, Law's Empire, (HUP 1986).

Ferencz B, Defining International Aggression, The Search for World Peace (Oceana 1975).

Finnis J, Natural law and natural rights (2nd edn, OUP 2011).

Fox H, The Law of State Immunity (3rd edn. OUP 2008).

Franck T, Fairness in International Law and Institutions (OUP 1995).

Franck T, Recourse to Force (CUP 2002).

Gelvin J, *The Israel-Palestine Conflict – One Hundred Years of War* (CUP 2014, 3rd edn).

Gentili A, De Jure Belli I, para VI, 48-52 (Clarendon 1933).

Green L, International Law (1984).

Grotius H, De Jure Belli ac Pacis II (Clarendon 1925).

Harris D J, Cases and Materials on International Law (7th edn, Sweet & Maxwell 2010).

Hart H.L.A, Essays in Jurisprudence and Philosophy (OUP 1983).

Hart H.L.A, The Concept of Law (2nd edn. OUP 1994).

Hathaway O and Shapiro S, *The Internationalists: How a radical plan to outlaw war remade the world* (Simon & Schuster 2017).

Henkin L, *How Nations Behave: Law and Foreign Policy*, 144-5 (Columbia University Press 2nd ed. 1979).

Henkin L, International Law: Politics and Values (Nijhoff 1995).

Higgins R, *Problems and Process, International Law and How We Use It,* (OUP 2010 reprint).

Hobbes T, Leviathan (Touchstone 2008).

Jian C, China's Road to the Korean War (Columbia University Press 1994).

Kelsen H, General Theory of Law and State (Harvard University Press 1945).

Koskenniemi M, From Apology to Utopia: The Structure of International Legal Argument (1989 CUP, reissue 2005).

Koskenniemi M, The Politics of International Law (Hart 2011).

Lauterpacht H, The Function of Law in the International Community (OUP 1933).

Lauterpacht H, Recognition in International Law (1947).

May L, Aggression and Crimes against Peace (OUP 2008).

McDougall C, *The Crime of Aggression under the Rome Statute of the International Criminal Court*, (CUP 2013).

McDougal M and Feliciano F, Law and Minimum World Public Order: The Legal Regulation of International Coercion, (YUP 1961).

McDougal M and Reisman WM, International Law Essays (Minneola 1981).

Murphy S, *Humanitarian Intervention: The United Nations in an evolving world order*, University of Pennsylvania Press (1996).

Murphy M, Natural Law in Jurisprudence and Politics (CUP 2006).

Nussbaum A, A Concise History of the Law of Nations (Macmillan, 2nd edn 1954).

Orakhelashvili A, *The Interpretation of Acts and Rules in Public International Law* (OUP 2008).

Rivier A, *Principes du Droit des Gens*, Tome I (Arthur Rousseau 1896).

Robb J, *Brave New War: The Next State of Terrorism and the End of Globalisation* (Wiley 2007).

Sayapin S, The Crime of Aggression in International Criminal Law – Historical Development, Comparative Analysis and Present State (Asser 2014).

Schachter O, International Law in Theory and Practice, 123-5 (Nijhoff 1991).

Edited Books

Barriga S and Kress C (eds), *Travaux Preparatoires of the Crime of Aggression*, Crime of Aggression Library (CUP 2012).

Bassiouni M C (ed), *International Criminal Law* (Nijhoff, 2nd edn. 1999).

Dinstein Y, Tabory M (eds), *International Law at a Time of Perplexity: Essays in Honour of Shabtai Rosenne* (Nijhoff 1989).

Donnelly E, McGoldrick D and Rowe P (eds), *The Permanent International Criminal Court: Legal and Policy Issues* (Hart 2004).

Evans M (ed), International Law (OUP 2010 3rd edn).

Frishman M, Muller S. (eds), *The Dynamics of Constitutionalism in the Age of Globalisation* (Hague Academic Press 2010).

Kammerhofer J and d'Aspremont J (eds), *International Legal Positivism in a Post-Modern World* (CUP 2014).

Kress C and Barriga S (eds), *The Crime of Aggression: A Commentary*, Crime of Aggression Library, Vol. I and II (CUP 2017).

Lauterpacht E (ed), *International Law: Collected Papers of Hersch Lauterpacht*, (CUP 2009).

Lillich R (ed), Humanitarian Intervention and the United Nations (UPVA 1973).

Mc Cormack T and Simpson G (eds), *The Law of War Crimes, National and International Approaches* (Kluwer 1997).

Olusanya O (ed), Rethinking International Criminal Law: The Substantive Part (Europa Law, 2007).

Pegis A (ed), Basic Writings of St. Thomas Aquinas (Hackett Publishing Co. 1997).

Perreau-Saussine A, Murphy J (eds), *The Nature of Customary Law* (CUP 2007). Peters A, Klabbers J and Ulfstein G (eds), *The Constitutionalization of International Law* (OUP 2009).

Politi M and Nesi G (eds), *The International Criminal Court and the Crime of Aggression* (Routledge 2004).

Weller M and Rylatt J (eds), *The Oxford Handbook of the Use of Force in International Law* (OUP 2015).

Wet, de E. & Vidma J. (eds) *Hierarchy in International Law: The Place of Human Rights* (OUP 2012).

Wilson R (ed), $Human\ Rights\ in\ the\ 'War\ on\ Terror'\ (CUP\ 2005).$

Journal Articles

Akande D, 'The Jurisdiction of the ICC over Nationals of Non-Parties: Legal Basis and Limits', JICJ 1 (2003), 618.

Akande D, 'Prosecuting Aggression: The Consent Problem and the Role of the Security Council', (2011) Oxford Legal Studies Research Paper No. 10/2011, 37.

Akande D and Tzanakopoulos A, 'Treaty Law and ICC Jurisdiction Over the Crime of Aggression' (2018), 29 EJIL, 939.

Barriga S and Grover L, 'A Historic Breakthrough on the Crime of Aggression' (2011) 105 (3) AJIL, 517.

Bassiouni M C, 'A Functional Approach to 'General Principles of International Law' (1990) 11 Mich. J. Int'l. L. 768.

Bassiouni M C, in 'International Crimes: Jus Cogens and Obligatio Erga Omnes' (1996), 59 Law and Contemporary Problems (4).

Bellamy A, 'Responsibility to Protect or Trojan Horse? The Crisis in Darfur and Humanitarian Intervention after Iraq' (2005), 19 (2) Ethics Int. Aff. 31.

Besson S, 'State Consent and Disagreement in International Law-Making' (2016), 29 Leiden J Int'l L 289.

Bianchi A, 'Immunity versus Human Rights: The Pinochet Case' (1999) EJIL 10, 217.

Bianchi A, 'Security Council's Anti-Terror Resolutions and their Implementation by Member States' (2006) 4 J Int'l Crim. Just. 1044.

Bianchi A, Gazing at the Crystal Ball (again): State Immunity and *Jus Cogens* beyond *Germany v Italy* (2013), 4 J Int'l Dispute Settlement (3).

Blokker N, 'The Crime of Aggression and the United Nations Security Council' (2007), 20 Leiden JIL, 881.

Caplan L M, 'State Immunity, Human Rights, and *Jus Cogens:* A critique of the Normative Hierarchy Theory (2003) 97 AJIL 741.

Carswell A, 'Unblocking the UN Security Council: The *Uniting for Peace* Resolution' (2013) 18 (3) J.C.&S.L, 453.

Cassese, 'On Some Problematical Aspects of the Crime of Aggression' (2007), 20 Leiden J. Int'l L., 841.

Clark R, 'Ambiguities in Articles 5 (2), 121 and 123 of the Rome Statute', (2009) 41 Case W. Res. J. Int'l L., 413.

Clark R, 'Amendments to the Rome Statute of the International Criminal Court' (2010), 2 Go. J. Int'l L., 696.

Crowcroft B, 'Egypt's Other Nationalists and the Suez Crisis of 1956', (2016) 59 (1) Hist. J.

Davies T, 'France and the World Disarmament Conference of 1932-4' (2004) Dipl. Statecraft 15 (4) 765.

DeNicola C, 'A Shield for the "Knights of Humanity": The ICC should adopt a Humanitarian Necessity Defense to the Crime of Aggression', (2009) 30 (2) U. Pa. J. Int'l. L., 641.

Drumbl M, 'The Push to Criminalize Aggression: Something Lost Amid the Gains?' (2009), 41 (2) Case W. Res. J. Int'l L. 291.

Dworkin R, 'Natural Law Revisited' (1982), 34 (2) U. Fl. L. Rev. 165.

Dworkin R, 'A New Philosophy for International Law', (2013) Wiley, Phil. & Pub. Aff.

Ferencz B, 'Defining Aggression – The Last Mile' (1973) 12 Colum. J. Transnat'l L.

Ferencz B, 'The UN Consensus Definition of Aggression: Sieve or Substance?' (1975) 10 J Int'l & Econ. 701.

Ferencz B, 'Enabling the International Criminal Court to Punish Aggression' (2007) 6 Wash. U. Global Stud. L. Rev. 551.

Ferencz B, 'Speaking Frankly about the Crime of Aggression: Reconciliating Legitimate Concerns and Removing the Lock from the Courthouse Door', (2008), http://www.benferencz.org/arts/93b.html, accessed 19 March 2020.

Ferencz B, 'Ending Impunity for the Crime of Aggression' (2009) 41 Case W. Res. J. Int'l L. 281.

Ferencz D, 'Aggression Is No Longer a Crime in Limbo', FICHL Policy Brief Series No. 88 (2018), http://www.toaep.org/pbs-pdf/88-ferencz/, accessed 25 October 2020.

Fitzmaurice M, 'Consent to be Bound – Anything New Under the Sun' (2005) 74 Nord. J. Int'l L. 484.

Fuller L, 'Positivism and fidelity to law: A reply to Professor Hart' (1958) 71 (4) Harv. L. Rev. 630.

Glennon M, 'The Blank Prose Crime of Aggression' (2010) 35 Yale J. Int'l L. 71.

Gray C, 'The Use and Abuse of the International Court of Justice: Cases concerning the Use of Force after *Nicaragua*' (2003) 14 (5) EJIL, 867.

Graziano M and Mei L, 'The Crime of Aggression Under the Rome Statute and Implications for Corporate Accountability' (2017), 58 (4) Harv. Int'l L. J. 55.

Guzman A, 'Saving Customary International Law', (2005) 27 Mich. J. Int'l L., 115.

Guzman A, 'The Consent Problem in International Law' (2011), UC Berkeley: Berkeley Program in Law and Economics, https://escholarship.org/uc/item/04x8x174, accessed 6 November 2020.

Guzman A, 'Against Consent' (2012), 52 Virginia J Int'l L., 747.

Hammes T, 'Fourth Generation Warfare Evolves, Fifth Emerges' (2007) Mil. Rev., 14.

Hart H.L.A, 'Positivism and the Separation of Law and Morals' (1958) 71 (4) Harv. L. Rev. 593.

Hehir A, 'The Impact of the Security Council on the Efficacy of the International Criminal Court and the Responsibility to Protect' (2015), 26 Criminal Law Forum, 153.

Heinsch R, 'The Crime of Aggression after Kampala: Success or Burden for the Future' (2010), 2 Go. J. I. L., 713.

Heller K, 'Retreat from Nuremberg: The Leadership Requirement in the Crime of Aggression' (2007) 18 EJIL, 486.

Hosli M, Moody R, O'Donovan B, Kaniovski S and Little A, 'Squaring the circle? Collective and Distributive Effects of United Nations Security Council Reform' (2011) 6 (2) Rev. Int. Organ. 163.

Jacobs D, 'The Sheep in the Box: The Definition of the Crime of Aggression at the International Criminal Court' (2010) The Review Conference and the Future of the International Criminal Court, Proceedings of the First AIDP Symposium for Young Penalists in Tuebingen, Germany, Kluwer Law International, 131.

Joyner C, 'Arresting Impunity: The Case for Universal Jurisdiction in Bringing War Criminals to Accountability' (1996), 59 LCP 153.

Kingsbury B, 'The Concept of 'Law' in Global Administrative Law', (2009) 20 (1) EJIL, 23.

Koh H and Buchwald T, 'The Crime of Aggression: The United States Perspective' (2015) 109 AJIL, 267.

Koskenniemi M & Leino P, 'Fragmentation of International Law? Postmodern Anxieties' (2002) 15 Leiden J. Int'l L. 553.

Krasner S, 'The Hole in the Whole: Sovereignty, Shared Sovereignty, and International Law' (2004) 25 Mich. J. of Int'l L. 1075.

Kress C and von Holtzendorff L, 'The Kampala Compromise on the Crime of Aggression' (2010) 8 (5) JICJ, 1179.

Kress C, 'On the Activation of ICC Jurisdiction over the Crime of Aggression' (2018) 16 (1) JICJ 1.

Laettilae V and Yloenen A, 'United Nations Security Council Reform Revisited: A Proposal' (2019), 30 (1) Dipl. & Statecraft, 164.

Langan J, 'The Elements of St. Augustine's Just War Theory' (1984), 12 The Journal of Religious Ethics (1), 20.

Leclerc-Gagne E and Byers M, 'A Question of Intent: The Crime of Aggression and Unilateral Humanitarian Intervention' (2009), 41 (2) Case W. Res. J. Int'l Law, 378.

Mancini M, 'A Brand New Definition for the Crime of Aggression: The Kampala Outcome' (2012) 81 NordJIL 227.

Maogoto J, 'Aggression: Supreme International Offence still in search of definition' (2009), 5 (1) South. Cross Univ. L. Rev. 1.

McDougal M and Feliciano F, 'International Coercion and World Public Order: The General Principles of the Law of War' (1958), https://digitalcommons.law.yale.edu/fss_papers/2451/, accessed 25 November 2020.

Melling G and Dennett A, 'The Security Council veto and Syria: Responding to mass atrocities through the "Uniting for Peace" Resolution' (2017) 57 Indian J. Int'l L., 285.

Meron T, 'Defining Aggression for the International Criminal Court' (2001), 25 Suffolk Transnat'l L. Rev. 1

Milanovic M, 'Aggression and Legality: Custom in Kampala' (2012), 10 (1) JICJ, 165.

Morris M, 'High Crimes and Misconceptions: The ICC and Non Party States', (2001) 64 LCP, 13.

Orakhhelashvili A, 'Jurisdictional Immunities of the State (*Germany v Italy: Greece Intervening*)' (2012) AJIL 106.

Osiel M, 'The Banality of Good: Aligning Incentives against Mass Atrocity', (2005) 105 Colum. L. Rev. 1751.

Paulus A, 'Second Thoughts on the Crime of Aggression', (2009) 20 (4) EJIL, 1117.

Petty K, 'Sixty Years in the Making: The Definition of Aggression for the International Criminal Court' (2008), 31 (2) Hastings Int'l & Comp. L. Rev. 1.

Petty K, 'Criminalizing Force: Resolving the Threshold Question for the Crime of Aggression in the Context of Modern Conflict' (2009), 33 Seattle University Law Rev. 105.

Ponzetto G and Fernandez P, 'Case Law versus Statute Law: An Evolutionary Comparison' (2008), 37 (2) JLS, University of Chicago Press, 379.

Randall K, 'Universal Jurisdiction under International Law', (1988) 66 Tex. L. Rev., 758.

Reisinger-Coracini A, 'The International Criminal Court's Exercise of Jurisdiction Over the Crime of Aggression – at Last...in Reach...Over Some' (2010) 2 GoJIL, 745.

Roth K, 'Was the Iraq War a Humanitarian Intervention?' (19 August 2006), 5 (2) J. Mil. Ethics.

Scharf M, 'The ICC's Jurisdiction over the Nationals of Non-Party States: A Critique of the US Position', (2001) 64 Law and Contemporary Problems, 67.

Scharf M, 'Universal Jurisdiction and the Crime of Aggression', (2012) 53 (2) Harv. Int'l L. J., 358.

Scheffer D, 'The Complex Crime of Aggression' (2011), 43 Stud. Transnat'l Legal Pol'y 173.

Schmitt M (ed), 'International Law across the Spectrum of Conflict', 75 International Law Studies, US Naval War College, 539.

Simma B and Paulus A, 'The Responsibility of Individuals for Human Rights Abuses in Internal Conflicts: A Positivist View' (1999) 93 AJIL, 302.

Stein M, 'The Security Council, the International Criminal Court, and the Crime of Aggression: How Exclusive is the Security Council's Power to Determine Aggression?' (2005), 16 Indiana Int'l. & Comp. L. Rev. 1.

Reisman M, 'Reflections on the Judicialization of the Crime of Aggression', https://digitalcommons.law.yale.edu/fss_papers/4926, accessed 14 March 2020.

Stone J, 'Hopes and Loopholes in the 1974 Definition of Aggression' (1977) 71 AJIL 224.

Trahan J, 'The Rome Statute's Amendment on the Crime of Aggression: Negotiations at the Kampala Review Conference' (2011) 11 Int'l Crim L. Rev. 50.

Trapp K & Mills A, 'Smooth Runs the Water where the Brook is Deep: The Obscured Complexities of Germany v Italy' (2012), 1 Cambridge J Int'l Comp L (1).

Tzevelekos V, 'The Use of Article 31 (3) (c) in the Case Law of the ECtHR: An Effective Anti-Fragmentation Tool or a Selective Loophole for the Reinforcement of Human Rights Teleology – Between Evolution and Systemic Integration (2010), 31 Mich. J. Int'l L. 621.

Ulfstein G and Christiansen H, 'The Legality of the Nato Bombing in Libya' (2013) 62 (1), ICLQ, 159.

Weil P, 'Towards Relative Normativity in International Law' (1983), 77 AJIL, 413

Weisbord N, 'Prosecuting Aggression' (2008) 49 Harv. Int'l. L. J. 161.

Weisbord N, 'Conceptualizing Aggression' (2009), 20 Duke J. Comp. & Int'l. L. 1.

Weisbord N, 'Judging Aggression' (2011) 50 (1) Col. J. Transnat'l L., 82.

Wing S, 'French intervention in Mali: strategic alliances, long-term regional presence?' (5 February 2016) 27 (1) Small Wars & Insurgencies, 59.

Contributions in Yearbooks

Akehurst M, 'Custom as a Source of International Law' (1975) 47 BYIL 1.

Ambos K, 'The Crime of Aggression after Kampala' (2010) 53 GYIL, 487.

Lauterpacht H, 'The Law of Nations and the Punishment of War Crimes' (1944), 21 BYBIL 58.

International Law Blogs

Akande D, 'What Exactly was agreed in Kampala on the Crime of Aggression?' 21 June 2010, available at http://www.ejiltalk.org/what-exactly-was-agreed-in-kampala-on-the-crime-of-aggression/, accessed 20 October 2019.

Akande D, 'Treaty Interpretation, the VCLT and the ICC Statute: A response to Kevin Jon Heller & Dov Jacobs', August 25, 2013, http://www.ejiltalk.org/treaty-interpretation-the-vclt-and-the-icc-statute-a-response-to-kevin-jon-heller-dov-jacobs/, accessed 20 March 2020.

Akande, 'The ICC Assembly of States Parties Prepares to Activate the ICC's Jurisdiction over the Crime of Aggression: But Who Will be Covered by that Jurisdiction', (EJIL:Talk! 26 June 2017), https://www.ejiltalk.org/the-icc-assembly-of-states-parties-prepares-to-activate-the-iccs-jurisdiction-over-the-crime-of-aggression-but-who-will-be-covered-by-that-jurisdiction/, accessed 23 November 2019.

Akande D, 'The International Criminal Court Gets Jurisdiction Over the Crime of Aggression' (EJIL:Talk! 15 December 2017), https://www.ejiltalk.org/the-international-

<u>criminal-court-gets-jurisdiction-over-the-crime-of-aggression/</u>, accessed 20 February 2018.

Ambos K, "Solid jurisdictional basis"? The ICC's fragile jurisdiction for crimes allegedly committed in Palestine', EJIL:Talk! 2 March 2021, https://www.ejiltalk.org/solid-jurisdictional-basis-the-iccs-fragile-jurisdiction-for-crimes-allegedly-committed-in-palestine/, last accessed 3rd September 2021.

Barriga S, 'The Scope of ICC Jurisdiction Over The Crime of Aggression: A Different Perspective', (EJIL:Talk! 29 March 2017), https://www.ejiltalk.org/the-scope-of-icc-jurisdiction-over-the-crime-of-aggression-a-different-perspective/, accessed 20 October 2019.

Guilfoyle D, 'Is the International Criminal Court destined to pick fights with non-state parties? EJIL:Talk!, 14 July 2020, available at https://www.ejiltalk.org/is-the-international-criminal-court-destined-to-pick-fights-with-non-state-parties/, accessed 5th September 2021.

Hathaway O, 'Reengaging on Treaties and Other International Agreements (Part I), President Donald Trump's Rejection of International Law', (2 October 2020), https://www.justsecurity.org/72690/reengaging-on-treaties-and-other-international-agreements-part-ii-a-path-forward, accessed 1 December, 2020.

Heller K, 'The Sadly Neutered Crime of Aggression' (OpinioJuris, 13 June 2010), http://www.opiniojuris.org/2010/06/13, accessed 7 January 2018.

Heller K, 'My UN Presentation on the Aggression Amendments' (OpinioJuris, 5 May 2017), http://opiniojuris.org/2017/05/05/my-un-presentation-on-the-aggression-amendments/, accessed 30 October 2020.

Heller K, 'The Draft Resolution's Curious Paragraph 3', (OpinioJuris, 15 December 2017), http://opiniojuris.org/2017/12/15/the-curious-paragraph-3/, accessed 3 January 2018.

Hurd I, 'How not to argue against the Crime of Aggression: A Response to Michael Glennon' (2010) http://buffett.northwestern.edu/documents/working-papers/Buffett_10-001_Hurd.pdf, accessed 15 March 2020.

Jacobs D, 'Why the Vienna Convention should not be applied to the ICC Rome Statute: a plea for respecting the principle of legality' (August 24, 2013), *Spreading the Jam*, https://dovjacobs.com/2013/08/24/why-the-vienna-convention-should-not-be-applied-to-the-icc-rome-statute-a-plea-for-respecting-the-principle-of-legality/, accessed 20 March 2020.

Khin T, 'Universal Jurisdiction, the International Criminal Court, and the Rohingyan Genocide', OpinioJuris, 23 October 2020, available at

http://opiniojuris.org/2020/10/23/universal-jurisdiction-the-international-criminal-court-and-the-rohingya-genocide/, accessed 5th September 2021.

Marchuk I & Wanigasuriya A, 'The ICC concludes its preliminary examination in Crimea and Donbas: What's next for the situation in Ukraine?' EJIL:Talk! 16 December 2020, available at https://www.ejiltalk.org/the-icc-concludes-its-preliminary-examination-in-crimea-and-donbas-whats-next-for-the-situation-in-ukraine/, last accessed 3rd September 2021.

Milanovic M, 'The Syria Strikes: Still Clearly Illegal' (EJIL:Talk! 15 April 2018) http://ejiltalk.org.the-syria-airstrikes-still-clearly-illegal/ accessed 12 December 2020.

Reeve R, 'War in Yemen' (17 April 2015), available at https://www.oxfordresearchgroup.org.uk/war-in-yemen-the-african-dimension, accessed 12 December 2020.

Reisinger-Coracini A, 'More Thoughts on "What Exactly was Agreed In Kampala on the Crime of Aggression", (EJIL:Talk! 2 July 2010), https://www.ejiltalk.org/more-thoughts-on-what-exactly-was-agreed-in-kampala-on-the-crime-of-aggression/, last accessed 10 May 2020.

Reisinger-Coracini A, 'The Kampala Amendments on the Crime of Aggression Before Activation' (OpinioJuris, 29 September 2017 and 2 October 2017) http://opiniojuris.org/2017/09/29/the-kampala-amendments-on-the-crime-of-aggression-before-activation-evaluating-the-legal-freamework-of-a-political-compromise/ accessed 19 October 2019.

Stuerchler N, 'The Activation of the Crime of Aggression in Perspective', (EJIL:Talk! 26 January 2018), https://www.ejiltalk.org/the-activation-of-the-crime-of-aggression-in-perspective/, accessed 20 October 2019.

Trahan J, 'One Step Forward for International Criminal Law; One Step Backwards for Jurisdiction', (OpinioJuris, 16 December 2017), http://opiniojuris.org/2017/12-16/one-step-forward-for-international-criminal-law-one-step-backwards-for-jurisdiction-the-perspective-of-someone-present-at-the-un-during-negotiations/ accessed 7 January 2018.

Trahan J, 'Activation of the International Criminal Court's Jurisdiction over the Crime of Aggression and Challenges ahead' (OpinioJuris, 18 July 2018), https://opiniojuris.org/2018/07/18/33604, accessed 6 August 2018.

Triscone J, 'Universal Jurisdiction, the only hope for prosecuting international crimes committed in Syria', 6 September 2021, Trial International, available at https://trialinternational.org/latest-post/universal-jurisdiction-the-only—-hope-for-prosecuting-international-crimes-committed-in-syria/, accessed 8th September 2021.

Reports

Amnesty International, Universal Jurisdiction: *A Preliminary Survey of Legislation Around the World*, 2012 update, available at www.amnesty.org/en/library/info/IOR53/019/2012, accessed 15th November 2020.

Bensouda F, 'Prosecutor's Report on Preliminary Examination Activities 2020', 14 December 2020, available at https://www.icc-cpi.int/itemsDocuments/2020-PE/2020-pe-report-eng.pdf, accessed 5th September 2021.

Chilcot J, *The Report of the Iraq Inquiry*, Executive Summary, 128-135, http://www.iraqinquiry.org.uk/media/247921/the-report-of-the-iraq-inquiry_executive-summary.pdf, accessed 11 August 2020.

Commission on the Responsibility of the Authors of the War and on Enforcement of Penalties (1919-1920), Report of 29 March 1919.

House of Commons, *Disinformation and 'fake news': Interim Report: Government Response to the Committee's Fifth Report of Session 2017-19*, (HC 363), published 23 October 2018, available at

https://www.publications.parliament.uk/pa/cm201719/cmselect/cmcumeds/1630/163002.html, accessed 30 October 2020.

Report of the International Commission on Intervention and State Sovereignty (ICISS), 'The Responsibility to Protect' (2001).

The Kosovo Report - Conflict - International Response - Lessons Learned, (OUP 2000).

Lie T, UN Secretary-General's Report on the Nuremberg Trials (1949) UN Doc. A/CN.4/5, UN Sales No. 1949 V.7.

Paul-Boncour J, Report of Commission III, Enforcement Arrangements on Chapter VIII Section B, at the United Nations Conference on International Organization, Security Council, Committee 3, Doc. 881 III/3/46, 10th June, 1945

Schnabel A and Thakur R (eds), *The Kosovo Report*, Peace and Governance Programme, Kosovo and the Challenge of Humanitarian Intervention, (2000) available at http://archive.unu.edu/p&g/kosovo_full.htm, accessed 12 August 2020.

Other Sources

Annan K, 'Address by Kofi Annan' (Kampala, Uganda, 31 May 2010), https://www.kofiannanfoundation.org/speeches/kofi-annan-addresses-the-first-review-conference-of-the-assembly-of-states-parties-to-the-rome-statute-of-the-international-criminal-court/, accessed 12 April 2018.

Background on Security Council Reform – Global Policy Forum, http://globalpolicy.org/security-council/security-council-reform.html, accessed 20 January 2021.

Bassiouni M C, 'Draft Code of Offenses Against the Peace and Security of Mankind', *Proceedings of the Annual Meeting* (ASIL), 80 (9-12 April 1986) (OUP 1986) 120.

Center for UN Reform, 'Security Council Update 2015', https://centerforunreform.org/2015/01/12/security-council-reform-update-2015-similar-scenarios-same-positions-new-outcomes/, accessed 21 January 2021.

Charbonneau B, An Interview: 'The Military Intervention in Mali and Beyond', 28 March 2019, http://www.oxfordresearchgroup.org.uk/blog/the-french-intervention-in-mali-an-interview-with-bruno-charbonneau, accessed 20 August 2020.

Coalition for the ICC, 'Palestine', https://www.coalitionfortheicc.org/country/palestine, accessed 12 July 2020.

Coalition for the ICC, 'Historic activation of the Crime of Aggression', https://www.coalitionfortheicc.org/explore/icc-crimes/crime-aggression, accessed 12 July 2020.

Council on Foreign Relations, Timeline of the Iraq War, http://www.cfr.org/timeline/iraq-war, accessed 20th January 2020.

Status of ratification and implementation, Crime of Aggression Info, https://www.crimeofaggression.info/the-role-of-states/status-of-ratification-and-implementation/, accessed 20th January 2021.

Federal Bureau of Investigation, Russian Interference in 2016 US elections, http://www.fbi.gov/wanted/cyber/russian-interference-in-2016-u-s-elections, accessed 30 October 2020.

Gberie L, 'The Special Court for Sierra Leone rests - for good', (April 2014) Africa Renewal, available at https://www.un.org/africarenewal/magazine/april-2014/special-court-sierra-leone-rest---good, accessed 3rd September 2021.

G4 Ministerial Joint Press Statement, 23 September 2020, German Federal Foreign Office, http://www.mofa.go.jp/files/100095828.pdf, accessed 21 January 2021.

Geneva Internet Platform Digital Watch Observatory for Internet Governance and Digital Policy, 'UN GGE and OEWG', http://www.dig.watch/processes/un-gge, accessed 20 January 2021.

German Federal Foreign Office, 22nd September 2020, 'Reform of the United Nations Security Council', German Federal Foreign Office, http://www.auswaertiges-amt.de/en/aussenpolitik/internationale-organisationen/vereintennationen/reformsr/231604, accessed 21 January 2021.

Global Campaign for Ratification and Implementation of the Kampala Amendments on the Crime of Aggression, 'The Council of Advisers on the Application of the Rome Statute to Cyberwarfare – Project Background and Goals', 29 October 2019, http://crimeofaggression.info/the-campaign/the-council-of-advisers-on-the-application-of-the-rome-statute-to-cyberwarfare/, accessed 30 October 2020.

International Justice Resource Center, *Universal Jurisdiction*, available at https://ijrcenter.org/cases-before-national-courts/domestic-exercise-of-universal-jurisdiction/, accessed 20th November 2020.

Jackson R, United States Representative to the International Conference on Military Trials, 'Opening Statement before the International Military Tribunal', 21 November 1945, HMSO, (London 1945).

Ki-moon B, 'An Age of Accountability', Address to the Review Conference on the International Criminal Court, Kampala, 31 May, 2010, https://www.un.org/africarenewal/web-features/secretary-general's-"-age-accountability"-address-icc-review-conference, accessed 12 April 2018.

Koh H, Legal Adviser for the US Department of State at Kampala, Statement on behalf of the United States, June 4, 2010, available at https://2009-2017.state.gov/s/l/releases/remarks/142665.htm.

Litvinoff M, Union of Soviet Socialist Republics, 'Definition of "Aggressor": Draft Declaration', Doc. Conf D/C.G.38, *Monthly Summary of the League of Nations*, XIII: 2 (1933), Annex.

National Security Strategy and Strategic Defence and Security Review 2015, United Kingdom, (23 November 2015), http://www.gov.uk/government/publications/national-security-strategy-and-strategic-defence-and-security-review-2015-annual-report-2016, accessed 30th October 2020.

National Security Strategy of the US in 2019, http://www.dni.gov/files/ODN/documents/National_Intelligence_Strategy_2019.pdf, accessed 30 October 2020.

Pace W, '21 Member States Launch New Initiative to Improve the Working Methods of the Security Council' (21 May 2013), available at http://www.centerforunreform.org/?q=node/541, accessed 21 January 2021.

Paulet V, 'The ICC & universal jurisdiction: Two ways, one fight', 23 May 2017, https://www.coalitionfortheicc.org/news/20170519/icc-universal-jurisdiction-two-ways-one-fight, accessed 12 December 2020.

Stevenson D, 'Europe before 1914', British Library archives, https://www.bl.uk, accessed 4th March 2020.

Trial International, Make Way for Justice #3, Universal Jurisdiction Annual Review 2017, https://trialinternational.org/wp-content/uploads/2017/03/UJAR-MEP_A4_012.pdf, accessed 23 January 2021.

Trial International, Terrorism and international Crimes: Prosecuting Atrocities for what they are, Universal Jurisdiction Annual Review 2020, https://trialinternational.org/latest-post/universal-jurisdiction-annual-review-2020-atrocitites-must-be-prosecuted-soundly-and-rigorously/ accessed 23 January 2021.

United Nations Foundation, 'Five Global Issues to Watch in 2021', available at http://www.unfoundation.org/blog/post/five-global-issues-to-watch-in-2021/, accessed 5th February 2021.

Weisbord N, 'Activation of the Crime of Aggression', Presentation before the UN General Assembly, 2 June 2017, available at http://sls.sbg.ac.at/data/Weisbord.pdf, accessed 20 October 2019.

News articles

BBC News, 'UK combat operations end in Iraq', 30 April 2009, http://news.bbc.co.uk/1/hi/8026136.stm, accessed 14 December 2020.

New York Times (20th January 2012), J. Goldsmith, 'The Shadow of Nuremberg', http://www.nytimes.com/2012/01/22/books/review/justice-and-the-enemy-nuremberg-9-11-and-the-trial-of-khalid-sheikh-mohammed-by-william-shawcross-book-review.html, accessed 15 October 2020.

BBC News, 'African Union backs mass withdrawal from ICC', 1 February 2017, http://www.bbc.com/news/world-africa-38826073, accessed 18 November 2020.

BBC News, 'Syria Torture: German court convicts ex-intelligence officer), 24 February 2021, http://www.bbc.co.uk/news/world-europe-56160486, accessed 27 February 2021.

Packham C, 'Exclusive: Australia concluded China was behind hack on parliament', September 16, 2019, available at http://www.reuters.com/article/us-australia-china-cyber-exclusive-idUSKBN1W00VF, accessed 30 October 2020.

Pomfret J, 'Opinion: Despite election result, China won't give up on Taiwan', Washington Post, 13 January, 2020,

https://www.washingtonpost.com/opinions/2020/01/13/despite-election-result-china-wont-give-up-taiwan/, accessed 30 October 2020.

Reuters, 'Timeline: Invasion, surge, withdrawal; US forces in Iraq', https://www.uk.reuters.com/article/iraq-usa-pullout-idINDEE7BH05520111218, December 18th 2011, accessed 14 December 2020.

Wilmshurst E, 'Wilmshurst resignation letter' (18 March 2003), BBC News, 24 March 2005, http://new.bbc.co.uk/1/hi/4377605.stm, accessed 11 August 2020.